Emergency Food Security Assessment Handbook

Handbook second edition

WFP World Food Programme

vam wfp food sec

second edition

Emergency Food Security Assessment Handbook



second edition







World Food Programme Via Cesare Giulio Viola, 68-70 Parco de' Medici Nysis 00148 Rome - Italy

www.wfp.org

Emergency Food Security Assessment Handbook

second edition / January 2009





Emergency Food Security Assessment Handbook - second edition

© January 2009, World Food Programme (WFP), Food Security Analysis Service

This handbook has been prepared by the United Nations World Food Programme (WFP) for use by staff and its partners. All rights are reserved. Reproduction is authorized, except for commercial purposes, provided that WFP is acknowledged as the original source.

United Nations World Food Programme

Via Cesare Giulio Viola 68/70, Parco de' Medici 00148, Rome - Italy

Food Security Analysis Service

Chief: Joyce Luma Tel: +39-06-6513-2168 E-mail: joyce.luma@wfp.org or wfp.vaminfo@wfp.org

EUROPEAN COMMISSION



Humanitarian Aid

ECHO provided financial support for the development and production of this Handbook.

Foreword

For decades, WFP has been on the frontline of the fight against hunger. But today, the global context in which WFP operates is rapidly changing. New challenges are emerging. High commodity prices, the increasing number of natural disasters, and the rapid urbanization of populations are leading to higher levels of food insecurity, but also to more complex and multi-faceted emergencies.

To address the changing nature of food insecurity and its nutritional outcomes, WFP has shifted from a food aid to a food assistance agency and developed a more diverse and innovative set of tools.

Food Security Assessments are one of the fundamental tools of the organization. They provide the analysis to make well informed decisions on the most appropriate type and scale of interventions. They determine who are the food insecure people, how many they are, where they live and why they are food insecure. They also make recommendations on what needs to be done to assist them.

WFP is committed to excellence in the field of food security analysis. In the last three years, through an ambitious project to strengthen WFP's needs assessment practice, needs assessments have become the backbone of WFP's operations. Every year WFP conducts more than 90 assessments in partnership with governments, United Nations Agencies, national and international non-governmental organizations (NGOs), and donors. WFP's assessments inform not only WFP but also the wider humanitarian community.

This commitment to excellence is reflected in the Emergency Food Security Assessment (EFSA) Handbook. This second edition has benefited from all the latest methodological advances made by WFP under the guidance of leading food security experts from academia, NGOs and United Nations organizations, and building on decades of field practice and lessons learned from major assessments ranging from the yearly in-depth assessments in Darfur to the rapid assessments conducted after the floods in West Africa in 2007. It has been made possible thanks to the generous support from the European Commission's Humanitarian Aid department (ECHO). This edition contains information on the latest progress made to understand how markets impact on household food security, to identify chronic and transitory food insecurity, and measure food insecurity. It integrates food security and nutrition analysis and explores the population's vulnerability and risks to livelihoods. It also facilitates the analysis of response options, looking at food and non-food options.

The EFSA Handbook provides WFP staff and partners with the most up-to-date and advanced guidance on how to conduct accurate, timely and transparent food security assessments. This will enable us to address the new hunger challenges of this century.

Manail 1

Ramiro Lopes da Silva Deputy Chief Operating Officer and Director of Operations

Acknowledgements

The Emergency Food Security Assessment (EFSA) Handbook was made possible thanks to the generous support of the European Commission's Humanitarian Aid department (ECHO). The German Government supported a preparatory technical meeting.

The Handbook was written under the umbrella of the Strengthening Emergency Needs Assessment Implementation Plan (SENAIP). The purpose of the Plan (2004-2007) was to reinforce WFP's capacity to assess humanitarian needs in the food sector during emergencies and their aftermath through accurate and impartial needs assessments. The Plan and complementary projects were funded from internal and external sources, including ECHO, the Canadian International Development Agency (CIDA), the Citigroup Foundation, the Department for International Development, United Kingdom (DFID), the French, Danish, and German Governments.

This Handbook benefited from several peer review and consultation mechanisms. Valuable comments and feedback were received from WFP staff members and partners consulted through an email-based discussion involving more than 150 people including the Food and Agriculture Organization of the United Nations (FAO), World Vision International, *Action contre la Faim (ACF)*, the Famine Early Warning Systems Network (FEWS NET), the Office for the Coordination of Humanitarian Affairs (OCHA), the United Nations Development Programme (UNDP), Système d'Information sur les Marchés Agricoles (SIMA) Niger, Catholic Relief Services (CRS) Niger, Direction Régionale de Développement Agricole (DRDA) Niger, Service des statistiques de l'élevage (SIM BETAIL) Niger, CARE Niger, Comité Permanent inter-Etats de lutte contre la sécheresse dans le Sahel (CILSS), Médecins d'Afrique, Agro-Action Allemande (AAA) DRC, El Centro Latinoamericano de Medicina de desastres (CLAMED), Oficina Nacional de Istatísticas de Cuba (ONE), Instituto de Investigaciones de Riego y Drenaje, (IIRD) Cuba, Instituto de Planificación Fisica (IPF) Cuba, Adventist Development and Relief Agency (ADRA), the Nepal Red Cross Society (NRCS) and HelpAge International.

The SENAC¹ Advisory group of experts was also invited to be part of a working group established for the preparation of the Handbook. Valuable advice was received from the Institute of Development Studies (University of Sussex), the Michigan State University, the International Food Policy Research Institute (IFPRI), Tulane University, the European Commission Joint Research Centre, and the Feinstein International Centre (Tufts University).

The working group also included a number of WFP staff members, including the Regional Assessment Officers, VAM (vulnerability analysis and mapping) and programme staff from several regional bureaux and country offices, and staff from Headquarters – especially the Food Security Analysis Service, the Policy, Planning and Strategy Division and Service, and the Programme Design and Support Division.

WFP also thanks the VAM officers and Regional Assessment Officers who provided additional inputs and contributed to the consolidation of the Handbook after a final workshop in April 2008.

WFP expresses special thanks to Agnès Dhur and Jeremy Loveless for their substantial contribution to the development of the Handbook and to the consulting group, Technical Assistance to Non-Governmental Organizations (TANGO), for consolidating the final draft.

All these contributions are gratefully acknowledged although responsibility for the present text, including errors, remains with WFP.

^{1.} Strengthening Emergency Needs Assessments Capacity (SENAC), which is the main component of SENAIP.



Contents

	Acronyms 10 Introduction 12			
	art I - Conceptual Framework, objectives and types of EFSAs			
	Introduction	17		
	Key questions that an EFSA should answer	18		
3	Conceptual Framework of an EFSA	20		
	3.1 Livelihoods	20		
	3.2 Food security	22		
	3.3 Nutrition security	24		
	3.4 The Food and Nutrition Security Conceptual Framework	25		
	3.5 Local adaptation of the Conceptual Framework	28		
	Objectives of an EFSA	29		
5	Types of EFSA and partnerships	31		
	5.1 Types of EFSA	31		
	5.2 Partnerships and inter-agency assessments	33		
D	art II - Data, indicators and sources of information			
	Introduction	37		
	The analysis plan	38		
	Information needs	41		
	Contextual information	46		
	Indicators and data	49		
	5.1 Definitions	49		
	5.2 The three key sets of indicators used in an EFSA	50		
	5.3 Proxy indicators	50		
	5.4 Linking indicators to EFSA objectives	51		
	5.5 Interpretation of indicators using thresholds	51		
	5.6 Cross-tabulation and comparison of indicators	53		
	5.7 Prioritization of indicators	54		
6	Data sources	55		
7	Choice of data and indicators	56		
	7.1 Mortality indicators	56		
	7.2 Nutrition status indicators	57		
	7.3 Food security status indicators	62		
D	DADT III Diaming and implementing on 5504			
	ART III - Planning and implementing an EFSA Introduction	84		
	Planning an EFSA	85		
2		00		

	2.1	Trigger mechanisms for an EFSA	86
	2.2	Objectives and terms of reference	87
	2.3	Budget	88
	2.4	Analysis plan and information requirements	88
	2.5	Collation and review of secondary data and reference material	88
	2.6	Methodology for primary data collection	92
	2.7	Sampling	95
	2.8	Human resources	113
	2.9	Administration and logistics	115
	2.10	Communications, security and emergency procedures	115
	2.11	Briefing	116
	2.12	Assessment schedule	116
3	Field	dwork	118
	3.1	Daily preparation	118
	3.2	Discussion with community leaders	118
	3.3	Collection of primary data	119
	3.4	Team meetings	119
	3.5	Final community meeting	119
	3.6	Daily analysis	119
4	Prin	nary data collection	120
	4.1	The importance of gathering good quality primary data	120
	4.2	Approaches to primary data collection	122
	4.3	Undertaking primary data collection	125
	4.4	Participatory tools for primary data collection	138

PART IV - Analysing EFSA data

1	Intro	oduction	146
2	Ess	ential concepts and methods	148
	2.1	Vulnerable groups, including livelihood groups	148
	2.2	Triangulation and convergence of evidence	150
3	Con	ducting a situation analysis	152
	3.1	Step 1: Synthesize contextual information	153
	3.2	Step 2: Estimate the numbers of food-insecure and malnourished	153
	3.3	Step 3: Determine the characteristics of the food-insecure and	
		malnourished	156
	3.4	Step 4: Identify the reasons for food insecurity and malnutrition	
		and for risks to livelihoods	160
	3.5	Step 5: Determine the nature of food insecurity and malnutrition	
		(chronic or transitory)	162
	3.6	Step 6: Estimate the severity of food insecurity and malnutrition	164
4	Con	ducting a forecast analysis	170
	4.1	Identification of future opportunities and shocks	170
	4.2	Developing scenarios	175

	4.3	Identification of population groups affected under		
		the most likely scenario	180	
	4.4	Estimation of the impact of shocks and opportunities on livelihoods	185	
	4.5	Estimation of the numbers of people who will be affected		
		by shocks and opportunities	191	
5	Cor	ducting a response analysis	193	
	5.1	Factors related to risks to lives and livelihoods	194	
	5.2	Entry points for interventions	196	
	5.3	Other stakeholders' interventions, and remaining gaps	197	
	5.4	Response options	199	
6	For	nulating recommendations for interventions and follow-up	209	
P	ART	V - Reporting EFSA results		
1	Intr	oduction	213	
2	Rep	ort structure	214	
3	Rep	ort quality monitoring	218	
4	Cor	nmunicating the results of the EFSA	219	
	4.1	Users of the EFSA report and communication channels	219	
	4.2	EFSA executive brief	220	
	4.3	Presentation workshop or meeting	221	
	4.4	Dissemination by e-mail and hard copy	222	
	4.5	The news media	222	
	4.6	Websites and newsletters	223	
	4.7	Advocacy for non-food recommendations	223	
	nne			
		1: Sample household questionnaire	226	
		Sample key informant questionnaire	242	
		Sample checklist for focus group discussions	248	
Aı	nnex	Main advantages and disadvantages		
		of selected response options	255	
Aı	nnex	Main response options and information required		
		to judge their appropriateness	258	
Aı	nnex	6: Executive brief template	270	
G	lossa	iry	273	
In	Index28Reference list28List of boxes29List of figures29List of tables29			
R				
Li				
Li				
Li				
R	Resources on the Food Security Analysis Guidance DVD 2			

Acronyms

ACF	Action contre la Faim (Action Against Hunger)
AIDS	acquired immunodeficiency syndrome
BMI	body mass index
CAP	Consolidated Appeal Process
CARE	Cooperative for Assistance and Relief Everywhere
CDC	United States Centers for Disease Control and Prevention
CFSAM	crop and food security assessment mission
CFSVA	comprehensive food security and vulnerability analysis
CFW	cash for work
CIDA	Canadian International Development Agency
CMR	child mortality rate
CSI	coping strategy index
DFID	Department for Intenational Development (United Kingdom)
ECHO	European Commission's Humanitarian Aid department
EFSA	emergency food security assessment
EMOP	emergency operation
FAO	Food and Agriculture Organization of the United Nations
FCS	food consumption score
FEWS NET	Famine Early Warning Systems Network
FFR	food for recovery
FFW	food for work
GAM	global acute malnutrition
GDP	gross domestic product
GNI	gross national income
H/A	height-for-age
HC	Humanitarian Coordinator
HH	household
HIV	human immunodeficiency virus
IASC	Inter Agency Standing Committee
ICRC	International Committee of the Red Cross
IDP	internally displaced person

IFPRIInternational Food Policy Research InstituteIFRCInternational Federation of Red Cross and Red Crescent SocietiesIOMInternational Organization for MigrationIPCIntegrated Food Security Phase ClassificationIRAImmediate Response AccountJAMjoint assessment missionMUACmid-upper arm circumferenceNGOnon-governmental organizationOCHAOffice for the Coordination of Humanitarian AffairsODIOverseas Development Institute (United Kingdom)PCNApost-conflict needs assessmentPDApersonal digital assistantPPSprobability proportional to sizePRROprotracted relief and recovery operationQMCquality monitoring checklistSENACStrenghthening Emergency Needs Assessment Implementation PlanSWOTstrengths, weaknesses, opportunities and threats analysisUSMRunder-5 mortality rate
IOMInternational Organization for MigrationIPCIntegrated Food Security Phase ClassificationIRAImmediate Response AccountJAMjoint assessment missionMUACmid-upper arm circumferenceNGOnon-governmental organizationOCHAOffice for the Coordination of Humanitarian AffairsODIOverseas Development Institute (United Kingdom)PCNApost-conflict needs assessmentPDApersonal digital assistantPPSprobability proportional to sizePRROprotracted relief and recovery operationQMCquality monitoring checklistSENACStrenghthening Emergency Needs Assessment Implementation PlanSWOTstrengths, weaknesses, opportunities and threats analysis
IPCIntegrated Food Security Phase ClassificationIRAImmediate Response AccountJAMjoint assessment missionMUACmid-upper arm circumferenceNGOnon-governmental organizationOCHAOffice for the Coordination of Humanitarian AffairsODIOverseas Development Institute (United Kingdom)PCNApost-conflict needs assessmentPDApersonal digital assistantPPSprobability proportional to sizePRROprotracted relief and recovery operationQMCquality monitoring checklistSENACStrenghthening Emergency Needs Assessment Implementation PlanSWOTstrengths, weaknesses, opportunities and threats analysis
IRAImmediate Response AccountJAMjoint assessment missionMUACmid-upper arm circumferenceNGOnon-governmental organizationOCHAOffice for the Coordination of Humanitarian AffairsODIOverseas Development Institute (United Kingdom)PCNApost-conflict needs assessmentPDApersonal digital assistantPPSprobability proportional to sizePRROprotracted relief and recovery operationQMCquality monitoring checklistSENACStrenghthening Emergency Needs Assessment Implementation PlanSWOTstrengths, weaknesses, opportunities and threats analysis
JAMjoint assessment missionMUACmid-upper arm circumferenceNGOnon-governmental organizationOCHAOffice for the Coordination of Humanitarian AffairsODIOverseas Development Institute (United Kingdom)PCNApost-conflict needs assessmentPDApersonal digital assistantPPSprobability proportional to sizePRROprotracted relief and recovery operationQMCquality monitoring checklistSENACStrenghthening Emergency Needs Assessment Implementation PlanSWOTstrengths, weaknesses, opportunities and threats analysis
MUACmid-upper arm circumferenceNGOnon-governmental organizationOCHAOffice for the Coordination of Humanitarian AffairsODIOverseas Development Institute (United Kingdom)PCNApost-conflict needs assessmentPDApersonal digital assistantPPSprobability proportional to sizePRROprotracted relief and recovery operationQMCquality monitoring checklistSENACStrenghthening Emergency Needs Assessment Implementation PlanSWOTstrengths, weaknesses, opportunities and threats analysis
NGOnon-governmental organizationOCHAOffice for the Coordination of Humanitarian AffairsODIOverseas Development Institute (United Kingdom)PCNApost-conflict needs assessmentPDApersonal digital assistantPPSprobability proportional to sizePRROprotracted relief and recovery operationQMCquality monitoring checklistSENACStrenghthening Emergency Needs Assessment Implementation PlanSWOTstrengths, weaknesses, opportunities and threats analysis
OCHAOffice for the Coordination of Humanitarian AffairsODIOverseas Development Institute (United Kingdom)PCNApost-conflict needs assessmentPDApersonal digital assistantPPSprobability proportional to sizePRROprotracted relief and recovery operationQMCquality monitoring checklistSENACStrenghthening Emergency Needs Assessment Implementation PlanSWOTstrengths, weaknesses, opportunities and threats analysis
ODIOverseas Development Institute (United Kingdom)PCNApost-conflict needs assessmentPDApersonal digital assistantPPSprobability proportional to sizePRROprotracted relief and recovery operationQMCquality monitoring checklistSENACStrenghthening Emergency Needs Assessment Implementation PlanSWOTstrengths, weaknesses, opportunities and threats analysis
PCNApost-conflict needs assessmentPDApersonal digital assistantPPSprobability proportional to sizePRROprotracted relief and recovery operationQMCquality monitoring checklistSENACStrenghthening Emergency Needs Assessment CapacitySENAIPStrenghthening Emergency Needs Assessment Implementation PlanSWOTstrengths, weaknesses, opportunities and threats analysis
PDApersonal digital assistantPPSprobability proportional to sizePRROprotracted relief and recovery operationQMCquality monitoring checklistSENACStrenghthening Emergency Needs Assessment CapacitySENAIPStrenghthening Emergency Needs Assessment Implementation PlanSWOTstrengths, weaknesses, opportunities and threats analysis
PPSprobability proportional to sizePRROprotracted relief and recovery operationQMCquality monitoring checklistSENACStrenghthening Emergency Needs Assessment CapacitySENAIPStrenghthening Emergency Needs Assessment Implementation PlanSWOTstrengths, weaknesses, opportunities and threats analysis
PRROprotracted relief and recovery operationQMCquality monitoring checklistSENACStrenghthening Emergency Needs Assessment CapacitySENAIPStrenghthening Emergency Needs Assessment Implementation PlanSWOTstrengths, weaknesses, opportunities and threats analysis
QMCquality monitoring checklistSENACStrenghthening Emergency Needs Assessment CapacitySENAIPStrenghthening Emergency Needs Assessment Implementation PlanSWOTstrengths, weaknesses, opportunities and threats analysis
SENACStrenghthening Emergency Needs Assessment CapacitySENAIPStrenghthening Emergency Needs Assessment Implementation PlanSWOTstrengths, weaknesses, opportunities and threats analysis
SENAIP SWOTStrenghthening Emergency Needs Assessment Implementation Plan strengths, weaknesses, opportunities and threats analysis
SWOT strengths, weaknesses, opportunities and threats analysis
U5MR under-5 mortality rate
UNCT United Nations country team
UNDAC United Nations Disaster Assessment and Coordination Team
UNDP United Nations Development Programme
UNHCR Office of the United Nations High Commissioner for Refugees
UNICEF United Nations Children's Fund
USAID United States Agency for International Development
VAM vulnerability analysis and mapping
W/A weight-for-age
WFP World Food Programme
W/H weight-for-height
WHO World Health Organization

Introduction

This Handbook is intended for use in emergency situations or protracted crises, whether due to sudden natural disaster, drought, disease, economic collapse or conflict, and to address the needs of both resident and displaced persons.

The Handbook is intended for WFP VAM and food security analysts but it will also be useful for programme staff as well as for the governmental, NGO and United Nations partners with whom WFP collaborates in emergency food security assessments (EFSAs).

Background

Since 2003, through the support of the European Commission's Humanitarian Aid department (ECHO), WFP has made particular efforts to improve its capacity for EFSA. In the first edition of the EFSA Handbook, WFP sought to catalogue the different types of assessment approaches and methodologies undertaken in emergencies and provide normative guidance to staff on conducting food security assessments in the field.

This second edition is a continuation of a process through which WFP intends to improve its capacity to accurately identify and measure food insecurity, as well as formulate appropriate responses. The second edition of the EFSA Handbook moves beyond describing the various approaches to measuring food insecurity among households and populations affected by natural and man-made emergencies. It complements the first edition by providing guidance for collecting and analysing both rapid and in-depth emergency assessment data and for formulating recommendations for effective programming response.

This second edition represents WFP's corporate approach to assessing food insecurity at the household and community levels and should be used as a companion to the first edition of the EFSA Handbook. It will be useful to both WFP and partner staff who have a firm grasp of the concepts introduced in the first edition and their practical application. Given the five years and continued technical

development that has occurred between the two editions, some terms and tools have been refined. In cases where differences exist, the most recent guidance should be employed.

Complementary guidelines

The EFSA Handbook (second edition) is complemented by the Comprehensive Food Security and Vulnerability Assessment (CFSVA) Guidelines. A CFSVA provides an in-depth picture of the food security situation and the vulnerability of households in a given country during "normal times" and serves as such as the foundation for WFP programming at the country level. The Conceptual Food and Nutrition Security Framework for EFSA and CFSVA is identical and the analytical approach for the different types of EFSA and the CFSVA is consistent: effort has been made to harmonize sampling approaches, define indicators and follow a similar logic of analysis. An in-depth EFSA uses a very similar approach to the CFSVA to analyse food security in a crisis situation. The CFSVA guidelines will therefore assist with the conduct of in-depth EFSA. Moreover, any type of EFSA can draw on the data of the information base of a previously conducted CFSVA in the same country through understanding the deeper causes of chronic food insecurity and vulnerability and comparing the standardized crisis indicators with pre-crisis levels.

The upcoming UNHCR / WFP Joint Assessment Missions (JAM) guidelines, which guide the assessment of the situation of refugees, returnees and internally displaced persons are consistent with the new EFSA guidance. Specific guidelines exist for joint FAO / WFP Crop and Food Security Assessment Missions (CFSAMs) and these are also consistent with the EFSA Handbook.

Structure of the Handbook

The Handbook comprises five parts.

Part I presents the purpose and objectives of an EFSA, the Conceptual Framework, key food security concepts and the different types of EFSA.

Part II focuses on information requirements and indicators. More precisely it covers the analysis plan, explaining its purpose and components; the information needs; the requirements for the contextual information; the indicators and data necessary to estimate food insecurity and consequences to lives and livelihoods; and what sources of data to use.

Part III focuses on planning an assessment and collecting primary data. It presents the planning steps to undertake prior to fieldwork, logistics considerations for

fieldwork, and approaches and tools for primary data collection, with particular emphasis on the importance of good-quality primary data.

Part IV explains how to conduct a situation analysis, forecast and response analysis and then how to formulate recommendations for interventions.

Part V outlines the EFSA report structure, explains the EFSA quality monitoring system that builds on the EFSA; and provides guidelines for communicating EFSA conclusions and recommendations to decision-makers.

The Annexes provide supplementary tools such as standard questionnaires, templates and additional guidance and reference documents for food security analysis and response.

It should be noted that in January 2008 WFP's Emergency Needs Assessment Service (ODAN) and the Vulnerability Analysis and Mapping Branch (VAM) were merged into the Food Security Analysis Service (OMXF). The acronym VAM is still commonly used to refer to this service.

Part I Conceptual Framework, objectives and types of EFSAs



Contents of Part I

1	Intro	duction	17
2	Key c	uestions that an EFSA should answer	18
3	Conc	eptual framework of an EFSA	20
	3.1	Livelihoods	20
	3.1.1	Livelihood components	20
	3.1.2	Resilience, vulnerability and coping	22
	3.2	Food security	22
	3.2.1	Food availability	23
	3.2.2	Food access	23
	3.2.3	Food utilization	23
	3.3	Nutrition security	24
	3.3.1	Food security	24
	3.3.2	Health status and the public health environment	24
	3.3.3	Care practices	24
	3.4	The Food and Nutrition Security Conceptual Framework	25
	3.5	Local adaptation of the Conceptual Framework	28
		ctives of an EFSA	29
5	Types	s of EFSA and partnerships	31
		Types of EFSA	31
	5.1.1	Initial assessment	32
	5.1.2	Rapid assessment	32
		In-depth assessment	32
		Partnerships and inter-agency assessments	33
	5.2.1	Partnerships	33
	5.2.2	Inter-agency assessments	34

Introduction

The purpose of an emergency food security assessment (EFSA) is to assess the impact of shock on the food security of households and communities within the affected area. An emergency is a situation that causes widespread human, material, economic or environmental damage, threatening human lives and livelihoods and exceeding the coping capacities of the affected communities and/or government.²

An EFSA combines primary and secondary information to inform the decision-making process during rapid- and slow-onset emergencies. This Handbook provides guidance on analysing the food security and nutrition situation within a framework, and examines the linkages between them. Key outputs of an EFSA include:

- description of the current food and nutrition security situation;
- analysis of the ways in which the affected population, the government and other stakeholders are responding to the emergency;
- forecast of the future evolution of food and nutrition security;
- identification of response options, and recommendations for intervention or non-intervention.

An EFSA may be conducted as a rapid assessment or an in-depth assessment. A sequential EFSA is a series of rapid assessments, or a rapid assessment followed by in-depth assessments.

Part I of this Handbook provides an overview of the objectives and conceptual basis of an EFSA. It covers:

- the objectives of an EFSA and the questions that an EFSA should answer;
- the concepts that underlie EFSA analysis, and the ways in which these are combined within the Food and Nutrition Security Conceptual Framework.

Each section of the Handbook includes references to detailed coverage provided in other sections and other documents, including Technical Guidance Sheets.

^{2.} The WFP corporate definition of emergencies, adopted in February 2005, is as follows: "For purposes of WFP emergency projects, emergencies are defined as urgent situations in which there is clear evidence that an event or series of events has occurred which causes human suffering or imminently threatens human lives or livelihoods, and which the government concerned has not the means to remedy; and it is a demonstrably abnormal event or series of events which produces dislocation in the life of a community on an exceptional scale." WFP/EB.1/2005/4-A

Key questions that an EFSA should answer

Although EFSAs differ according to the context and nature of the emergency concerned, each should answer a series of core questions. These focus on measuring the impact of a shock in terms of the potential change in food security status among affected households.

Does the shock have an impact on food security?

 How does the crisis situation compare with pre-crisis? Is the population food-insecure?

Has the level of malnutrition among the affected population been exacerbated by the shock?³

 Is the population likely to remain or become food-insecure or malnourished in the future?

How severe is the problem?

· How severe is food insecurity and/or malnutrition?

How do people cope?

• Are the affected people able to cope with the problems on their own, without becoming more food-insecure and malnourished?

Who? How many? Where?

- Which population groups are food-insecure and/or malnourished now? Which groups may become so in the future?
- How many people are affected now? How many may be affected in the future?
- Where are these people located?

^{3.} In most EFSAs, the primary concern is undernutrition, i.e. the form of malnutrition associated with poor growth, loss of weight and/or vitamin and mineral deficiencies. Overnutrition is another form of malnutrition, which is increasing in many contexts and can also affect food security, but it is rarely a primary concern in an emergency.

Why?

• What are the causes of present and future food insecurity and malnutrition?

What is needed?

- Can the affected people cope with and recover unaided? Are they already receiving assistance?
- Is additional assistance needed? If so, what type? For whom? When? Where? How much? For how long?
- Can the government and national organizations provide this assistance or is international assistance required?
- What is the most appropriate response?

Within this outline, a more detailed set of questions is developed for each assessment (see **Part III, Chapter 4**), according to the nature of the emergency, the type of EFSA and the time and resources available.

Conceptual framework of an EFSA

An EFSA may be undertaken in response to a rapid- or a slow-onset emergency. In either case, **food and nutrition security** is analysed to determine the nature of the risks faced by individuals and households.

This section explains three core concepts – livelihoods, food security and nutrition – and examines these within the Food and Nutrition Security Conceptual Framework presented in **Section 3.4.**

3.1 Livelihoods⁴

3.1.1 Livelihood components

Livelihoods are defined in the Sphere Standards as: "Livelihoods comprise the capabilities, assets (including both material and social resources) and activities required for a means of living linked to survival and future well-being." See: "Humanitarian Charter and Minimum Standards in Disaster Response", The Sphere Project, Geneva, 2004.

An EFSA examines the **livelihoods** of households and individuals, and the ways in which these withstand problems or shocks. Analysis of livelihood security begins with examination of household **assets** and **livelihood strategies**.

Box 1.1: Assets and strategies

An **asset** can be defined as "anything that is considered valuable or useful, such as a skill, a quality, a person, etc."⁵ In the Sustainable Livelihoods Framework, assets are defined under six categories:

- human: health and nutrition status, physical capacity, skills, level of education, etc.;
- social: household, gender, kinship and other networks, community groups, values and attitudes, etc.;
- financial: income, credit and loans, savings, liquid assets, etc.;

^{4.} The description of livelihoods presented in this section is based on the Sustainable Livelihoods Approach (SLA). See: www.livelihoods.org/info/info_guidancesheets.html for more details.

^{5.} Chambers Compact Dictionary, Chambers Harrap Publishers Ltd, Edinburgh, UK, 2005.

- physical: productive assets such as tools and equipment, stores, housing, livestock, infrastructure, etc.;
- natural: land, water, forests, etc.;
- political: power relationships, access to and influence over local and higher-level government processes.

Strategies are the ways in which households utilize and combine their assets to obtain food, income and other goods and services, in the context in which they live. This is explained in the Sustainable Livelihoods Guidance Sheets as: "People's choice of livelihood strategies, as well as the degree of influence they have over policy, institutions and processes, depends partly upon the nature and mix of the assets they have available to them. Some combination of them is required by people to achieve positive livelihood outcomes – that is, to improve their quality of life significantly on a sustainable basis."

Example 1.1 gives a simplified illustration of how assets might be utilized within a household livelihood strategy.

Example 1.1: Assets and strategies

A household has the following **asset profile**:

- human: healthy and able-bodied men and women of working age; good knowledge of agriculture.
- social: well connected within the community.
- financial: low cash reserves.
- physical: few productive assets.
- natural: no land.
- political: no local representatives at the national level.

Assets are used to form a livelihood strategy:

• The household might use its strong **social** assets to obtain loans with which to rent land and buy tools and agricultural inputs. This enables it to utilize its human assets to cultivate and compensate for the low levels of financial and natural assets.

The assets to which a household has access and the strategies that it can employ are affected by the **context**, as follows:

- Policies may affect access to natural assets, through laws governing land tenure and property rights; the use of economic and physical assets, through trade policy; and the development of human assets, through payment for education and health services.
- Institutions include formal services, such as health, education and agricultural extension, and informal or customary institutions, such as those governing participation in the workforce for both men and women, the natural resource management of forests, pastureland and water, and conflict resolution.
- **Processes** are the ways in which actions take place and change occurs. They are affected by power relations and, in turn, affect the range of strategies available to a household. For example, access to influential people enhances livelihood strategies; informal relationships, such as traditions of trust and reciprocal support, are crucial to them.

Note: There are overlaps between some types of *assets* and *contexts*. For example, reciprocal support relationships could be categorized as "social assets" or "processes". In such cases, categorization is less important than *understanding* and *including* the issue in the analysis.

3.1.2 Resilience, vulnerability and coping

When a shock occurs, households and individuals within a community react in different ways. The extent to which they can withstand shocks without excessive disruption of their livelihoods depends on their *resilience* or *vulnerability*. **Vulnerability** is defined as exposure to risk and the lack of ability to cope with its consequences. **Resilience** refers to a person's or a community's "ability to bounce back or recover after adversity or hard times, and to be capable of building positively on these adversities".⁶

A household or individual with low resilience is considered to be vulnerable (see **Part IV, Chapter 2**). Resilience and vulnerability are determined by:

- the type of **shocks** that people are exposed to, for example, disease, economic problems such as unemployment, adverse climate, or conflict;
- the degree to which households and individuals can recover from shocks without compromising their long-term livelihood security.

When shocks push households beyond the difficulties faced in normal times, households and individuals employ **coping strategies**. Some of these may damage lives and livelihoods, thereby reducing resilience and increasing vulnerability.

In general, the more assets a household has and the more varied the strategies available to it, the more resilient that household will be. However, households with many physical, financial or natural assets may be targeted during conflict or civil unrest. In these cases, a strong asset profile increases vulnerability.

3.2 Food security

At the World Food Summit in 1996, **food security** was defined as: "Food security exists when all people, at all times, have physical and economic access to sufficient, safe and nutritious food to meet their dietary needs, and food preferences for an active and healthy life."

In an EFSA, the analysis of food security is based on three pillars: (i) **food availability**; (ii) **food access**; (iii) and **food utilization**. Indicators for analysis of the three pillars are provided in Part II.

^{6.} Mission Australia Research and Social Policy, 2005.

3.2.1 Food availability

Food availability is the **physical presence of food** in the area of concern through all forms of domestic production, commercial imports and food aid. Food availability might be aggregated at the regional, national, district or community level. In an EFSA, food availability is usually analysed at the district and community levels; national and regional food availability may be considered when developing future scenarios and discussing response options. Food availability is determined by:

- production: food produced in the area;
- trade: food brought into the area through market mechanisms;
- stocks: food held by traders and in government reserves;
- transfers: food supplied by the government and/or aid agencies.

3.2.2 Food access

Food access concerns a **household's** ability to **acquire** adequate amounts of food, through one or a combination of own home production and stocks, purchases, barter, gifts, borrowing and food aid. The following are some examples:

- own production crops, livestock, etc.;
- hunting, fishing and gathering of wild foods;
- purchase at markets, shops, etc.;
- barter exchange of items for food;
- gifts from friends/relatives, community, government, aid agencies, etc.

Food may be *available* but not *accessible* to certain households if they cannot acquire a sufficient quantity or diversity of food through these mechanisms.

3.2.3 Food utilization

Food utilization refers to **households'** use of the food to which they have access, and **individuals'** ability to absorb and metabolize the nutrients – the conversion efficiency of the body. Food utilization includes:

- the ways in which food is stored, processed and prepared, including the water and cooking fuel used, and hygiene conditions;
- feeding practices, particularly for individuals with special nutrition needs, such as babies, young children, the elderly, sick people, and pregnant or lactating women;
- the sharing of food within the household, and the extent to which this corresponds to individuals' nutrition needs growth, pregnancy, lactation, etc.;
- the health status of each member of the household.

Food may be *available* and *accessible* but certain household members may not benefit fully if they do not receive an adequate share of the food in terms of quantity and diversity, or if their bodies are unable to absorb food because of poor food preparation or sickness.

3.3 Nutrition security

Assessing the nutrition situation is an integral part of an EFSA.⁷ In addition to identifying the obvious health risks and problems affecting malnourished people, information on nutrition status provides objective and comparable indications of the extent of risks to lives and livelihoods. In particular:

- acute malnutrition is a clear sign that lives are in danger;
- **chronic malnutrition** indicates that there are long-standing problems in terms of food, health or care, generally related to poor livelihoods or deterioration of livelihoods.

The causes of malnutrition are summarized in the Sphere Standards as: "Food insecurity is one of three underlying causes of malnutrition [the others being health and care practices], and therefore wherever there is food insecurity there is a risk of malnutrition, including micronutrient deficiencies. Consideration of the impact of food insecurity on the nutrition situation is an essential part of food security assessment. However, it should not be assumed that food insecurity is the sole cause of malnutrition, without considering possible health and care causal factors."

3.3.1 Food security

Food security is defined in **Section 3.2**.

3.3.2 Health status and the public health environment

In relation to nutrition status outcomes:

- the health status refers to the body's ability to absorb and use the nutrients that are consumed; sickness, such as diarrhoea, can cause food to pass through the body without being absorbed;
- the public health environment refers to the conditions in which people live and the conduciveness or otherwise of these conditions to human health; water, sanitation, waste disposal systems and type of housing are key determinants of the public health environment.

3.3.3 Care practices

These are the ways in which dependent members of a household are looked after and fed. Among other factors, they include feeding practices for babies and young children, particularly breastfeeding and complementary feeding; food distribution priorities within the household, for example, children or adults first; and practices for the care of sick children and adults and the elderly. If care practices are inadequate, individual malnutrition can exist in households with good food access and a healthy environment.

^{7.} This does not necessarily mean that an EFSA must always carry out direct measurement of the nutrition status, such as through collecting anthropometric data.

3.4 The Food and Nutrition Security Conceptual Framework

The EFSA analysis is based on an understanding of food security and vulnerability. The Food and Nutrition Security Conceptual Framework informs not only the selection of indicators for analysis and use in geographical targeting, but also the design of field assessment instruments and the organization of standardized reporting formats. The household food security conceptual framework adopted by EFSAs considers food availability, food access and food utilization as core determinants of food security, and links these to households' asset endowments, livelihood strategies, and political, social, institutional and economic environment.



Figure 1.1: The Food and Nutrition Security Conceptual Framework

During an EFSA, the conceptual framework serves two purposes by providing:

- a basis for developing initial hypotheses on the emergency, its causes and effects;
- a way of visualizing the relationships among factors that affect food and nutrition security, which is helpful during data collection and analysis.

The food security status of any household or individual is typically determined by the interaction of a broad range of agro-environmental, socio-economic and biological factors. As with the concepts of health or social welfare, there is no single, direct measure of food security. However, the complexity of the food security problem can be simplified by focusing on three distinct but interrelated dimensions: aggregate food availability, household food access, and individual food utilization.

Vulnerability is a forward-looking concept for assessing community and household exposure and sensitivity to future shocks. Ultimately, the vulnerability of a household or community depends on its ability to cope with exposure to the risks associated with shocks such as drought, flood, crop blight or infestation, economic fluctuation and conflict. The ability to manage these risks is determined largely by the characteristics of a household or community, particularly its asset base and the livelihood and food security strategies it pursues.

The framework shows that exposure to risk is determined by the frequency and severity of natural and human-induced hazards, and by their socio-economic and geographical scope. The determinants of coping capacity include the levels of a household's natural, physical, economic, human, social and political assets, the levels of its production, income and consumption, and its ability to diversify its income sources and consumption to mitigate the effects of the risks it may face at any moment.

Coping behaviour often involves activities such as the sale of land or other productive assets, the cutting of trees for sale as fuelwood or, in extreme cases, the sale of girls into prostitution. These practices can undermine not only the long-term productive potential of vulnerable households, but also important social institutions and relationships. The extent of reliance on destructive practices is an indicator of vulnerability levels during a crisis.

An understanding of *how* households cope is an important aspect of analysis, but an understanding of *how well* they cope, or of their resilience, is even more important. How well the local economy can absorb the additional labour or products – such as livestock or fuelwood – that come on the market as the result of coping behaviour during a disaster, and the stability of wages and prices for these products are critical factors in understanding vulnerability.

Food security analysis is a static view of food access and household constraints

to food access, from either a short- or a long-term perspective. In contrast, vulnerability analysis views food access from a more dynamic, forward-looking perspective, because it includes the element of risk that households face in their day-to-day decision-making, and their capacity to respond effectively over time.

There is a significant overlap between households that are currently food-insecure and those who are at risk to the severe fluctuations in food access that threaten human well-being. Although all households may be considered vulnerable, from an operational perspective the primary emphasis of vulnerability analysis should always be on those households that are nearly or already food-insecure.

Early identification of problems clearly reduces the likelihood of malnutrition and excess mortality. Prompt action at the lower levels of the Conceptual Framework is therefore highly desirable. An EFSA should be undertaken as soon as potential problems are identified (see **Part III, Section 2.1**).

Two simplified ways in which the Conceptual Framework can guide EFSA analysis are given in Example 1.2.

Example 1.2: The EFSA analytical approach

- Poor financial assets lead a household to adopt damaging livelihood strategies, whereby all employable household members seek paid work. The mother is forced to leave her baby with the older children, who feed the baby irregularly using powdered milk mixed with contaminated water. These poor care practices create food utilization problems for the baby, who cannot absorb nutrients properly, even when the mother is present. The baby's condition worsens.
- 2. Weak health service institutions mean that when people become sick they are unable to get help. Poor health means that human assets deteriorate. Household productivity declines, with consequent reductions in economic assets and food access. Reduced food quantity and diversity exacerbate health and nutrition problems.

These examples demonstrate some of the ways in which problems in one component - livelihoods, food security or nutrition - can lead to problems in another. In other words, there are causal linkages among the factors. Causal linkages can also be two-way, as shown in Example 1.3.

Example 1.3: Two-way causal linkages

Food insecurity can cause malnutrition. For example:

- poor access to food may lead to inadequate dietary diversity and insufficient consumption of micronutrients;
- the use of dirty water in cooking may cause diarrhoea, and hence poor absorption of nutrients.

(cont...)

(...cont)

Malnutrition can cause food insecurity. For example:

- malnourishment diminishes a child's learning capacity, which may limit his/her ability to find well-paid work in the future;
- malnourishment in adults can lead to poor productive capacity and frequent sickness; these have severe economic effects on the household, leading to poor food access.

3.5 Local adaptation of the Conceptual Framework

All EFSAs should draw on the Food and Nutrition Security Conceptual Framework to answer each of the core questions identified in **Section 2**, but the relative significance of each element of the Framework depends on the region and the country, the nature of the crisis, and the groups that are affected. The Framework and questions must therefore be **adapted** to reflect the context of each EFSA. This is done **in the country**, based on consultations with as wide a spectrum of stakeholders as possible.

The Conceptual Framework *is adapted, prior to primary data collection in the field*, by carrying out the following steps:

- review of secondary information and discussion with key informants (see Part III, Section 2.5);
- identification of the factors most likely to affect food security and nutrition within the current context;
- identification of possible linkages among the factors affecting food security and nutrition.

Adapting the Conceptual Framework to the local context allows the formulation of **initial hypotheses** regarding the probable effects of the crisis and the issues to examine in the food and nutrition security assessment. These hypotheses also provide a basis for defining the **information requirements** (see **Part II**), inform the selection of assessment methodology (see **Part III**), and enable development of a preliminary analysis plan (see **Part III, Section 2.4**). Initial hypotheses may be *confirmed, refuted or adapted* during the course of the assessment.

Objectives of an EFSA

The objectives define the outputs the EFSA is expected to generate. General objectives are similar for all EFSAs, and encompass the core questions outlined in **Chapter 2**. As with the Conceptual Framework, however, these essentially standard objectives can be adapted to local circumstances. Context-specific details that may be added include the names of affected areas and an indication of priority population groups. It may also be necessary to insert additional objectives, depending on the context.

Objectives should be:

- as **specific** as possible: expectations should be clear;
- **realistic**: expectations should be feasible given the available time and resources, and the existing constraints.

Example 1.4 provides a set of general objectives that can be applied to most EFSAs.

Example 1.4: EFSA objectives

- Identify the prevalence and severity of food insecurity and malnutrition in the area.
- Estimate how many people are affected.
- Determine where the affected people are located.
- Describe the coping strategies utilized by the various population groups, and identify any that may have a negative impact on lives or livelihoods.
- Describe the food-insecure and/or malnourished population in terms of their individual and socio-economic characteristics gender, ethnicity, etc. and livelihoods.
- Establish the reasons why people are food-insecure and/or malnourished by identifying factors that are associated with food insecurity and malnutrition.
- Determine whether food insecurity and nutrition problems are chronic or transitory.
- Develop scenarios for the next three, six and twelve months, and use these to forecast the evolution of the food security and nutrition situation *if no intervention is made*. Consider:
 - the severity of current food insecurity and malnutrition;
 - factors associated with food insecurity and malnutrition;
 - chronic and transitory issues;
 - the likelihood of future shocks;
 - the resilience and vulnerability of individuals and households to future shocks.

(cont...)

(...cont)

- Evaluate the need or otherwise for external assistance food or non-food. Consider:
 ongoing and planned interventions by government or other agencies;
 - the role of food aid and whether or not it is appropriate in this situation;
 - the strengths, weaknesses, opportunities and threats (SWOT) analysis of various response options.
- Make recommendations for interventions, including: What? How much? For whom? When? For how long?

More details on the definition of objectives are given in Part III, Section 2.2.

chapter 5

Types of EFSA and partnerships

There are numerous ways of conducting an EFSA. The most appropriate approach depends on the objectives of the EFSA; the type of emergency; the stage in the emergency – early, late, etc.; the time available; access to the affected area; logistics; security; resources, especially human; and many other factors. The following sections present three types of EFSA. **In real situations, distinctions may not be as clear** as these categories imply, and it is not always possible to assign an EFSA precisely to one of the three types.

5.1 Types of EFSA

The three types of EFSA can be broadly summarized as:

- initial assessment;
- rapid assessment;
- in-depth assessment.

The analytical basis is the same for each type of EFSA. The essential differences lie in the time available for the assessment and the constraints to access to the areas concerned. These factors affect the scope of information that can be collected and the depth of the analysis.

In general, the three EFSA types can be categorized as follows:

- An initial assessment provides rough information quickly.
- A rapid assessment provides information that is collected and analysed using rigorous procedures, but time and access constraints lead to substantial reliance on assumptions, estimates and approximations.
- An in-depth assessment is based on a rigorous methodology, and collects a more substantial body of quantified information. The approach is time-consuming and generally inappropriate for sudden-onset crises.

Characteristics of each type of EFSA are summarized in the following sections.

5.1.1 Initial assessment⁸

An initial assessment is undertaken promptly following: (i) a sudden crisis; (ii) reports of deterioration in a long-term crisis; or (iii) improved physical access to an area experiencing an ongoing crisis.

The primary purpose of an initial assessment is to provide critical information for the formulation of emergency assistance plans, funding appeals and the design of more detailed follow-up assessments. Typically, all aspects of an initial assessment – from fieldwork in affected areas to production of the final report – should be completed within six to ten days. Decisions regarding where to carry out an initial assessment and which agencies to involve will usually be taken at the country level, through discussion between the United Nations country team (UNCT), led by the Humanitarian Coordinator (HC) or the United Nations Resident Representative, and the government concerned. Initial assessments are usually based on secondary data and key informant interviews, but some quick field visits may be undertaken.

The initial assessment addresses the following priority questions:

- Is there a food security and/or nutrition problem that poses an immediate threat to life? If so, what types of intervention should be launched immediately?
- Is there need for a more thorough rapid assessment? If so, when should this take place and what should its focus be?

5.1.2 Rapid assessment

A rapid assessment is undertaken following an initial assessment in a sudden crisis, or as a component of a reassessment. It provides more details than the initial assessment, and is based on a combination of secondary and primary data. Formal surveys and interviews may be used. Both quantitative and qualitative data may be collected. The rapid assessment typically provides information on:

- the nature and scale of the crisis: effects on food security, nutrition, and livelihoods;
- the affected population: estimated numbers and locations;
- access constraints: logistics, security, etc.;
- recommendations for immediate, short-term and, possibly, longer-term interventions.

A rapid assessment often takes place in a fast-changing context where results are needed quickly for decision-making, hence the need to compromise between information accuracy and timeliness.

5.1.3 In-depth assessment

An in-depth assessment is undertaken when more time, access and resources are available. It provides detailed and often statistically representative information that

^{8.} For more information, see: Technical Guidance Sheet No. 6, Initial Emergency Food Security Assessments, WFP Emergency Needs Assessment Service, September 2007.

can be extrapolated to wider population groups and areas. An in-depth assessment may be carried out when:

- the situation seems to be deteriorating slowly, and detailed information is required to inform programming decisions;
- an emergency has stabilized, and detailed analysis is necessary and feasible;
- baseline information needs to be created or updated for monitoring purposes.

In-depth assessments use rigorous methodologies that are adapted to the context and that include random-sampled, large-scale household food security and nutrition surveys (see **Part III**) and household economy baseline surveys⁹.

5.2 Partnerships and inter-agency assessments

5.2.1 Partnerships

As far as possible, assessments should be planned and implemented through partnerships involving multiple stakeholders; for example, WFP working with the national government, the Food and Agriculture Organization of the United Nations (FAO), the United Nations Children's Fund (UNICEF), the Famine Early Warning Systems Network (FEWS NET) and other agencies. Partnerships have the following **advantages**:

- The involvement of numerous stakeholders helps to ensure that many perspectives are taken into account.
- Collaboration in defining assessment objectives and analysing information ensures comprehensive ownership of the assessment conclusions and recommendations.
- The incorporation of different agencies can broaden the skills available to the assessment team.
- Transparency of the process enhances the acceptability of the conclusions. For example, when affected communities are included, they are more likely to understand the basis for recommendations.
- Assessment fatigue among affected communities is reduced, because they are subjected to questioning by one assessment team instead of several separate teams.

Potential **disadvantages** of partnerships include the following:

- It may take a long time to coordinate stakeholders, which is problematic in initial and rapid assessments.
- The incorporation of too many perspectives may compromise the methodology; if too many issues are included, the assessment may lose focus.

^{9.} For more information, see: Comprehensive Food Security and Vulnerability Analysis Guidelines, WFP, January 2009; and Technical Guidance Sheet No. 12, Complementary Methods and Tools for Emergency Food Security Assessment, WFP Emergency Needs Assessment Service, December 2007.

• Particularly in conflicts, the inclusion of certain partners may compromise the neutrality of the assessment team and affect informants' willingness to talk openly.

In most situations, however, the benefits of partnerships outweigh the drawbacks.

When a variety of partner agencies are to be included in the **field assessment team**, two points should be borne in mind:

- 1. Ensure that each member of the team has something to contribute to primary data collection. The inclusion of individuals for purely political reasons is counter-productive.
- 2. Do not make the team too large. An overlarge team can be difficult to manage and may pose a security problem and affect relationships with communities and the quality of the information collected. The arrival of a fleet of vehicles in a small village can accentuate the differences between the community and the team, making it impossible to develop dialogue.

5.2.2 Inter-agency assessments

Partnerships often take the form of inter-agency assessments, and are organized according to the context and objectives of the assessment. In general, all the key agencies should be involved in planning the assessment, and a range of agency representatives with complementary skills should be included in the assessment team.

WFP regularly participates in several inter-agency assessments in numerous countries. Some of these are listed in Table 1.1.

Table 1.1: Inter-agency assessments			
Assessment	Main partners	Context	
Joint assessment mission (JAM)	WFP – UNHCR	Refugee and IDP situations	
Crop and food security assessment mission (CFSAM)	WFP – FAO	Agricultural areas affected by crisis	
United Nations Disaster Assessment and Coordination Team (UNDAC)	WFP – governments – other United Nations agencies – NGOs	Large-scale natural disasters	
Consolidated Appeal Process (CAP)	WFP – OCHA – governments – other United Nations agencies – governments – NGOs	Large-scale natural or human-induced disasters	
Post-conflict needs assessment (PCNA)	WFP – World Bank – UNDP – other United Nations agencies – governments	Post-conflict recovery and reconstruction	

There are detailed guidelines for some of these partnership assessments, such as JAMs, CFSAMs and CAPs.

Part II Data, indicators and sources of information

deepa
Contents of Part II

1	Introd	duction	37	
2	2 The analysis plan			
3	3 Information needs			
4	Conte	extual information	46	
5	Indica	ators and data	49	
	5.1	Definitions	49	
	5.2	The three key sets of indicators used in an EFSA	50	
	5.3	Proxy indicators	50	
	5.4	Linking indicators to EFSA objectives	51	
	5.5	Interpretation of indicators using thresholds	51	
	5.6	Cross-tabulation and comparison of indicators	53	
	5.7	Prioritization of indicators	54	
6	Data	sources	55	
7	Choic	ce of data and indicators	56	
	7.1	Mortality indicators	56	
	7.2	Nutrition status indicators	57	
		Food security status indicators	62	
	7.3.1	Food consumption indicators	62	
	7.3.2	Food access indicators	66	
		Description of the current household food security situation	73	
	7.3.4	Coping strategy indicators	75	

PART II

Introduction

The first step in planning an EFSA is to define the objectives, as explained in **Part I**. The next step is to determine what information is required and what will have to be collected to achieve the objectives. This is covered in **Part II**.

An effective EFSA depends on defining information requirements early in the process. This ensures that data collection is focused and that only necessary information is collected and analysed. It minimizes the collection of redundant information that will not be used.

Part II covers:

- the analysis plan, explaining its purpose and components;
- **information needs**, using the Food and Nutrition Security Conceptual Framework to define broad information needs;
- the **contextual information** that is required to understand the background and causes of the crisis;
- **indicators and data**, introducing key concepts and presenting the indicators used in an EFSA to estimate food insecurity and risks to lives and livelihoods;
- data sources, both primary and secondary;
- choice of data and indicators, providing detailed guidance on how to choose indicators for a particular assessment.

The analysis plan

The analysis plan guides many of the decisions made during an EFSA. It is based on key hypotheses to be tested, and provides guidance on what data to collect, how to collect them, and what types of analyses will be required to interpret them. It may also guide which data need to be collected from primary and which from secondary sources. The analysis plan is best developed at the beginning of an assessment, *prior* to designing questionnaires, checklists and other data collection tools; there are three main reasons for this:

- Efficiency: It ensures that only *useful* information is collected. Time is not wasted in collection of information that will not help achieve the assessment objectives and will not be used.
- Thoroughness: It ensures that *all* the necessary information is collected.
- **Feasibility**: When all the information needs and available resources have been considered, it is possible to decide whether or not the EFSA can be undertaken as planned.

Box 2.1 provides further information about feasibility.

Box 2.1: Feasibility

While the analysis plan is being developed, it may become clear that the assessment cannot be carried out as *originally intended*, for reasons that include the following:

- There is insufficient time to collect all the information required.
- Access constraints affect the intended sampling approach.
- Too few personnel are available, or personnel do not have the requisite skills.
- Logistics or budgetary constraints restrict the scope of the assessment.

If analysis of the constraints shows that the original scope of the assessment is unrealistic, the approach will need to be modified. Depending on the context, the following are two of the main ways of doing this:

- The assessment objectives can be changed to reflect practical realities.
- The methodological approach can be modified. For example, random sampling might be replaced by purposive sampling; in extreme cases it may be necessary to rely on secondary sources only.

When designing the analysis plan, each of the EFSA objectives (see **Part I**) is considered with regard to the following questions:

- · What information is needed to meet the objective?
- · How can this information be collected?
- From what source(s) can the information be collected?

Table 2.1 shows a format that can be used to develop the analysis plan.

Table 2.1: Format for an analysis plan					
Information needs	Contextual information	Indicator(s)	Data required	Data source(s)	Analysis type
Objective 1:					
		Indicator 1.1			
		Indicator 1.2			
		etc.			
Objective 2:					
		Indicator 2.1			
		Indicator 2.2			
		etc.			
Objective 3:					
		Indicator 3.1			
		Indicator 3.2			
		etc.			

The components of the analysis plan are defined in Box 2.2, and described in more detail in the following sections.

Box 2.2: Definition of terms used to develop an EFSA analysis plan

Objectives: The outputs expected from the assessment (see Part I), for example, an estimate of the impact of a conflict on food access.

Information needs: The range of information that is needed to answer the questions posed by the objectives. For the objective stated above, the following information requirements might be identified:

- Details of the conflict: What are the causes? Who are the protagonists? Who are the victims, direct and indirect?
- Details of the population: What are the main livelihoods in the area? How do people normally obtain access to food and income?
- Effect of the conflict: What is the likely impact of the conflict on the food access strategies identified?

(cont...)

(...cont)

Contextual information: Details of the processes that led to the current emergency and the reasons why they did so. Factors that help explain the emergency and identify potential responses.

Indicator: A specific variable or combination of variables that gives insight into an aspect of the objectives. For example, if a livelihood group is expected to obtain access to food through cash crop sales and market purchases, the following indicators might be defined:

- the area currently planted with selected cash crops, compared with that under normal circumstances;
- the ratios of selling prices of selected cash crops to the costs of staple foods, now and under normal circumstances.

Data required: The information that must be collected to satisfy the broad information needs and the indicators. Examples include:

- qualitative information about a conflict, its causes and effects;
- qualitative information about livelihoods, social structure and politics;
- quantitative information about areas of land planted, average yields and market prices of items bought and sold.

Data sources: Potential sources of information; others can be added as the assessment progresses. Examples include:

- sources of qualitative data: key informants, such as political analysts, the staff of local non-governmental organizations (NGOs), etc.; focus groups within communities; and household interviews;
- sources of quantitative data: household surveys; key informants, such as agricultural extension workers, farmers and market traders; and focus groups within communities.

Analysis type: The type of parametric or non-parametric analyses that can be used to explore and interpret the data, for example:

- non-parametric analyses of primarily qualitative data;
- parametric tests for testing statistical hypotheses, such as analysis of variance and regression.

Data and indicators must be selected carefully for each EFSA. If too much information is collected, time is wasted during the data collection and analysis stages. If too little information is collected, it may be impossible to answer the assessment's key questions.

Information needs

Information needs are defined as **the data from primary and secondary sources that must be collected and processed in order to satisfy the assessment objectives**. As explained in **Part I**, the overall objectives of an EFSA are to measure the impact of a shock and the potential change this may have on households' food security situation.

Primary information needs are identified at the start of the assessment based on what secondary information is available. It is important to gather and review as much *relevant* secondary information as possible. This enhances understanding of the emergency and its consequences, and provides the basis for selecting data to be collected directly during the assessment.

Analysis of secondary information influences the choice of what primary information to collect in the following ways:

- When secondary information is recent, accurate and relevant, there may be less need to collect primary information. For example, if a well designed anthropometric survey has recently been undertaken, it is probably not necessary to repeat the exercise during the EFSA.¹⁰
- Secondary information provides a reference point. For example, if particular food access indicators were used in the past, it may be worthwhile to use the same indicators, so that the status of food security *now* can be compared with that in *the past*.
- There may be inconsistencies or gaps in the secondary information; one of the objectives of the EFSA may be to clarify the situation.

It is also necessary to decide how information may need to be *stratified*, i.e. collected in layers or classes. Stratification, or stratified sampling, involves dividing the population of interest into sub-groups – strata – that share something in common based on criteria related to the assessment objectives. Typical strata include geographical boundaries. Stratification is used when separate food security

^{10.} The use of secondary data usually requires the assessment team to judge the quality of the data, its relevance to the geographic areas and populations covered in the assessment, and its relevance to addressing the information needs of the EFSA.

estimates are desired for each of the sub-groups, at a predefined, minimum level of precision. Stratification can also allow more precise estimates of overall food security for the population of interest. For example, female-headed households may use different coping strategies from male-headed households; gender-based analysis is therefore important.

When pre-crisis data are not available, it is crucial that the EFSA collect information on the food security situation prior to the shock. Table 2.2 gives an example of how this can be done. Similar pre-crisis information can be collected through focus group discussions.

Table 2.2: Template for collecting pre-shock livelihood information				
Activities	What were your household's main activities before the shock? (rank up to 3 income activities)	What percentage contribution did each activity make to total household income?	What activities are you able to carry out now, after the shock?	What percentage contribution does each activity make to total household income now?
Main		%		%
Secondary		%		%
Tertiary		%		%
Total		100%		100%

Once the basic parameters of the crisis are understood, the factors that are likely to cause malnutrition and food insecurity are examined in more detail, to fine-tune the data collection requirements. The nature of the emergency and its possible impacts are considered. Tables 2.3 and 2.4 give examples of emergencies and their possible impacts.

Table 2.3: Sudden-onset crises and their impacts on nutrition status and food security			
Type of crisis	Impact		
Earthquake, flood, tsunami	 Destruction of infrastructure and equipment; consequent disruption of food production, markets and transportation systems Mass mortality and injury, leading to reduced participation in food production and distribution Destruction of medical, water and sanitation systems and injury/death of staff; increased incidence of disease Destruction of food stocks and other assets Destruction of housing, leading to population displacement; poor sanitation and shelter, leading to disease and death Loss of economic infrastructure: workplaces, roads, other infrastructure 		
Conflict: initiation or sudden escalation	 All of the issues cited above Targeted attacks on food production facilities – farms – and distribution systems: aid convoys and commercial trucking operations Hoarding of food, leading to increased prices Forced displacement to insecure and unsanitary locations, leading to disease and starvation 		

Emergency Food Security Assessment Handbook / second edition

Table 2.4: Slow-onset crises and their impacts on nutrition status and food security			
Type of crisis	Impact		
Drought	 Reduced food availability because of poor harvest Decline in assets such as livestock Decreased food access 		
Environmental degradation	 Reduced soil fertility, loss of topsoil and lowering of water table, leading to poor harvests and reduced food availability 		
Economic decline	 Reduced food access because of unemployment and declining terms of trade Deterioration in the nutrition status of poor people because of deterioration in diet 		
Long-term conflict	 Increased mortality Decline in food availability because agricultural land becomes unworkable and human assets are diverted towards the war Decline in food access because of rising prices and unemployment Deterioration of nutrition status because of poor food access, lack of health care, destruction of water/sanitation systems and absence of carers 		

As mentioned in **Part I**, knowledge of the context and probable effects of the emergency should enable adaptation of the Food and Nutrition Security Conceptual Framework and allow the EFSA to identify each of the factors directly influencing food security at the individual and household levels, and the linkages among these factors.

Starting at the top of the Conceptual Framework (see **Part I, Section 3.4**), each of the boxes is considered in turn, as explained in Box 2.3.

Box 2.3: Identification of priority information requirements using the Food and Nutrition Security Conceptual Framework

The Conceptual Framework is used to identify the main factors that are likely to affect nutrition status and food security. It helps to prioritize the information to be collected in a given context, and to analyse linkages among factors.

The factors are considered from the top of the diagram, starting with the outcomes. If secondary data review indicates problems at the outcome level – e.g. mortality or malnutrition – these must be addressed immediately. If the crisis has not reached the status of mass mortality and/or malnutrition, attention should be given to the next level down: immediate causes.

Starting at the top of the diagram, the process proceeds as follows.

Outcomes

• Mortality and nutrition status; Is there evidence or suspicion of mass mortality and/or widespread malnutrition? If so, urgent action is required. Information needs focus on the most immediate questions: What are the causes? How many people are affected? Are particular age or sub-groups affected? Where are they?

(cont...)

(...cont)

Immediate causes

- Individual dietary intake and disease: These are considered from two perspectives:

 (i) secondary data recent health and nutrition surveys, etc.; and (ii) knowledge of the context is this type of emergency in this particular context likely to result in poor dietary intake and/or disease or a further erosion of the current food security status? If it seems likely that dietary intake problems and/or disease exist, appropriate indicators should be chosen.
- Food utilization: It is important to look for information on feeding patterns, care and health practices, sanitation and access to clean water before the crisis, and then estimate the impact of the crisis on these.

Underlying causes: food security

An understanding of food consumption, access and utilization within the specific context forms the basis for analysing risks to lives and livelihoods (see **Part IV**).

- Household-level food consumption: The basic indicator in most EFSAs is the food consumption score (FCS). Other context-specific indicators can be used. For example, variation from the usual diet might be explored and explained.
- How households ensure access to food: This is context-specific and requires a thorough understanding of the local economy. For example, in an urban area where people buy most of their food, indicators may focus on the terms of trade between commodities and wages; the availability of jobs, disaggregated according to job type and population group; and how efficiently the market functions. In a rural or agricultural area, indicators would cover both market/employment issues and agricultural production: quantity of own production consumed and food stocks; quantity sold; and market prices. Coping strategies and their sustainability should also be considered. A common indicator is the coping strategy index (CSI), but this is not always appropriate and other indicators can be used (see Section 7).
- Livelihood assets and strategies: These provide important insights into people's access to and use of food. For example, details of household food production and cash earnings give an indication of food access; and the quality and availability of education and health services affect the support and advice available for preventing and treating disease.

Basic causes

These are issues that are outside the control of individuals and households and that fundamentally influence food access, consumption and utilization. The nature of the emergency and the context must be carefully considered when defining indicators at this level. For example:

- when exploring food availability there is no point in developing long lists of agricultural indicators if agriculture does not play a significant role in the food economy of the area;
- the quality and availability of education services will not have an immediate impact on nutrition and food security after a sudden-onset crisis such as a flood, but should be considered in follow-up assessments and when analysing chronic problems, because education services play a crucial role in recovery from crisis;
- health services are of primary significance during rapid-onset acute nutrition crises, when indicators related to health service provision become essential, such as ratio of qualified medics to population size, and quality and quantity of drugs and medical equipment;
- the political system at the local and national levels affects livelihood strategies and access to assets, so it is important to identify political structures and the ways in which they influence different groups within the society.

Linkages

At each level, linkages among the factors are suggested. For example, in an urban area household food access – at the underlying causes level – might be affected by the following factors:

- underlying causes: gifts/transfers, cash earnings/loans, and education level;
- basic causes: markets, education services, and health services.

These linkages are helpful when identifying information requirements. In this example, information about the following issues would be required:

- the market for labour and goods;
- health and education services;
- access to credit.

The process described in Box 2.3 is most effective when implemented with partners such as the staff of humanitarian organizations and government counterparts. This enables knowledge and experience to be shared, and key stakeholders to agree on the initial hypotheses and priorities for data collection (see **Part III**).

Information needs are specified at two levels:

- **Contextual information**: This provides the background to and helps build understanding of the effects of the emergency on nutrition and food security. It is crucial when analysing the causes of food insecurity and malnutrition and developing response options (see **Part IV**).
- **Data and indicators:** These provide the basis for analysing the nutrition and food security status now and in the future (see **Part IV**).

Contextual information

Contextual information is a crucial component of the primary data collected in an EFSA. As shown in **Part IV**, contextual information is essential to the interpretation of mortality, nutrition and food security data and the development of response options.

Ways of collecting contextual information include:

- observation;
- questionnaire-based household surveys and household interviews;
- key informant interviews;
- focus group discussions;
- secondary information review, at the outset and throughout the assessment.

It is best to use as many of these as possible.

Table 2.5 gives examples of contextual information that might be useful. The key issues should be selected with care, as not all will be relevant to every assessment. Additional issues should also be considered, as Table 2.5 does not cover every eventuality.

Table 2.5: Examples of contextual information		
Type of analysis	Key issues	
Current crisis	Is there a crisis? If so, what are its causes, nature and consequences?Who is most affected and why?What is likely to happen in the coming months?	
Historical	 What factors led to the current crisis? How do factors related to the history of the area and population groups affect the crisis? Has the area faced similar crises in the past? What were the impacts, and what interventions were undertaken? Are these experiences documented? What lessons were learned? 	
Conflict	 Is the area affected by conflict? If so, what caused the conflict? What is its nature – civil war, international conflict, etc.? How long has it been ongoing? What are the expectations for the future? What groups are most affected by the conflict? What are the direct and indirect effects of the conflict on food and nutrition security? Who are the actual and/or potential winners and losers of the conflict? 	

Type of analysis	Key issues
Security	 Is the area secure or insecure? If insecure, why? Who is in danger – residents, visitors, etc.? Why? What impact might the security situation have on responding to a food security crisis?
Political/ institutional	 Which government policies affect food and nutrition security – land tenure, price controls, wage rates, import/export taxes, subsidies, etc.? What are the effects of these policies? Have any of these policies been changed recently? Are they likely to be changed in the near future? Do socio-political factors, such as power struggles between groups, land nationalization or privatization, affect the crisis? What is the status of government service provision, particularly health, education, social security and agricultural extension? Are services improving, deteriorating or staying the same? Why? Does the government provide social security support to people who are unemployed, sick, etc.? If so, who qualifies? How much is the allowance and how does it compare with average incomes?
Social	 How many people live in poverty and absolute poverty, and who are they? What administrative systems exist? For example, do traditional leaders exert significant influence, or are national/local government systems more powerful? Are some groups marginalized, for example, on the basis of ethnicity or relationship with local leaders? How are gender relationships characterized? How do men and women participate in decision-making processes? Are gender relationships changing? What respective roles do men and women have in controlling household assets, including land, crops, livestock, food and cash? Is local society stable? Are social institutions such as power relations changing rapidly? If so, why? What social support systems exist, traditional/non-formal and State? Who has access to social support, and who does not? What are the reasons for inclusion and exclusion?
Agricultural	 What are the main food and cash crops? What are the average production levels? Where are crops cultivated and under which farming system – small farms, commercial farms, etc.? What are the main livestock species raised? What are their main uses? What are the main risks to agriculture – drought, flood, crop disease, etc.? How important are fishing and aquaculture to people's livelihoods?
Economy and markets	 What is the basis of the regional/national/local economy? What is the status of the economy: good/bad, growing/declining? What is the inflation rate? What have been the trends in the consumer price index¹¹ over recent months and years? Are long-term changes taking place, for example, from dependence on subsistence agriculture to industry? If so, why? Where are the main markets located? How accessible are they to people affected by the crisis? Do men and women have equal access to markets? Before the crisis, did markets function well? Were they well integrated and competitive? What was the status of market food availability and access? Have markets been affected by the crisis? If so, how?

(cont...)

^{11.} Market Analysis in Emergency Food Security Assessments: Guidelines on Market Situation Analysis and Forecast and Response Protocol, WFP Emergency Needs Assessment Service, September 2007.

(...cont)

1 /	
Type of analysis	Key issues
Education	 What are the level and quality of the education services in the area? Who has access to education services? For example, do boys and girls have equal access? What impact has the crisis had on education infrastructure, services, enrolment, retention, etc.?
Cultural	 What are the main cultural factors that might affect food and nutrition security, for example, breastfeeding practices, food taboos?
Environmental	 Is the natural environment changing – deforestation, water resources, etc.? If so, what are the causes and what are the consequences? Are natural resources, or scarcity thereof, a cause of food and nutrition insecurity? If so, what is the nature of the problem? Is there conflict over natural resources? Are natural resources and/or health being affected by human activity, for example, industry?
Geography and infrastructure	 What transportation infrastructure is there – roads, railways, airports, etc.? Is infrastructure affected by the season? Where are services located – hospitals, clinics, schools, etc.? Where are government offices located – administrative, water board, etc.?

Indicators and data

5.1 Definitions

In an EFSA, indicators are used to describe and measure household food security status, individual nutrition status, and immediate and underlying causes of food insecurity, and to provide insights into the context.

For example:

- the prevalence of malnutrition among children 6 to 59 months of age gives an indication of the nutrition status of the population as a whole;
- the rate of unemployment indicates the state of the economy.

Indicators are distinct from data. **Data** are the pieces of information that are collected from primary or secondary sources. **Indicators** are *compiled from data* and are interpreted through comparison with standard or context-specific thresholds.

For example, terms of trade between the cost of wheat flour and the wage for daily labour might be used as an indicator of food access. This indicator is defined by combining two variables:¹² the cost of 1 kg of wheat flour and the wage for one day's labour. The indicator may be tracked over time to establish trends in food access, compared with a benchmark level that indicates an acceptable ratio between the two variables, and/or compared among different sub-groups or across different geographical areas.

Indicators are defined at the start of the assessment, and may be modified as information is collected and analysed. The data needed for compiling and interpreting indicators are collected during the assessment. The data used in an EFSA may be qualitative or quantitative, as explained in the following sections.

^{12.} Market Analysis Tool: Terms of Trade, WFP Economic Analysis Unit, August 2007.

5.2 The three key sets of indicators used in an EFSA

In an EFSA, three key sets of indicators are used to estimate the dimensions of the food security problem caused by an emergency:

- Mortality rates give an indication of risks at the population level.
- Nutrition indicators are used to estimate nutrition status at the *individual level*.
- Food security indicators focus on assessing access to food and food consumption at the *household level*. The coping strategy index (CSI) is an important indicator of food security at the *household level*.

Other data and indicators are used to construct and interpret the key indicators, for example:

- market indicators, integrated with food security indicators (see Chapter 7);
- age and sex of individuals, used in the interpretation of mortality rates and nutrition status;
- household characteristics, used in the compilation and interpretation of nutrition, food security and coping strategy indicators. Common characteristics include households' size and composition, such as age and gender profiles and education level; their residential status, such as host population or displaced; and the presence of chronically sick individuals, such as those with HIV and AIDS.

5.3 Proxy indicators

Not all indicators provide a direct measurement of the factor to which they are related. Proxy indicators provide indirect information about a factor.

For example, the CSI (see **Section 7.3.4**) is sometimes used as a proxy indicator of household food security. The different coping strategies – behaviours – used by households in an emergency are used to estimate the severity of food insecurity. Extensive field testing has demonstrated that coping strategies correlate closely with food security. Box 2.4 describes proxy indicators that may prove useful in an EFSA.

Box 2.4: Proxy indicators of food security

Proxy indicators to estimate the severity of food insecurity include the following:

- The coping strategies that people and households adopt: These may damage health, nutrition status, productive capacity, etc., such as through drastic changes in food consumption, depletion of assets or unusual migration.
- Diversity of the food items consumed by households: This is an indication of macroand micronutrient intake.
- The size of the food gap: This measures the difference between households' food requirements and the food to which they have access.
- Previous crises: These might provide insights into the potential evolution of the crisis.

There are various ways of combining indicators to estimate severity. For example, the **Integrated Food Security Phase Classification** (IPC) system combines information from different sources to position a crisis on a scale of severity (see **Part IV, Chapter 3**).

5.4 Linking indicators to EFSA objectives

Indicators must be chosen carefully. They should provide information about the issues identified in the objectives, as described in Example 2.1.

Example 2.1: Linking indicators to EFSA objectives

One objective of an EFSA is to identify the prevalence, as a percentage, and the degree – severe or moderate – of food insecurity and malnutrition in the area (see **Part I**).

The core EFSA indicators – mortality, nutrition, food security and coping strategies – are used to address this objective. For example:

- nutrition: mid-upper arm circumference (MUAC) measurement for children 6 to 59 months of age;
- food availability: food production in the district plus food imports from other districts minus food exports to other districts;
- food access: terms of trade between the costs of wheat and daily labour;
- food utilization: level of individual knowledge about the cooking of newly introduced relief food;
- **coping strategies**: type of coping strategies currently used, and the significance of these for food security.

All of these indicators are **context-specific**. In this example:

- MUAC measurements are taken if there is a reasonable expectation of malnutrition, or a need to ascertain the nutrition status of the affected population quickly but insufficient time or resources to undertake a full nutrition survey;
- the choice of food availability indicator in this example implies that the area is agricultural; it would not be useful in an urban setting;
- the choice of food access indicator implies that wheat is a main staple and daily labour is a significant source of livelihood;
- the food utilization indicator is used if recent food distributions have included foods with which people are unfamiliar.

5.5 Interpretation of indicators using thresholds

Indicators are compared with thresholds or pre-crisis information to estimate the current status of nutrition and food security.

Some thresholds are established internationally and are universally applicable, for example:

 wasting: a weight-for-height ratio of minus 2 z-scores of the median of reference is used as a threshold to define global acute malnutrition in children 6 to 59 months of age; • crude mortality rate: a threshold of one death per 10,000 people per day denotes an alert; two deaths per 10,000 people per day indicates a critical emergency.

Other indicators and thresholds are context-specific and must be defined for each situation. For example:

- the ways in which people obtain access to food vary widely; indicators and thresholds for food access can be defined only when the local context is understood;
- coping strategies are also highly context-specific; for example, the collection of wild plants for eating might be a normal activity in one society, but indicates an extreme level of crisis in another.

Context-specific thresholds are defined through **value judgements**; much depends on the experience and knowledge of the people making the judgement. Thresholds can be defined in one or a combination of the following ways:

- using pre-crisis data¹³, when knowledge of normal conditions forms the basis for comparison;
- using surveys carried out by other agencies in the same area and during the current crisis;
- based on the judgement of local key informants and/or experts; a group discussion with several informants facilitates consensus.

When establishing thresholds in any of these ways, **transparency** is paramount. The rationale for the threshold and the limitations to its application must be clearly stated in the assessment report.

Example 2.2: Establishing thresholds

During the Darfur assessment in 2006, the following information was collected:

- Most people in the area under study depended primarily on their own agricultural production for food and income.
- According to FAO/WFP references, an average individual in Darfur required 150 kg of cereal per year for consumption.
- According to the 2005 EFSA, average yields of cereal were 450 kg/ha.
- To produce enough food, a household had therefore to cultivate at least 0.33 ha of cereal per household member.

On the basis of this analysis, simple thresholds can be established:

< 0.3 ha per household member = critical.

0.3 to 0.4 ha per household member = borderline.

> 0.5 ha per household member = satisfactory.

These thresholds are arbitrary and inserted for the purpose of illustration. In a real situation, numerous factors would be considered when assigning thresholds. For example, if rainfall is very unpredictable, the thresholds for borderline and satisfactory might be raised to provide a wider margin of safety.

^{13.} For example, the comprehensive food security and vulnerability analysis (CFSVA) conducted by WFP.

Notes:

- This type of threshold must always be analysed within the broader context. On its own, the information these categories provide is insufficient. For example, a household with 0.5 ha per household member might have problems if the rains were less than usual, or if conflict restricted access to fields.
- Variations in the data also need to be considered. In this example, average yield is estimated at 450 kg/ha, but this could mask a wide variation: perhaps half the farms yield 750 kg/ha and the other half only 150 kg/ha.

5.6 Cross-tabulation and comparison of indicators

In cross-tabulation and comparison, two or more indicators are combined to gain insights into the prevalence and causes of malnutrition and food insecurity. Box 2.5 provides a summary of this process.

Box 2.5: Cross-tabulation and comparison

Possible linkages among factors are identified during local adaptation of the Conceptual Framework. Indicators that are to be collected during fieldwork to investigate these linkages are defined. The following are some examples:

- The link between main household income source(s) and household food security status: Do households with poor food security have specific income sources? Indicators to be collected for this analysis would be related to food access, food consumption and income sources.
- 2. The link between water source and malnutrition: Are malnourished individuals mainly found in households with poor access to water, in terms of quantity and/or quality? Data to be collected would include nutrition indicators, such as MUAC; water quality and source of water; water quantity in litres per person per day; and household water usage for storage, personal washing, etc.

Indicators are **cross-tabulated** during the analysis to provide insights into the factors that affect food security and nutrition status. The results are used in the response analysis (see **Part IV**).

When computerized statistics programmes and skills are available, regressions and multivariate analyses can be undertaken to combine numerous variables simultaneously. Principal component analysis is an example of this approach.¹⁴

When advanced computing capacity is lacking, simple cross-tabulation of two or three variables or indicators can yield valuable information.

Note: Cross-tabulation can only be applied to quantitative data from one sample. If qualitative data are used, or if information is collected from unrelated sources, *comparisons* can be made but will not have the same statistical validity as cross-tabulations. For example, the area planted from a household questionnaire survey can be *compared* with statements about seed availability from focus group discussions with people in the same community, but the two pieces of information cannot be *cross-tabulated*.

^{14.} The 2009 Guidelines for Comprehensive Food Security and Vulnerability Analysis (CFSVA) (WFP Food Security Analysis Service) include guidance on principal component and cluster analysis.

5.7 Prioritization of indicators

An EFSA should not collect too many indicators. It is better to have a few carefully selected indicators than many, if some will not be useful for the analysis.

Minimum information requirements can be determined according to the following list, which should be reviewed for each context:

- **Mortality**: Mortality data should be used if they are available and/or highly relevant to the crisis being assessed. Consider the timeframe to which the data refer. For example, mortality rates compiled over the course of a year are not useful for estimating the impact of a tsunami a week after its occurrence.
- Nutrition status: If there is reason to believe that malnutrition exists, but a full nutrition survey, including weight and height measurements, cannot be carried out and is not available from other sources, MUAC measurements can be used instead.
- Food security: This is assessed from household food consumption, taking into account food access. The food consumption score (FCS) – see Section 7.3 – should be calculated for each household interviewed. At least one relevant food access indicator should be defined and used for each livelihood or other group of households, such as refugees in camps. Selected coping strategies should be compiled, or the CSI adapted, for each household interviewed (see Section 7.3.4).

Each indicator should be supported by contextual and qualitative information gathered through focus group discussions, key informant interviews and observation.

Contextual information should always be presented to support the analysis. The depth and scope of this information depend on the time available and the objectives of the assessment.

Data sources

PART II

The data sources are specified in the second to last column of the analysis plan (see **Table 2.1**). These may be secondary or primary:

- Secondary data have been collected prior to the EFSA, often by other people, and are used to inform the EFSA. They may come from baseline surveys, previous assessments, government information offices, such as for economic and agricultural data, or any other source that is not consulted directly during the assessment.
- **Primary data** are collected during the EFSA from key informant interviews, focus group discussions, observation, household interviews and/or household questionnaires.

Data sources should be identified in the analysis plan as village focus group, household survey, district agricultural officer, etc.

chapter 7 Choice of data and indicators

Chapter 3 described how to identify the broad information requirements based on the available secondary information and using the Food and Nutrition Security Conceptual Framework to identify key factors of food insecurity and the linkages among them. The next step is to define more precisely the data and indicators that will be used in the EFSA.

The assessment design should allow scope to adjust indicators if they prove to be inappropriate, or if additional indicators are identified during the course of the assessment:

- For a **questionnaire-based survey**, the questionnaire should be pre-tested before the assessment starts. This allows problems with the structure, questions and indicators to be corrected before the fieldwork begins (see **Part III**).
- For an EFSA based on **qualitative data**, the data and indicators can be adjusted as the assessment progresses. All field teams should be informed about changes and the reasons for making them.
- Indicators may be adjusted during the analysis stage, for example, by reorganizing the cross-tabulations among data or indicators.

Indicators are categorized into three sets (see Section 5.4):

- mortality;
- nutrition status;
- food security.

An additional set of indicators reflecting the *broader context* is also defined (see **Chapter 4**). The other three sets of indicators are described in the following subsections.

7.1 Mortality indicators

Mortality is measured at the *population level*. In rapid EFSAs, mortality data are often obtained from secondary sources, often local institutions such as hospitals, statistics offices, etc. Primary mortality data can also be collected, *but a representative random*

sample must be used. It is rarely possible to obtain accurate mortality data in the early stages of an emergency, owing to the lack of reliable data collection and reporting mechanisms, poor access and rapidly changing circumstances. Moreover, in many disaster-prone countries, reliable data for normal times do not exist.

Table 2.6: Crude mortality rate		
Crude mortality rate	Significance	
< 1 death per 10 000 people per day	Reasonable health situation	
1-2 deaths per 10 000 people per day	Elevated mortality	
> 2 deaths per 10 000 people per day	Health emergency	

Table 2.6 identifies the crude mortality rate thresholds that can be used.¹⁵

Qualitative information on mortality normally comes from local key informants sharing their perceptions of excess death in the area, or from direct observation of bodies, new graves, etc. This sort of information should be used with care, as it is highly susceptible to bias. For example, a member of a community that has been the target of a military attack will probably overstate the level of mortality in the region as a whole.

Qualitative information about mortality should be triangulated as widely as possible, and should not be extrapolated to make conclusions about overall mortality rates. Qualitative reports about high and unusual mortality rates provide support to other analyses of nutrition, food security and coping strategies. Alarming reports of mortality can also act as the trigger for a more rigorous assessment and the initiation of a response.

7.2 Nutrition status indicators

Weight, height and age, and micronutrient data are usually collected through random or census anthropometric surveys, covering children 6 to 59 months of age and/or adults, often women. It is possible to gather useful data on nutrition status *without* undertaking a full anthropometric survey, but the reliability of the analysis will be reduced.

The following are examples:

• MUAC measurements on a convenience sample of individuals might produce alarming results and trigger concerns about the nutrition situation, but

^{15.} A Manual: Measuring and Interpreting Malnutrition and Mortality, United States Centers for Disease Control and Prevention (CDC) and WFP, July 2005.

extrapolation to the population as a whole would not be possible. In such cases, the assessment should be followed by a rigorous nutrition survey.

• Growth monitoring through the collection of weight-for-age data at a health clinic may show deteriorating trends, but the children measured may not be representative of the wider population.

As with mortality rates, there are internationally accepted thresholds for determining the extent to which nutrition status has deteriorated at the individual and population levels (see Table 2.7). As long as data have been collected properly, analysis of nutrition indicators is therefore relatively straightforward. Table 2.7 includes guidance on interpreting nutrition thresholds for risks to lives.

Three of the most useful indicators are weight-for-height of children 6 to 59 months of age, MUAC of children and adults, and body mass index (BMI) of adults, which is calculated as the ratio of weight to the square of height.

As a general guide, the following thresholds can be used to define risks to lives at the individual level:

- Wasting: There is risk to lives when weight-for-height scores are < -2 z for children.
- MUAC: There are risks to lives when MUAC is < 12.5 cm for children, and < 22.5 cm for women.

For adults, BMI thresholds indicative of risks to lives are less clear, but individuals' lives can be considered to be at risk if they have a **BMI below 16** combined with an infectious disease such as HIV/AIDS or tuberculosis.

The guidance note on "Strengthening Rapid Food and Nutrition Security Assessments"¹⁶ provides additional advice on organizing an EFSA to ensure proper analysis of the nutrition situation, with or without anthropometric measurement. However, it is advisable to consult a nutritionist to identify the most appropriate nutrition indicators and interpret results.

^{16.} Guidance Note, Strengthening Rapid Food and Nutrition Security Assessments, WFP Emergency Needs Assessment Service, July 2007

Table 2.7: Anthropometric and clinical indicators of the nutrition situation				
Type of malnutrition	Indicators at the individual level	Public health significance		
Undernutrition	Wasting: Iow weight for height (W/H) → Global acute malnutrition (GAM): W/H < -2 z-scores → Severe acute malnutrition: W/H < -3 z-scores → Moderate acute malnutrition: W/H between -3 and -2 z-scores	Benchmarks of prevalence at the population level (WHO): GAM < 5%: acceptable GAM 5-9%: poor GAM 10-14%: serious $GAM \ge 15\%$: critical • Increased risk of morbidity • Increased risk of morbidity		
	Stunting: low height for age (H/A) → Global chronic malnutrition: H/A < -2 z-scores → Severe chronic malnutrition: H/A < -3 z-scores → Moderate chronic malnutrition: H/A between -3 and -2 z-scores	Benchmarks of prevalence at the population level (WHO): Stunting < 20%: acceptable Stunting 20-29%: poor Stunting $30-39\%$: serious Stunting $\geq 40\%$: critical • Increased risk of morbidity • Increased risk of morbidity • Decreased performance at school		
	Underweight: low weight for age (W/A), combining wasting and stunting → Global underweight: W/A < -2 z-scores → Severe underweight: W/A < -3 z-scores → Moderate underweight: W/A between -3 and -2 z-scores	Benchmarks of prevalence at the population level (WHO): Underweight < 10%: acceptable Underweight 10-19%: poor Underweight 20-29%: serious Underweight ≥ 30%: critical • Increased risk of morbidity		
	Underweight: MUAC • In children: → Global: MUAC < 12.5 cm → Severe: MUAC < 11.0 cm → Moderate: MUAC 11–12.5 cm • In women: → Global: MUAC < 22.5 cm → Severe: MUAC < 21 cm → Moderate: MUAC 21–22.5 cm	 In children: Increased risk of mortality In women: Increased risk of low birthweight babies 		
	BMI in adults: W/H ² → Severe: BMI < 16.0 → Moderate: BMI 16-16.9 → Mild: BMI 17-18.4 → Normal: BMI 18.5-24.9	 Benchmarks of prevalence at the population level (WHO): BMI below 18.5 for 5-9%: low BMI below 18.5 for 10-19%: mild BMI below 18.5 for ≥ 40%: very high BMI below 18.5 for ≥ 40%: very high For women: Increased risk of low birthweight babies For all adults: Increased risk of mortality with very low BMI 		

(...cont)

Type of malnutrition	Indicators at the individual level	Public health significance
Undernutrition	 Anaemia: low blood haemoglobin → Standard thresholds available for adults and children 	 For women: Increased risk of: mortality when giving birth low birthweight babies For children: increased risk of stunting decreased performance at school For all: decreased physical capacity decreased resistance to disease
	Vitamin A deficiency: low serum retinol → Standard thresholds available for adults and children	 Decreased resistance to disease Impaired or loss of vision
	Iodine deficiency: low urine iodine → Standard thresholds available	 Increased risk of mental and physical disabilities Decreased performance at school
Overnutrition and obesity	BMI in adults: W/H² → Overweight: BMI 25-29.9 → Obese: BMI ≥ 30	 Increased risk of chronic diseases: diabetes, cancer, hypertension Increased risk of mortality

Additional data and indicators complement the indicators in Table 2.7 and provide further insights into nutrition status (see Table 2.8). These indicators serve two functions:

- 1. They provide possible explanations of the nutrition problems identified through the indicators in Table 2.7.
- 2. They act as proxies for the indicators in Table 2.7. If anthropometric and clinical information is not available, or measurements cannot be taken directly, proxy indicators can help determine whether or not the nutrition situation is hazardous.

Table 2.8: Examples of additional indicators that inform about nutrition status		
Category	Indicator	
Water access	 Quality: potable/non-potable, treated/untreated Quantity: litres per person per day Distance to water source Time taken for round trip to collect water 	
Water usage	Storage capacity in house: litresType of storage: covered/uncovered	
Sanitation	Type of sanitation used: household latrine, communal latrine, etc.Hand washing: always, sometimes, never	
Health status	 Prevalence of infectious disease: i.e. percentage of children who have been sick over the previous 2 weeks Prevalence of chronic diseases Trends in infectious and chronic diseases: seasonal and long-term 	
Health care	 Nearest staffed and equipped clinic or hospital: distance and time to reach it Presence of emergency health services: government, United Nations, NGO or other Immunization coverage, particularly measles 	
Health practices	Food handling practices: hygienic/unhygienicExtent to which people seek professional health care when sick	
Care	 Feeding practices: breastfeeding, complementary feeding, etc. Age and education level of child carers, i.e. mother Personal hygiene of children and their carers: acceptable/risky Relationship between children and their carers Relationship between heads of household and children Other occupations undertaken by carers: casual labour, collection of water, etc. 	

It is difficult to analyse nutrition status using **qualitative data**. *Only highly experienced staff such as nutritionists or other health experts* may be able to make useful qualitative observations based on people's physical aspect or clinical status, including the extent of emaciation or weight loss and signs of micronutrient deficiency, such as scurvy, pellagra and night blindness. Such observations cannot be extrapolated to the wider population, but can be used as triggers for implementing a proper nutrition survey, including anthropometric measurements.

Qualitative information about health and nutrition status can also be obtained from key informants. For example, a focus group of mothers of young children might provide:

- descriptions of the illnesses and symptoms that affect children, and comparison with the past, particularly the same season in the previous year;
- information about the disease cycle using a historical timeline (see Part III, Chapter 4): major events – seasons, natural and human-induced disasters, etc.
 – are plotted on a timeline and disease outbreaks are inserted.

Information from focus groups can be shared with nutritionists and other health experts who may be able to develop hypotheses about nutrition problems and their causes, and to recommend a course of action, such as an anthropometric survey.

7.3 Food security status indicators

An EFSA should employ indicators of household food security status. These indicators will enable assessment of the *current* quality and quantity of food consumption, and of the household's access to adequate food and nutrition. Food security is a broad concept encompassing many factors, and there are currently no internationally recognized qualitative or quantitative indicators for most of these factors. There are however established quantitative and qualitative methods for obtaining reliable information on household food security status. Food consumption indicators, food access indicators and the CSI can all be considered proxy indicators for food security.

7.3.1 Food consumption indicators

Food consumption indicators are designed to reflect the quantity and/or quality of people's diets. In EFSAs, the most commonly used food consumption indicator is the **food consumption score (FCS)**. This is a proxy indicator that represents the dietary diversity, energy and macro and micro (content) value of the food that people eat. It is based on dietary diversity – the number of food groups a household consumes over a reference period; food frequency – the number of days on which a particular food group is consumed over a reference period, usually measured in days; and the relative nutritional importance of different food groups. The FCS is calculated from the types of foods and the frequencies with which they are consumed during a seven-day period.

Although it provides essential information on people's current diet, the FCS is of limited value for in-depth analysis of food consumption patterns, for the following reasons:

- It is based on a seven-day recall period only. This is insufficient for a full analysis of food consumption for longer periods, which is likely to vary by season, for example.
- It provides no indication of the quantity of each foodstuff consumed.
- It does not give information on intra-household food consumption, such as who eats first and last.
- It does not show how food consumption has changed as a result of the crisis, unless previous FCS for the same types of household are available.

More information is needed if food consumption practices and trends are to be fully understood. For example, questions on usual food consumption should complement the seven-day household FCS. The calculation of the FCS is explained in Box 2.6¹⁷ and Example 2.3.

Box 2.6: Calculation of the FCS

In the household questionnaire

Households are asked to recall the **foods that they consumed in the previous seven days** (see the list of items in Table 2.9). Each item is given a score of 0 to 7, depending on the number of days on which it was consumed. For example:

- if potatoes were eaten on three of the last seven days, they are given a frequency score of 3;
- if potatoes were eaten on three of the last seven days, even if they were eaten twice on each of those days, at two meals, they are still given a frequency score of 3.

In the analysis

Food items are listed according to **food groups** (see Table 2.9), and the frequencies of all the food items surveyed in each food group are summed. Any summed food group frequency value over 7 is recoded as 7.

Each food group is assigned a **weight** (see Table 2.9 and its note), reflecting its **nutrient density**. For example:

- beans, peas, groundnuts and cashew nuts are given a weight of 3, reflecting the high protein content of beans and peas and the high fat content of nuts;
- sugar is given a weight of 0.5, reflecting its absence of micronutrients and the fact that it is usually eaten in relatively small quantities.

The household FCS is calculated for each household by multiplying each food group frequency by each food group weight, and then summing these scores into one composite score.

The household score can have a maximum value of 112, implying that each of the food groups was consumed every day for the last seven days.

The household score is compared with pre-established **thresholds** that indicate the status of the household's food consumption. WFP applies the following thresholds in a wide range of situations:

- poor food consumption: 0 to 21;
- borderline food consumption: 21.5 to 35;
- acceptable food consumption: > 35.

These thresholds can be adjusted if there is clear justification for doing so. For example, in some populations, consumption of sugar and/or oil may be frequent among nearly all households surveyed, even when consumption of other food groups is rare and the food score is otherwise low. In these cases, when the base diet of oil and sugar is combined with frequent (seven days) consumption of starch base only, the score already arrives at 21, but this clearly cannot be classified as even a borderline diet. The thresholds can therefore be raised from 21 and 35 to 28 and 42 – adding 7 to each threshold to account for the daily consumption of oil and sugar, which adds 7 points to the FCS.

When the overall population's consumption of oil and sugar is high, the FSC thresholds should be changed to:

- poor food consumption: 0 to 28;
- borderline food consumption: 28.5 to 42;
- acceptable food consumption: > 42.

^{17.} For further information about application of the FCS, see: Food Consumption Analysis – Calculation and use of the Food Consumption Score in food consumption and food security analysis, WFP Vulnerability Analysis and Mapping Branch, January 2008.

Table 2.9 provides a template for calculating the FCS. Example 2.3 gives an example of a completed template.

Table 2.9: Template for calculating the FCS				
Food item (examples)*	Food group	Weight (A)	Days eaten in past 7 days (B)	Score A x B
Maize, rice, sorghum, millet, bread and other cereals	Cereals and tubers	2		
Cassava, potatoes and sweet potatoes	and tubers			
Beans, peas, groundnuts and cashew nuts	Pulses	3		
Vegetables, relish and leaves	Vegetables	1		
Fruits	Fruit	1		
Beef, goat, poultry, pork, eggs and fish	Meat and fish	4		
Milk, yoghurt and other dairy products	Milk	4		
Sugar and sugar products	Sugar	0,5		
Oils, fats and butter	Oil	0,5		
			Composite score	

* Food items relevant to the context should be inserted.

Example 2.3: A completed FCS template				
Food item	Food group	Weight (A)	Days eaten in past 7 days (B)	Score A x B
Maize, rice, sorghum, millet, bread and other cereals	Cereals and tubers	2	7	14
Cassava, potatoes and sweet potatoes	and tubers			
Beans, peas, groundnuts and cashew nuts	Pulses	3	1	3
Vegetables, relish and leaves	Vegetables	1	2	2
Fruits	Fruit	1	0	0
Beef, goat, poultry, pork, eggs and fish	Meat and fish	4	0	0
Milk, yoghurt and other dairy products	Milk	4	1	4
Sugar and sugar products	Sugar	0,5	4	2
Oils, fats and butter	Oil	0,5	2	1
			Composite score	26

The FCS is a continuous variable, so standard statistics such as the mean and variance can be calculated, and trends of means over time and across categories determined. Frequencies and cross-tabulations can be determined for food consumption groups. Example 2.4 shows a trend analysis for the FCS of non-beneficiaries and beneficiaries.



Example 2.4: A trend analysis for food consumption scores

Box 2.7: Validation of the FCS

Recent research by the International Food Policy Research Institute (IFPRI) attempted to validate the use of the FCS for classifying household food security status, based on survey data from three countries – Burundi, Haiti and Sri Lanka.

The study found the usefulness of the dietary diversity and food frequency indicators encouraging. There are positive and statistically significant associations with calorie consumption per capita, particularly when small quantities are excluded from food frequencies. However, the cut-off points currently used by WFP to define poor, borderline and adequate food consumption groups correspond with energy intake that is considerably below the usual average 2,100 kcal/capita/day benchmark used to define undernourishment. Hence, the poor food consumption group corresponds with extreme undernourishment, and some households in the acceptable food consumption group have consumption below 2,100 kcal/capita/day.

Table 2.10: Food consumption groups and corresponding FCS thresholds and energy intake levels				
Food consumption	Burundi		Haiti	
group	FCS	Corresponding energy consumption in kcal/capita/day	FCS	Corresponding energy consumption in kcal/capita/day
Poor Borderline Acceptable	≤ 23 > 23 and ≤ 37 > 37	≤ 1 550 > 1 550 and ≤ 1 800 > 1 800	≤ 28 > 28 and ≤ 42 > 42	≤ 1 600 > 1 600 and ≤ 1 900 > 1 900

These data reinforce the notion of context specificity in formulating FCS. In EFSAs, it is recommended that the thresholds illustrated in Box 2.6 be used.

7.3.2 Food access indicators

Food access is a measure of a household's ability to acquire available food over a given period. People's access to food varies widely among and within areas; it is therefore impossible to define a single standard food access indicator that can be used in all situations. Food access indicators must be tailored to the livelihood strategies employed by the population of the area in which the EFSA takes place.

Food access indicators as measures of individuals' ability to acquire food are associated with livelihood activities, sources of food consumption, food stocks, food- or income-related coping strategies, asset wealth and expenditures.

Knowledge of livelihoods is used to identify the food access strategies of different livelihood groups.¹⁸ For example:

- farming households might obtain their food from a combination of own production and purchases, using income from sales of their produce;
- pastoral households might consume animal products, such as milk, and sell animals to buy grain and other necessities;
- labouring households might buy all of their food at the market, using money that they have earned from a variety of jobs.

^{18.} A livelihood group is a group of people who share the same basic means of livelihood and lifestyles – the same main subsistence and income activities, and the same social and cultural practices – and face similar risks of food and nutrition insecurity.

An example of defining food access indicators based on livelihood strategies is given in Example 2.5.

Example 2.5: Definition of food access indicators associated with livelihoods

In a certain area, some households depend on daily labour for their income. They use the money earned to buy food and other items at the market. All of their food is bought at the market.

Appropriate food access indicators in this case should reflect the purchasing power of households based on:

- prices of key commodities;
- wage rates;
- frequency with which labourers can find work.

Table 2.11 gives other examples of food access indicators and the circumstances in which they might be used.

Table 2.11: Examples of food access indicators			
Category	Indicator	Explanation	Circumstances and comments
Food consumption	Sources of food and income	Identifies the reliability and sustainability of food and income sources	All types of emergency
Food consumption	Consumption of "famine foods"	Within a society, some foodstuffs may be consumed only during periods of food insecurity; regular consumption of these indicates that there is a problem	Slow-onset emergencies that have reached a critical stage
Food stocks	Diversity of food products available	The variety of food items that are available and accessible will, in part, determine the quality of the diet	Stocks may be limited by production failures, transportation blockages, embargoes

(...cont)

(cont) Category	Indicator	Explanation	Circumstances and comments
Food stocks	Food self-sufficiency	Duration of staple food stocks for household's own consumption	Where agricultural production is an essential livelihood activity and food purchases are constrained by lack of access to markets or lack of income
Income/ livelihood	Purchasing power	Comparison of household income or expenditure with the minimum cost of living, for food and other essential expenditures	All types of emergency, but it is often difficult to obtain accurate estimates of income or expenditure, and costs of living may vary
Income/ livelihood	Remittances	The size and frequency of transfers from migrants may represent an important source of income for households, but it is usually difficult to quantify remittances. The location of the person sending the remittance and her/his relationship to the household can sometimes act as proxy indicators	Situations in which significant numbers of people travel out of the area to find work
Expenditures	Terms of trade ¹⁹	Comparison of, for example: • cost of staple food with daily wage rate • price of livestock with price of cereal Relevant terms of trade need to be defined for each livelihood group	Any emergency in which economic exchange is significant: i.e. most emergencies
Asset wealth	Asset ownership	Ownership of productive assets that facilitate food and income generation, i.e. land, animals, skills. Relevant assets are identified from knowledge of livelihood groups and pre- and post- emergency economic activities	Any emergency, but depends on having a good knowledge of local livelihoods
Coping strategies	Food-related	The different coping strategies (behaviours) adopted by households in an emergency are used to estimate the severity of food insecurity	Some behaviours are highly reversible, others are not

^{19.} Market Analysis Tool: Terms of Trade, WFP Economic Analysis Unit, August 2007; Market Analysis in Emergency Food Security Assessments: Guidelines on Market Situation Analysis and Forecast and Response Protocol, WFP Emergency Needs Assessment Service, September 2007.

Examples of food access indicators are given in Example 2.6. Food access indicators should always be defined according to the economic context.

Example 2.6: Sample applications of food access indicators

Sources of food and income

- A household acquires most of its food from the relief assistance provided by an
 international humanitarian organization. This source is considered poor, as it is
 unreliable and unsustainable. The household has no income, as it has recently moved
 to a camp for displaced people. Its income source is therefore also considered poor.
 The combination of poor food sources with poor income sources leads to the
 conclusion that the household has poor food access.
- Another household is receiving most of its food from relief; it too has poor food sources. However, this household retains access to some of its fields and is able to harvest and sell some cash crops. In addition, the household receives regular remittances from a relative working in the capital. This household's income sources are good. The combination of poor food sources with good income sources leads to the conclusion that this household has average food access.

An illustration of the development of this indicator is given in Example 2.7.

Consumption of famine foods

- Households in a certain area acquire food from cultivation and market purchases. Some local wild plants are nutritious but are not usually consumed because they taste bad and indicate that a household cannot obtain food through the normal channels, making these plants socially unacceptable as food. An EFSA reveals that an increasing number of households are consuming these plants; this indicates that access to normal foods is declining.
- In an urban environment there is stigma about using government soup kitchens. Households that use soup kitchens are considered to have failed. An EFSA reveals that an increasing number of households are using soup kitchens.

It may be difficult to obtain accurate data about these strategies because people are reluctant to admit that they are using them. Data should be cross-checked, for example, by reviewing attendance records at soup kitchens.

Access to natural resources

A pastoral community depends on cattle for its consumption of milk products and for sales to buy food and other essential items. Conflict has reduced mobility and the amount of pasture available to the pastoralists, who are forced to cut the sizes of their herds by selling animals. The price of cattle decreases and the pastoralists' purchasing power declines, as does their direct access to milk products. Food access for the community as a whole has, therefore, deteriorated.

Purchasing power

 Market surveys indicate that the minimum per capita cost of living is US\$60 per month. In a household survey, informants are asked to describe their income sources and the monthly income that they receive from each. The results are compared with the US\$60 minimum cost of living.

It can be extremely difficult to obtain accurate data on income and expenditure, particularly when much of the economy is informal and people derive their incomes from multiple sources. Informants frequently underestimate their incomes and expenditures, either because they do not know how much they earn and spend in a month – these figures are variable – or because they are reluctant to reveal such information to a stranger. The survey may also miss crucial non-cash components of household income, such as in-kind gifts. This indicator should therefore be used with caution.

(cont...)

(...cont)

Terms of trade

- Households depend on rice purchased at market as their staple food. The livelihoods of the poorest people are based on casual labour. Workers earn an average of US\$60 per month. If rice costs US\$0.5 per kilogram, a worker's monthly salary equates to 120 kg of rice.
- If this indicator is monitored over time, trends in food access can be ascertained. For example, if the rice price doubles to US\$1 per kilogram and wages remain constant, a worker's salary will equate to 60 kg of rice, indicating a sharp decrease in food access.
- A minimum rice/wage ratio can be determined below which household food access is considered to be insufficient. This ratio will depend on the amount of rice that an average household requires and the other expenses that must be covered from the wages, among other factors.

Food self-sufficiency

In a community, households keep about half of their harvested crops for their own consumption. This does not cover all of their food needs, so the households must have a source of income to buy food and other necessary items. Income comes partly from the sale of crops and partly from other activities, such as fishing and livestock sales. To estimate food access, the post-harvest duration of food stocks is compared with the reliability of the main source of income. Food stocks in this example are classified as follows:

- poor: up to two months household food supply;
- average: three to seven months household supply;
- good: more than seven months household supply.

Income sources are also rated as poor, average and good. The two variables are then combined. For example, a household with poor food stocks and poor income sources is considered to have poor food access. A household with good food stocks but poor income sources is considered to have average food access.

Asset ownership

- In an agricultural area, access to land may be the primary determinant of food access. A suitable indicator might be area of land per household member.
- Displaced people who have lost most of their possessions depend on their skills for their livelihoods. In this case, education level or professional skills might therefore be used as indicators of human assets.
- Access to assets may vary according to gender. Female-headed households may be at a disadvantage compared with male-headed households.

Remittances²⁰

An area is affected by a drought – a slow-onset emergency – which has led to the widespread loss of livelihood assets. It is known that people with close relatives working in other parts of the country or abroad are better off, as they generally receive some remittances. Although it is difficult to gain accurate data about the size of remittances, the location of the migrant worker gives an indication of the significance of this source of income for the household. For example, a relative in:

- the district capital, seeking casual labour, implies a minor enhancement of household food access;
- mines in another part of the country implies a medium-level enhancement of household food access;
- the Gulf States, working on oil installations, implies a major enhancement of household food access.

^{20.} For more information, see: Technical Guidance Sheet No. 1 Integrating Migration and Displacement into Emergency Food Security Assessments, WFP Emergency Needs Assessment Service, May 2007; Remittances during crises: implications for humanitarian response, K. Savage and P. Harvey, eds., Humanitarian Policy Group Report No. 25, Overseas Development Institute (ODI), May 2007.

Markets are of critical significance to food access in most situations. Many of the food access indicators described in Table 2.11 are based on market interactions, so it is essential to have indicators illustrating the ways in which markets function.

Common market indicators are shown in Table 2.12. Some of these may not be appropriate in every situation, and additional market indicators can be added, according to the context.²¹

Many of the indicators should be reviewed over time, through comparison with the same period in previous years and with trends over recent weeks or months. If baseline surveys or previous assessments are not available, market traders can usually provide reliable information about trends and the reasons for them.

Table 2.12: Market indicators			
Households' interaction with markets Sources: household survey, focus group discussion, key informant interview			
Own production sold: staple food, cash crops, livestock, livestock products	 Quantity sold Proportion of own production sold Price obtained Reasons for selling 		
Food bought from market: disaggregated according to different foods	 Quantity per week/month Proportion of total household food consumption Price Seasonal variation Access to credit from traders 		
Participation in labour market	 Household members involved in casual or seasonal labour Access for men and women to markets Type of work and season(s) Daily wage rate(s) Proportion of annual income from this source 		
Market functioning Source: Market trader interview			
Prices of key commodities: staple foods, cash crops, livestock, fuel, etc.	 Main commodities available Prices now and at same time last year Margins now and at same time last year Price variation and trends over recent weeks or months Impact of food aid on prices Perception of future evolution of prices 		
Sources of key commodities	Local, other parts of country or importedProblems with movement of commodities		
Trade volume	 Quantity of commodities sold, and seasonal variation Variation in supply and demand of key commodities over recent weeks/months Speed of response in case of changes in supply or demand Trends over recent weeks/months Reasons for trends 		

21. For more guidance on market analysis see: Market Analysis in Emergency Food Security Assessments, WFP Emergency Needs Assessment Service, August 2007. (cont...)
| Terms of trade | Staple food/livestockCash crop/cerealLabour wage rate/cereal |
|--|---|
| Labour market | Number of people seeking work compared with number finding work each day Variation in supply and demand for labour, according to season Daily wage rate and seasonal variation |
| Traders | Number of traders and trend over recent weeks/months Access to credit Size of stocks of key commodities Impact of food aid on willingness to trade Transaction costs: transportation, taxes, etc. |
| District/national level
Sources: Secondary data | review, key informant interview |
| Characteristics
of markets | Location Wholesale, retail, etc. Areas covered by markets Distance between markets Frequency of markets |
| Consumer price index | Recent trends, disaggregated as far as possible by district |
| Main trade routes | Commodities traded In-country and international trade routes Ease of movement of commodities: physical, administrative |
| Price variation
among markets | Prices in different marketsTransaction costs |
| Proportion of
country's / district's
food imported | Percentage |
| Exchange rate | Fluctuation and impact on imports/exports |
| National/district data | Inflation Poverty rate Per capita trends in gross national income (GNI) and gross domestic product (GDP) Unemployment rate Interest rates |
| Policy | Significant changes in trade policy |

Examples of using market indicators to develop food access indicators are given in Example 2.7.

(...cont)

Example 2.7: Using market indicators to define food access indicators

For each of the following three food access indicators, examples of market data that would be incorporated in the indicator are given. Other useful market data should be determined according to the context.

Food access indicator: sources of food and income

- Price stability: Are the prices of food in the market and the prices paid to producers stable?
- Food sources: Where does the food in the market come from? If it is imported internally or from abroad – how reliable is the supply?
- Labour market: How many days per month can a casual labourer expect to find work? Is this stable?

Food access indicator: purchasing power

Price stability: Is the cost of essential food and non-food items increasing, decreasing
or remaining stable in relation to normal for this time of year?

Food access indicator: terms of trade

- Staple food/labour: The cost of staple food is monitored at the market and compared with the average monthly wage of a casual labourer. This indicator is used to estimate the status of food access for livelihood groups that depend primarily on casual labour.
- Livestock prices/cereal prices: A decline of livestock prices against cereal prices has proved to be a strong indicator of deteriorating food access and general food security in pastoral communities.

7.3.3 Description of the current household food security situation

A key part of the description of household food security in an EFSA or a CSFVA is derived from a short-term household food security classification. This is based on the household's current food consumption as a proxy for its current food security. It gives a snapshot picture of the household's situation at the time the data are collected.

This is an essential step in both the EFSA and CFSVA processes and is the starting point for situation analysis and scenario-building exercises (see **Part IV, Section 4.2**).

Households are classified according to the FCS – poor, borderline or acceptable. For some households, the FCS may not reveal their current food security situation. In such cases, information about household access to and sources of food is crucial in allowing these households to be reclassified.

Description of the current household food security is therefore based on the FCS and its thresholds, as described in Box 2.6. This usually²² means that households with an FCS of 21 or less have poor food security, those with an FCS between 21.5 and 35 have borderline food insecurity, and those with an FCS of more than 35 have acceptable food security.

^{22.} These thresholds can be increased by 7 points each, as described in Box 2.6.

The household food consumption classification serves as a standardized, objective and replicable tool for describing **short-term food security**. This classification can be standardized by using household FCS as the basis for comparison. Although differences in context must be considered when interpreting the FCS, this method of standardization is acceptable because the FCS is well-defined²³ and objectively measurable. When FCS thresholds are appropriately defined, the resulting food consumption groups match the corresponding levels of food intake, to a certain extent (see Box 2.7).

The short-term household food security description may need to be adjusted if the FCS does not properly reflect the food security situation of the moment. This is the case for households with unsustainable sources of food, or with food access strategies that are uncertain, damaging to their future livelihoods or so severe that they endanger the health of household members. Typical examples are food aid recipients, who may be benefiting from acceptable food consumption at the time of the assessment, but who would probably have poor food security without that food.

The use of the FCS-based classification to describe current food security should be triangulated with other food security indicators, such as the CSI and income and production indicators.

In an EFSA, the description of current food security should always go beyond this method. A complete situation analysis should also include statements about the evolution of the overall food security context and about critical livelihood factors related to resilience, coping mechanisms and how income generation and food production will define future access to food.

The description of the current household food security situation therefore serves only as a starting point and for reference. **Part IV, Chapters 3 and 4**, on conducting situation and forecast analysis, go beyond this snapshot analysis and include projections for the future.

The situation analysis starts from the current household food security description, which is based on the FCS complemented by a livelihoods analysis to make it a true, forward-looking food security classification. The analyst decides what adjustments need to be made, based on other food security indicators and a livelihoods analysis, and concentrating on the outlook for households' access to food in the near or more distant future. Ideally, the analyst should draw on a combination of quantitative indicators and qualitative information. The context-specific indicators used for household food security classification are similar to those identified in Table 2.11:

^{23.} It will often be necessary to design a country-specific questionnaire, to ensure that household food consumption is evaluated in a way that is appropriate to the local context.

- income sources, unsustainable or harmful coping strategies, debt, distress indicators;
- production, stocks, reserves;
- food sources, including aggregate food supply; and
- asset ownership, access to natural resources.

Analysis based on household food consumption alone should therefore **not** serve as a simplistic approach to targeting food assistance during programme implementation. The description of current household food insecurity does not automatically equate to food assistance requirements: not everyone with poor food consumption at the time of the data collection will need assistance, and some households with currently good consumption may need assistance later. To define assistance, it is essential to have a good understanding of how households obtain access to food, their livelihoods, the effects of shocks on these, and the macro trends for the future.

7.3.4 Coping strategy indicators

The **coping strategy index (CSI)** is often used as a proxy indicator for food security. Its elements can be used to analyse the structure of coping strategies.²⁴ The index is based on the many possible answers to the question: "In the past seven days, if there have been times when you did not have enough food or enough money to buy food, how many days has your household had to…"

A summary of the procedure for establishing the CSI is given in Box 2.8.

Box 2.8: Process for establishing the CSI

a) **The specific community's** usual food-based coping strategies are recorded from focus group and key informant interviews.

b) Local key informants assign a **weight** to each coping strategy, based on the **severity** of the circumstances under which it is used. For example, a slight reduction in food consumption by adults might be a response to short-term food insecurity entailing no major problems in the long term. On the other hand, the selling of prime productive assets, such as livestock or machinery, might indicate an extreme level of food insecurity.

c) During the field survey, the **current** food-based coping strategies that people use and the frequency with which they use each strategy are established.

d) For each household, a score is given to each coping strategy: Score = (frequency with which coping strategy is used) x (weight).

(cont...)

^{24.} Detailed guidance on the CSI is given in The Coping Strategies Index – Field Methods Manual, second edition, CARE, Feinstein International Center, Tango, United States Agency for International Development (USAID), WFP, January 2008

(...cont)

insecurity.

e) The scores for each coping strategy are added together to give a composite score for each household.

A household's composite score is meaningless unless it is compared with some other factor:

- Comparing the scores of different households at the same time gives an indication
 of their relative food security status; for example, household X is more severely foodinsecure than household Y.
- Comparing the scores of the same household, or group of households, over time gives a useful indication of the food security trend: improving, deteriorating or stable.
 The composite score can also be calibrated against other food security indicators. For example, if a score of 95 correlates directly with severe food insecurity as established by other reputable means, this score can be used in the future to indicate severe food

An example of calculating the CSI is given in Example 2.8, taken from a study in Kenya cited in the Cooperative for Assistance and Relief Everywhere (CARE) and WFP Coping Strategies Index, Field Methods Manual.

Example 2.8: Calculating the CSI								
In the past 30 days, as a result of not having enough food, how often has your household had to:	All the time/ every day	Fairly often/ 3-6 times per week	Occasionally/ 1-2 times per week	Rarely/ less than once a week	Never	Raw score	Severity weight	Score = relative frequency x weight
Relative frequency score	7	4.5	1.5	0.5	0			
a. Rely on less preferred and less expensive foods?		х				4.5	2	9.0
b. Borrow food, or rely on help from a friend or relative?			Х			1.5	4	6.0
c. Purchase food on credit?			Х			1.5	4	6.0
d. Gather wild food, hunt, or harvest immature crops?					Х	0	8	0
e. Consume seed stock held for next season?					х	0	6	0
f. Send household members to eat elsewhere?				Х		0,5	4	2.0
g. Send household members to beg?					Х	0	8	0
h. Limit portion sizes at meal times?	Х					7	2	14.0
i. Restrict adults' consumption so that children can eat?			Х			1.5	6	9.0
j. Feed working household members at the expense of non-working members?					Х	0	4	0
k. Ration the money available and buy prepared food?					Х	0	N.A.	-
I. Reduce number of meals eaten in a day?		Х				4.5	2	9.0
m. Pass entire days without eating?					Х	0	8	0
Total household score								55.0

As noted in Box 2.8, the CSI provides a score for each household, which in Example 2.8 is 55.0. However, unless the significance of the score has been established through reliable calibration, or CSIs have been collected over time, this score alone does not explain much about the **absolute** level of food insecurity experienced by the household. Instead, it allows comparison of the **relative** food security of different households whose CSIs were calculated during the assessment.

This does not mean that the CSI should not be compiled; it is a useful reference for future assessments. The information about coping strategies and the circumstances in which they are employed is used to estimate the **absolute** level of food security, as explained in Box 2.9.

Box 2.9: Developing coping strategy indicators

A method for analysing coping strategies during an EFSA

This approach relies on the combination of qualitative and quantitative data collection:

- Baseline information about local coping strategies is collected through a focus group discussion.
- The information collected is used in the questionnaire design. This leads to the collection of quantitative data that are analysed as follows.

During a focus group interview in the community being assessed, the following questions are asked:

- In this community, what strategies do households adopt when they do not have enough food, or do not have enough money to buy food?
- Which groups within the community might adopt each strategy?
- Under what circumstances is each strategy adopted?

From this it is possible to deduce the food security status and the severity associated with each strategy.

A table is then compiled. The following table uses examples of coping strategies.

Strategy	Groups using the strategy	Implication
1. Purchase of less expensive food	All	Alert
2. Withdrawal of children from school	All	Risk to future livelihoods
3. Reduction of number of meals	All	
4. Migration of whole household to look for work	Landless households	
5. Selling land	Landowners	Risk to livelihoods

Based on this example, the focus group might agree to the following:

- Strategies 4 and 5 are adopted during periods of severe food insecurity.
- Strategies 2 and 3 are adopted when food insecurity is moderate or deteriorating, but not yet severe.
- Strategy 1 corresponds to a household that is not currently at risk, but whose situation must be monitored.

The frequency with which the various coping strategies are used can also be considered when discussing and interpreting the severity of the food security situation.

Having established the types of strategies that people *might* use, either qualitative or quantitative data are collected to determine the strategies that they are *currently* using. When quantitative data are used, questions about the coping strategies identified during the focus group are included in the questionnaire. The resultant data are analysed to determine which households are resorting to strategies that indicate moderate or severe food insecurity, according to the information provided by the focus group.

Recent research on the CSI has led to a reduced version being developed. The **reduced coping strategy index** (reduced CSI) compares food security across different contexts. It is a subset of the context-specific CSI, calculated on the basis of a specific set of behaviours each with its own universal severity weighting. The reduced index is less valuable in identifying the most vulnerable households in a location, but it is very useful for comparisons across crises or for geographical targeting because it measures the same set of behaviours and uses the same weights. The behaviours measured by the reduced CSI are:

- eating less preferred/expensive foods;
- borrowing food or relying on help from friends and relatives;
- limiting portion sizes at meal times;
- limiting adult intake so that small children can eat;
- reducing the number of meals per day.

Example 2.9: Calculating a reduced CSI				
In the past 7 days, if there have been times when you did not have enough food or money to buy food, how often has your household had to:	Raw score	Universal severity weight	Weighted score = frequency X weight	
Relative frequency score				
a. Rely on less preferred and less expensive foods?	5	1	5	
b. Borrow food, or rely on help from a friend or relative?	2	2	4	
c. Limit portion sizes at meal times?	7	1	7	
d. Restrict consumption by adults so that small children can eat?	2	3	6	
e. Reduce the number of meals eaten in a day?	5	1	5	
Total household score – reduced CSI		the totals strategy	27	

Part III Planning and implementing an EFSA

Contents of Part III

1	Introc	luction	84
2	Plann	ing an EFSA	85
	2.1	Trigger mechanisms for an EFSA	86
2.	1.1	Sudden-onset crises	86
2.	1.2	Slow-onset or protracted crises	87
	2.2	Objectives and terms of reference	87
	2.3	Budget	88
	2.4	Analysis plan and information requirements	88
	2.5	Collation and review of secondary data and reference material	88
	2.6	Methodology for primary data collection	92
	2.7	Sampling	95
	2.7.1	Sampling frame	95
	2.7.2	Types of sampling	96
	2.7.3	Purposive sampling	97
		3.1 Principles of purposive sampling	97
		3.2 Determining the sample size	98
		3.3 Selection of geographical zones	100
		3.4 Selection of locations within geographical zones	101
		3.5 Selection of households and individuals within locations	101
		3.6 Example of purposive sampling	102
		3.7 Snowball sampling	103
		3.8 Convenience sampling	103
	2.7.4	Random sampling	104
		4.1 Principles of random sampling	104
		4.2 Geographical stratification	105
		4.3 Determining the sample size	106 107
		4.4 Selection of locations/clusters within geographical zones4.5 Selection of households or individuals within locations	107
		4.6 Example of random sampling	111
	2.7.	Human resources	113
	2.9	Administration and logistics	115
	2.10	Communications, security and emergency procedures	115
	2.11	Briefing	116
	2.12	Assessment schedule	116
3	Field		118
Ĩ	3.1	Daily preparation	118
	3.2	Discussion with community leaders	118
	3.3	Collection of primary data	119
	3.4	Team meetings	119
	3.5	Final community meeting	119
	3.6	Daily analysis	119

4 Pr	imary data collection	120
4.1	The importance of gathering good quality primary data	120
4.	2 Approaches to primary data collection	122
4.	3 Undertaking primary data collection	125
4.	3.1 Formal questionnaires	125
	4.3.1.1 Conducting the interview	125
	4.3.1.2 Designing the questionnaire	126
4.	3.2 Semi-structured interviews	127
	4.3.2.1 Conducting the interview	127
	4.3.2.2 Designing the checklist	128
4.	3.3 Household surveys	129
4.	3.4 Community group discussions	129
4.	3.5 Focus group discussions	131
4.	3.6 Key informant interviews	133
4.	3.7 Observation	136
	4.3.7.1 Casual observation	136
	4.3.7.2 Observation within households	136
	4.3.7.3 Transect walks	136
	4.3.7.4 Structured observation	137
4.	4 Participatory tools for primary data collection	138
4.	4.1 Daily calendars	138
4.	4.2 Seasonal calendars	139
4.	4.3 Historical time lines	140
4.	4.4 Proportional piling	140
4.	4.5 Pair-wise ranking	141

Introduction

The three types of EFSA were defined in **Part I**. Information requirements and indicators for an EFSA were defined in **Part II**. **Part III** focuses on planning an assessment and collecting primary data. It addresses the following issues:

- planning an EFSA, outlining the steps undertaken prior to fieldwork;
- logistics considerations, regarding the management of fieldwork and daily activities;
- **primary data collection**, emphasizing the importance of good quality primary data and describing approaches to and tools for primary data collection.

The specific requirements for planning, logistics and primary data collection vary depending on whether the EFSA is to be an initial, rapid or in-depth assessment. This part of the handbook provides general guidance for an EFSA, but it is important to consider the different requirements for each type of assessment and to adjust planning, logistics, and other functions accordingly.

Planning an EFSA

Any assessment must be carefully planned. Activities should be identified in advance and a timeframe established.

Participation in assessment planning should be as broad as possible depending on the time available. Planning the EFSA in consultation with key partners ensures that knowledge of the situation is shared, transparency is enhanced and constraints are identified. It also makes it more likely that stakeholders accept and use the assessment findings in programme responses. Representatives of the national/local government and the organizations participating in the assessment should take part in the planning. Organizations with a stake in the assessment's outcome should also participate; these include organizations working in nutrition and food security sectors, and donors.

The amount of consultation depends on the urgency of the situation and the capacity of stakeholder organizations. For example, if detailed contingency plans have already been developed, extensive consultation may not be necessary. In a quick-onset crisis, common sense should be applied. The objectives of this type of assessment are usually obvious;²⁵ it is likely that only key stakeholders will be consulted rapidly, in person or by telephone, and a set of objectives drafted and circulated.

The key objectives of planning an EFSA include:

- 1. determining whether or not an assessment is required, assisted by trigger mechanisms;²⁶
- 2. defining the assessment objectives and terms of reference;
- 3. preparing the budget;
- 4. developing an analysis plan and identifying the information requirements;
- 5. reviewing secondary information and collecting reference material;

^{25.} For more guidance, see: Technical Guidance Sheet No. 6 Initial Emergency Food Security Assessments, WFP Emergency Needs Assessment Service, November 2007.

^{26.} For more guidance, see: Technical Guidance Sheet No. 4 Trigger Criteria for an Emergency Food Security Assessment in Slow-Onset Crisis Situations, WFP Emergency Needs Assessment Service, November 2007.

- 6. determining the data collection methodology;
- 7. determining the sampling approach;
- 8. establishing the staffing structure for the assessment, and training the teams;
- 9. making administrative and logistics arrangements;
- 10. establishing communications, security and emergency procedures;
- 11. briefing the assessment teams; and
- 12. preparing the assessment schedule.

The order in which these tasks are carried out is not fixed and the process is *iterative*. This means that completion of each task may lead to repetition and/or adaptation of other tasks; for example, consideration of logistics issues might lead to adaptation of the sampling approach.

2.1 Trigger mechanisms for an EFSA

An EFSA should be launched when trigger mechanisms show evidence that an assessment is needed. A *trigger* is an event or series of events indicating that the nutrition or food security situation is deteriorating, or has already reached a level of crisis. Trigger indicators are based on the data already collected by early warning and food security monitoring systems, and consist of a few indicators at the macroand micro-levels that signal the need for a closer look at the situation. Such indicators include the harvest levels of staple crops, the gap between harvests and food consumption requirements at the national or sub-national level, staple food market prices, terms of trade of staple foods against key assets such as livestock and labour, acute malnutrition rates, mortality rates, and context-specific coping strategies that indicate food stress.²⁷ If any of the data monitored give cause for concern, an initial assessment should be launched.

There are three types of EFSA: initial assessment, rapid assessment, and in-depth assessment (see **Part I**).

2.1.1 Sudden-onset crises

Sudden-onset crises are often the result of large-scale disasters such as earthquakes, floods and the outbreak of conflict. Typically, an **initial assessment** is carried out to provide a rough idea of the nature and scope of the crisis and, in urgent situations, the information on which the first stages of the response plan are based.²⁸ After the initial assessment, a follow-up **rapid assessment** is

^{27.} Details of an indicator-based monitoring system for triggering EFSAs in slow-onset crises are given in Technical Guidance Sheet No. 4 Trigger Criteria for an Emergency Food Security Assessment in Slow-Onset Crisis Situations, WFP Emergency Needs Assessment Service, November 2007.

^{28.} Extensive guidance on conducting this type of assessment is provided in Technical Guidance Sheet No. 6 Initial Emergency Food Security Assessments, WFP Emergency Needs Assessment Service, November 2007.

undertaken if more in-depth analysis is needed or the initial assessment information is out of date owing to rapid changes in the context.

A follow-up rapid assessment is not necessary if:

- there is compelling evidence that the crisis will have no impact on nutrition and food security;
- other agencies have conducted assessments that provide sufficient reliable information; or
- it is clear that the government or other organizations are capable of covering all the needs.

2.1.2 Slow-onset or protracted crises

Slow-onset or protracted crises include drought, economic or environmental decline, long-term conflict, and pandemics such as HIV/AIDS. **In-depth assessments** gather detailed, representative information about the affected population. The trigger for an EFSA is less clear than in rapid-onset crises because the situation deteriorates *gradually* and there may be no sudden and dramatic escalation that alerts attention. Frequent, large-scale EFSAs may be undesirable or unfeasible, so it is essential to monitor the deteriorating situation and take action before crisis status is reached.

Slow-onset food security crises should be monitored continually through as many different mechanisms as possible, including:

- field offices, where staff members monitor the situation during field visits and through discussions with stakeholders;
- partner organizations, through regular communication with government offices of health, agriculture, meteorology, etc. and organizations working in relevant sectors, such as specialized United Nations agencies and NGOs;
- formal monitoring systems that compare selected trigger indicators with preestablished baselines.

2.2 Objectives and terms of reference

The EFSA planning process is based on the assessment objectives and the terms of reference. These are defined in consultation with the intended users of the results, including country office decision-makers and partners:

- The objectives describe the outputs expected from the assessment (see Part I).
- The terms of reference indicate how the assessment will be carried out, and define the roles and responsibilities of the assessment team.

2.3 Budget

The budget is based on the planned data collection methods, sample methodology, number of assessment staff, and vehicles and other equipment required during the assessment schedule. These in turn depend on the assessment's objectives and terms of reference. Revisions to the assessment methodology may be necessary if the resulting budget exceeds the available funding.

The available budget for EFSA activities will determine the following:

- Professional profiles of participating staff: If the budget is tight, it may be
 possible to use only a few experienced staff members, which limits the scope of
 the assessment and the methodology used; experienced staff are needed to
 manage teams in the field, analyse data, etc.
- Logistics arrangements: Budget considerations will likely determine the amount and type of equipment available and the logistics arrangements for field activities.

WFP has established policies and procedures for funding at least a portion of assessment costs.²⁹ In some cases, such as a high-profile emergency, it may be possible to fund an assessment completely from external donor resources.

2.4 Analysis plan and information requirements

The analysis plan provides a framework for collecting information and interpreting findings based on the EFSA objectives. It specifies:

- the information needed to meet the objectives;
- how this information should be collected;
- the source(s) from which it should be collected.

Part II provides guidance on how to develop and use the analysis plan to identify information requirements.

2.5 Collation and review of secondary data and reference material

Secondary information includes any available contextual information about the area and the local population prior to the current crisis, and information on the current crisis collected by other organizations.

A thorough review of secondary information is crucial for defining assessment objectives and determining further information requirements. It also minimizes

^{29.} Policies and Procedures for the Use of the Immediate Response Account (IRA), WFP Operations Department and Administrative Department Joint Directive, 18 October 2005 (OD2005/005, A 2005/009).

Examples of second	dary data and sources are	given in Table 3.1.

Table 3.1: Secondary data sources			
Type of information	Example of sources		
Information about the area and its population before the current crisis			
Nutrition and food security baseline surveys	- CFSVA: WFP - Food economy baseline: Save the Children - Nutrition survey: Ministry of Health, UNICEF - FEWS NET		
Food production baseline surveys	- CFSAM: WFP/FAO - National government: e.g. Ministry of Agriculture		
Market surveys	Government, universities, World Bank, United Nations, NGOs		
Social, political, historical and anthropological reports	Universities, literature		
Information about	t the current crisis collected by other organizations		
Recent assessments	Government, other organizations: United Nations, NGOs		
Education baseline surveys	- Government statistics - United Nations, NGO reports - WFP baseline and follow-up surveys		
Media reports	Local and international news agencies		

If resources allow, one person or a small team should be assigned to go through secondary data and identify useful material. These individuals must be fully conversant with the assessment objectives and terms of reference.

The **reliability** of secondary sources must be carefully appraised in terms of the source, methodology used, potential bias, age of the information, relevance, and agreement with other sources. Once the secondary information has been reviewed, the remaining gaps constitute the information that will be collected directly as primary data during the EFSA.

A template for secondary information review and identification of gaps is shown in Table 3.2.

Table 3.2: Identification of information gaps, with examples					
Question	Information requirements	Secondary source 1	Secondary source 2	Secondary source	Ways to fill information gaps (with primary information)
Is there a food security or nutrition problem?	curity of excess 6 mo tion mortality? show		Doctor interviewed in local press says that mortality rate "seems to be declining"	Recent NGO rapid assessment indicates increased burial ceremonies in last 3 months	Collect data on number of deaths over last 6 months in sampled villages
	Is there evidence of acute malnutrition?	An NGO specialized in nutrition carried out an anthropometric survey in 3 of the 5 affected districts 3 weeks ago	Livelihoods and the emergency's impact in the other 2 districts seem to differ from those in the 3 surveyed districts		Undertake an anthropometric survey in the remaining 2 districts, using the same methodology as the NGO used
	Have people lost land or access to land?	Government economic data show increased land sales and decreased land prices in drought-affected areas	A market survey by the local university shows no increase in the number of people looking for casual labour opportunities	A report by the International Organization for Migration (IOM) indicates no unusual migration within or out of the affected area	Identify the people selling land. Are these crisis sales? If so, how are people compensating for their lost land assets?
	How do people currently obtain food?	A market survey by a local NGO indicates that demand for expensive foods such as meat has declined	A WFP rapid EFSA undertaken 3 months ago shows most households' food consumption measured – through the FCS – was acceptable	The local clinic reports increased micronutrient deficiency among children under 5	Evidence suggests that food access has deteriorated over the last 3 months. The EFSA will check the current situation and look for the causes of this

As Example 3.1 illustrates, the availability of reliable secondary data helps determine the types of primary data that must be collected during the EFSA.

Example 3.1: Using secondary data to determine primary information needs

Drought is leading to widespread crop failure and deteriorating livestock health.

One of the objectives of the EFSA is to estimate the impact of crop failure on food access. The following information is available from secondary sources:

- Source 1: A reliable and recent market survey shows that prices of staple crops have doubled since the same season last year, and prices of livestock have declined by one-third.
- Source 2: A livelihoods assessment from five years ago indicates that 25 percent of the population buy staple food using the proceeds of livestock sales, and 50 percent combine consumption of own production with sales of cash crops.
- Source 3: Key informants say that the relative proportions of livestock owners and farmers have changed over the last five years, but percentages cannot be reliably estimated.

Question	Information requirements	Source 1	Source 2	Source 3	Ways to fill information gaps
How has crop failure	Price of crops now	2X			
affected	Price of crops this time last year	Х*			
food access?	Price of livestock now	0.67Y			
	Price of livestock this time last year	Y**			
	Proportion of households buying food through sale of livestock		25%	Changed	Find current proportion and number of households
	Proportion of households buying food through production and sale of crops		50%	Changed	Find current proportion and number of households

This information is summarized in a table, as follows.

 X^* = price of crops this time last year.

 Y^{**} = price of livestock this time last year.

The table indicates the following:

- Accurate price data are available, because Source 1 is highly reliable, so it is not necessary to collect price data during the assessment.
- Data on the proportion and number of households utilizing each of the two livelihood strategies are out-of-date, so this information must be sought during the assessment.

Price information from secondary data is combined with information about livelihood strategies from primary *and* secondary data to estimate the impact of the crop failure on the two livelihood groups.

Secondary data also provide contextual information that is essential to the analytical process (see **Part IV**).

Reference material collected from secondary sources is also useful for planning an EFSA. Reference information includes the following:

- **Population data**: Accurate and up-to-date data on population numbers and locations are valuable when determining the approaches to information collection (see **Section 2.6**) and the sampling strategy (see **Section 2.7**). The sources and estimated accuracy of the information must be considered when planning fieldwork and reporting results.
- Maps: These are useful for designing the sampling approach, and planning and implementing the fieldwork. Any available information relevant to the emergency, such as population movements, damaged roads and airstrips, is plotted on the most up-to-date maps available of the affected area.
- **Crisis updates and bulletins**: In many emergencies, regular bulletins from the national government, the Office for the Coordination of Humanitarian Affairs (OCHA) and others provide the latest news about population movements, humanitarian operations, security, logistics constraints, etc.

Primary data are collected for most EFSAs, but occasionally secondary data are sufficiently comprehensive, reliable and up-to-date to cover all the information requirements. *In such cases, primary data collection is unnecessary*.

2.6 Methodology for primary data collection

The data collection methodology determines the ways in which primary data are collected during the assessment. The following are some of the critical issues that must be considered:

- Is this an initial, rapid or in-depth assessment?
- Will the assessment be based on standard questionnaires, semi-structured interviews or a combination of both?
- What sort of interviews will be used: household interviews, community group discussions, focus group discussions, key informant interviews, etc.?
- Will sampling follow a random, purposive or other approach?

Details of primary data collection methods and techniques are given in Chapter 4.

Practical constraints always affect the sampling approach and methods of primary data collection. The following are some of the most common constraints.

Security environment

 If security in the survey area is poor, the number of people and the time spent in the field should be limited. The sampling approach and data collection will be affected, as fewer households and locations can be visited. Instead of household interviews, focus groups might be used to collect information more quickly. If it is impossible to visit the affected area, the entire assessment must be based on secondary data and interviews with people who have recently come from the area.

Physical access

- Poor transportation and communication infrastructure may constrain field activities. Assessment team members should consider such factors as road conditions, alternative ways to travel, time taken to travel to and from affected areas, and whether or not teams will be able to maintain contact with field offices.
- If some areas are not accessible, it may be possible to interview individuals who have recently come from these areas, such as displaced people and people working for humanitarian organizations.

Available time and resources

- Programme priorities may dictate a specific period for the assessment. For instance, results of the assessment may be needed for preparing project proposals and appeals such as an emergency operation (EMOP), a protracted relief and recovery operation (PRRO), a flash appeal or a United Nations Consolidated Appeal. Partner organizations may also have specific information needs and deadlines.
- Resource constraints may prevent the country office and partners from sustaining a lengthy assessment, in which case support from the regional bureau or Headquarters may be required. It should be kept in mind that an assessment always places demands on the country office and draws resources for staff, administration, vehicles, etc. away from other activities.

Nature of the emergency

- In a quickly evolving emergency the assessment should focus on collecting essential data: What is the cause of risk? Which groups/people are affected? Where are they? Rapid information collection uses key informants, observation and purposive sampling. Quick assessments are undertaken until the situation stabilizes, at which point a more thorough assessment is carried out.
- In a slow-onset emergency, information needs are less urgent. More time can therefore be spent on developing a rigorous methodology and undertaking an in-depth survey using two-stage sampling.

EFSAs vary depending on the type of assessment and the context; some common scenarios and related approaches to data collection are presented in Example 3.2.

Example 3.2: Choice of data collection and sampling methodology for different scenarios

Rapid-onset emergencies

Initial assessment:

Scenario 1: The beginning of a sudden-onset emergency, such as an earthquake, population displacement, etc.

- It is an **emergency**. Lives are in danger. Information is needed within one day so that relief operations can be initiated.
- The crisis is **highly volatile**. The situation is constantly changing. Assessments must be updated daily.
- There is **little time**. Emergency health and feeding programmes may be needed immediately.
- The **area is small**. Most of the affected people can be reached quickly with available means of transport.
- All parts of the area are accessible.

Under this scenario, a snapshot assessment approach is followed:

- Key informants are the core source of information. They include local relief workers and people coming from the affected area who can explain the details and locations of problems. Quick visits to the most severely affected locations are undertaken, based on information from key informants. Observation and short interviews with people in these areas are carried out. Clinics, mortuaries, etc. provide the basis for injury and mortality estimates.
- Ideally, this type of assessment is updated at least once a day until the situation stabilizes (see next scenario).

Rapid assessment:

Scenario 2: A quick-onset emergency has stabilized, such as when mass population movement ceases and mortality and injury rates decline, but there are major fears about disease because of unsanitary living conditions.

- Although lives are not at the same level of risk as in Scenario 1, the situation is still **urgent**, as shelter and water supplies are inadequate and there is a constant risk of disease.
- The crisis is **less volatile**. Population movement has stopped and there is no imminent danger of further shocks.
- There is **little time**, but the reduced volatility of the situation makes it possible to be more rigorous with information collection than in Scenario 1.
- The area and accessibility remain the same as in Scenario 1.

The assessment approach is still rapid, but information is collected more rigorously than in Scenario 1:

- If the affected area is small and compact, such as a refugee camp, a town or a small group of villages, the assessment is completed within one to two days. In larger areas, more time is needed, and it is important to prioritize the areas to visit according to the impact of the emergency. Sampling is therefore mostly **purposive**.
- Key informants continue to provide an overview of what is happening. Relief services such as health, water and food have been established; staff members working on these projects are also consulted.
- Consultation with the affected people is more rigorous. If there are population lists, such as in a refugee camp, a simple random sample is taken (see Box 3.2) for household visits. If there are no population lists, the area is mapped with the help of key informants, and priority areas are identified. Households are identified within these areas, using one of the random sampling techniques explained in Section 2.7.3.5.
- Semi-structured interviews with households, focus groups and key informants may be combined with a questionnaire survey.

Slow-onset emergencies

In-depth assessment:

Scenario 3: A slow-onset emergency, such as a drought or long-term conflict.

 The crisis is neither urgent nor volatile. The situation has been slowly deteriorating for some time, and some factor has now triggered an assessment (see Section 2.1).

- There are **no time constraints**, but the assessment results are needed soon for programming.
- The area may be large or small.

Stratified two-stage sampling is applied (see Box 3.2): zones (strata) \rightarrow locations \rightarrow households. Where possible, random samples of locations and households within those locations are used. If random samples are not feasible, locations are selected purposively, and households within these are selected randomly or purposively. Both semi-structured interviews and questionnaires surveys are undertaken.

2.7 Sampling

In an EFSA, it is normally neither feasible nor desirable to survey every location and household affected by the emergency. A **sample** must therefore be drawn. A sample is a selection of households or individuals from the total affected population. The sample should represent the larger population and reduce the time and cost of data collection. If a sample is representative, generalizations about the total population can be extrapolated from the results of the **sample survey**.

It is extremely important that the sample be drawn in a methodologically rigorous way. This section explains the key terms used in sampling, and provides guidance on choosing the most appropriate sampling methodology for a given situation.

2.7.1 Sampling frame

The sampling frame represents the area and population that the assessment is intended to cover, for example, a region within a country or a particular population group, such as displaced people. The sampling frame must be defined at the start of the assessment planning process.

The sampling frame may cover only areas and groups **directly affected** by the emergency. Alternatively, it may also include **indirectly affected** areas and groups, where the impact on the population can be just as severe. These include the areas into which displaced people have moved;³⁰ host populations for displaced people; and areas suffering economically as a result of the emergency, such as those whose markets depend on produce from a drought-affected area.

^{30.} Technical Guidance Sheet No. 1 Integrating Migration and Displacement into Emergency Food Security Assessments, WFP Emergency Needs Assessment Service, May 2007.

The directly and indirectly affected areas and population groups are identified from secondary information and key informant interviews.

2.7.2 Types of sampling

The choice of sampling methodology depends on the time and resources available, the level of access and the specific objectives of the assessment. There are two major types of sampling approach, **probability sampling** and **non-probability sampling**. The following two sampling approaches are commonly used in EFSAs:

Purposive sampling (non-probability sampling): The researcher decides which
particular groups to interview. Non-probability sampling does not involve random
selection, so the results cannot be used to characterize the wider population. Its
value lies in selecting information-rich cases to gain a deeper understanding of
the situation when random sampling is not possible. The researcher selects what
she/he regards as representative sampling units, but the generalization of
findings from such a sample can always be contested.

Purposive sampling techniques are normally used in initial and rapid assessments for rapid-onset emergencies.

• Random sampling (probability sampling): All members of the population have a known, non-zero chance of being selected. Random sampling is based on formal statistical theory, which allows reliable estimates to be calculated and minimizes bias. Results can be extrapolated to the entire population with a degree of accuracy that depends on the sample size and the variability of the indicator.

Random sampling techniques are normally used for in-depth assessments, usually in slow-onset emergencies, and in emergency situations once conditions have stabilized.

Table 3.3 shows the circumstances under which each type of sampling is used.

Table 3.3: Circumstances under which each type of sampling is used				
Circumstances				
Initial and rapid assessments				
 Population data do not exist or are not reliable Access is limited to certain areas There is little time The situation is unstable and changing continuously Can be used for initial, rapid and in-depth EFSAs 				
 Very little information is available about the location of groups or individuals People are reluctant to be registered on a list Time is short In an initial EFSA it can provide background information for a subsequent rapid or in-depth EFSA 				
 Time or access is limited In an initial EFSA it can provide background information for a subsequent rapid or in-depth EFSA 				
In-depth assessments				
 Unbiased estimates with known precision are needed Population data exist All parts of the affected area are accessible The situation is reasonably stable There is sufficient time to visit all the selected households and carry out the required number of interviews Can be used for rapid and in-depth EFSAs in slow-onset emergencies or once an emergency situation has stabilized 				

2.7.3 Purposive sampling

This section presents methods for initial and rapid assessments using purposive sampling.

2.7.3.1 Principles of purposive sampling

Purposive sampling might be used when any of the following conditions apply:

- It is difficult to reach every area, household or individual member of the population.
- Reliable information about population locations and numbers is not available.
- There is insufficient time to visit the number of households or individuals needed for statistical analyses using a random sample. The most severely affected areas are prioritized, for example, agricultural areas during a drought, or villages inhabited by targeted groups during a conflict.
- There is strong evidence that nutrition and food security risks are concentrated in certain areas or population groups, such as particular livelihood groups or displaced people. In many cases, it is not possible to set up a sampling frame that allows these groups or areas to be stratified.

PART III

• The assessment objectives necessitate in-depth investigation of specific issues. For example, if the EFSA is expected to provide detailed information about the impact of market disruptions, groups of people who depend on the markets will be prioritized, such as traders, cash crop growers and people who rely on purchasing their food at markets.

Although the information collected through purposive sampling cannot be extrapolated to the entire population – as it can be in random sampling – generalizations can be extended to the wider population under the following circumstances:

- Triangulation is applied to all data (see **Part IV**). In triangulation, data are collected from numerous sources and their convergence, or otherwise, is appraised. If data from at least three reliable sources converge, the conclusions can be treated with reasonable confidence.
- Purposive sampling is combined with semi-structured interviews for primary data collection (see Section 4.3.2). This allows in-depth discussion and, consequently, a greater understanding of the crisis, its causes and effects. Conclusions from all the semi-structured interviews provide a sound basis for judging the extent to which they can be extrapolated.

It is not known how representative the sample is, so generalizations made through the purposive sampling approach are subjective (or biased), and confidence intervals for the estimates would be irrelevant.

Good purposive sampling depends on having a thorough knowledge of the context – the type of emergency and the characteristics of the population. Such knowledge is initially gained through secondary data review (see **Section 2.5**). If secondary data are incomplete or inaccurate, or if the situation is changing rapidly, sampling can be adjusted as the assessment progresses. For example, population groups and areas may be added or removed as more information about the crisis is obtained.

Purposive sampling can be combined with random sampling techniques. For example, households might be selected through random sampling in a location selected through purposive sampling. Note that this does not make the sample statistically representative, which requires that *the entire sampling process follow random sampling principles* (see **Section 2.7.4**).

2.7.3.2 Determining the sample size

There is no formula for setting the sample size for purposive sampling. Instead, judgements must be made, based on the expected heterogeneity of areas, population groups, locations, households and individuals. If heterogeneity is high and units are very different from each other, a large sample is needed, but the sample size also depends on the time and resources available. If heterogeneity is low and units are similar to each other, a smaller sample will suffice. This is illustrated in Example 3.3.

Example 3.3: Heterogeneous populations: implications for purposive sampling

An EFSA is undertaken in the urban and rural areas of a region affected by economic collapse. One of the objectives is to determine the impact of the crisis on livelihoods. The assessment team decides that to do this, they need to sample according to livelihood groups.

In the urban area, six main livelihood groups are identified:

- civil servants;
- traders;
- professional workers: doctors, engineers, teachers, etc.;
- salaried workers: shop assistants, storekeepers, construction supervisors, etc.;
- · casual labourers;
- unemployed/beggars.

To understand the situation, and taking the time constraints into account, the team decides that ten households must be interviewed from each livelihood group. This gives a total sample size for the urban area of $6 \times 10 = 60$ households.

In the rural area, three main livelihood groups are identified:

- farmers producing crops for consumption and sale;
- pastoralists living from the consumption and sale of animal products;
- landless farm workers.

Again, the team decides to interview ten households per livelihood group. The total sample size for the rural area is therefore $10 \times 3 = 30$ households.

If areas and population groups are heterogeneous, a separate sample size should be estimated for each. The **rule of thumb** explained in Box 3.1 can be used,

Box 3.1: Rule of thumb for estimating sample size in purposive sampling

As a rule of thumb based on empirical experience of household food security surveys, **between 50 and 150 households** per reporting domain can be included in a purposive sample, and the following guidance applied.

If locations are clearly very different from each other:

- divide them into groups of locations with similar characteristics;
- within each group, visit seven to fifteen locations;³¹
- within each location, interview seven to ten households or individuals.³²

If locations seem to resemble each other:

- check that there are no less obvious characteristics that may influence household food security (see Section 2.7.3.4), such as:
 - geographical dispersion of the locations, which could affect crop production patterns, access to markets and other services, and roads;
 - population size;

(cont...)

^{31.} However, whenever possible, random probability sampling is preferable to purposive sampling.

^{32.} Here too, random sampling should be considered if it fulfils the assessment requirements.

(...cont)

distribution of ethnic or social groups within the locations, such as different ethnicities or social strata within communities, which may affect access to assets, including food and income;

- direct and indirect impacts of events, etc;
- if differences exist, group the locations according to the chosen characteristic and proceed as indicated above.

If there is no information for identifying differences among locations:

random sampling is applied, as described in **Section 2.7.4.4**; fewer locations and households can be sampled because the statistics derived cannot be extrapolated with known confidence intervals, and time is limited. The number of locations can be ten to fifteen, with seven to ten households within each location.

In purposive sampling, the sample size can be adjusted during the assessment. For example, if it becomes apparent that all locations within a group are very similar, the number of locations to be visited can be reduced to seven or eight, instead of ten to fifteen. When households are seen to be very similar, the number of households visited in each location can be reduced in the same way. Conversely, if there is *more* heterogeneity than expected, groups of locations may be subdivided and additional groups added.

2.7.3.3 Selection of geographical zones

In purposive sampling, the geographical zones to be covered are determined according to the nature of the crisis and the assessment objectives:

- If the crisis is small-scale and concentrated, the assessment may cover only the directly affected area, such as when flooding affects only one isolated part of a country.
- In a sudden-onset crisis, early assessments will prioritize the zones directly affected, where lives are most at risk. Subsequent assessments may consider wider economic, social and health implications.
- In a slow-onset emergency affecting the whole country, such as in a drought in a small country that relies on agriculture, zones may be selected in any part of the country, but random sampling would be more appropriate if time and resources permit.

Wherever possible, it is useful to select a range of zones that include some directly affected, some indirectly affected and some minimally affected, to provide the basis for comparison. This is rarely possible in sudden-onset emergencies, however, so zones are prioritized according to the expected severity of the situation in each.

In some situations, it is worth visiting areas covered by previous assessments to allow qualitative comparisons between current and previous situations.

2.7.3.4 Selection of locations within geographical zones

The locations - villages, communities or neighbourhoods - to be visited can be identified in either of the following ways:

- They can be selected according to the expected severity of the situation, or the characteristics of the locations and the people living in them. This method is used when there are significant differences among the locations in a geographical zone, or when constraints such as time and accessibility limit the number of locations that can be visited.
- When locations are known to be similar, or when little is known about their characteristics, those selected should be more dispersed, and hence more representative. The number of locations should also be increased i.e. from ten to fifteen (see Box 3.1).³³

When choosing locations, it is important to minimize the "**hub effect**"³⁴ In many emergencies, humanitarian hubs develop in the affected areas, typically main towns, where aid agencies congregate, set up field offices and stockpile resources. Villages close to a humanitarian hub tend to receive more attention and services than those further away. When choosing locations to visit for an EFSA, it is important to bear this in mind and visit some locations that are less easily accessible.

The list of purposively selected locations to be visited can be changed during the assessment, for example, if it is found that initial assumptions about the most affected locations were wrong or incomplete. New locations can be added, and existing ones removed from the list.

As with random sampling, it is better to visit a relatively large number of locations and interview a few households in each, than to visit fewer locations and interview many households in each.

2.7.3.5 Selection of households and individuals within locations

Within each of the locations visited by the assessment team, households, individuals and groups are selected for interview. The aim is to achieve as valid and accurate an impression of the location's entire population as possible.

If the location is reasonably homogeneous and there is sufficient time, households and individuals can be selected through random sampling (see **Section 2.7.4.5**).

When time is limited, the groups expected to be most severely affected are prioritized. Within these vulnerable groups, households may be selected purposively, such as the most severely affected within the group. This gives the sample a high bias

^{33.} However, when the means allow, random sampling is still preferable.

^{34.} Guidelines for Emergency Assessment, International Federation of Red Cross and Red Crescent Societies (IFRC), 2005.

towards the most affected. In this case *no* generalizations of findings to the general population can be made, only statements about the worst-affected households.

Alternatively, within the vulnerable groups, representative households can be selected by the analyst or, preferably, at random.

For some social stratifications it is often difficult to develop a proper sampling frame, so purposive sampling is used. These groups include:

- displaced people living in a host community;
- the community hosting the displaced people;
- the landless and the poor;
- the livelihood groups that are expected to be most severely affected by the crisis, such as pastoralists during droughts or traders during market collapse;
- members of an ethnic group that is targeted during conflict.

It is advisable also to consider groups that are less severely affected. This gives a basis for comparison, and helps to confirm or reject initial assumptions about vulnerability.

2.7.3.6 Example of purposive sampling

Example 3.4 describes purposive sampling from an EFSA carried out in Rwanda in 2006.

Example 3.4: Purposive sampling in the 2006 Rwanda EFSA

In April 2006, WFP and partners undertook an EFSA in Rwanda to assess the impact of drought on household food security. The EFSA was triggered by indications of severe hardship after two years of poor rainfall. Sampling was based on Rwanda's administrative system: the country is divided into districts, which are subdivided into sectors, and then into cells of 20 to 40 households each.

Sampling approach

Five of the most drought-affected districts were identified from a review of field mission reports by various agencies and government services, and discussion with the country office.

In each of the five districts, a purposive sample of eight cells was drawn. The cells were identified as follows:

- The most affected sectors were selected by the district executive secretaries.
- The head of each of the most affected sectors selected the most affected cells.

Within each selected cell, interviews were held with four to eight of the worst-off households. Key informant interviews, focus group discussions and market and health centre surveys were undertaken, to complement and triangulate the information.

This sampling approach did not provide an overview of the situation, as it focused on the most severely affected areas and households. However, given the time and information limitations, this type of sampling enabled conclusions to be drawn about the severity of the situation and how the worst-affected people can or cannot cope with it.

A sample of the "worst of the worst" is severely biased towards the most foodinsecure and is not at all representative of the five districts as a whole; no estimates of numbers of affected households can be made.³⁵

When purposive sampling is used, the assessment report should explain why it has been used and describe the process for selecting which people to interview.

2.7.3.7 Snowball sampling

Snowball sampling can be used if neither random nor purposive sampling is feasible. This may be the case when the population of interest cannot easily be found, such as refugees scattered across a large city, or when people do not wish to be identified, such as illegal immigrants or people affected by HIV/AIDS who do not want to be stigmatized.

Snowball sampling is undertaken as follows:

- Key informants are identified on the basis of their knowledge of the emergency and the people affected by it.
- The key informants are interviewed, and act as entry points by recommending other people who can provide useful information.
- The next group of informants is interviewed and the process continues.

Snowball sampling incurs a high risk of bias. Each informant is likely to refer the assessment team to people that he/she knows. These people may all belong to a particular sub-group within a larger population. For example, if the assessment team is looking for refugees, they may end up interviewing people from only a particular region, or belonging to certain political parties or ethnic groups.

When snowball sampling is used, the assessment report should explain why it has been used and describe the process for selecting which people to interview.

2.7.3.8 Convenience sampling

Convenience sampling is the least effective of the sampling approaches, but if time and access are very short, such as during initial assessments, it may be the only option. In convenience sampling, informants are chosen because they are accessible. For example, if an initial assessment involves only one short visit to the affected area, with little or no preparation, the EFSA team should talk to everyone who is available; team members should apply judgement, such as by visiting health

^{35.} As an alternative, local key informants can be asked to stratify the districts according to whether they are "most affected", "moderately affected" or "least affected". Within each of these strata, further substrata can be defined. Households can be randomly selected within each substratum. If the population of each substratum is known, the appropriate sample weights can be applied to make unbiased estimates of the prevalence of food-insecure households.

clinics and other places where they can expect to find good quality information.

With convenience sampling, the risk of bias is very high. Generalizations to other areas and population groups must therefore be made with caution.

2.7.4 Random sampling³⁶

This section presents methods for in-depth assessments using random sampling.

2.7.4.1 Principles of random sampling

Random sampling is based on the principle that each unit in a population has exactly the same chance³⁷ of being selected as every other unit. In an EFSA, a unit is usually a household when analysing food security, or an individual when collecting anthropometric measurements.

In random sampling approaches, **all selections within a stratum are made randomly**, including the selection of:

- locations within the geographical strata;
- households and individuals within the chosen locations.

Random sampling is the preferred method because, theoretically, it is the only one that allows findings to be generalized to the entire sampling frame. It is used when there is need for statistically representative data that can be extrapolated to the wider population with a known degree of confidence, such as for estimating the prevalence of malnutrition or food insecurity.

Random sampling requires the following:

- There must be sufficient information about the population, location and numbers to construct the sampling frame from which a random sample is to be drawn, and it must be possible to find each of the households selected for interview. Population lists or maps showing the location of each residence must exist, or it must be possible to construct them.
- It must be possible to assess every area, household and individual within the sampling frame. There can be no physical or security constraints to access.
- There must be sufficient time and resources to visit the selected areas and interview each selected household or individual, including travel time.

^{36.} For more detailed guidance, see: Guidelines for Comprehensive Food Security and Vulnerability Analysis (CFSVA), WFP Food Security Analysis Service, January 2009, and Thematic Guidelines on Sampling, WFP Vulnerability Analysis and Mapping Branch, December 2004.

^{37.} The probability of being selected does not have to be equal for each unit, see: Guidelines for Comprehensive Food Security and Vulnerability Analysis (CFSVA), WFP Food Security Analysis Service, January 2009. It is enough that the probability is known and not zero, so that sample weights can be constructed to calculate unbiased population estimates.

2.7.4.2 Geographical stratification

Normally an EFSA targets a specific area of a country, and a stratified two-stage sampling approach is often used within that part of the country.

Depending on the expected homogeneity of the survey area, geographical zones can be selected in either of two ways:

- If the characteristics of the population and the impact of the emergency are expected to vary across the sampling frame, geographical zones are defined according to such characteristics as the extent to which each is expected to be affected by the crisis. The survey area is **stratified**, meaning that zones are grouped according to characteristics that are important for the food security situation. Random sampling of locations according to population size is then carried out *within* each stratum. Results are representative of each stratum, results from the various strata can be compared, and the number of households to be targeted can be calculated for each zone separately. Such stratification is designed to ensure that each stratum is as homogenous as possible and that there are important differences among strata. A typical stratification could be the agro-ecological or livelihood zones in a country, or the areas that have been affected by a flood vs. those that have not. These approaches are illustrated in the worked example in **Section 2.7.4.6**.
- If no particular food security pattern is expected across the survey area, the area can be subdivided using the existing administrative divisions. If there are too many of these, they can be regrouped. For example, in a survey covering eight provinces of a country, three geographical strata could be maintained: stratum A covering the three western provinces; stratum B the three central provinces; and stratum C the two eastern provinces.

Within a stratum, households and individuals can be chosen through either **two-stage sampling** or **simple random (direct) sampling**.

Box 3.2: Two-stage and simple random sampling

Two-stage stratified sampling (see Section 2.7.3)

- This is the most common approach used in an EFSA. The sample is defined in stages:
- Stratification: The geographical zones to be assessed are determined. In an EFSA these are based on the expected extent of the emergency's impact, and include both directly affected and indirectly affected areas. For instance, for an assessment after the passage of cyclone Nargis in Myanmar, the delta region was divided into a "severely affected", a "moderately affected" and a "mildly affected" zone. Geographical zones may also correspond to those chosen in previous assessments, for the purpose of comparison.
- First stage: The locations to be visited within the chosen geographical zones are selected. These might be villages, groups of villages, towns, neighbourhoods within towns, livelihood zones, camps or any other unit appropriate to the local context and

(cont...)

(...cont)

type of crisis. The locations are selected with probability proportional to size (PPS).³⁸

3. Second stage: Within each location, groups, households or individuals are randomly chosen for interview, based on an existing or specially prepared list.

Two-stage sampling saves resources and time, as only selected locations are visited. However, the **sample size must be increased** to compensate for the relative homogeneity of households within locations.³⁹

The number of locations to visit and the number of households or individuals to interview within each location depend on the total sample size required. This is estimated differently for random and purposive sampling (see Sections 2.7.3 and 2.7.4). In all cases, it is better to select a large number of locations and interview relatively few households/individuals at each location than to select a small number of locations and interview many households/individuals at each.

Simple random sampling

Households and individuals are selected from lists covering the entire area under study – i.e. all the units within the sampling frame. This avoids the need to select locations from which to draw the sample. This approach depends on having up-to-date population lists and accurate data on the locations of households and individuals. It also depends on access; if the population is widely dispersed, it may not be possible to reach all units within the time available for the assessment.

Because of these restraints, EFSAs rarely use simple random sampling, unless the sampling frame is small and geographically concentrated, such as in an internally displaced person (IDP) camp.

2.7.4.3 Determining the sample size

The sample size is the number of surveyed units – generally households or individuals – required to give the desired level of precision.

The sample size is calculated on the basis of the main characteristic of interest for the assessment. For example, to estimate acute malnutrition among children under 5 years of age, the sample size is calculated on the basis of the *expected prevalence of wasting*. The number of households to be visited is derived⁴⁰ from the average number of children under 5 per household.

This approach is problematic for food security assessments however, because no single food security indicator can be used to estimate the size of the sample. If baseline studies exist,⁴¹ the proportion of food-insecure households in the baseline study can be adjusted according to the expected impact of the crisis.

^{38.} Guidelines for Comprehensive Food Security and Vulnerability Analysis (CFSVA), WFP Food Security Analysis Service, January 2009.

^{39.} *ibid.*

^{40.} For more details see: Emergency Nutrition Assessment: Guidelines for Field Workers, Save the Children, 2004.

^{41.} For example, the CFSVAs carried out by WFP.

There are formulae and statistical software for calculating sample size once the characteristics to be measured and the expected prevalence have been determined.⁴²

If there are no relevant data from which to estimate prevalence, the rule of thumb explained in Box 3.3 can be used.

Box 3.3: Rule of thumb for estimating the sample size in a household survey

Based on experience from many household food security surveys, a simple rule of thumb is to consider a **sample size of between 150 and 250 households for each reporting domain**. Ideally, the sample size should be towards the upper end of this scale, to increase the reliability of the results and the validity of their extrapolation to other households in the sampling frame. For example, in one reporting domain the sample size is 200 households (ten households interviewed in each of twenty villages). If the sample prevalence of food insecurity is 40 percent, it could be generalized, with 95 percent confidence, that for the entire reporting domain the prevalence of food insecurity is between 31 and 49 percent.⁴³

The following points should be noted when using this rule of thumb:

- The sample size applies to each geographical area, population group or other reporting domain. For example, if prevalence of food insecurity is to be estimated for both IDPs and residents, a sample of 150 to 250 households must be drawn *from each*. If food insecurity is to be compared among districts, a sample of 150 to 250 households should be drawn *from each district*.
- 2. If sampling is done in stages (see Box 3.2), the upper limit of 250 households must be used to factor in the homogeneity of the households within each location sampled during the first stage.⁴⁴
- 3. Anthropometric measurements to estimate the prevalence of malnutrition are sometimes combined with indicators to estimate the prevalence of food insecurity. In this case, the size of the household sample is based on the sample size required for estimating the prevalence of malnutrition with the required degree of precision. Typically, the sample required for an anthropometric survey is larger than the 150 to 250 households required for household food security analysis, and a sample of up to 900 children may be recommended.

2.7.4.4 Selection of locations/clusters within geographical zones

When simple random sampling is used, there is no need to select locations, or clusters, as the sample of households or individuals is drawn directly from the entire sampling frame (see Box 3.2).

^{42.} Guidelines for Comprehensive Food Security and Vulnerability Analysis (CFSVA), WFP Food Security Analysis Service, January 2009; and Thematic Guidelines on Sampling, WFP Vulnerability Analysis and Mapping Branch, December 2004.

^{43.} Assuming a design effect of 2.

^{44.} This phenomenon is known as the "design effect". In EFSAs it may require a doubling of the sample size.
The first step is to determine the number of clusters to visit within each zone or stratum. This depends on:

- the sample size: the total number of households or individuals to interview (see **Section 2.7.3.5**);
- practical considerations, such as time, access, logistics and human resource constraints.

As explained in Box 3.2, it is better to select a large number of clusters and interview a relatively small number of households/individuals in each than to select fewer clusters and hold more interviews in each. The **rule of thumb** explained in Box 3.4 can be applied.

Box 3.4: Rule of thumb for determining the number of clusters in a random sample

Based on experience of assessing household food security, a total of **ten households per cluster** is usually sufficient. The main result of adding households in each cluster is an increase in the design effect. The sample size can therefore be decided according to the following, based on the level of precision desired:

- if the sample size is 150 households, at least 15 clusters should be selected, and at least 10 households within each cluster;
- if the sample size is 250 households, at least 25 clusters should be selected, and at least 10 households within each cluster.

Clusters, often villages, within each zone are selected randomly with probability according to their size (see **Section 2.7.4.6**). The communities or villages to visit are determined according to the selected clusters. There may be more or less than one cluster in a village.

In nutrition surveys of children under 5, it is common to use a 30 x 30 sample: 30 locations are selected, and anthropometric measurements of 30 children are taken in each location. The total sample size is therefore 900 children. It is advisable to fine-tune this sample size according to the expected prevalence of malnutrition.⁴⁵

2.7.4.5 Selection of households or individuals within locations

In each location/cluster, simple random sampling of households or individuals is carried out (see Box 3.2) in either of two ways:

• Simple random sampling is preferred and can be used when accurate population lists exist or can be created and when each of the listed households can be located and reached easily within the time available. All the households are listed. Households are then selected at random, using a random number table or by picking names out of a hat. This approach is feasible in small

^{45.} The 30 x 30 sample is valid for an expected prevalence of 50 percent and a desired confidence (at 95 percent) interval of +/- 5 percent, with a non-response rate of about 10 percent. For more guidance, see: A Manual: Measuring and Interpreting Malnutrition and Mortality, CDC and WFP, 2005.

PART III

communities and camps with known boundaries.

 Systematic sampling is used when population lists do not exist, but each house can be identified and reached within the time available. Households can be chosen by walking through the location and visiting houses at pre-selected intervals, such as every tenth house, or by spinning a pen and counting off the houses in the direction indicated. Both of these approaches are illustrated in Example 3.5.

Example 3.5: Selection of households or individuals within locations

Option 1: Houses are laid out in streets and each can be identified from either a map or observation.

Example: A sample of 15 households is to be selected from a total of 186 houses in the location:

- Calculate the sampling interval by dividing the total number of houses by the sample size. In this case: 186/15 = 12.4 houses.
- Round the sampling interval up/down. In this case: down to 12 houses.
- Agree a starting point for the assessment, usually one end of a particular street.
- Choose a number between one and the sampling interval randomly. In this case: between one and 12, say five.
- Count off five houses from the first identified. Interview this household.
- Count off the next 12 houses and undertake the second interview in the twelfth.
- Repeat the process at intervals of 12 houses until the entire location has been covered, and the sample of 15 households has been completed.
- If there are fewer than 12 houses in the street, continue the count in the next street. Clear instructions about the procedure should be given; for example, "at the end of the street, turn right into the next street".

The process is illustrated in the following diagram.



(...cont)

Option 2: Houses are not laid out in streets, but each can be identified by observation on the ground. $^{\rm 46}$

Example: A sample of three households is to be selected:

- Stand in the centre of the location.
- Spin a bottle on the ground or throw a pen in the air and watch where it lands.
- Walk in the direction indicated by the end of the bottle or pen as far as the edge of the location, counting the houses passed. In this case, there are 15 houses.
- Calculate the sampling by dividing the number of houses by the sample size. In this case: 15/3 = 5 houses.
- Choose a number between one and the sampling interval randomly. In this case, between one and five. This is the first house to be visited.
- After this house, walk in the same direction and count another five houses. The fifth is the second household to be visited. Carry out the same procedure to identify the third household in the sample.

The process is illustrated in the following diagram.



^{46.} Adapted from Guidelines for Emergency Assessment, IFRC, 2005.

2.7.4.6 Example of random sampling

An EFSA is being carried out in a conflict-affected area. Some people have been displaced to camps, while others remain in their homes. Two of the objectives of the EFSA are to determine the rate of under-5 malnutrition and to determine the levels of food insecurity among both displaced and resident populations.

In this case, the affected area is the basis for **the sampling frame** and it is large, with villages and IDP camps widely dispersed. Although each of these locations could be reached, there is insufficient time to do so. The dispersion means that simple random sampling of the whole population cannot be carried out. Instead, a **two-stage sampling** approach is chosen (see Box 3.2).

The following steps are undertaken:

- 1. The population is divided into two strata: resident population and IDPs.
- 2. For each stratum, a sample of **localities** is needed. For residents, localities are defined as villages; for IDPs they are camps.
- 3. Because the rates of malnutrition are unknown and therefore cannot be used to calculate the sample size more accurately, it is decided that a 30 x 30 cluster sample will be drawn for each stratum, to obtain a sufficient number of children for statistical analysis of the anthropometric data. Thus, 30 clusters are identified (stage 1), and 30 units households or individuals are selected in each cluster (stage 2). In each locality, 30 households will be interviewed to estimate the level of household food insecurity, and 30 children under 5 will be examined to determine their nutrition status.
- 4. The samples to be drawn are therefore:
 - resident population: 900 households and 900 children under 5;

• IDP population: 900 households and 900 children under 5. Children under 5 are selected from the households where food security interviews take place. The sample size for each stratum is therefore 900 households.

- 5. A table similar to Table 3.4 is constructed and its columns filled in as explained in the following steps. A separate table is constructed for each stratum: residents and IDPs.
- To identify where the clusters will be located, the names of the localities the villages and camps – are entered in the second column. To ensure a truly random procedure, each location is assigned a random number and ranked in the table according to that number.

- 7. The best available population data for each of these localities are entered in the third column. For the villages, this information may come from a population census or district-level records. Information on camps is obtained from the NGOs, United Nations agencies and/or government organizations that manage them. Camp records should indicate the villages from which IDPs come; this information may be used to modify the data on village populations.
- 8. The best available data on the number of children under 5 in the population are used. In this example, secondary data sources indicate that approximately 20 percent of the total population is under 5. These data are entered in the fourth column of the table.

Table 3.4: Table for identifying clusters										
Randomized order	Geographical unit	Estimated total population	Estimated children 6-59 months	Cumulative population 6-59 months	Attributed numbers	Number of clusters				
1	Locality F	2500	500	500	1-500	1				
2	Locality N	1000	200	700	501-700	1				
3	Locality D	800	160	860	701-860	0				
4	Locality A	3250	650	1510	861-1510	2				
etc.	etc.				1511					
	Total	50000	10000	10000						

- 9. The **sampling interval** is calculated by dividing the population of children under 5 by the number of clusters. In this case, the population of under 5 children is 10,000 and the number of clusters is 30. The sampling interval is therefore 10,000/30 = 333.
- 10. The location of the first cluster is determined by randomly selecting a number within the sampling interval of one to 333. Say that the randomly chosen number is 256, the first cluster is in locality 1 to which the number range 1 to 500 has been attributed.
- 11. The remaining clusters are identified by adding the sampling interval sequentially to the starting number until 30 clusters have been selected. In this example:
 - the first cluster is at 256, in locality 1
 - the second cluster is at 256 + 333 = 589, in locality 2
 - the third cluster is at 589 + 333 = 922, in locality 4
 - the fourth cluster is at 922 + 333 = 1,255, in locality 4

The process continues until the required number of clusters have been chosen. Note that the larger the population of the locality, the more clusters it contains.

Therefore, locality 3, with a small population, has no cluster, while locality 4, with a large population, has two clusters. Hence the approach is called **probability proportional to size (PPS)**.

12. Within each of the selected localities, households are identified by one of the methods explained in Example 3.5.

2.8 Human resources

Human resource requirements for an EFSA depend on the assessment methodology that has been chosen:

- For an in-depth large-scale household survey, enumerators, field managers, data analysts, vehicles, drivers, translators, etc. will be needed.
- For an initial or rapid assessment based on purposive sampling and semi-structured interviews, fewer staff will be needed, but the people carrying out the interviews will have to be very well trained and experienced.

Human resources should be drawn from the country office and partners in the country. If capacity is lacking at the national level, additional resources may be requested from the regional bureau and Headquarters, or consultants may be employed.

Human resource needs also depend on the assessment type. The following staff may be required:

- Assessment managers and team leaders: See following paragraph on management structure.
- Enumerators for questionnaire-based surveys: These individuals need to be well-educated, resourceful and prepared to travel for the assessment period. They do not need in-depth knowledge of food security, nutrition or assessment, but should have at least a basic understanding of these issues and preferably some experience of field-based research.
- **Interviewers** for semi-structured interviews and focus group discussions: These people require a good understanding of food security and nutrition, and must be experienced in interview techniques.
- One translator should be available for each member of the field team who does not speak the local language. Ideally, translators should be hired at the same time as the rest of the team, and should go through the same pre-assessment training. Translators are sometimes hired in the field, in which case time should be set aside to brief them on the assessment objectives and data collection methods.

- **Drivers** are required for the vehicles to be used.
- In large assessments, dedicated administrators and logisticians may be used. In smaller assessments, administrative and logistics capacity may be shared with the country office, in which case the time it allocates to the assessment must be clearly agreed with the country office.
- Data analysts and data managers are needed if much statistical analysis will be required.

To ensure that gender issues are addressed during field activities, **the enumerators, interviewers and translators should include both men and women**, preferably in equal numbers.

The **management structure** for an EFSA depends on the scope of the assessment. For a large-scale in-depth assessment involving a household survey, different layers of management are needed: overall assessment manager, team leaders, etc. For an assessment in an insecure area, a single team with an experienced team leader is more appropriate.

Management positions typically include the following:

- Assessment manager with overall responsibility for the assessment: In addition
 to strong technical assessment skills, this person should also be able to manage
 people, organize multiple, simultaneous tasks, and deliver results within a
 specified period. She/he is ultimately responsible for ensuring that the
 assessment fulfils its objectives.
- Field team leaders: If multiple field teams are deployed to different parts of the affected area, such as in a household survey, each team needs a leader. This person ensures that his/her team works according to the plan agreed with the assessment manager, and is responsible for the team's administrative and logistics arrangements in the field.
- Analytical team leader: If large quantities of data are to be processed, an analytical team is established. The analytical team leader ensures that data are properly entered and cleaned, analysis is completed on time, and problems are identified and rectified quickly. This involves collaboration with field teams throughout the assessment.
- Administrative and logistics team leaders: For a large-scale assessment, these
 people are assigned to provide full-time support to other management personnel
 and assessment team members. For small-scale assessments, they may work
 part-time on the assessment and part-time in their usual country office roles. In
 the latter case, it is important to assess the time requirements for each role, and
 agree these with the country director.

In assessments involving external staff, the country director should assign a senior manager from the country office to liaise with the assessment team. This person does not have a hands-on role in the assessment, but is kept up-to-date with its

progress and any changes in the planning. She/he meets the assessment manager regularly and helps to resolve any operational problems that arise.

All the people working on the assessment must receive training, even if they have undertaken assessments in the past. This is particularly important for enumerators. **Trainers** must be deployed for the time needed, which is typically about two days for enumerators' training and one week for interviewers'.

2.9 Administration and logistics

Most assessments involve a lot of administrative work, such as:

- organizing contracts and payment arrangements for the staff employed specifically for the assessment;
- managing cash flow during the assessment, including paying hotel bills, purchasing food, and other incidental expenses; reliable and *safe* procedures must be developed for teams working in the field, as it is best to avoid carrying large quantities of cash;
- booking hotels, flights, etc.;
- booking office space for training, briefings, debriefings, etc.;
- arranging security clearance, visas and internal travel permits when needed; this can take several weeks, so it must be addressed as soon as the assessment is agreed to.

Teams must have all the equipment and systems they need to operate efficiently and safely in the field, including:

- fully functional and equipped vehicles, with spare wheels, tools and first aid equipment;
- radios, satellite and/or mobile telephones, depending on the context;
- supplies of food and drinking water, where needed;
- camping equipment, where needed;
- adequate quantities of checklists, questionnaires and all the stationery required.

2.10 Communications, security and emergency procedures

Communications procedures must be established to ensure that assessment teams remain in contact with assessment managers and country offices. Regular contact times should be set; these might be once a day in a secure situation and more frequently when security is poor. Fall-back plans should be made, in case a team cannot report for some reason.

Emergency and security procedures are important, and should include:

• obtaining the latest version of the agency's security procedures for the specific

context, and ensuring that all team members read them;

- training all staff in the agency's security procedures;
- providing an up-to-date briefing from the security officer;
- establishing field security procedures, such as travel times and no-go areas;
- identifying medical facilities in the areas to be visited;
- establishing evacuation procedures.

2.11 Briefing

All members of the assessment team, including translators and drivers, are briefed before fieldwork commences. Briefing includes:

- the assessment objectives and methodology;
- techniques and tools to be used, such as semi-structured interviews and proportional piling;
- the time schedule;
- communications, security and emergency procedures;
- administrative and logistics arrangements, such as transport and accommodation.

2.12 Assessment schedule

An assessment schedule must be drawn up during the early stages of the EFSA planning process, giving all the major activities and the dates on which they should start and finish. The assessment manager and team members consult the schedule regularly to ensure that their work is progressing according to plan. If the schedule has to be changed during the assessment, the assessment manager consults the country office and partners to ensure that the proposed changes are feasible. Once agreed, the revised schedule is circulated among all assessment team members and partners.

Table 3.5 shows a sample EFSA schedule. The schedule may expand or contract depending on whether it is an initial, rapid or in-depth assessment.

	Time (days)																				
Activity		Week 1					Week 2						Week 3								
Activity	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20	21
1. Establish/confirm working arrangements with partners																					
2. Collate and review secondary data																					
 Review working scenario from initial investigation 																					
 Define assessment objectives and time frame 																					
5. Draw up assessment plan																					
6. Establish analysis plan and define information requirements																					
Decide data collection methods and sampling procedure																					
7. Design/customize data collection instruments																					
Pre-test assessment tools, with teamtraining																					
Finalize assessment tools																					
8. Prepare briefing kit, supplies and equipment																					
9. Identify and recruit team members																					
Train and brief team(s)																					
10.Arrange transport, security and communications																					
11.Collect data at field sites																					
12.Process and analyse data																					
13.Identify and analyse response options																					
14.Write report																					
15.Present findings																					
Finalize and disseminate the report																					

Table 3.5: Example of an EFSA schedule

chapter 3 Fieldwork

Fieldwork must be carefully managed to ensure that:

- teams work efficiently and cohesively;
- the assessment schedule is adhered to;
- all the information requirements are fulfilled;
- information is collected rigorously and consistently;
- unanticipated issues relevant to the food security situation are identified and incorporated.

The assessment manager has overall responsibility for fieldwork. She/he remains in contact with team leaders, who make sure that their teams are working efficiently and collecting all the information required. This involves daily planning. The following section describes a typical day in the field.⁴⁷

3.1 Daily preparation

The assessment team should make the following preparations every day, typically during the evening before field activities are to be carried out:

- Identify the location(s) to be visited for the day.
- Make a preliminary list of interview types and informants. For a household survey using questionnaires, the approach is standard for all locations. For an assessment based on qualitative data, the types of focus group and key informant interviews are agreed, and can be adjusted later, at the location.
- Define responsibilities who will carry out each interview.

3.2 Discussion with community leaders

Assessment teams meet community leaders when they arrive at a location. During this meeting, the team leader explains the reason for the visit and the assessment methodology to be used, and asks for the leaders' support. The meeting provides

^{47.} Section 3.1 is adapted from Guidelines for Emergency Assessment, IFRC, 2005.

an opportunity to obtain the leaders' perspective on the crisis, including its causes and effects on the community in general, and vulnerable households in particular.

3.3 Collection of primary data

Enumerators and interviewers disperse and collect information according to the day's plan. Detailed guidance on primary data collection is given in **Chapter 4**.

3.4 Team meetings

Where possible, each team meets at midday to discuss progress and agree any necessary changes to the approach. This is especially relevant for surveys using qualitative data and semi-structured interviews. Interviewers often identify important pieces of information that had not been anticipated. They share this additional information with their colleagues, who then examine the issue(s) raised during the afternoon interviews.

At a minimum, the whole assessment team must meet every evening to discuss progress and share opinions and initial conclusions.

3.5 Final community meeting

Whenever possible, it is good practice to convene a short meeting with community representatives at the end of the visit to a location. This gives the team an opportunity to offer feedback to the host community about the process and, possibly, the conclusions, thereby enhancing transparency.

3.6 Daily analysis

Analysis is undertaken throughout the assessment and not just at the end, when all the data are available. The extent of this real-time analysis differs according to whether the data are quantitative or qualitative:

- Wherever possible, quantitative data are sent to data analysts every day, to allow them to start data entry, cleaning and preliminary analysis. This saves time at the end of the assessment and enables assessment teams to make use of the analysis during their fieldwork. For example, if the preliminary analysis indicates a particular type of malnutrition, field teams might adjust their approach to probe this issue.
- If personal digital assistants (PDAs) are used, the questionnaires completed on the PDAs must be downloaded to a central computer each day.
- Qualitative data are analysed by field teams continuously. Discussion among team members and assessment managers can lead to daily adjustment of checklists and data collection methodologies.

Primary data collection

4.1 The importance of gathering good quality primary data

The primary data collection methodology is critical to the accuracy of an EFSA. **Even excellent analysis cannot produce good results if the data that feed it are not accurate and reliable**.

Box 3.5 describes some common problems of primary data collection.

Box 3.5: Common problems with primary data collection

Primary data used in EFSAs come from interviews with key informants, focus groups, households and individuals. The following are some common sources of inaccuracy in primary data.

Interviewers lack knowledge or skills

- Interviewers are insufficiently knowledgeable about the context in which they are working or the issues they are discussing. This makes them miss essential information or interpret responses inappropriately.
- Interviewers lack the skills to ask questions about sensitive subjects, such as gender issues, violence or coping strategies.

Information is incomplete or inaccurate

This is often the case when discussing income and expenditure, for the following reasons:

- Informants do not know exactly how much they earn and spend. This is particularly common when households depend on several income sources, each of which is variable. It is difficult to estimate an average monthly income.
- The interviewer and the interviewee interpret terms differently. For example, does
 income include in-kind gifts from neighbours? What is an average month? The latter
 is especially relevant in seasonal economies such as farming.
- Informants are reluctant to reveal all sources, because they are suspicious of the motives for the assessment and fear that results may be used by taxation or legal authorities; some of their activities are illegal; or they do not want to reveal details of their livelihoods to a stranger.
- Informants become aware of the types of information that are most likely to result in their community receiving assistance, and tailor their responses accordingly. If information is taken at face value, a highly misleading impression might be created.

Questionnaires or checklists neglect key issues

If the questions have not been defined carefully and with good knowledge of the context, important issues may be left out. For example, some communities rely heavily on remittances from relatives working in other places. If the assessment concentrates solely on local income generation, the situation may be portrayed as being worse than it actually is.

Interviewers and informants are biased

- Particularly during semi-structured interviews, interviewers can affect responses by asking leading questions or by their attitude. For example, if interviewers are condescending, respondents might feel embarrassed about revealing the extent of their poverty and the coping strategies to which they resort.
- Understandably, informants frequently see the assessment as an opportunity for personal or community gain. They may give the impression that the situation is worse than it actually is in order to obtain relief or other assistance. They may also give misleading information about the impact of the crisis on different groups.

Interviewers and informants become bored

- It is often observed that the first semi-structured interviews of the day take much longer and go into more depth than those carried out later in the day. Interviewers tire and feel as though they are hearing the same responses repeatedly. They become less inclined to query the information given to them, and are less alert to inconsistencies and new information than they are at the start of the day.
- Informants also suffer from boredom, particularly during a long interview and when they have other things to do. At the start of the interview they may be happy to discuss their lives, but lose interest as the interview progresses; the quality of their responses therefore declines.

Informants experience assessment fatigue

In high-profile emergencies, numerous organizations may be conducting assessments during the same time period. Sometimes communities are visited by several assessment teams asking similar questions in the course of a few days. This may have the following effects:

- Informants are diverted from other tasks and find the process tedious; the quality of the information they provide will therefore decline.
- Informants learn how to adapt their information to their own advantage.

Any of the problems described in Box 3.5 could result in inaccurate and unreliable information. If this is fed into the analysis system (see **Part IV**), the final conclusions may be misleading and the recommendations inappropriate.

In order to minimize these problems, the following principles should be observed:

- Collect primary information using an approach that is appropriate to the assessment objectives, the context and the skills of the personnel carrying out the fieldwork.
- Undertake thorough secondary data review to ensure that questionnaires and checklists are as relevant as possible, and that they address fully the specific objectives of the EFSA.
- Train the enumerators and interviewers well. Ensure that they are able to ask questions in a non-leading and sensitive way, to probe during interviews and to triangulate information. Ensure that interviewers have a thorough knowledge of

the context and the issues to be discussed, so that they know what to look for and are alert to unusual information.

- Do not try to carry out too many interviews in one day. It is better to undertake a few high-quality interviews than many interviews that produce inaccurate and incomplete data.
- Discuss impressions and conclusions regularly among the assessment team, at least once a day during the fieldwork.
- Use a variety of informants. Ensure that the perspectives of both women and men are included.
- Triangulate the information (see **Part IV, Section 2.2**). If information from diverse sources converges, it is likely to be accurate; if it is inconsistent or diverges, it is likely that at least one of the sources is providing inaccurate information, or that perspectives differ across the sample.

Application of these principles cannot eliminate all inaccuracies, but it will reduce them and help assessment teams to recognize when problems exist. If it is clear that the primary information is inaccurate and none of the approaches suggested in this section reduces the inaccuracies to an acceptable level, it may be necessary to adapt the assessment or sampling approach, such as by:

- removing interview data that are clearly incorrect, and including additional informants as necessary;
- including more qualitative semi-structured interviews, to probe issues more deeply and identify the reasons for inconsistencies;
- adding new locations to the sample, particularly when using purposive sampling; for example, adding areas that have not been exposed to aid agencies might help clarify inconsistencies brought about by strategic behaviour among the beneficiaries.

4.2 Approaches to primary data collection

During an EFSA, primary data are collected in the following ways:

- Household surveys: Household representatives are interviewed using questionnaires or semi-structured interview checklists.
- Community group discussions: A mixed group of community members is interviewed.
- Focus group discussions: A group of people sharing at least one common characteristic is interviewed.
- Key informant interviews: Individuals with good knowledge of aspects of the community or the present emergency are interviewed.
- **Observation**: Visible and significant aspects of the affected area are noted.

Each of these approaches has strengths and weaknesses and is useful for collecting specific types of information; **the approaches should therefore be combined**. This improves the quality of the information and provides the basis for triangulation.

Four of these five approaches to primary data collection are based on interviews or discussions. These can be conducted in a variety of ways, ranging from highly structured questionnaire approaches, through semi-structured interviews, to open-ended conversations. The two main approaches used in EFSAs are **questionnaires** and **semi-structured interviews**. Each has merits and drawbacks, as explained in Boxes 3.6 and 3.7.

Box 3.6: Main characteristics of questionnaires

Features

A questionnaire consists of a series of questions, carefully formulated and ordered. Questions are coded to facilitate data entry and statistical analysis. The same questionnaire is used for all the households or other informants selected from the sample.

Advantages

- Each informant answers the same series of questions, so the results are comparable.
- Enumerators' training is standardized, so they are all likely to record the information in the same way.
- The structured and coded nature of the questionnaire facilitates data management and analysis.

Disadvantages

- All issues to be investigated must be planned in advance so that a standard series of questions can be designed; there is little opportunity to adjust questions on the basis of information received during the fieldwork.
- It is difficult to collect sensitive information using a questionnaire, such as information about coping strategies that are socially unacceptable or illegal.
- It is difficult to check the accuracy of the collected information before the analysis stage, by which time it is usually too late to repeat fieldwork.

Key issues when planning an assessment that uses questionnaires

- The questionnaire should be designed according to knowledge about the specific context, otherwise redundant questions are likely to be included, which wastes time, and important questions left out, which reduces the usefulness of the questionnaire. Thorough secondary data review is essential prior to designing the questionnaire (see Section 2.5). When time allows, secondary data can be supplemented by primary information collected through semi-structured interviews with key informants, focus groups and households, before the questionnaire was designed.
- The questionnaire must be prepared in accordance with the EFSA analysis plan (see **Part II**). Questions should be designed to allow the cross-tabulations and comparisons required for the analysis.
- The questionnaire should be translated into the language in which it will be administered. Once translated, it should be translated back into the original language by a different translator to check the accuracy of the translation.
- Before it is applied, the questionnaire should be pre-tested in a zone of the assessment area. Testing helps to identify: (i) questions that are redundant and do not provide useful information; (ii) questions that need to be adapted; (iii) problems with the structure of the questionnaire; and (iv) new questions that should be added.
- Enumerators the people who administer the questionnaires in the field should be well trained and supervised. They should be taught how to ask the questions in a culturally sensitive way, and advised on how to pick up inconsistencies and inaccuracies in responses. They should understand the rationale behind questions so that they can ask them in an appropriate way and explain them to informants.

Box 3.7: Main characteristics of semi-structured interviews

Features

A semi-structured interview is based on a pre-prepared series of questions – a **checklist or topical outline**. Unlike a questionnaire interview, the phrasing, order and form of the questions for a semi-structured interview are not fixed. Interviewers use a conversational approach and encourage respondents to explain issues in their own words and their own time. Interviewers must be alert to additional information that was not anticipated when the checklist was compiled, so that they can pursue relevant topics as they arise during the interview.

Semi-structured interviews can be carried out with individuals, household groups, gatherings of individuals such as community leaders, and focus groups.

Advantages

- The conversational style encourages people to be more forthcoming with information than in the formal style of a questionnaire interview.
- Sensitive issues can be addressed by formulating and timing questions to suit the informant.
- Information can be cross-checked during the interview.
- Additional information can be collected and the checklist of questions adjusted during the fieldwork.

Disadvantages

- The questions and interpretations of answers are not standardized, making comparison among interviews difficult.
- Interviewers require considerable skills and experience.

Key issues when planning an assessment that uses semi-structured interviews

- Secondary data must be thoroughly reviewed to identify information requirements and design the checklist.
- Interviewers and translators must be briefed about the context, and must understand the issues and the reasons why the information is needed.

The two approaches to primary data collection described in Boxes 3.6 and 3.7 can be combined. For example, the first phase of an assessment might use semi-structured interviews for community and focus group discussions and key informant interviews. This provides a good understanding of the problems affecting the population. Information from this phase can then be used to design a questionnaire for the second phase of the assessment. This approach has the following advantages:

- Qualitative understanding of the emergency and its impacts is developed through the semi-structured interviews.
- Statistical information is collected through the questionnaire-based survey. This
 information can be used to draw conclusions about the severity and scope of the
 problems identified.
- The semi-structured interviews and questionnaire-based survey can be cross-checked against each other.

Such an approach requires considerable time, even when the affected population and area are small. Therefore, it may not be practical for a rapid EFSA conducted over a three-week period.

Alternatively, semi-structured interviews can be undertaken at the same time as the questionnaire survey, with one group of enumerators carrying out the questionnaire survey, while another carries out semi-structured interviews with individuals and groups. It is important that the two groups consult each other at least once a day to discuss findings and adjust the methodology as necessary.

The most appropriate way of collecting primary data depends on the specifics of the emergency context.

4.3 Undertaking primary data collection

This section explains how to carry out **questionnaire interviews** and **semi-structured interviews**, and how to undertake the **five main approaches** to primary data collection identified in **Section 4.2**:

- household surveys;
- community group discussions;
- focus group discussions;
- key informant interviews;
- observation.

It also describes **tools** used to help informants express themselves, and interviewers to analyse the information they receive.

4.3.1 Formal questionnaires

4.3.1.1 Conducting the interview

A questionnaire contains a standard series of questions that is asked of every respondent. Although the approach is standardized, the way in which enumerators interact with respondents and the manner in which they ask the questions can have major impacts on the quality of the data collected. Table 3.6 outlines some common pitfalls and suggests some solutions.

Consequence	Solution
Respondent does not understand the reason for the visit and may not wish to answer questions. This may lead to incomplete, inaccurate or misleading answers.	Explain the reason for the visit and how the information will be used. Ask whether the informant is happy to answer the questions and whether the time is convenient.
Information is inaccurate.	The respondent should be an adult member of the household – not a guest – and preferably the household head or her/his spouse. If nobody suitable is available, skip this household and move to the next on the list. Return later to interview the household if possible.
Information is often incorrect, as revealed during data cleaning. For example, the reported food consumption may be incompatible with the physical status of household members.	Take time over the interview. Ask each question carefully, ensuring that the respondent has understood it properly. Think about the response, and compare it with the responses to previous questions. If necessary, repeat the question and probe to be sure that the answer is accurate.
Important information is missed.	During the introduction, take time to look around. Ask questions about points of interest and note the responses. Then go through the questionnaire.
	Respondent does not understand the reason for the visit and may not wish to answer questions. This may lead to incomplete, inaccurate or misleading answers. Information is inaccurate.

Table 3.6: Common pitfalls and solutions in questionnaire interviews

4.3.1.2 Designing the questionnaire

A household questionnaire consists of a series of questions addressed to the selected households. The following principles should be applied when designing a household questionnaire:

1.The questionnaire should be specific to the context: Food access and consumption patterns differ markedly among countries and among areas within a country. Questions should be based on a thorough understanding of the context, which can be developed through secondary information review, primary data collection, or – ideally – a combination of the two.

2. The questionnaire should be focused: Each question must have a purpose. Each should contribute directly to fulfilling the information requirements and should fit into a clear analysis plan (see **Part II**). The inclusion of numerous questions that will not be used in the analysis wastes time, and can lead to poor quality responses to the questions that are needed, as both interviewers and informants become bored with very long questionnaires.

3. Questions should provide the opportunity for cross-checking: It is useful to include questions that provide different perspectives on the same issue, to facilitate triangulation. For example, several questions may address sources of income, receipt of remittances and migration. Questions related to household expenditure and/or consumption can be used as proxies for income.

4. Questionnaires should be in the local language: Questionnaires are frequently designed in a language such as English, French or Spanish that is not widely spoken in the assessment area. The questionnaire should be translated into the local language, even if the enumerators speak both languages. Back-translation is important: the questionnaire is translated into the local language, and then translated back into the original language *by a different translator*. If the re-translated version is different from the original, there is a problem with one of the translations. The process is repeated until an accurate re-translation is achieved. If necessary, translators can be changed during the process.

Annex 1 gives an example of a good questionnaire. Note that this questionnaire was developed for a specific emergency and should not be used elsewhere without being adapted. Other examples of questionnaires can be found on the DVD that accompanies this handbook.

4.3.2 Semi-structured interviews

4.3.2.1 Conducting the interview

Semi-structured interviews are more difficult to conduct than questionnaire-based interviews. A **checklist or topical outline** is used instead of a questionnaire, and the interviewer tries to build a relaxed and constructive relationship with the informant, through a conversational approach. This is not always easy, especially if the interviewer is not familiar with the culture and does not understand the language. Box 3.8 gives tips for carrying out semi-structured interviews.

Box 3.8: Tips for semi-structured interviews

Preparation

- Study the checklist thoroughly before setting out for the field, to avoid having to refer to it continually during interviews.
- Work in teams of two people. One asks the questions, while the other takes notes.
- If translators are used, brief them thoroughly about the purpose of the interviews and give them copies of the checklist. Explain any tools that will be used (see Section 4.4).
- Dress respectably for the culture, and do not overdress. Wearing expensive clothes in a poor neighbourhood can increase the psychological distance between the interviewer and the informants.

Conducting the interviews

- Ask the informant if he/she is happy to talk. Explain the reason for the visit and the ways in which the information will be used.
- Try to fit in with the household as much as possible. If the household members are sitting on the floor, do the same. Accept offers of tea, snacks, etc.
- Start the interview slowly. Talk about general issues such as the weather or the family. Be sympathetic and friendly.
- Gradually lead into the questions on the checklist. Give the informants time to express
 themselves in their own words. If answers are unclear, let the informant finish, then
 ask for clarification. Do not worry about the order in which the issues on the checklist
 are addressed; it is more important to let the conversation flow.
- Be alert to information that was not anticipated.
- Think about the responses that people give, and compare them with the information received previously from this interview and other sources. If there seem to be contradictions, ask for more explanation or ask the same questions in different ways.
- Avoid looking at the checklist frequently, as this disrupts the flow of the interview.
- Be alert to signs that the informant is becoming bored or irritated, or has other things to do. Terminate the interview politely if this happens; people have no obligation to talk to EFSA teams.

A good interviewer should be **patient, sympathetic and curious**. It takes a lot of practice to become good at interviewing. If the EFSA depends on semi-structured interviewers, the team should include experienced interviewers. These people can be paired with less experienced colleagues who can learn from them.

4.3.2.2 Designing the checklist

A checklist – also referred to as a topical outline – is similar to a questionnaire in that it consists of a list of questions that need to be answered during an interview. As in questionnaires, the questions are based on the assessment objectives and knowledge of the context. There are important differences, however, including the following:

- A checklist is used to remind the interviewer to cover all the issues identified in the analysis plan (see **Part II**).
- Questions are not asked in a particular order, and the way in which they are asked is adjusted to suit the conversational approach used during semi-structured interviews.

- The interviewer should be ready to discuss relevant issues that are not on the checklist. Additional discussion points might come from observation or from the people being interviewed.
- Unlike questionnaires, checklists can be changed as the assessment progresses and understanding of the situation improves.
- Analysis from checklists is more difficult because some answers, such as narratives, cannot easily be coded.

Annex 3 gives an example of a checklist. As with the questionnaire example in Annex 1, this is a context-specific checklist that should not be used elsewhere without adaptation. Additional examples can be found on the DVD.

4.3.3 Household surveys

The term "household survey" usually refers to the administration of a standard questionnaire to a random sample (see **Section 2.7**). For a household survey:

- a standard questionnaire is designed based on the information requirements identified in **Part II**;
- enumerators are trained to administer the questionnaire;
- the questionnaire is field-tested and modified as necessary;
- the final questionnaire is administered according to the chosen household sampling approach (see **Section 2.7.2**).

Household surveys can also use semi-structured interviews administered to a random or purposive sample.

Standard questionnaire surveys are most useful when there is good knowledge about the population size and location, and good physical access. Under these circumstances, a random sampling approach can be used, and statistical analyses performed on data that are coded in a standard format. Less structured household surveys are useful when access is limited, knowledge of the population is poor, or the required information is sensitive. In such cases, a purposive sampling approach and checklists are generally used, with less emphasis on statistical analysis.

In some cases, households are interviewed outside the formal household survey. This occurs frequently during initial or rapid assessments, when the depth of the information is more important than its statistical representation.

4.3.4 Community group discussions

Community group discussions can be particularly useful when time is limited. Community groups consist of men and women with mixed backgrounds, who are not chosen on the basis of common characteristics, as is the case with focus groups. Community group discussions generally take place in the following circumstances:

- 1.A discussion is organized soon after the assessors arrive at a location, as an entry point. A selected cross-section of the community takes part.
- 2. A discussion occurs spontaneously, when assessors are talking to an individual or small group and other people join in. Such groups can become quite large.

It is advisable to be prepared for community group discussions, as they almost always occur at some point during an EFSA. When talking to a community group, it is best to aim for breadth rather than depth of information; issues can be addressed in more depth during focus group and individual interviews. The information that can be gathered from a community group includes:

- the overall situation within the community, and the impact of the emergency;
- the livelihood strategies used in the area, and how they are affected by the emergency;
- preliminary identification of the most affected areas and groups;
- insight into the status of services and infrastructure, such as health, education, markets, water installations and roads;
- other context-specific issues.

Interviewers should be alert to new information. Community groups are excellent opportunities for uncovering information that can be probed later during key informant and focus group discussions.

Box 3.9: Data collection in the context of sexual and gender-based violence

The following is extracted from the **Inter-Agency Standing Committee's** guidance⁴⁸ for gender-based violence.

During emergencies, it is unlikely that there will be any reliable data about sexual violence, most information is likely be based on anecdotal evidence obtained through a variety of sources.

During an emergency, many assessments are undertaken by humanitarian organizations, donors and government authorities. These assessments should include information about sexual violence, regardless of sector or organizational interests, and should be shared with the Gender-Based Violence working groups. This will avoid duplicate assessments and repetitive interviews with the community.

It is advised to collect and compile information related to the nature and extent of sexual violence; policies, attitudes, and practices of multisectoral actors; and existing prevention / response services and gaps. Information should include:

- demographic information, including disaggregated age and sex data;
- description of population movements (to understand the risk of sexual violence);
- description of the setting(s), organizations present, and types of services and activities under way;
- overview of sexual violence (populations at higher risk, any available data about sexual violence incidents);

^{48.} Guidelines for Gender-based Violence Interventions in Humanitarian Settings, Focusing on Prevention of and Response to Sexual Violence in Emergencies, IASC, September 2005.

- national security and legal authorities (laws, legal definitions, police procedures, judicial procedures, civil procedures);
- community systems for traditional justice or customary law;
- existing multisectoral prevention and response action (coordination, referral mechanisms, psychosocial, health, security/police, protection/legal justice).

Collect information in accordance with guiding principles for safety, confidentiality, respect and non-discrimination, and bear in mind the sensitive nature of this issue in communities and among service providers.

- Methods for collecting information should involve the community and may include semi-structured interviews, site visits and observation of the environment.
- Secondary information sources that may be useful include existing needs assessments, reports and available data related to sexual violence.
- Use techniques that will gain rather than alienate community and individual trust, incorporating cultural sensitivity and extreme care in discussing sensitive topics.
- Ensure anonymity and safety of all information sources.
- Use same-sex interviewers and interpreters.
- Information gathering should ideally be conducted by multidisciplinary teams.

Community groups can be difficult to manage, as they often become large very quickly. Many people may want to speak, causing a chaotic atmosphere. Interviewers should try to maintain good humour and explain that people have to speak one by one. They should encourage the quieter people to contribute, without insisting if somebody does not want to talk. If the conversation gets out of control, with too many people talking, arguments breaking out, etc., the interviewers should thank all the participants and close the discussion.

4.3.5 Focus group discussions

A focus group consists of people who have attributes in common, and who are able to provide information about the topic or subject that is the focus of discussion. Focus groups are extremely useful for obtaining detailed information about a topic. They are also a useful complement to a household survey, because they can provide information about sensitive subjects that are not easily addressed in a questionnaire survey (see **Section 4.3.1**), and they can be used for triangulation and cross-checking of information.

When talking to a group of no more than ten people, the opinions expressed by one person can be cross-checked immediately with the other members of the group. The group dynamics often result in more lively debate than would be achieved during individual interviews. There should not be significant power differentials among group members, as this often results in influential people dominating the discussion. Examples of focus groups include:

- farmers who use similar agricultural systems and have similar assets, such as area of land cultivated;
- traders who work with similar commodities and have similar turnovers;

 mothers and caregivers, who have primary responsibility for child care and food preparation in many societies; these are issues of crucial interest to EFSA teams.

In societies where women are less powerful or less educated than men, or where women exercise influence out of the public arena, it is advisable to talk to them separately. Focus groups are selected according to the type of information that is sought, and the nature of the emergency. Example 3.6 gives some illustrations.

Example 3.6: Focus group discussions in the context of a drought

The assessment team wants to discover how the drought affects different livelihood groups. Focus groups may be of:

- farmers who plant crops and sell their produce to purchase other goods;
- livestock owners who sell animals and animal products to buy food;
- traders who buy and sell crops and animals.

It is important to talk to these people separately because, among other issues, the drought may affect them differently:

- Farmers have reduced harvests. Consumption of their own production decreases and they have less output to sell. However, the prices they receive for their crops increase.
- Animals are in poor health because of the state of grazing and water supplies. They
 therefore fetch low prices at the market. Livestock owners have to sell their animals
 at low prices, but must buy food at high prices.
- **Traders** have stockpiled food crops when prices were low; they are now able to sell them at high prices, thereby making a good profit.

A semi-structured approach is used when conducting a focus group discussion. A checklist is developed, as explained in **Section 4.3.2**, and the interviewer encourages a relaxed and constructive atmosphere (see Box 3.8). Box 3.10 explains features of focus groups that must be taken into account.

Box 3.10: Tips for facilitating focus group discussions

Preparation

- Identify which groups and which individuals from those groups to talk to, through consultation with local people.
- Ensure that there are no large power differences among the people in the group, such as between a local chief and a member of a marginalized clan. When this occurs, the more powerful people tend to dominate the discussion, and the less powerful may be reluctant to express their opinions openly.
- The group should consist of between six and ten people. If there are fewer than six, the benefits of accumulated knowledge and group dynamics are reduced. If there are more than ten, the group becomes difficult to manage and individuals do not get enough time to speak.
- Draw up a checklist for the discussion. This should be specific to the context and the focus group. Section 4.3.2 explains how to compile a checklist.
- Agree a time and place for the discussion. A secluded place is better than a public area, where onlookers may observe or participate, and the group may quickly become unmanageable.

- Dress respectably but not too formally.
- Where possible, have two interviewers: one to ask questions, and the other to record the answers.
- The assessment team should agree on a format for recording the information received during focus group discussions. The details are not important, but the format must be used consistently by different teams, to facilitate analysis.⁴⁹

Conducting the discussion

- Arrange the seating so that everyone is at the same level, either all on the floor or all on chairs. Sit in a circle.
- The interviewer introduces herself/himself and explains the objectives of the exercise.
- The interviewer explains the rules for the discussion: everybody has an equal opportunity to talk, and any views can be expressed.
- The interviewer uses a checklist: topics are introduced into the conversation and group members are invited to give their inputs. The interviewer ensures that all the issues are covered, and looks out for additional information that was not expected.
- The recorder writes down the responses.
- The interviewer needs to use his/her judgement to decide when to let the discussion take its own course, and when to bring it back to the topics on the checklist.
- The interviewer tries to ensure that all the people in the group participate. Often one
 or two people dominate a discussion. If this starts to happen, the interviewer politely
 asks these people to wait, and invites input from the less forceful members of the
 group. Considerable tact may be needed.
- There are usually disagreements among group members. For example, one person might say that everybody is receiving relief assistance, while another says that people in certain districts receive more than others. The interviewer works through the issue by bringing other people into the discussion and probing to discover the source of the disagreement. One of the main advantages of using focus groups is this opportunity for instant cross-checking. There is often no single "truth". Group members have different perspectives, and the emergency may affect them differently. A focus group provides an opportunity to learn about these perspectives.
- The interviewer gradually moves through the topics on the checklist. At the end of each she/he summarizes the conclusions to ensure that he/she has understood what the group has been saying.
- At the end of the discussion, the interviewer summarizes all the conclusions and explains how the information will be used.

After the discussion

• The interviewer and the recorder agree on the results and structure them in the pre-agreed format.

4.3.6 Key informant interviews

Key informants are people with specialist knowledge about some aspect of the area, the population or the emergency. They should be consulted in all EFSAs. Typical key informants include:

- health workers and other service providers, such as water engineers and teachers;
- traders;
- farmers and agricultural extension workers;

^{49.} For more details, see: Technical Guidance Sheet No. 9, Qualitative Data Collection and Analysis for Food Security Assessments, L. Morinière, WFP Emergency Needs Assessment Service, September 2007.

- government, United Nations and NGO employees;
- community leaders.

Anyone who might have an interesting perspective and knowledge about the situation can be interviewed as a key informant. The assessment team should always look out for such people during assessment planning and implementation.

Selection of key informants depends on the context and the information requirements defined in **Part II**. For example:

- if the crisis is triggered by long-term economic decline and the closure of trade links with traditional partners, key informants might include economic analysts from the local university, and market traders from different types of market – wholesale, retail, etc.;
- if the crisis is caused by drought, key informants might include farmers, agricultural extension officers and market traders.

There are no set rules for selecting key informants or the number to interview during the assessment, but the rules of thumb described in Box 3.11 can be applied.

Box 3.11: Rules of thumb for identifying key informants

Select **primary key informants** who are directly involved with the issues identified in the analysis plan. For example, all EFSAs cover nutrition and health status, and food availability and access, so key informants could include:

- health workers, for information about health and nutrition status and health services;
- market traders, for information about food availability and access, through inputs on price trends, supply and demand, market integration, etc.;
- fishers, crop producers, pastoralists, for information about production expectations, prices, demand for produce, etc.

Identify as many key informants as time allows, and try to select a diverse range. For example, if there is time to interview five key informants about nutrition problems, it is not advisable to talk to five nutritionists working at the same feeding centre. It would be more useful to interview:

- one nutritionist from the feeding centre, for information about the type and severity of malnutrition among children and mothers attending the feeding centre;
- one health extension worker, for information about care practices, food preparation and water usage;
- one representative of the local water authority or an NGO working in water and sanitation, for information about the quality and quantity of water available, and the sanitary environment;
- one market trader, for information about the types of food available, price trends and, hence, food accessibility;
- one teacher, for information about the nutrition and health status of children attending school.

Consulting a diverse range of key informants allows issues to be studied in depth. In this example, the different informants provide information about both the symptoms of the problem, that is the level of malnutrition seen in the feeding centre, and its possible causes – poor water quality leading to disease, poor dietary diversity and so on.

Select **secondary key informants** who are knowledgeable about the context and who may have indirect links with the issues identified in the analysis plan. These people may not be directly involved with the core sectors of interest – nutrition and food security – but they can provide contextual information and insights into the core sectors. They include:

- community leaders;
- teachers;
- local journalists;
- local NGO workers.

Secondary key informants should be interviewed as time allows. Assessment teams should look out for people who can provide interesting perspectives, and use their judgement about how long to spend talking with these people.

Key informant interviews can be undertaken individually or in groups. In the example in Box 3.11, a discussion with all five informants together would probably be interesting. Another alternative is to facilitate a **focus group discussion** among, for example, health workers or traders.

In general, key informant interviews follow the semi-structured approach (see **Section 4.3.2**). **The checklist is specific to the type of informant**, as in the following examples:

- **Health workers**: Interviews focus on factors that affect the health of the population, disease incidence and causes of poor health.
- **Traders**: The functioning of local and national markets is discussed, using the indicators in **Part II** as guidance.
- **Relief workers**: Discussion covers the crisis' impact on populations with whom the relief workers are in contact, and the type of assistance that is already being provided.

In addition to sector-specific information, it is often useful to discuss general topics with the key informants, who are often well-informed about the situation and may travel frequently to different parts of the affected area. For example:

- traders may work in several markets and can therefore provide information about the relative impact of the crisis in different parts of the affected area;
- professional people such as doctors and engineers often have friends and contacts in other sectors, such as local politicians, so may be able to talk knowledgeably about the political or social situation.

Annex 2 gives an example of a key informant questionnaire. As with the examples in Annexes 1 and 2, this is context-specific and should not be used elsewhere without adaptation.

A word of caution when working with key informants: It is easy to give undue weight to the opinions of certain key informants, particularly if they are very articulate and accessible. For example, an assessment team leader who does not come from

the affected country might find it easy to talk to a university professor who has strong opinions and speaks English, French, etc. There is a danger that this person's input becomes more influential to the assessment than that of (say) a market trader who speaks only the local language. Assessors must be aware of this form of bias and should try to work with as wide a variety of key informants as possible.

4.3.7 Observation

Observation is an important tool in allowing a great deal of information to be assimilated quickly. It can also guide which information to collect. For example, if flood damage or adverse road conditions are observed on the way to the assessment area, these would be obvious subjects for enquiry. Observations must be used with caution, however; a factor with a strong visual impact may assume undue weight. For example, flood damage to homes and roads may be limited to a specific location with no implications for other areas.

Observation can take several forms. Some of the most useful approaches are described in the following sections.

4.3.7.1 Casual observation

Fieldworkers are constantly exposed to a vast number of visual images. When carrying out an assessment, it is advisable to take advantage of every opportunity for observing the context:

- **During journeys** the features that can be observed include economic activities, such as farming, manufacture and trading; population movement along roads; climatic conditions, such as rainfall, state of pasture and crops; road conditions; security, such as road blocks and armed people; and numerous other things.
- On arrival in a location: After introductions have been made and before interviews begin, it is a good idea to walk around the location with members of the community. This helps to develop a feel for the location and a rapport with the local people; it also provides material for questions, through observation of people's activities, etc.

4.3.7.2 Observation within households

During a household survey, using either questionnaires or semi-structured interviews, it is useful to look around the households where interviews take place. This can provide useful information about the foods consumed, cooking and hygiene practices, the types of asset that people have, and so on. Assessment teams should not be intrusive, however, and should ask permission to examine foodstuffs, visit the latrine, etc.

4.3.7.3 Transect walks

The assessor walks in as straight a line as possible across the location - village,

urban district, etc. – in the company of one or more local people. The assessor records noteworthy features, such as the locations of a river, a grinding mill and different types of shelter, and discusses these with the local people. This can provide an interesting cross-section of the location. If time allows, additional transect walks can be taken along different axes.

4.3.7.4 Structured observation

Observation can be used to provide specific quantitative data. For example, an observer can sit near the village water point during the times when people collect water – usually morning and evening. He/she counts the number of people coming to the water point, notes the size of containers that they use, and measures the average length of time that they spend at the water point queuing, filling, etc.

Observations should be recorded, otherwise it is easy to overlook the information they provide for analysis. In structured observation, this is straightforward. A standard form should be developed to be completed by all observers. In less structured approaches to observation, it is important to record the issues noted, usually in two stages:

- 1. The observer notes in her/his field notebook the points of interest.
- 2. Later, usually at the end of the day, the observer collates the information in a format agreed with the rest of the assessment team.

Table 3.7: Form for recording observations, with examples									
Location	Observation	Significance	Follow-up						
Village X	 Poor drainage around well; spilled water flowing back into the well Animals walking around the well 	 Water contamination, likely to lead to diarrhoeal disease and malnutrition, particularly among young children 	 Investigate household water usage: do people boil and/or treat water? Talk to health workers about nutrition situation Undertake anthropometric measurements 						
	 Healthy livestock observed on outskirts of the village: informants have mentioned livestock health as a major problem 	 Some villagers have maintained their animals' health through access to better pasture, veterinary care, etc. The healthy animals belong to people from other areas The informants have misrepresented the situation 	 Ask local people who owns the animals Try to identify the factors that enable the owners of this herd to keep their animals healthy 						

Table 3.7 illustrates a form for recording observations.

4.4 Participatory tools for primary data collection⁵⁰

The tools presented in this section can be used in any of the interview and discussion formats described in Section 4.3. The tools:

- help informants express themselves, by structuring the way in which issues are discussed; this is especially useful when informants are unfamiliar with quantitative concepts, such as percentages, and are not used to categorization according to priorities, timeframes, etc.;
- help assessors to analyse the information received, by converting data often qualitative – into standard frameworks;
- facilitate discussion, because working through a task together can help break down barriers between informants and interviewers; in focus groups, the tools can stimulate discussion.

The tools should be used selectively. Not all of them are useful in every situation; in some situations, none may be useful. For example, they are unlikely to be useful in a key informant interview with a highly educated professional, such as an engineer or a doctor.

Assessment teams should use their judgement to decide which tools, if any, to use in a given situation. Additional tools not included in this handbook, such as Venn diagrams showing institutional relations, may also be used.⁵¹

4.4.1 Daily calendars

Informants are asked to describe a typical day, giving as much detail as possible about the activities that they carry out and the amount of time each takes.

Daily calendars help the assessment team to find out how different members of a community spend their time, and ways in which daily routines may change in response to a shock. They can also guide programme design. For example, if people spend five hours a day collecting water, the development of an improved water supply should be considered. Comparing current daily schedules with past ones helps identify trends. For example, if people who used to find fuelwood within half an hour now have to walk for two hours to find it, it can be concluded that there may be a deforestation problem.

^{50.} Section 4.4 is adapted from Guidelines for Emergency Assessment, (IFRC), 2005; and Participation by Crisis-Affected Populations in Humanitarian Action: Practitioners' Handbook, Active Learning Network for Accountability and Performance in Humanitarian Action. See also Participatory Techniques and Tools - A WFP Guide, 2001.

^{51.}Technical Guidance Sheet No. 9 Qualitative Data Collection and Analysis for Food Security Assessments, L. Morinière, WFP Emergency Needs Assessment Service, September 2007; and Participatory Techniques and Tools - A WFP Guide, 2001.

It is often useful to carry out separate exercises with different members of a household, such as children, men and women.

4.4.2 Seasonal calendars

Informants are asked to identify events that take place at particular times of a normal year. These include climatic events such as rains or cold weather, livelihood activities such as planting, harvesting or labour migration, cultural events such as religious festivals, and other events that are significant to the community. These are plotted on a calendar, and unusual events resulting from the current crisis are superimposed on this.



A seasonal calendar can indicate whether something is normal, in that it happens regularly, or new. For example, some agricultural areas always have a hunger gap just before the harvest. The people living in these areas have developed systems for coping with this difficult period. Limited food availability at this time of year is therefore much less significant than it would be immediately after the harvest. Seasonal calendars are also useful if activities have to be coordinated and timed to fit in with local schedules, such as seed or food distributions, which may be affected by the state of roads at certain times of year. People's workloads should also be taken into account when planning activities. For example, people tend to be very busy during planting and harvesting periods.

4.4.3 Historical time lines

A historical time line provides a useful overview of how the current crisis fits into a historical perspective. It shows whether this type of crisis is a regular or a one-off occurrence. It also helps identify trends. For example, a time line might show that drought is affecting an area more frequently than in the past.



There are numerous ways of constructing a historical time line. When informants are not used to attributing specific dates to events, the following approach can be used:

- A line is drawn to represent a given period, say 20 years.
- Informants are asked to identify two or three important events that have occurred within the time period, such as the start/finish of conflict, national independence, elections, major floods or volcanic eruptions. These are located in chronological order on the line.
- People then think about significant events, both positive and negative, and locate these on the line, explaining their causes and impacts.

4.4.4 Proportional piling

This is useful for estimating quantities and proportions, especially when working with people who are not used to quantifying data. For example, to discover the proportions of a livelihood group's annual income to come from different sources, the procedure is as follows:

- 1. Collect 100 dried beans, pebbles or anything similar that are all more or less the same size.
- 2. Working with a focus group drawn from a specific livelihood group, ask the informants to divide the beans into piles relative to the income received from each source.
- 3. Count the number of beans in each pile; this number is equivalent to the percentage of annual income to come from that source.



As well as quantifying data, proportional piling is also a good facilitation tool. An activity of this kind can break down barriers within a group of people. It can focus discussion, as there is usually considerable debate about the relative sizes of the piles; this encourages participation and enhances accuracy.

4.4.5 Pair-wise ranking

This is a good way of analysing the relative importance of different factors, such as when identifying which problems people consider the most severe. For example, four major problems have been identified: lack of rain, lack of health care, poor domestic water sources, and insecurity. Each problem is inserted on a grid, with the cells along and below the diagonal blanked out, as shown in Table 3.8, to ensure that questions are not asked twice. Each pair of factors is then considered in turn, and the responses to the following questions noted on the grid:

- Which is the more severe problem, rain or health care? In this example, the answer is health, which is noted in the relevant box.
- Which is the more severe problem, rain or domestic water? Rain, in this example.
- And so on...

Table 3.8: Example of pair-wise ranking									
	Rain	Health care	Domestic water	Insecurity					
Rain		Health care	Rain	Insecurity					
Health care			Health care	Insecurity					
Domestic water				Insecurity					
Insecurity									

Count the results, which in this case are as follows:

- Lack of rain: 1.
- Lack of health care: 2.
- Poor domestic water sources: 0.
- Insecurity: 3.

This indicates that for this group, insecurity is the most serious problem, and poor domestic water sources the least serious. Domestic water's score of zero does not mean that it is not a problem, but that the group considers it to be less severe than the other three problems.

Part IV Analysing EFSA data


Contents of Part IV

1	Introdu	ction	146		
2	2 Essential concepts and methods				
	2.1	Vulnerable groups, including livelihood groups	148		
	2.2	Triangulation and convergence of evidence	150		
3	Conduc	cting a situation analysis	152		
	3.1	Step 1: Synthesize contextual information	153		
	3.2	Step 2: Estimate the numbers of food-insecure and malnourished	153		
	3.2.1	Households facing major health risks	155		
	3.2.2	Estimate the number of households that are food-insecure	155		
	3.3	Step 3: Determine the characteristics of the households and			
		individuals facing food insecurity and malnutrition	156		
	3.3.1	Using quantitative data	157		
	3.3.2	Using qualitative data	159		
	3.4	Step 4: Identify the reasons for food insecurity and malnutrition			
		and for risks to livelihoods	160		
	3.5	Step 5: Determine whether food insecurity and malnutrition			
		are chronic or transitory	162		
	3.6	Step 6: Estimate the severity of food insecurity and malnutrition	164		
	3.6.1	Indicators of risks to lives and livelihoods	164		
		1 Mortality and nutrition indicators	164		
		2 Food security and coping strategy indicators	165		
	3.6.2	Using convergence of evidence from a series of indicators	165		
	3.6.3	Using the food consumption or food access gap	168		
4		cting a forecast analysis	170		
	4.1	Identification of future opportunities and shocks	170		
	4.2	Developing scenarios	175		
	4.2.1	Procedures	176		
	4.2.2	Contingency planning	179		
	4.3	Identification of population groups affected under	100		
	404	the most likely scenario	180		
	4.3.1	Population groups currently at risk	180		
	4.3.2	Additional population groups expected to become at risk	181		
	4.3.3	Combining current and predicted population groups	183		
	4.4	facing risk to food and nutrition security Estimation of the impact of shocks and opportunities	103		
	4.4	on livelihoods	185		
	4.4.1	Conducting a food gap analysis	187		
	4.4.1	Estimation of the numbers of people who will be affected	107		
	4.5 Estimation of the humbers of people who will be affected by shocks and opportunities 191				
	4.5.1	Final estimate of numbers at risk	191		
	4.5.1	That estimate of numbers at lisk	192		

5 Cond	ucting a response analysis	193		
5.1	Factors related to risks to lives and livelihoods	194		
5.2	Entry points for interventions	196		
5.3	Other stakeholders' interventions, and remaining gaps	197		
5.4	Response options	199		
5.4.1	Identification of response options	200		
5.4.2	SWOT analysis	203		
5.4.3	Ranking and prioritization of response options	206		
5.4.4	Targeting	207		
6 Formulating recommendations for interventions and follow-up				

Introduction

The EFSA analytical process adopts a dynamic approach to interpret food and nutrition security by examining the *current* situation, as well as the *past* and the *future*:

- Analysis of the **current situation** determines whether or not food security and nutrition are compromised now.
- Analysis of the **past** establishes **trends**: is the situation improving, deteriorating or remaining constant?
- Scenarios are developed for forecasting the direction and magnitude of future trends.

Throughout this analysis, the **risks** faced by the population are balanced against the population's **capacities** – coping and resilience – and **vulnerabilities**.

To analyse the population's food security and nutrition situation, communities are disaggregated into groups that share similar **livelihoods** and are likely to be affected by **shocks** in similar ways. These groups are considered individually; their specific livelihood assets and strategies, the shocks to which they are exposed, and their capacities and vulnerabilities are analysed in relation to the overall context. Other relevant issues, such as gender, HIV/AIDS and displacement, are also considered.

The EFSA analytical process is divided into two key stages, which are summarized in Table 4.1.

Table 4. 1: Stages of EFSA analysis			
Stage	Objectives		
Situation analysis	 Describe the food security and nutrition status of the population Identify the population's coping strategies Determine whether food insecurity and malnutrition are chronic or transitory 		
Forecast and response analysis	 Forecast the evolution of the situation, using scenarios – projections of how the situation is likely to develop Determine the need, or otherwise, for external intervention Determine the types of intervention that are appropriate, and their characteristics: level, duration, target groups, etc. Determine the capacities of the various stakeholders to provide assistance, and identify the remaining gaps 		

Table 4.1: Stages of EFSA analysis

The following sections explain these stages. It should be noted that analysis is iterative. Steps are often repeated as more information becomes available and understanding improves. The order presented here is generally the order in which the steps are taken. In a real assessment, numerous feedback loops exist, as information is refined and new questions are raised.

Essential concepts and methods

This section explains some of the concepts that are essential to EFSA analysis, building on the guidance given in **Parts II** and **III**.

2.1 Vulnerable groups, including livelihood groups

In an EFSA, it is useful to distinguish population groups according to the characteristics that make them vulnerable. A household's level of vulnerability reflects the extent to which it can cope with shocks affecting nutrition and food security.

During the assessment, the identification of vulnerable groups assists:

- the sampling and identification of which groups to assess (see **Part III**), such as people displaced during a conflict;
- the identification of issues that affect vulnerability; for example, if adaptation of the Food and Nutrition Security Conceptual Framework (see **Part I**) indicates that vulnerability may be related to livestock ownership, groups would be distinguished according to the size of their herds.

During the situation and response analyses, identification of the characteristics that affect vulnerability is used to:

- 1. Estimate the impacts of a shock on different segments of the population;
- 2. Target assistance, if provided; identification of the observable characteristics of vulnerable groups is essential to targeting.

The characteristics that make a group vulnerable depend on the nature of the crisis. For example, in a conflict, the primary indicator of vulnerability might be displacement. Households would therefore be categorized as IDP or resident, and most data analysis would be based on these two groups. In other situations, gender, education level or the presence of a chronic disease such as HIV/AIDS may be used to categorize groups.

In EFSAs, vulnerable population groups are usually defined according to livelihoods, particularly when the shock has an economic impact and its overall impact differs according to households' livelihood assets and strategies. A **livelihood group** is

defined as: A group of people who share similar basic means of livelihood and life styles – the same main subsistence activities, main income activities and social and cultural practices – and face similar risks to food and nutrition insecurity.

Livelihood groups can be defined in various ways. When possible, **pre-crisis definitions** of livelihoods should be used,⁵² as this facilitates comparison. However, this may not be feasible, either because livelihood groups have not been previously defined, or because pre-crisis definitions are no longer relevant, such as when large-scale displacement or other social change has occurred.

If pre-crisis definitions cannot be used, livelihood groups are defined according to **primary productive activity**. This helps to ensure that the groups distinguished *are relevant to the local context*. Which activities are defined as "primary productive activities" depends on the area and the ways in which local people describe themselves. Wherever possible, people from the affected area should be involved in defining livelihood groups, through focus group or key informant interviews.

Examples of primary productive activities include:

- subsistence farming;
- cash crop farming;
- pastoralism;
- fishing;
- petty trade;
- daily labour.

Note that these categories do not denote the *only* productive activities undertaken by the households. For example, although subsistence farmers are likely to produce a substantial proportion of the food they eat, they will probably supplement this through other activities, such as selling cash crops or engaging in daily labour.

Once the livelihood groups have been defined, details of each group's livelihood strategies are identified. For example, subsistence farmers may gain 60 percent of their food from own production, 20 percent from selling produce, 10 percent from handicrafts, and 10 percent from casual labour.

The number of livelihood groups defined depends on the complexity of the economic environment and the extent to which the crisis effects differ among groups, as illustrated in the following examples:

• If locusts have destroyed crops, livelihood groups involved with crop production and sales should be defined, such as the farmers who produced the damaged crops, crop traders, labourers who are normally employed during the harvest, and people who purchase the crop in local markets.

^{52.} For example, the baseline information provided by WFP CFSVAs usually includes a description of livelihood groups.

• The outbreak of widespread conflict affects all sectors of the economy, but the impacts are likely to vary according to livelihood group. Numerous livelihood groups should therefore be identified, and the impact of the crisis on each analysed.

When defining population groups, livelihoods can be combined with other characteristics. For example, if refugees are identified as a group at risk, they could be subdivided into livelihood groups: those whose livelihoods are based on daily labour, those depending on food aid, etc.

Having established *who is vulnerable to food insecurity*, it is necessary to understand *why* they are vulnerable. This means examining the factors that affect food security and nutrition, and the risks that food insecurity poses to the livelihoods of affected households. It involves:

- identifying the characteristics of food-insecure households and malnourished individuals;
- identifying the factors that contribute to food insecurity, malnourishment, coping mechanisms and specific vulnerabilities;
- determining the extent to which food security and malnutrition problems are directly related to the current crisis, or are persistent: are they **chronic** or **transitory**?

Box 4.1: Chronic and transitory food insecurity

A state of food insecurity may be chronic or transitory, depending on its evolution over time:

- Chronic food insecurity is a long-term or persistent inability to meet minimum food requirements. As a rule of thumb, food insecurity lasting for at least six months a year can be considered chronic.
- **Transitory food insecurity** is a short-term or temporary inability to meet minimum food requirements, indicating a capacity to recover. As a rule of thumb, limited periods of food insecurity related to sporadic crises can be considered transitory.

These definitions do not presuppose the **severity** of the food insecurity; for example, transitory food insecurity may be short-lived but very severe. People who are chronically food-insecure are likely to be particularly vulnerable to transitory food insecurity. Repeated periods of *transitory* food insecurity may lead to a situation of *chronic* food insecurity if people do not have time to recover fully from one crisis before the next one arrives.

(See **Section 3.5** and Technical Guidance Sheet No. 5 *Distinguishing between Chronic and Transitory Food Insecurity in EFSAs*, WFP Emergency Needs Assessment Service, December 2007.)

2.2 Triangulation and convergence of evidence

Triangulation is the process through which information from different sources is compared to determine whether or not **evidence converges**.

Example 4.1 illustrates the triangulation process.

Example 4.1: Triangulation

An EFSA team is investigating the functioning of markets after a hurricane. Access to the affected area is impossible, so the assessment team relies on information from key informants and secondary data.

- Woman who travelled from the area yesterday: "The main road is flooded. The usual market area is not accessible. In some of the smaller markets, food seems to be available, but in much smaller quantities than usual."
- Market trader who has not been in the affected area since the hurricane: "I usually go to the market every Wednesday. It is my main source of income for the week. However I have not been able to get there since the hurricane, because of the flooded roads. My produce is rotting, and my income is seriously reduced."
- Woman who lives in an area that was not affected by the hurricane: "I don't think there is such a big problem with markets there. We are all poor and we all need help."
- Reputable evaluation report from the previous major hurricane, which hit the area five years ago: "Markets were seriously affected by the flooded roads. Food access was severely disrupted for up to four weeks in some areas. This led to widespread economic loss and malnutrition among young children."
- Local highway engineer: "The drainage systems in the affected area are overdue for maintenance. We have been telling the local government this for years, but no funds have been available. This means that floods will recede more slowly than usual, and the state of emergency will persist longer than it did after the hurricane five years ago."

Informants 1 and 2 indicate that the floods are disrupting the markets. Both seem to be reliable: informant 1 because she travelled from the area recently; and informant 2 because she is intimately involved with the local market.

These statements about the current situation are supported by information from the evaluation report, which shows that markets were disrupted in a previous similar crisis, and by the engineer's statement that the flooding problem will probably be worse than last time. Both of these sources seem to be reliable.

Informant 3 contradicts the other four sources. She can be considered less reliable, however: her information is not first hand, and she seems to be trying to convince the assessors that her area needs assistance, even though it is not directly affected by the hurricane.

On the basis of this evidence it can be concluded, *with reasonable confidence*, that markets will be seriously disrupted, and that this will have consequences on lives and livelihoods.

Triangulation is essential to the analysis of both qualitative and quantitative data:

- with qualitative data, information from different focus groups and key informants is compared;
- with quantitative data, conclusions from different cross-tabulations are compared.

Triangulation can also be used to check consistency between qualitative and quantitative data. For example, surveys using quantitative data can be cross-checked against surveys using qualitative data among the same population.

chapter 3 Conducting a situation analysis

Situation analysis focuses on the *current* food security and nutrition status of the population *at the time of the assessment*. Ultimately, an EFSA situation analysis should result in answers to the following core questions:

- 1. How many people are food-insecure?
- 2. Which population groups are at risk?
- 3. Who are the people at risk?
- 4. Why are they at risk?
- 5. How severe is the situation?

A situation analysis should answer these questions by considering the following factors:

- The status of food security and nutrition, and the factors that affect them availability, access, utilization, health and care among different communities, groups and individuals.
- The severity of food insecurity and nutrition problems among different communities, groups and individuals.
- The type of coping strategies currently used to withstand the crisis.
- The affect(s) of the shock on livelihoods.
- The number of people affected, and their locations.

The severity of a food security or nutrition crisis depends on the extent to which health and/or livelihoods are threatened. It is difficult to measure severity, or potential severity, at the start of a crisis. It may be easier to measure it later, once the crisis has unfolded, for example, through nutrition surveys. By then, however, it may be too late to avert suffering and death. To avoid waiting until mortality and malnutrition have reached unacceptable levels, severity can be estimated using **proxy indicators**, as described in **Part II, Section 5.3**.

This section describes critical steps in conducting a situation analysis, each of which is intended to answer the core questions. In an EFSA, the steps of situation analysis are often carried out concurrently, and may be repeated several times as understanding of the situation improves. The steps for conducting a situation analysis are:

- **Step 1:** Synthesize contextual information to gain a broad understanding of the nature of the crisis.
- **Step 2:** Use quantitative and qualitative data to estimate the numbers of households and individuals that are food-insecure and malnourished.
- **Step 3:** Determine the characteristics of the households and individuals facing food insecurity and malnutrition, and define their livelihood or other relevant characteristics.
- **Step 4:** Identify the reasons why people are food-insecure and malnourished and why their livelihoods are at risk.
- Step 5: Determine whether food insecurity and malnutrition are chronic or transitory.
- **Step 6:** Estimate the severity of food insecurity and malnutrition.

3.1 Step 1: Synthesize contextual information to gain a broad understanding of the nature of the crisis

A sound knowledge of the context is essential if the factors causing malnutrition and food insecurity and the linkages among these are to be understood. Contextual information is gathered from both secondary and primary sources (see **Part II**).

Contextual information is used continually to inform the analysis, particularly during Step 5. **Part II** provides useful information for this. The analysis includes food availability from crop production and market supplies, and access to key services, particularly health and education.

Contextual information is constantly updated throughout the assessment. Assessors should always look for people who can enhance the assessment team's understanding of the situation.

3.2 Step 2: Use quantitative and qualitative data to estimate the numbers of households and individuals that are food-insecure and malnourished

The population groups that are likely to be facing food insecurity can be identified:

- Select key indicators that measure food insecurity and coping strategies (see **Part II**).
- Define thresholds that indicate the degree of severity for each indicator or combination of indicators (see **Part II**).
- Determine the numbers of individuals and households that are at risk according to each of the chosen indicators and thresholds.
- Estimate the numbers of individuals and households that are likely to be suffering from food insecurity, as a proportion of the population sample.

• Estimate *the total number of people in the population* whose food security is at risk by multiplying the proportion of the sample that is at risk by the total population size.

It is often difficult to obtain accurate and up-to-date data on the total population size in crisis-affected countries. Technical Guidance Sheets⁵³ Nos. 7, 10 and 11, provide guidance on methods for estimating population size when data are unavailable, unreliable or contested.

In some emergencies, it is impossible to obtain a random sample. In such cases, purposive sampling may be the best option available.

Example 4.2 illustrates this process.

Example 4.2: Extrapolation of conclusions when no random sample is available

For an EFSA in a conflict-affected region, access to some parts of the region is impossible. Large-scale displacement has occurred.

A household survey is undertaken, based on a purposive sample of three villages where both IDPs and residents live:

- Residents and IDPs are consulted about social, economic, cultural and other variables in the conflict-affected region during normal times.
- IDPs are consulted about their areas of origin, and when they travelled from these areas. The most up-to-date information about these areas and their similarities with those in which the survey is taking place are collected.

It is necessary to **judge the extent to which the sample is representative of the wider population**. If discussions with residents and IDPs indicate that the situation in other areas is broadly similar to that in the villages sampled, the conclusions of the assessment can be extended to the wider population *with caution*.

In the assessment report, the process of extrapolation should be explained in detail, and the limits to its statistical validity stated clearly.

A more thorough assessment will be carried out as soon as access is possible.

^{53.} Technical Guidance Sheet No. 7 Area Method to Estimate Population Size and Demographics in Emergency Food Security Assessments, A. Henderson, WFP Emergency Needs Assessment Service, September 2007; Technical Guidance Sheet No. 10 Using the Delphi Method to Estimate Population Size and Demographics in Emergency Food Security Assessments, A. Henderson, WFP Food Security Analysis Service, January 2008; Technical Guidance Sheet No. 11 T-Square Method to Estimate Population Size and Demographics in Emergency Food Security Assessments, A. Henderson, WFP Food Security Analysis Service, December 2008.

3.2.1 Households facing major health risks

Nutrition data can be used to identify individuals and households that may be facing major health risks. When collecting nutrition data, statistical software is typically used to calculate the proportions of malnourished individuals. These proportions are extrapolated to the whole population of interest, if the sample is representative.

Households that have poor food consumption at the time of the assessment and are expected to remain in that situation for the next month or more are likely to face major health risks, particularly among vulnerable members with added nutrition needs, such as young children, the sick and the elderly. Pregnant women's expected newborns are also likely to face major health risks.

Households that are using certain coping strategies and are expected to continue using these in the next month or more could also face health risks. The coping strategies concerned are context-specific, and include sustained skipping of meals, dropping of health treatment to save money for food, consumption of contaminated water for drinking and food preparation, and consumption of wild resources that are inappropriate for consumption.

3.2.2 Using qualitative data to estimate the number of households that are food-insecure

In qualitative data analysis, the current status of food insecurity is investigated using information gathered from focus group discussions and key informant interviews. Box 4.2 shows the procedure for this (see **Part III** for guidance on semi-structured interviewing and the use of data collection tools).

Box 4.2: Qualitative identification of population groups, households and individuals facing food insecurity and malnutrition

The criteria for food security – food access and food consumption – used in quantitative data analysis can also be used in qualitative data analysis; **Part II** provides guidance on which indicators to discuss. These criteria can be analysed during a focus group discussion as follows:

- 1. Ask the group how people obtain access to food.
- 2. Determine whether each means of access is considered poor, average or good.
- 3. Using proportional piling (see **Part III, Section 4.4.4**), estimate the proportions of households in the community that rely on poor, on average and on good ways of obtaining access to food.
- 4. There will now be three piles. Keep these piles.
- 5. Ask the group to explain the diets the types of food consumed by different groups in the community.
- 6. Determine whether each diet indicates poor, borderline or acceptable food consumption.
- 7. Take each of the piles from step 3 in turn. Ask the group to divide each pile according to the proportions of poor, borderline and acceptable food consumption.
- 8. Count the beans and enter the numbers in a table.

(cont...)

(cont...)

Coping strategies are investigated in a similar way to that proposed for analysing quantitative data, adapted as follows:

- 1. List the coping strategies that people use, and the severity related to each, as explained in **Part II, Section 7.3.4**.
- 2. For each group, check whether its members are using damaging coping strategies. This may result in re-categorization of some groups. For example, people who are found to use highly damaging coping strategies may be re-categorized as severely food-insecure.

Having determined the **proportion** of people at risk in the community, the **number** can be estimated using population figures for the community. These may come from a census, a village/town/district register, estimation by community members, or any other source that is appropriate to the context.

It is advisable to harmonize, as much as possible, the ways in which different communities describe food consumption and access, to facilitate comparison among communities.

In an assessment, similar exercises are undertaken in different communities. The results are **triangulated** to determine the level of confidence with which they can be treated and the extent to which generalizations can be made about the wider area and population.

The qualitative approach can be applied to investigating other types of risk to livelihoods that were not anticipated when the assessment was planned. **This is a major strength of the qualitative approach**. Informants are asked to explain risks in their own terms and to identify the individuals and groups that are vulnerable to these risks. Numbers of people at risk are estimated as described in Box 4.2.

3.3 Step 3: Determine the characteristics of the households and individuals facing food insecurity and malnutrition, and define their livelihood or other relevant characteristics

Step 2 defined the numbers of individuals and households who are food-insecure. In Step 3, **profiles** of these people are developed. The aim of this step is to:

- categorize the livelihoods of the people who are food- and nutrition-insecure;
- define easily recognizable characteristics that can be used to identify at-risk people – IDPs, women-headed households, pastoralists, etc. – for targeting, should an intervention be necessary.

Profiling is done by:

- cross-tabulating, or matching in the case of qualitative data, the food-insecure people with their livelihood characteristics, at the individual, household, community and national levels;
- defining population groups that can be used for targeting, such as livelihood groups.

When nutrition status data are available, it is useful to cross-tabulate or compare these with the information collected during the nutrition and food security components of the assessment. This provides information about the characteristics of the households where malnutrition exists, and the possible causes of malnutrition. The way in which cross-tabulations and comparisons are made depends on the data collection approach. Table 4.2 provides guidance on this aspect of analysis.

Table 4.2: Characteristics of households where malnutrition exists			
Methodology for collecting nutrition information	Cross-tabulations and comparisons		
Anthropometric data collected simultaneously with food security, health and care information: Same households Same geographical area Same time	 Direct cross-tabulation/comparison of malnutrition with: Household food security status Food access and consumption indicators The health and public health environment Care information 		
Anthropometric data collected in parallel with food security, health and care information: Different households Same geographical area Same time	 Comparison between the geographical concentration of malnutrition and geographical information on: Household food security status The health and public health environment Care information 		
Only household food security, health and care information available: Anthropometric data not collected and not available from other surveys	 Only hypotheses of the nutrition situation can be made, based on: Household food security status Food consumption patterns, preferably at the individual level The public health environment Care information 		

3.3.1 Using quantitative data to create profiles of households and individuals facing food insecurity and malnutrition

With quantitative data, each food- and nutrition-insecure individual and household is cross-tabulated against a range of livelihood characteristics, such as:

- key livelihood characteristics, which should be included in every assessment;
- context-specific livelihood characteristics, which depend on the communities and areas being assessed and the nature of the crisis.

The following key livelihood characteristics should *always* be considered:

- Locations of individuals and households: name of area, village, town, etc.
- **Residential status** of households: resident, IDP, refugee, returnee, hosting displaced people, etc.
- Sex, age and health status of individuals whose nutrition status is measured.
- Sex, age and education level of heads of households.

- Size and age composition of households, including dependency ratio⁵⁴ or proportion of dependants,⁵⁵ where appropriate.
- Sources of food for individuals and households.
- Sources of income for individuals and households.
- Markets: physical access to markets in distance and time, proportions/amounts of food and other items purchased at markets.
- Coping strategies used by individuals and households.
- Health and disability status of household members, with details of diseases, chronic sickness and disability.
- Health care access: physical access to health services, constraints to health care access.
- Water access: quantity and quality of households' water sources, distance(s) to them.
- Sanitation facilities: types and extent of usage by household members.
- Housing: type and quality, protection from heat, cold, rain, wind, etc.
- Assets: types, numbers and values (when/if possible).

Other livelihood characteristics and their relevance to risks of food insecurity are context-specific. These are identified through:

- contextual analysis of the specific community and the factors that affect it;
- analysis of the changes affecting the community as a result of the current crisis.

Context-specific characteristics are identified on the basis of the background information gathered in Step 1 (see **Section 3.1**).

Cross-tabulations are used to develop a series of **profiles** of individuals and households at risk. Other statistical techniques such as regression analysis can also be used.

Having profiled the food- and nutrition-insecure individuals and households, the next step is to define easily identifiable at-risk groups. These may be **livelihood groups** or other relevant groups (see **Section 2.1**). If possible, groups should be defined in the same ways as they were before the crisis, to facilitate comparison. If groups were not defined before the crisis or if the definitions of groups are no longer relevant, livelihood groups must be defined on the basis of the survey data.

^{54.} The household dependency ratio is the number of individuals aged under 15 or over 64 years, divided by the number of individuals aged 15 to 64 years, expressed as a percentage. Age thresholds may have to be adapted to the norms in the country concerned.

^{55.} The proportion of dependants is the number of dependent household members divided by the total number of household members, expressed as a percentage. Definition of a dependant is context-specific, and may include people who are under a certain age, over a certain age, chronically sick, or disabled.

3.3.2 Using qualitative data to create profiles of households and individuals who are facing food insecurity and malnutrition

In Step 2, the analysis of qualitative data was based on focus group discussions and key informant interviews investigating aspects of risk to people's food security and livelihoods. The numbers of individuals and households facing food or nutrition insecurity were also estimated. Information on the *characteristics* of the people at risk is usually collected during the same interviews, as described in Box 4.3.

Box 4.3: Defining the characteristics of households and individuals facing food insecurity and malnutrition, using qualitative data

For more guidance on conducting focus group discussions, see Part IV and Technical Guidance Sheet No. $9.^{\rm 56}$

During a focus group discussion, the interviewer poses a question similar to the following: "You have explained the types of risks to which people are exposed. Can you now tell me what sorts of people are vulnerable to these risks?"

The group may start by giving vague descriptions of "people who live over there", or "farmers". They may say that everyone is exposed to the risks. It is usually necessary to **probe** extensively. Interviewers ask detailed questions, continually **cross-checking** the answers with different members of the focus group, and with information collected from previous interviews.

Look for the same information as recommended for cross-tabulations in quantitative data analysis (see Section 3.3.1). Key characteristics that should always be investigated are:

- Location of individuals and households: name of area, village, town, etc.
- Residential status of households: resident, IDP, refugee, returnee, hosting displaced people, etc.
- Sex, age and health status of individuals whose nutrition status is measured.
- Sex, age and education level of heads of households.
- Size and age composition of households, including dependency ratio or proportion of dependants, where appropriate.
- Sources of food for individuals and households.
- Sources of income for individuals and households.
- Markets: physical access to markets in distance and time, proportions/amounts of food and other items purchased at market.
- Coping strategies used by individuals and households.
- Health and disability status of household members, with details of diseases, chronic sickness and disability.
- Health care access: physical access to health services, constraints to health care access.
- Water access: quantity and quality or households' water sources, distance(s) to them.
- Sanitation facilities: types and extents of usage by household members.
- Housing type and quality: protection from heat, cold, rain, wind, etc.
- Assets: types, numbers and values (when/if possible).

In addition, informants are asked to explain other context-specific characteristics that increase vulnerability to food insecurity.

During the focus group discussion, the interviewer lists the characteristics and links them

(cont...)

56. Technical Guidance Sheet No. 9 Qualitative Data Collection and Analysis, L. Morinière, WFP Emergency Needs Assessment Service, September 2007

(...cont)

to the specific risks identified in Step 2. At the end of the discussion, she/he reads out the list and the linkages to ensure that he/she has understood correctly.

Information from different discussions and interviews must be compared and **triangulated**. If there is solid **convergence of evidence**, it can be concluded that these characteristics contribute to food insecurity. If evidence does not converge, the analysts must look for reasons why. The following are some possible explanations:

- Informants from different groups are talking about different at-risk populations.
- Informants have different perceptions of risk.
- Some interviews and discussions were conducted more thoroughly than others.
- Information has been misinterpreted.

Wherever possible, inconsistencies should be identified during the interview or discussion (see guidelines on interview technique in **Part III**). This facilitates on-the-spot revision of data.

Groups that are at risk are then profiled on the basis of the characteristics described by the informants (see **Section 2.1** for ways of defining groups).

3.4 Step 4: Identify the reasons why people are food-insecure and malnourished and why their livelihoods are at risk

This step is the same for both quantitative and qualitative data analysis, although statistical analysis can be applied to quantitative data to help determine risk factors.

Knowledge of the context (Step 1) is combined with the characteristics of the people facing food insecurity and malnutrition (Step 3) to draw conclusions about the *reasons* why people are at risk:

- The characteristics of **each group** are compared with the nature of the risk affecting the group and the characteristics of the people who are not at risk.
- Judgement and a thorough knowledge of the context are used to draw conclusions about which characteristics are associated with risk.

Example 4.3 illustrates this process.

Example 4.3: Identifying the reasons why people are at risk

A crisis is characterized by:

- high rates of malnutrition, leading to health risks among under-5 children in group X;
- no obvious problem with food security: food consumption, food access and coping strategies all indicate that households are not food-insecure or facing risk to livelihoods.

Households in group X have the following characteristics:

- 1. Livelihoods are based on agriculture, growing crops for own consumption and sale.
- 2. Most household heads are male, aged between 25 and 60 years.
- 3. Education levels are low, particularly among women.
- 4. Health access is good; there is a well-stocked and well-staffed clinic close by.
- 5. Market access and functioning are good, and prices are normal for the season.
- 6. Water quality and quantity are poor; water is collected from a well with a declining yield.

Each of these characteristics is compared with the nature of the risks:

- 1. The type of livelihood does not seem to be a risk-causing factor, as food security is satisfactory.
- 2. There is no obvious causal link between malnutrition and the sex of the household head.
- 3. The low education level among women may lead to poor care, food preparation and water handling practices.
- 4. Households appear to have good access to curative health care.
- 5. Effective markets indicate that farmers should be able to sell their produce and buy essential items.
- 6. Poor water supplies could lead to disease and malnutrition among young children.

On the basis of this analysis, it would appear that the main risk-related characteristics among group X are:

- low levels of female education;
- poor water supplies.

Once these hypotheses are made, further evidence is sought to corroborate or refute them. If random sampling and quantitative data are used, the primary factors associated with malnutrition can be ascertained through regression analysis. Triangulation with information from other sources – key informants, health data, etc. – can strengthen this analysis.

The factors associated with risk are more effectively determined when consultations are held among members of the assessment team, representatives of the affected communities and technical experts. If shortage of time makes it impossible to convene this ideal group, a group of assessment team and country office representatives can undertake the analysis.

Table 4.3 gives some *examples* of people who may face risks to health and livelihoods, and possible reasons why they are at risk.

Individuals or group at risk	Possible reasons for vulnerability	
Children under 5 living in area X	 Mothers spend many hours a day collecting fuel and water, so are unable to dedicate sufficient time to preparing food and feeding their young children 	
School-aged children	Long-term livelihoods are compromised because: • children are taken out of school to work; • children who attend school are malnourished, and therefore unable to learn well	
Nomadic livelihood group: households' main productive activity is raising animals for the sale of animal products	 Drought causes deaths and poor health among animals, making owners eager to sell; livestock prices decrease, and nomads also have fewer dairy products to sell Low livestock prices and reduced sales of products reduce income; food access declines and animals are sold beyond the regenerative capacity of the herd 	
Petty traders livelihood group: households' main productive activity is the trade of small quantities of food and other commodities	 Insecurity makes roads unsafe, reducing the movement of goods and people; income from trade disappears 	
Internally displaced people	Loss of livelihood assets and strategies leads to loss of income and food	

 Table 4.3: Examples of individuals and households who may face risks to health and livelihoods, and possible reasons why

Causality is often **two-way**, and **feedback loops** frequently exist. The following are examples:

- Food insecurity might cause malnutrition, because household members are unable to obtain and consume sufficient quantities or diversity of foodstuffs.
- Malnutrition might cause food insecurity, because productive household members are weakened, become unable to work and, hence, are unable to earn enough money with which to obtain food.

Analysts should be careful not to confuse **association** with **causality**, as illustrated in Example 4.4.

Example 4.4: Association and causality

An agricultural area is affected by floods just before the harvest. Most of the staple crops are lost. Soon after this event, abnormally high levels of acute malnutrition are found among children in the area.

It might be assumed that flood damage to crops has resulted in low food availability and, hence, malnutrition. However, closer analysis reveals that food availability is satisfactory because markets are functioning well. The households with malnourished children are those that depend on daily labour – in normal times they rely on harvesting work to earn money for food purchases. They are unable to work because of crop destruction. The primary cause of the problem is therefore low food access among certain livelihood groups, and not low food availability due to harvest failure.

The choice of appropriate cross-tabulations depends on the context, the hypotheses made when adapting the Food and Nutrition Security Conceptual Framework, and the distribution of the variables in the sample.

3.5 Step 5: Determine whether food insecurity and malnutrition are chronic or transitory

The distinctions between chronic and transitory food insecurity were explained in **Section 2.1**. It is important that an EFSA distinguish between chronic and transitory food insecurity as they are likely to require different types of response, in terms of both the design and the duration of interventions. Interventions aimed at addressing *chronic* food insecurity typically last for several years and focus on the underlying and basic causes of food and livelihood insecurity. Responses to *transitory* food insecurity may focus on the immediate causes of food insecurity and last several months. In some instances, however, it may be important for short-term interventions also to address underlying causes of food insecurity in order to prevent repeated transitory food insecurity that may lead to chronic food insecurity.

To determine whether food insecurity and malnutrition are chronic or transitory,

information about the situation *before* the crisis must be collected. The following questions should be addressed:

- Did food insecurity and/or malnutrition exist before the current crisis?
- If so, what were the nature, underlying causes, extent and severity of the food insecurity and/or malnutrition? How different were these from the current situation?
- Which groups were affected, and how different were they from the groups that are currently facing food insecurity and/or malnutrition?

To answer these questions, it is necessary to analyse:

- pre-crisis data;
- crop production, market and price data for the previous three to five years;
- data on all relevant issues such as malnutrition rates (including stunting), disease prevalence, livelihoods, poverty, food production and market features – for the equivalent period in other years, to account for seasonal factors.

If the pre-crisis situation is not documented, it must be compiled retrospectively.⁵⁷ This is best done using qualitative data from focus group discussions, key informant interviews and secondary data review (see Box 4.4), possibly complemented with retrospective questions included in a formal household questionnaire.

Box 4.4: Compilation of pre-crisis information using a focus group discussion

The characteristics of the food- and nutrition-insecure population were defined in Step 3. In this step, the focus group discusses the pre-crisis situation. Interviewers may ask the following questions:

- Who was affected by food insecurity before the crisis? Locally adapted terminology and concepts should be used to define food insecurity. What coping strategies were used, and by whom? Where possible, groups of individuals and households are defined according to the same criteria used in Step 4.
- If pre-crisis food- and nutrition-insecure groups were similar to those at risk during the current crisis, have the proportions and/or numbers of people facing food insecurity and malnutrition increased?
- If pre-crisis food- and nutrition-insecure groups were different from those that are currently food- and nutrition-insecure, what are the reasons for this?

Information about the nature of the risks (Steps 2 and 4) can also provide indications about whether food insecurity and malnutrition are chronic or transitory. For example:

 Stunting is a sign of long-term malnutrition, and therefore indicates a chronic problem that could be caused by persistent food insecurity and/or a poor health environment.

^{57.} It is important to anticipate the need for these data and to review secondary data early on during the EFSA.

Wasting is a sign of short-term malnutrition, and therefore might indicate a transitory problem of food insecurity and/or infectious disease; wasting can also be due to recurrent, possibly seasonal, problems.

Distinctions between chronic and transitory food insecurity are also linked to the type of factors associated with malnutrition and livelihood insecurity. These variables can be either structural or dynamic:

- **Structural** variables relate to the underlying contextual factors that affect individuals and communities in the area in which they live. These variables do not change quickly, and can influence livelihood outcomes. Such factors can include: local climate, soil type, local governance system, public infrastructure roads, drainage, etc., land tenure, and inter-ethnic relations.
- **Dynamic** variables relate to features that can change quickly. They tend to be indicators of **transitory** problems, which may exacerbate existing chronic problems. Examples include: infectious disease, displacement, change of market functioning, fluctuation in labour demand, ownership of assets, level of indebtedness, labour migration patterns, and size of harvest.

Further guidance on chronic and transitory issues is provided in Technical Guidance Sheet No. 5.58

3.6 Step 6: Estimate the severity of food insecurity and malnutrition

Severity at the population level can be estimated in three ways:

- 1. Through the prevalence of food insecurity and borderline food insecurity, and analysis of food access gaps.
- According to the numbers of individuals and households found to have health and livelihoods at risk, based on nutrition, mortality and food security indicator information (Step 2, see Section 3.2).
- 3. Through convergence of evidence, using multiple indicators.

Where possible, these methods should be combined.

3.6.1 Indicators of risks to lives and livelihoods

3.6.1.1 Mortality and nutrition indicators

There are standard thresholds for mortality and nutrition indicators (see **Part II**). The analysis results can be collated in a template, as illustrated in Table 4.4.

^{58.} Technical Guidance Sheet No. 5 Distinguishing between Chronic and Transitory Food Insecurity in EFSAs, WFP Emergency Needs Assessment Service, November 2007.

Table 4.4: Interpretation of the population-level severity of mortality and nutrition status indicators Indicator Rate (%) Severity at the population level (based on standard thresholds) Crude mortality rate Under-5 mortality rate Prevalence of wasting alobal acute malnutrition in under-5 children Prevalence of stunting global chronic malnutrition in under-5 children Prevalence of low BMI in non-pregnant, non-lactating women

3.6.1.2 Food security and coping strategy indicators

There are no universal standards for food security and coping strategy indicators; the severity of the situation is estimated according to the proportion of the population with an FCS below a certain threshold. However, WFP uses thresholds of FCS \leq 21 for "poor food consumption" and FCS \leq 35 for "borderline food consumption" (see **Part II, Section 7.3.1**).

There are also no universal benchmarks for the CSI. However, in a specific context, some coping strategies used by households show that they endanger livelihoods (selling productive assets, for instance) or even lives (working in conditions or places where physical safety is not guaranteed, for instance).

3.6.2 Using convergence of evidence from a series of indicators

Comparison of a variety of different indicators is an effective way of determining the severity of a crisis. If numerous indicators lead to the same conclusion, and evidence converges (see **Section 2.2**), it is probable that their conclusion about severity is correct.

The Integrated Food Security Phase Classification (IPC) system is a way of compiling indicators systematically and consistently. The IPC approach is summarized in Box 4.5.

IPC indicates when, in a certain area, there are households whose livelihoods are at risk of damage or loss – the "acute food and livelihoods crisis" phase. When the lives of households are at risk, the crisis is in the "humanitarian emergency" and "famine/humanitarian catastrophe" phases.

Box 4.5: The IPC approach

IPC was developed for Somalia by the Somalia Food Security Analysis Unit, and is now being implemented in other countries. It is not an assessment methodology, but a way of collating information from the assessments of several organizations to produce conclusions that are rigorous, transparent and comparable. The results of an EFSA could be *included* in IPC, along with the assessment results of other agencies such as Save the Children, UNICEF, CARE and government bodies. The end result of IPC is a *phase classification* of the crisis in question, according to one of five phases:

- 1. generally food-secure;
- 2. moderately/borderline food-insecure;
- 3. acute food and livelihood crisis;
- 4. humanitarian emergency;
- 5. famine/humanitarian catastrophe.

Classification of the crisis is based on indicators and thresholds. Where possible, internationally recognized standards are used, such as for nutrition data. Where the indicator is context-specific, judgements are made using standard guidance (see references at the end of this box).

The following indicators are used for classification: crude mortality rate; acute malnutrition; disease; stunting; food access/availability; dietary diversity; water access and availability; destitution/displacement; hazards; civil security; coping strategies; livelihood assets; and structural issues.

In addition to the classification of phases, IPC also provides:

- a strategic response framework, with guidance on the priority types of intervention in each phase;
- early-warning levels: watch, moderate risk, high risk;
- colour-coded maps showing the relative levels of food security across a country or area and including information about immediate hazards, key underlying causes, estimated populations, criteria for social targeting, the usual phase prior to the current one, projected trends, and the confidence level of the analysis.

IPC is a useful way of bringing together the various actors involved in food security analysis and combining their conclusions into a standard framework that can be interpreted easily and compared among different crises, and over time in a single crisis. IPC is *not* an assessment methodology, so it is not an alternative to the WFP EFSA; as explained, EFSA results should be a *component* of IPC.

For a detailed explanation of using IPC, see:

- Integrated Food Security Phase Classification: Technical Manual Version 1.1, IPC Global Partners, FAO, 2008.
- Integrated Food Security Phase Classification (IPC) User Guide. Draft-in-progress for feedback. FAO, July 2008.

Additional information is available on www.ipcinfo.org

Table 4.5 gives the indicators and thresholds used in the IPC approach. These can also be used during EFSA analysis to estimate severity. Convergence of evidence from a number of indicators enhances the confidence with which conclusions can be stated.

Table 4.5: Key reference indicators and thresholds used in the IPC approach

Phase Classification			Indicators and thresholds
1A	Generally	Crude Mortality Rate	< 0.5 / 10,000 / day
	Food Secure	Acute Malnutrition	<3 % (w/h <-2 z-scores)
		Stunting	<20% (h/age <-2 z-scores)
		Food Access / Availability	usually adequate (> 2,100 kcal ppp day), stable
		Dietary Diversity	consistent quality and quantity of diversity
1B	Generally	Water Access / Avail.	usually adequate (> 15 litres ppp day), stable
10	Food Secure	Hazards	moderate to low probability and vulnerability
		Civil Security	prevailing and structural peace
		Livelihood Assets	generally sustainable utilization (of 6 capitals)
2	Moderately /	Crude Mortality Rate	<0.5 / 10,000 / day; U5MR<1 / 10,000 / day
~	Borderline	Acute Malnutrition	>3% but <10 % (w/h <-2 z-score), usual range, stable
	Food Insecure	Stunting	>20% (h/age <-2 z-scores)
		Food Access / Availability	borderline adequate (2,100 kcal ppp day); unstable
		Dietary Diversity	chronic dietary diversity deficit
		Water Access / Avail.	borderline adequate (15 litres ppp day); unstable
		Hazards	recurrent, with high livelihood vulnerability
		Civil Security	unstable; disruptive tension
		Coping	'insurance strategies'
		Livelihood Assets	stressed and unsustainable utilization (of 6 capitals)
		Structural	pronounced underlying hindrances to food security
3	Acute	Crude Mortality Rate	0.5-1 / 10,000 / day, U5MR 1-2 / 10,000 / day
	Food and	Acute Malnutrition	10-15 % (w/h <-2 z-score), > than usual, increasing
	Livelihood	Disease	epidemic; increasing
	Crisis	Food Access / Availability	lack of entitlement; 2,100 kcal ppp day via asset stripping
		Dietary Diversity	acute dietary diversity deficit
		Water Access / Avail.	7.5-15 litres ppp day, accessed via asset stripping
		Destitution / Displacement	emerging; diffuse
		Civil Security	limited spread, low intensity conflict
		Coping	'crisis strategies'; CSI > than reference; increasing
		Livelihood Assets	accelerated and critical depletion or loss of access
4	Humanitarian	Crude Mortality Rate	1-2 / 10,000 / day, >2x reference rate, increasing;
	Emergency		U5MR > 2 / 10,000 / day
		Acute Malnutrition	>15 % (w/h <-2 z-score), > than usual, increasing
		Disease	pandemic
		Food Access / Availability	severe entitlement gap;
			unable to meet 2,100 kcal ppp day
		Dietary Diversity	regularly 3 or fewer main food groups consumed
		Water Access / Avail.	< 7.5 litres ppp day (human usage only)
		Destitution / Displacement	concentrated; increasing
		Civil Security	widespread, high intensity conflict
		Coping	'distress strategies'; CSI significantly > than reference
		Livelihood Assets	near complete & irreversible depletion or loss of access
5	Famine /	Crude Mortality Rate	> 2/10,000 / day (example: 6,000 / 1,000,000 / 30 days)
	Humanitarian	Acute Malnutrition	> 30 % (w/h <-2 z-score)
	Catastrophe	Disease	pandemic
		Food Access / Availability	extreme entitlement gap; much below 2,100 kcal ppp day
		Water Access / Avail.	< 4 litres ppp day (human usage only)
		Destitution / Displacement	large scale, concentrated
		Civil Security	widespread, high intensity conflict
		Livelihood Assets	effectively complete loss; collapse
			· · · · · · · · · · · · · · · · · · ·

EFSA results can also be represented on maps, as in IPC (see Box 4.5).

The IPC process requires extensive consultation with partners and the use of standard templates. In a rapid EFSA, consultation may be limited by time constraints, especially if some partners are unfamiliar with the approach. Moreover, some of the indicators listed in Table 4.5 may not be available.

3.6.3 Using the food consumption or food access gap

The severity of food insecurity can be confirmed by two additional indicators:

- the food consumption dap:
- the food access gap.

Further guidance on calculating food gaps is provided in **Section 4.4.1**. When using either of these indicators, the season must be taken into account. In many areas, household food consumption varies during the course of a normal year, and a temporary food consumption gap may not be cause for alarm.

The food consumption gap gives a direct indication of the severity of food **insecurity** at the aggregate population level (see Box 4.6).

Box 4.6: Estimation and use of the food consumption gap

The gap between aggregate food consumption required to meet nutrition needs and actual aggregate food consumption is estimated, by comparing households' food intake with the intake and thresholds established from reference nutrition requirements. The difference between the reference threshold and the households' score indicates the severity of the gap. This approach can be used to estimate the number of households called food-insecure because of a deficient diet.

However, food intake data are hardly ever available.

The **food access gap** (see Box 4.7) can be estimated by comparing household food expenditure with the cost of a minimum food basket, taking into consideration the proportion of food that is not purchased, such as food coming from own production. The food access gap can be a useful indicator in livelihood groups that purchase most of their food.

Box 4.7: Estimation and use of the food access gap

Average current expenditure on food for a given livelihood group is estimated through a questionnaire survey or focus group interviews. The quantity and monetary value of food produced and consumed by households are also estimated, and compared with the cost of a minimum local food basket, estimated through a market survey.

This approach can give an indication of **major food access shortfalls**. People often under-report their food expenditure, so it can be difficult to obtain accurate figures in less extreme circumstances, such as when there is a relatively small food access shortfall.

There are no standard references against which to judge the severity of the food consumption or food access gap. The following rule of thumb can be used:

- If more than 10 percent of the population is facing a severe food gap, there is a critical crisis.
- If more than 30 percent of the population is facing a moderate food gap, there is a severe crisis.

Conducting a forecast analysis

This section explains the following steps for forecasting and scenario development:

- Identify opportunities and shocks that are likely to affect the area in the future.
- Combine an analysis of shocks and opportunities with their influence on household food security to develop scenarios that describe how the situation might develop in the future.
- Identify the most likely scenario and the groups whose food security will be at risk under this scenario.

Forecasting is, by nature, uncertain. Uncertainty can be reduced by using the best information available and rigorous analytical procedures. There is always an element of **judgement**, however. Analysts must decide what they consider the most likely outcome, based on the available information. The forecast should then be qualified by the reliability of the information on which it is based. A forecast includes **assumptions**, which must be clearly documented in the assessment report, along with the process through which conclusions were developed. At a minimum, a forecast analysis should result in the following outputs:

- a forecast of the future opportunities and shocks that are likely to affect the food security and nutrition situation;
- scenarios that forecast the evolution of the food security and nutrition situation in the absence of assistance; and
- identification of the groups that will be most at risk in the most likely scenario.

4.1 Identification of future opportunities and shocks

In order to develop reasonably accurate forecasts, it is necessary to identify the range of **opportunities** and **shocks** that may affect the future nutrition status and food security of a particular population.

Box 4.8: Opportunities and shocks

In EFSA analysis, **opportunities** are events that ameliorate or improve an adverse situation. They can be:

- natural opportunities, such as resumption of rain after a dry season;
- human-induced opportunities, such as the next harvest, peace accords, opening of previously closed borders, and improvement of market infrastructure.

Opportunities can arise through a combination of natural and human-induced occurrences. For example, a peace agreement means that roads will be opened and markets may become better integrated. Good rains contribute to an improved harvest. These two opportunities will combine to enable farmers to sell larger quantities of produce at good prices in the coming months.

In an EFSA, **shocks** are events with a negative impact on nutrition status and/or food security. They may be:

- natural shocks, such as earthquakes, drought or floods;
- human-induced shocks, such as conflict or economic recession.

Natural and human-induced shocks are not always easy to differentiate. For example:

- floods can be caused by seasonal climatic fluctuation, exacerbated by deforestation;
- conflict can be caused by political tension, exacerbated by drought.

Opportunities and shocks do not necessarily occur at distinct times; some **evolve** gradually. For example:

- an improvement of physical infrastructure and services takes effect slowly;
- the impact of a drought increases slowly.

When analysing opportunities and shocks, both sudden-onset and evolutionary variants should be considered.

The importance of assessing the **current** risks to food security during the situation analysis was discussed in **Section 3**. In forecasting, each of the factors related to food security is reviewed, to determine whether the same situation is likely to persist in the future. Additional events – opportunities or shocks – that are not currently present should also be identified during a forecast analysis.

The analysis is informed by the following:

- The nature of a potential shock: Is it a one-off event, such as an earthquake, or a long-term and complex process, such as conflict or environmental degradation?
- The opinions of experts and key informants: For example, meteorologists and environmental experts might provide input on trends related to the natural environment; local NGO and social workers might provide input on social trends; economists might help with market predictions; and political analysts might provide input on the evolution of a conflict.

A procedure for predicting future opportunities and shocks is explained in Box 4.9

Box 4.9: Predicting opportunities and shocks

1. Start with the existing situation. What are the features of the current crisis and are they likely to persist into the future? For example, households' food consumption is found to be poor because recurrent droughts have reduced agricultural production and limited the opportunities for daily labour. *This problem will probably persist for the foreseeable future*.

2. Consider various time periods, such as three, six and twelve months. What new shocks and opportunities are likely to arise? The following are examples.

The harvest is due in one month. Farmers usually sell 80 percent of their production to traders, who sell it overseas. However, neighbouring countries have closed their borders for the export of agricultural produce. It is probable that when the harvest comes, farmers will be forced to sell their produce at reduced prices, thus curtailing their income. A **shock** to local livelihoods can be predicted.

The harvest is due in one month, and it looks as though it will be very good. Markets are recovering, and there is high demand for local produce. This should represent a good **opportunity** for farmers, who can expect their income to rise within the next month.

Some shocks, such as tsunamis, earthquakes and volcanic eruptions, are extremely difficult to predict. Certain parts of the world are far more susceptible to this type of shock than others. In these areas, **contingency plans** should be developed to ensure preparedness in case of shock (see **Section 4.2**). Historical trends of shocks can allow greater confidence in forecasting future shocks.

EFSAs generally take place in areas that are already facing a crisis or in which a crisis is predicted. The approach explained in Box 4.9 is therefore appropriate for most EFSAs.

The following information should be noted for each opportunity or shock:

- Recurrent/persistent or occasional: This defines the nature of the shock or opportunity and provides important insight into its likely evolution. Persistent shocks are continuous or recurrent, such as long-term drought. Occasional shocks are one-off, such as earthquakes. An equivalent approach is applied to opportunities.
- Probability of occurrence: Some shocks and opportunities can be predicted with more confidence than others. For example, the arrival of the rainy season can be predicted with reasonable confidence – it occurs at more or less the same time every year – although the amount of rain that falls may vary greatly from year to year. The return of refugees following a conflict may be less easy to predict as it depends on numerous factors, all of which are uncertain, such as the signing of a peace accord, the availability of transport, and refugees' perception of the security situation.
- **Expected time of occurrence**: The timing of some opportunities and shocks, such as a harvest or a hunger gap, can be predicted reasonably accurately. Others, such as earthquakes, are much more difficult to predict.
- Scale of the severity of a shock or the benefit of an opportunity: some shocks are more severe than others, and some opportunities bring greater benefits than

others. For example, a pest attack on crops affecting a small proportion of farms is less severe than a flood that destroys large swathes of farmland; a comprehensive peace agreement conveys more benefits than a temporary drop in food prices due to a one-off localized food distribution.

Table 4.6 gives examples of how shocks and opportunities may be recorded for a forecast analysis. This template should be adapted to reflect the context of the shocks and opportunities that are likely to occur in a particular region.

Table 4.6: Documentation of opportunities and shocks, with examples				
Expected event	Recurrent / persistent or occasional	Probability of occurrence 1 = low 5 = high or ongoing	Expected time of occurrence	Scale of severity or benefit
Shocks				
Earthquake	Occasional	1	Any time	High severity
Drought	Recurrent	Ongoing	Ongoing	Medium severity
Attack by armed groups	Occasional	3	Any time	High severity
Opportunities				
Harvest	Recurrent	4	7 months from now	High benefit
Establishment of rural health service	Occasional	3	1 month from now	Medium benefit
Peace accord	Occasional	2	Unknown	High benefit

Examples of different types of shock and their effects on food security are given in Table 4.7. Shocks affect people in different ways, depending on their individual or group characteristics. The following are examples of this:

- During periods of insecurity, women may be at greater risk than men because they have to walk long distances to collect water, fuelwood, etc. and are targets for violence.
- When people are displaced, women's coping strategies may have more severe consequences on their lives and livelihoods than men's.
- Children are more vulnerable to diseases than adults, such as when water supplies are contaminated during floods.
- Certain ethnic groups may be targeted during conflict. Other groups may be denied access to areas where they farm or carry out other livelihood activities.

It is therefore essential to undertake a **disaggregated** analysis of the potential impact of shocks. Populations should be disaggregated according to:

- **sex** always;
- age always;

- livelihood group, particularly for slow-onset shocks and conflict;
- **health status**, particularly regarding chronic diseases such as HIV/AIDS and acute infections in young children always;
- ethnic or social group, such as IDPs, refugees and host families only if relevant to the specific emergency;
- other, locally relevant criteria, for example, by location, such as coastal or mountain, when it implies different exposures to risk.

Table 4.7: Shocks and their potential impacts on food security				
Shock	Potential direct effects on food security			
	Main immediate effects	Food availability and access	Food consumption	
Drought Crop pest, such as locust invasion, crop disease	Loss of harvest Loss of food stocks	 Decreased staple food availability Decreased staple food access from own production, for sale and consumption Decreased food access from purchase, due to increased market food prices, decreased sales 	 Decreased amounts of food consumed, due to decreased availability and/or increased prices Lower quality of diet by choice (coping) and/or availability 	
 Flood Tsunami Hurricane, cyclone 	 Loss of harvest Loss of food stocks Loss of economic infrastructure: workplaces, roads, etc. Loss of assets 	 Decreased staple food availability Decreased staple food access from own production Decreased food access from purchase, due to increased market food prices, loss of income, decreased sales Decreased employment opportunities 		
Earthquake	 Loss of food stock Loss of assets Loss of animals 	 Decreased staple and animal food access from own stocks Decreased food access from purchase, due to loss of income Decreased food availability 		
 Animal disease Livestock thefts and looting 	 Loss of animal products Loss of animals 	 Decreased animal food access Decreased food access from purchase, due to loss of income, decreased sales 		
 Market food price rise Economic collapse 	 Deterioration of terms of trade for livestock or labour Loss of purchasing power 	Decreased food access from purchase, due to loss of income		
 Forced displacement Conflict 	 Loss of harvest Loss of animals Loss of assets 	 Decreased staple food and animal product availability Decreased food access from purchase, due to increased market food prices, loss of income, decreased sales of own production 		
Epidemics, such as cholera HIV / AIDS	• Disease	 Decreased food access from purchase, due to increased health expenditures, decreased income earnings because of loss of physical capacity and extra time required to care for the sick 	 Decreased amounts of food consumed, due to loss of appetite, lack of time to care for vulnerable individuals Loss of nutrients 	

4.2 Developing scenarios

In forecast analysis, assessors develop possible future scenarios. A **scenario** is "a *description of situations that could occur; it is a set of informed assumptions about a situation*".⁵⁹ Scenarios indicate alternative ways in which the situation might evolve, based on: (i) **current food insecurity**; (ii) **assumptions** about possible **future shocks** and **opportunities**, taking into account the type of emergency and its volatility; and (iii) people's **resilience and vulnerability**. The *most likely* scenario is chosen as the basis for predicting the coming three, six and/or twelve months; the period depends on the purpose of the EFSA and the data that have been collected. This process is explained in detail in following sections. A simplified example is given in Example 4.5.

Example 4.5: Scenario development

Note: This is a simplified example using the hypothetical situation of a rural area affected by conflict and recurrent drought.

Step 1: Identify future shocks and opportunities that could affect the food security and nutrition situation.

- The harvest is expected in one month: rains have been good, although the area planted has been 20 percent smaller than usual. The harvest is expected to be average compared with long-term trends, but much better than the average for the last ten years.
- Increased conflict is expected in area X because of its strategic importance: troops from both sides are massing in this area; and populations are moving from area X to the border, which is currently closed.
- Other parts of the country seem to be relatively stable.

Step 2: Develop scenarios to anticipate the evolution of the food security and nutrition situation in the absence of assistance.

One or more scenarios is/are developed, depending on the volatility of the situation. In general, the *most likely scenario* is used for planning, but in some cases a worst-case scenario could be used for additional contingency planning. In this example, the most likely scenario might be as follows:

- In most parts of the country, food availability will improve because of the relatively good harvest. Food access will improve because of enhanced labour and trade opportunities. Food utilization is unlikely to change significantly because long-term health issues have not been addressed: access to health care and water quality.
- In area X, all food security factors are likely to deteriorate. Nutrition problems are probable, especially in IDP settlements.

Step 3: Identify the population groups affected by the most likely scenario, and the impact of the shocks and opportunities on their livelihoods.

- In general, vulnerability is decreasing because of improved harvests and reduced conflict.
- In area X, vulnerability is expected to increase because of the fighting. IDPs moving from area X are extremely vulnerable during their move and when established in makeshift camps on the border.

59. Contingency Planning and Humanitarian Action, a Review of Practice, R. Choularton, Overseas Development Institute (ODI) Network Paper No. 59, March 2007.

4.2.1 Procedures

One or more scenarios can be developed, depending on the level of uncertainty that surrounds future events. If the future can be predicted with a high level of confidence, one scenario might be sufficient. If the crisis is complex and has the potential to evolve in several different ways, it might be necessary to develop more than one scenario and to judge which is the most likely. In certain cases, shocks are almost impossible to predict accurately, so contingency plans are developed (see Section 4.2.2).

Wherever possible, scenarios should be developed through consultation; the extent to which this is possible depends on the time available and the degree of collaboration among stakeholders. It is critical that EFSA team members work with local key informants and counterparts from partner agencies to decide the most likely evolution of future events. The likelihood of a given scenario is based on the **collective judgement** of the group; there is no standard way of determining likelihood.

The period covered by a scenario depends on the following:

- The type of emergency: For example, a rapid-onset emergency that is limited in geographical scope, such as a flood, may necessitate a scenario covering the coming three months, which might be updated later. A slow-onset, persistent emergency, such as a drought, may demand a scenario covering a year or more.
- The type of operation that WFP is planning: For example, an EMOP may cover a period of six to twelve months, while a PRRO may last for three years.

Scenarios are developed by considering all the potential opportunities and shocks identified in **Section 4.1** and making assumptions about their combined influence on food security in the near future. For example, the harvest is imminent, which is a good opportunity for food-insecure households. Fighting is currently intensifying, however, and soldiers frequently loot or destroy crops. This is likely to continue at least until the onset of the rainy season in five months' time, and the shock caused by the fighting is expected to obliterate the potential opportunity of the harvest.

To develop realistic scenarios that are not too complex, it is necessary to identify the **dominant opportunities and shocks** that will have the greatest influence on how the food and nutrition security situation evolves in the coming months. In the example in the previous paragraph, the fighting is the dominant event, because its effects overcome those of the coming harvest. Identifying the dominant shock saves time; the fighting is expected to obliterate the benefits of the harvest, so there is no point in engaging in in-depth analysis of the harvests' benefits for the population. The following are some examples of dominant opportunities and shocks:

- Large-scale droughts and floods are natural shocks that are not affected by other events, at least in the short term.
- Major armed violence is likely to override most other events.
- Government policies that are directly related to food and nutrition security may represent supportive opportunities or harmful shocks.

Dominant opportunities and shocks steer the development of scenarios. **Secondary opportunities and shocks** might be: (i) *caused* by the dominant event, as a cascade effect; (ii) unrelated to the dominant event but *enhancing* or *mitigating* its effects, as a synergistic effect; or (iii) subsumed by the dominant event. The following are simplified examples of these three possibilities:

- Cascade effect: Armed violence is the dominant shock leading to large-scale displacement; displaced people move to town, where they live in overcrowded slums, resulting in the spread of disease and increased severe malnutrition as secondary shocks.
- **Synergistic effect**: A drought is the dominant shock, which is exacerbated by the introduction of government policy that constrains the movement of food among districts in a country as a secondary shock.
- **Subsumed effect**: An earthquake is the dominant shock causing mass loss of life and destruction of infrastructure. This subsumes the effects of a localized pest infestation as a secondary shock. It may not be worth analysing the effects of the pest infestation because the earthquake and its effects dominate the scenario.

The existence of a dominant shock does not mean that secondary shocks should not be analysed. Many scenarios are characterized by a variety of different shocks interacting to produce a **composite shock**, as in the following examples:

- Low rainfall leads to a poor harvest; conflict results in reduced mobility, loss of access to fields, and looting of crops; and deterioration of roads leads to increased market transaction costs. These shocks combine to cause an escalation of grain prices, with each factor exacerbating the negative impacts of the others.
- Low rainfall and deterioration of roads put upwards pressure on the grain price because of poor harvests and high market transaction costs. However, resolution of conflict mitigates these negative effects to some extent. The three factors are analysed together to determine the net impact on the population.

The population groups that might be affected in a scenario are identified broadly, based on their main livelihood characteristics, particularly their sources of food and income, and their geographical location, as in Example 4.6. More detailed characteristics and numbers of affected people are estimated after the most likely scenario has been identified (see **Sections 4.3, 4.4** and **4.5**).

Example 4.6 illustrates how different types of shock interact.

Example 4.6: Formulating a scenario in a relatively straightforward situation

A rural, primarily agricultural area has been affected by drought for the last three seasons. The population is poorer than the national average, and people's purchasing power has declined since the drought began. This year's rains have been good. Large-scale seed distributions have allowed farmers to plant the same area as they plant in a normal year:

- A good harvest is expected in one month.
- Physical access to the area is poor because of bad roads, which are sometimes unusable during the rainy season. Poor physical access and low profit margins make the area unattractive to traders. A road construction programme started a year ago, and should be completed within the next six months. This will allow year-round access to the area.
- The area has been beset by low-level insurgency against the government for the last ten years. Since the recent collapse of peace negotiations, the conflict has escalated severely in district X, which has experienced many casualties and widespread looting.

1. Identification of opportunities and shocks

Opportunities and shocks are first identified in isolation, and the probability of each occurring is judged and assigned a value of 1 to 5, with 5 representing a certainty.

Opportunities

Good harvest in one month – probability: 4. Improved road in six months – probability: 4.

Shocks

Escalation of conflict, with high loss of life and looting – probability: 3.

2. Combination of shocks and opportunities

The positive effects of both the harvest and the road improvement have a high probability of occurring *if these events are considered in isolation*. However, escalation of the conflict, although slightly less likely to occur, would probably outweigh the benefits of the harvest and road construction. With a probability of 3, conflict escalation has to be taken seriously.

3. Development of the scenario

Conflict escalation is judged to be the dominant event, which overrides the others in its effect on lives and livelihoods. In this scenario, it is assumed that conflict escalation will occur. The interaction between the dominant shock and the two **secondary opportunities** is then considered:*

- The harvest is one month away. Although it is expected to be good, its positive impact will be reduced by: (i) an expected loss of 20 percent of the crop to looting in district X; (ii) disruption of markets by conflict, making it more difficult for farmers in the area to sell produce; and (iii) a forecast displacement of 30 percent of the population of district X because of the fighting.
- Road construction is due to finish within six months. Given the escalating security threats, it is likely that the contractor will halt construction pending resolution of the problems.
- Markets are unlikely to pick up as previously hoped, because of the continued bad physical access with no road, the poor security, incurring risks to lives and high transaction costs, and the low purchasing power of the population.

Based on this, the probable scenario is as follows:

 Conflict escalates, particularly in district X, where it leads to displacement of 30 percent of the population. Food availability declines, owing to loss of harvested crops, and food stocks throughout the area are stretched further by the presence of IDPs from district X. Markets do not provide an effective response.

* Percentages in this example are arbitrary and provided for the sake of illustration. The information on which these estimates are based could come from key informants, focus groups and/or household surveys. Consideration of dominant and secondary opportunities and shocks and their relative probabilities helps to determine whether the situation is likely to **improve**, **deteriorate** or **stay the same**. The situation in Example 4.6 is likely to deteriorate drastically. Sometimes the outcome is less obvious, as shown in Example 4.7.

Example 4.7: Formulating a scenario in a relatively unclear situation

The situation is the same as in Example 4.6, but the probability is different.

Opportunities

Good harvest in one month – probability: 4. Improved road in six months – probability: 4.

Shocks

Escalation of conflict, with high loss of life and looting - probability: 1.

In this case, the outcome is less clear than in Example 4.6. The probability of severe escalation is low but still significant; given the implications of such an escalation, it would be unwise to discount this possibility altogether. Conflict does not constitute the **most likely scenario**, however, which could instead be postulated as follows:

 The harvest is good, leading to a significant improvement in food availability. Prices in the market decrease, improving food access. Some traders are prepared to traverse the poor roads to buy produce from the area, which now has a food surplus. Over the next six months, trade increases greatly, owing to completion of the road and integration of the area's markets into the national market system. In areas where conflict continues, there are looting and small-scale, temporary displacement.

In this type of situation, two or more scenarios might be developed:

- Most likely scenario: This describes the situation most likely to occur, given the likelihood of each of the opportunities and shocks and the interactions among them.
- Worst-case scenario: When an alternative, worse scenario is less likely to occur, but still has a possibility of occurring in this example, conflict it should also be considered for contingency planning and preparedness (as in Example 4.6).

The situation should be monitored constantly to identify promptly any deterioration that might lead to the worst-case scenario.

4.2.2 Contingency planning

It is advisable to base recommendations for response planning (see **Chapter 5**) on the most likely scenario, but to make **contingency plans** according to the worst-case scenario.

The reliability of the information used to develop the scenarios should also be taken into account. If confidence in the information is low, such as in a rapid EFSA carried out in a short time without full access, it is advisable to prepare two or three scenarios, select the most likely, and include the worst-case for contingency planning. If information is known to be accurate and can be treated with a high level of confidence, a single planning scenario can suffice.
The scenarios can be summarized as in Table 4.8, which includes simplified examples of two scenarios.

Table 4.8: Scenarios and their impacts, examples				
Scenario and probability (5 = certainty)	Period	Opportunity or shock	Location	
1. Situation will improve: probability 4	0-6 months	Good harvest: opportunity	Districts A and B	
		Floods: shock	Villages along river	
2. Situation will deteriorate: probability 2	0-6 months	Major crop failure due to late failure of rains: shock	Districts A and B Town C	

Of the two scenarios in Table 4.8, one is considerably more probable than the other. However, given the grave consequences of scenario 2 - major crop failure contingency plans should be made for this scenario. Indicators should be defined and closely monitored to determine whether or not scenario 2 is developing.

4.3 Identification of population groups affected under the most likely scenario

After developing the most likely scenario, the next step is to identify the groups and numbers of people likely to be negatively affected under this scenario. This involves:

- comparing the most likely scenario with the existing situation, and determining whether the food security of the same groups would be at risk in the future;
- identifying additional groups whose food security would become at risk in the future owing to the effects of each of the opportunities and shocks identified in the scenario.

4.3.1 Population groups currently at risk

The most likely scenario is compared with the existing situation. The scenario may predict a continuation of an existing situation, such as a long-lasting drought or conflict. In this case, the profiles of the groups whose food security is at risk will be similar to those developed in the situation analysis (see **Section 3.3**), with the following modifications:

- If the crisis is expected to become more severe, the numbers of people at risk are likely to rise, and vice versa.
- Additional population groups may be put at risk if the same crisis persists, such as during a drought. To begin with, only the poor with few assets are at risk. As the drought continues, the assets of more wealthy groups are depleted, putting these people at risk. Identification of these groups is explained in Section 4.3.2.

4.3.2 Additional population groups expected to become at risk

The **main characteristics** of additional groups likely to be affected by each of the opportunities and shocks in the scenario are identified. The following are examples:

- If armed attacks are expected to target particular ethnic groups, those ethnic groups are likely to be the most affected.
- If it is predicted that the market system will be disrupted, such as through border closures, the people whose livelihoods are based on the market, such as producers and traders, and those whose food access is based on purchase at the market will be affected.
- If particular rivers are expected to flood, the people living nearby will be affected first, followed by those who depend on the rivers for their food and income.

Having determined the main characteristics of the groups that will be at risk in the future, detailed **profiles** are compiled using information about the population gathered from primary and secondary sources:

- With quantitative data, group profiles are developed through cross-tabulation of the primary livelihood characteristics – IDP, farmer, etc. – of populations whose food security is at risk, as identified during the situation analysis (see Section 3.3).
- With **qualitative data**, information collected during focus group discussions is analysed to identify the groups that are likely to be affected in the future. Their profiles are developed in the same way as in the situation analysis.

Example 4.8 illustrates this procedure.

Example 4.8: Forecasting the effects of opportunities and shocks on population groups

In a particular area, much of the food in the markets comes from neighbouring countries.

Note: This is a simplified example for illustration purposes. In a real situation the analysis will be more complex, although the principles remain the same.

Under the most likely scenario, it is predicted that border closures will halt food imports for at least six months. This will result in the doubling of staple food prices. This shock is expected to have the following effects on the population:

- People who depend primarily on market purchases for their food will be adversely affected.
- Local farmers who are able to sell their produce during this period will benefit from the raised prices.

If the EFSA data are **quantitative**, households that are primarily dependent on market purchase for their food are identified by looking at the share of household food that comes from this source. These households are then cross-tabulated against other key characteristics, such as gender, age and displacement status. The profiles developed are used for subsequent targeting and monitoring.

If the data are **qualitative**, the livelihood profiles developed during focus group discussions are used to identify the groups that are particularly dependent on market purchases.

The same procedure is used to identify the groups that will *benefit* from the shock: in this case, the farmers who are able to sell their produce.

The next step in forecast analysis is to categorize groups according to the degree to which their lives and livelihoods are likely to be at risk under the scenario (see **Section 3.3**), based on the expected impacts on mortality, nutrition status, food security and coping strategies.

Table 4.9 provides examples of the groups that are likely to be affected by particular shocks. These examples are for the purpose of illustration; the effects of shocks on different groups should always be analysed within the specific context. Both **directly** and **indirectly affected** groups should be identified. For example, during a conflict in part of a country:

- people who are targeted by warring factions are directly affected;
- food traders and consumers who depend on produce from the conflict-affected area are indirectly affected, as their livelihoods and food consumption are damaged by the conflict.

Shocks	Livelihood groups likely to be most affected	Potential effects on livelihood assets	Possible alternative / complementary livelihood strategies
Drought Crop pest: such as locust invasion, crop disease	 Subsistence farmers Landless agricultural labourers: loss of labour Pastoralists: drought Consumers dependent on markets for food 	 Human: malnutrition from decreased food consumption Physical: sale of tools, equipment, animals - decapitalization Financial: decreased income from decreased sales of crops and animals; decreased access to credit from difficulties with reimbursement; decreased access to food Natural: overexploitation of grazing areas and other natural resources, such as forest 	 Decreased expenditures on food and essential non-food items and services, such as health and education Increased indebtedness Use of savings Preferences for lower-yielding bu more drought- and pest-resistant crops and animals Migration in search of labour, grazing land, water
Flood Tsunami Hurricane, cyclone	 Subsistence farmers Landless agricultural labourers: loss of labour 	 Human: malnutrition from decreased food consumption; disease from unsafe water; missed education opportunities from destruction of schools, longer distances to school Physical: loss of tools, equipment, animals, housing - decapitalization; loss of infrastructure such as roads, bridges, health services, schools Financial: decreased income from decreased sales of crops and animals; decreased access to credit from difficulties with reimbursement Natural: losses from erosion, landslides 	 Decreased expenditures on food and essential non-food items and services, such as health and education Increased indebtedness Use of savings Migration in search of shelter and labour
• Earthquake	 All livelihood groups with no/limited assets: human, financial, physical 	 Human: missed education opportunities from destruction of schools, longer distances to school Physical: loss of equipment, animals, sometimes housing - decapitalization; loss of infrastructure such as roads, bridges, health services, schools Financial: decreased access to credit from difficulties with reimbursement 	 Decreased expenditures on food and essential non-food items and services, such as health and education Increased indebtedness Use of savings

Table 4.9: Shocks and their impacts on different groups, examples

Shocks	Livelihood groups likely to be most affected	Potential effects on livelihood assets	Possible alternative / complementary livelihood strategies
 Animal disease Livestock theft and looting 	 Pastoralists Small agropastoralists 	Human: malnutrition from decreased animal food consumption Physical: loss of animals Financial: decreased income from decreased sales of animals and their products; decreased savings; decreased access to credit from difficulties with reimbursement	 Decreased expenditures on food and essential non-food items and services, such as health and education Decreased cultivation from loss of draught power or increased cultivation Migration in search of security
 Rise in market food prices Economic collapse 	 Casual workers Landless agricultural labourers Pastoralists 	Human: malnutrition from decreased food consumption Financial: decreased value of cash savings; decreased access to credit from difficulties with reimbursement	 Decreased expenditures on food and essential non-food items and services, such as health and education Sale of assets - decapitalization Increased indebtedness Use of savings Migration in search of labour
 Epidemic, such as cholera HIV / AIDS 	 All livelihood groups with no/limited assets: human, financial, physical 	 Human: malnutrition, disease, mortality Social: disruption of community, kinship and other social networks - solidarity stretched Financial: decreased access to credit from difficulties with reimbursement 	 Decreased expenditures on food and essential non-food items and services, such as education Sale of assets - decapitalization Switch to lower-earning and lower-yielding crops that are less labour-intensive, because of poor physical capacity, time needed to care for sick individuals Increased indebtedness Use of savings Sending children to relatives
Forced displacement Conflict	 All livelihood groups, particularly those with weak social assets/solidarity networks 	 Human: malnutrition, missed education opportunities Social: disruption of community, kinship and other social networks from dispersion of members; creation of war economies Physical: loss of tools, equipment, animals, housing Financial: decreased income from decreased sales of crops and animals; loss of savings from theft Natural: overexploitation 	 Decreased expenditures on food and essential non-food items and services, such as health and education Migration in search of security and labour

4.3.3 Combining current and predicted population groups facing risk to food and nutrition security

As defined in **Sections 4.3.1** and **4.3.2**, groups that will be at risk under the future scenario(s) include:

- groups whose food security is currently at risk and will remain so;
- groups whose food security is not currently at risk, but *will become so* in the future.

Population groups that will *remain* at risk and groups that will *become* at risk are *added together* to provide the total of groups whose food and nutrition security is expected to be at risk in the future. Groups that are *currently* at risk but whose situation is expected to improve to the point at which they are *no longer* at risk are not included on the list.

Caution: It normally takes a long time for risk levels to decline because crises typically entail loss of assets, displacement, etc. In the short term – say, for the next three months – groups *currently* at risk are therefore usually expected to *remain* at risk. Each situation has to be carefully analysed to determine the point at which the risks have declined to a level where affected groups are no longer at risk. For example, following a drought, farming communities may remain at risk for several years after the first harvest, because they have to rebuild the assets that were lost during the drought, or sold to cope with the situation.

Example 4.9 illustrates how this analysis is applied.

Example 4.9: Combining forecast opportunities and shocks with current population groups at risk

The situation analysis indicates that agricultural labourers are among the groups whose food security is currently at risk, because last year's harvest was destroyed by floods and labourers were unable to accumulate enough money to buy food during the dry season, when little work is available. Agricultural labourers are predominantly from ethnic group X.

Forecast analysis indicates that the following events are likely to happen in the coming months:

- The planting season is expected to start in one month. Farmers have already received seed from agricultural relief programmes, so are expected to plant substantial areas in spite of last year's poor harvest.
- There is growing civil unrest, and local leaders predict that low-intensity armed conflict will erupt within the next three months. Last time this happened, ten years ago, nearly all members of ethnic group Y, who are currently among the more affluent members of the population, were displaced to the neighbouring region.

In this simplified example, the following conclusions might be drawn:

- Although the agricultural labourers are at risk now, within the next month their situation should improve, as there will be abundant work opportunities.
- The lives and/or livelihoods of ethnic group Y are not currently at risk. If conflict erupts as predicted, however, group Y will be forced to move and its situation will deteriorate dramatically.
- Therefore, although at the moment the livelihoods of agricultural labourers are most at risk, within the next three months the situation is likely to change, with members of ethnic group Y becoming the most at risk.

This example is simplistic, for the purposes of illustration. In reality, each group must be analysed in terms of its **vulnerability** to the threats that it currently faces and is likely to face in the future. In this example, the following issues might be taken into account:

- Will the labourers find sufficient work to counteract the deterioration in their assets that resulted from last season's problems? The situation analysis may indicate that they have sold productive assets. To regain their livelihood security, the labourers will therefore need to replace their assets, including savings, *and* supplement these with enough additional savings to sustain themselves through the next lean period.
- Ethnic group Y is relatively affluent and owns some of the land on which the labourers find work. Will the expected displacement of group Y affect labour opportunities?
- How will the coming conflict affect agricultural activity? Will the expected benefits be achieved? For example, landowners might delay planting their fields because of the political uncertainty.

The fact that ethnic group Y will probably relocate does not necessarily mean that it
will be more vulnerable than other groups. People in this group are currently relatively
affluent, and probably have savings and other resources in different locations. As the
conflict is predicted for three months from now, it is probable that members of ethnic
group Y are making provisions and moving assets out of the area.

4.4 Estimation of the impact of shocks and opportunities on livelihoods

The selected scenario's impact on livelihoods must be estimated for each group concerned. This can be done by estimating the ways in which existing sources of food and income will be affected. Coping strategies are taken into account during this analysis. Examples 4.10 and 4.11 illustrate two ways of making this estimation, one based on quantitative data, the other on qualitative data.

Example 4.10: Estimating the impact of a shock using quantitative data

During a household survey, the following data about sources of food and income for a given livelihood group are collected.

Food sources

- 50 percent from own production.
- 50 percent bought at market.

Income sources

- About 10 percent from sale of handicrafts.
- About 50 percent from seasonal labour.
- About 40 percent from sale of livestock products.

Expenditures

- Food: XYZ 3,000 per year, representing 67 percent of total expenditure.
- Other essential expenditure, such as health care, school and clothing: XYZ 1,500 per year, representing 33 percent of total expenditure.

Total essential expenditure: XYZ 4,500 per year.

According to the **most likely scenario**, the labour market will collapse because of fighting in the area where people go to work. As a result, access to labour is expected to drop by 75 percent, with only 25 percent of the labour market remaining accessible. This means that instead of providing 50 percent of total income, seasonal labour will now cover only about 12 percent: that is, 25 percent of 50 percent. This leaves an income shortfall of about 38 percent.

The situation analysis has shown that people cope with such a shortfall in the following ways:

- Selling animals: At current market prices and assuming that no more than two animals can be sold before herd sizes are reduced to unsustainable levels, this may bring an extra XYZ 1,000, or 22 percent of total expenditure.
- Reducing food consumption and diet diversity: This may save between XYZ 500 and XYZ 1,000, or 11 to 22 percent of total expenditure.

It is therefore predicted that there may still be a 10 percent shortfall in household income, depending on how drastically food consumption is adjusted.

Example 4.11: Estimating the impact of a shock using qualitative data

Poor rainfall leads to the prediction that the harvest will be 50 percent less than normal: a shock. A major road construction project is about to start in the area: an opportunity. The combined impact of the shock and the opportunity is estimated for farming households.

The following details of farmers' livelihoods *in a normal year* are identified during focus group discussions:

- Farming households eat 50 percent of their produce and sell the rest.
- Own produce covers consumption needs for seven months a year.
- For the remaining five months, farming households buy food using the money from their food sales, which is estimated to cover about three months of food needs; and undertake casual labour on other people's farms and construction projects, which is estimated to cover about two months of food needs.
- In a normal year, these strategies enable farming households to cover all their food and non-food needs, while retaining savings equivalent to 10 percent of their annual expenditure.

The **impact** of the predicted 50 percent reduction in the harvest is estimated as follows:

- It is assumed that households will continue to eat 50 percent of their produce and sell the rest – this can be checked in the focus group discussions. Own produce will therefore now account for 3.5 months of food instead of seven. Because of the poor harvest, food prices in the market are relatively high. Food sales will now enable the household to buy approximately half the usual amount of food, that is 1.5 months of food needs.
- Own produce, both consumption and sales, is therefore expected to cover approximately five months of food needs instead of the usual ten months.
- Because of the poor harvest, households are unable to find any seasonal labour on other people's farms.
- Road construction will provide employment for at least one member of each farming household for some portion of the coming year. It is estimated that this will provide approximately four months of food needs per household.
- Households will experience a food shortfall and will mobilize some of their savings from previous years, estimated at 1.5 months of food needs.

Combining all of these figures, the extent to which farming households might be expected to cover their food needs in the coming year is estimated as follows: Own consumption: 3.5 months

It is therefore predicted that farming households will experience a food gap of 1.5 months in the coming year.

4.4.1 Conducting a food gap analysis

Calculating the food gap is an essential step in estimating the food needs of the affected population in an emergency. The food gap has three distinct elements:

- the food availability gap, which is the shortfall between a region's aggregate food needs and its aggregate food availability;
- the food access gap, which is the shortfall at the household level; and
- the food consumption gap, which is the shortfall between nutrition needs and actual food consumption.

Estimating the food gap is a relatively straightforward process. In simple terms, the food access gap is the difference between the level of household food stores, or access to food, and the actual amount of food needed to ensure adequate nutrition and health for every household member. By quantifying this difference, an EFSA can arrive at reasonably accurate estimates of emergency food needs.

For a population affected by a shock, the food access gap – or food need – is equal to the aggregate food deficits of vulnerable households who are unable to meet their own minimum requirements without endangering their health, their access to essential non-food items such as income and education, and their own or the community's resource base.

Every EFSA should estimate the expected food access gap resulting from an emergency, but the specific methodology for doing so differs according to the type of EFSA being undertaken. In an initial assessment, there is not usually enough time to gather in-depth food access information on distinct livelihood groups. As a result, estimates of the food gap developed during initial assessments typically rely on general information obtained from group discussions and key informant interviews. A rapid assessment should be able to outline how the food gap differs among distinct livelihood groups, but may have limited ability to gather detailed quantitative information at the household level. An in-depth assessment should focus on obtaining detailed, household-level information and should calculate food gaps with relative precision.

Table 4.10 outlines the general process for determining the food gap.

Table 4.10: General process for determining the food gap

Process	Sources / methods
→ Identify distinct vulnerable groups within the population of concern	 Key informants Informal discussions with different groups: gender-based, livelihood groups, community-based organizations, etc.
→ Describe the ways in which each vulnerable group currently obtains access to food and meets its non-food needs; how members of each group use the food and other resources available to them; and how access and use vary within and across groups, according to the wealth or status of households	 Focus group discussions Proportional piling and seasonal calendars with groups of participants representing the different livelihood groups Observation (transect walks) Household visits and/or household surveys
→ Examine the general availability of food and other resources in the locality, and use the findings of the forecast analysis (Section 4) to predict how changes in context – e.g. market access, climatic conditions, sources of conflict – are likely to affect people's access to and use of food, their nutrition status or their indebtedness	 Key informants: local authorities, traders, etc. Food aid distribution data Monitoring reports Nutrition surveys and surveillance data Market surveys/analysis FEWS information VAM analyses
 → For each livelihood group, estimate the difference between food needs and the amount of food that individuals expect to be able to provide for themselves The difference represents the <i>food gap</i>: the level of food assistance recommended by the EFSA should be sufficient to fill this gap 	 Household surveys that enable analysis of household income and expenditure Quantitative and qualitative data that provide empirical information on the nutrition outcomes of household resource use

Wherever possible, the recommended method for calculating the food gap follows five basic steps.

Step 1: Determine baseline consumption and income levels

To estimate the food gap, an EFSA must derive baseline consumption and income levels that reflect a normal year – one in which emergency food aid is not required. Assessors should ensure the accuracy of baseline information by triangulating household information with other data sources, such as district-level agricultural production data.

Step 2: Convert basic food requirements to cereal equivalents

A common measure is needed so that consumption, income and expenditure data can be combined, and food gaps or food aid needs calculated. The most common measure is the cereal equivalent. As cereals account for the bulk of energy needs for food-insecure households and of food assistance provided to these households, it is convenient to use them as the measure. Income and expenditures are converted to cereal equivalents using the local market values of cereals, or their substitutes, collected during the assessment from secondary sources, key informant interviews or market visits. Typically, a minimum consumption requirement of 2,100 kcals/person/day is used as a basis for calculating the food gap.⁶⁰ If this minimum requirement is derived from cereals only – which is unlikely, even during emergencies – each person will require approximately 18 kg of cereals per month. If the 2,100 kcal comes from cereals only, it can be assumed that the diet is not diverse enough and that there are potential problems with nutrition quality.

Step 3: Convert non-cereal foods into cereal equivalents

In many situations, the predominant portion of beneficiaries' food intake comes from non-cereals, for example from root or tuber crops such as cassava or potatoes. An EFSA must determine the extent to which cereals such as wheat or rice can substitute the deficit of the non-cereal staples. To determine the nutrition deficits of these areas, the non-cereal foods can be converted into nutritional cereal units. Using the example in step 2, the theoretical cereal deficit is 18 kg per person per month.

It should be noted that when a cereal substitutes a non-cereal diet staple, beneficiaries will often trade or sell the cereal food aid, frequently at unfavourable terms of trade. If the EFSA finds this response, it may be possible to account for these poor exchanges and terms of trade by supplying additional cereal to make up the trade deficit, or by introducing ways of using cereals in locally preferred dishes.

If the substitute cereal is likely to be consumed by beneficiaries, the nutritional cereal equivalent applies. If it is traded, the economic equivalent applies.

Step 4: Develop an income/expenditure balance sheet

The most common approach to calculating the food gap uses a balance sheet of household income and expenditures. This is because many households rely on purchasing food for at least part of their dietary needs, and the assessment needs to determine the degree to which households normally fulfil their food requirements from such transfers. Using an income/expenditure balance sheet enables direct comparison of households' available incomes with their expenditures. It also provides insight into the degree of food insecurity (if any) and determines the household deficits that will be taken into account when calculating food aid needs.

Food produced by a household for consumption is considered to be both a source of income and an expenditure. In both cases, it is estimated in terms of quantity, for example, weight. Other sources of cash income, such as wage labour and remittances, must initially be estimated in terms of their actual market values. Again, each must be converted into a cereal equivalent so that all income sources and expenditures can be compared. This should be done using the current market price of the cereal as the conversion rate.

^{60.} A general food ration of 2,100 kcal/person/day is based on the mean per capita energy requirement for the "normal" population distribution of a developing country. This estimate is designed to include the needs of vulnerable sub-groups: infants, young children, pregnant and lactating women, the elderly.

In Example 4.12, a number of household income sources are converted into cereal equivalents, and the total income of the household is estimated for the current year. In this particular year, food produced for home consumption was 400 kg; if this were a drought scenario, this value would likely be significantly lower than the normal or average income value of food production.

	Income		
Average household income	Cash value /	Cereal price =	Cereal equivalent
Food produced for home consumption			400 kg
Cash crop sales	\$70	0.35	200 kg
Livestock sales	\$150	0.35	300 kg
Off-farm cash income	\$35	0.35	100 kg
Remittance income	\$18	0.35	50 kg
Savings	\$18	0.35	50 kg
Total income capacity			1100 kg
E			
Household expenditures	Cash value /	Cereal price =	Cereal equivalent
Cereal seeds	n/a		100 kg
Cereal storage losses	n/a		50 kg
School fees	\$18	0.35	50 kg
Medical expenses	\$35	0.35	100 kg
Clothing	\$18	0.35	50 kg
Fuel	\$53	0.35	50 kg
Total expenditures			1,500 kg
	Balance		
Total household requirements			1500 kg
Total expenditure capacity			1100 kg
Total needs: requirements minus capacity			400 kg

Example 4.12: Household income and expenditure balance sheet

The balance sheet in Example 4.12 shows a household that has expenditures of approximately 1,500 kg cereal equivalent for the current year, but the capacity to provide only 1,100 kg of cereal equivalent based on all estimated income sources. It will therefore fall short of meeting its minimum food needs by approximately 400 kg. In this example, an estimated 400 kg of food aid (in cereal equivalent) is needed to alleviate the household's food deficit.

Step 5: Calculate aggregate food needs for different socio-economic groups It is recommended that an EFSA calculate balance sheets for households within different livelihood groups or wealth categories, depending on the homogeneity of the population being considered. This is because different groups living in the same geographical area, such as the same food economy zone or agro-ecological zone, typically have different incomes and expenditures, and hence have different food deficits. Overall, regional or

PART IV

national deficits are the weighted sum of the deficits of the different groups that have been defined and for which balance sheets have been developed. Example 4.13 illustrates an aggregate food gap estimate for various livelihood groups.

Livelihood group	Yearly household deficit (kg)	Number of households	Total food needs (mt of cereal equivalents)
Coastal fishers	200	4000	800
Highland coffee farmers	150	6000	900
Highland subsistence farmers	400	10000	4000
Total		20000	5700

Example 4.13: Aggregate food gap estimate

The use of different livelihood or other groups will help to target food aid, but only makes sense if the differences among groups are reflected in the actual distribution of food aid. It must be operationally and politically feasible to allocate different food aid rations to specific sub-groups living in the same area. For each group, the number of months that food aid will be needed should also be estimated, and factored into the overall estimate of the food gap. This requires consideration of how different groups will be able to recover from whatever shock created the emergency, and how future income and expenditures will be affected.

4.5 Estimation of the numbers of people who will be affected by shocks and opportunities

The groups whose food security is likely to be at risk in the future were identified in **Section 4.3**. During the situation analysis of an EFSA, the **numbers** of people expected to be at risk in the future must be estimated. How this is done depends on whether the data used are quantitative or qualitative. This section provides guidance on estimating the *affected* population; for guidance on ways of estimating the *total* population size, see the desk review on estimating population size in emergencies and Technical Guidance Sheets Nos. 7, 10 and 11.⁶¹

^{61.} Desk Review, Estimating Population Size in Emergencies, A. Henderson, WFP Emergency Needs Assessment Service, December 2006; Technical Guidance Sheet No. 7 Area Method to Estimate Population Size and Demographics in Emergency Food Security Assessments, A. Henderson, WFP Emergency Needs Assessment Service, September 2007; Technical Guidance Sheet No. 10 Using the Delphi Method to Estimate Population Size and Demographics in Emergency Food Security Assessments, A. Henderson, WFP Emergency Needs Assessment Service, January 2008; Technical Guidance Sheet No. 11 T-Square Method to Estimate Population Size and Demographics in Emergency Food Security Assessments, A. Henderson, WFP Emergency Needs Assessment Service, January 2008; Technical Guidance Sheet No. 11 T-Square Method to Estimate Population Size and Demographics in Emergency Food Security Assessments, A. Henderson, WFP Food Security Assessments, A. Henderson, WFP Food Security Assessment Service, December 2008; Technical Guidance Sheet No. 11 T-Square Method to Estimate Population Size and Demographics in Emergency Food Security Assessments, A. Henderson, WFP Food Security Assessments, A. Henderson, WFP Food Security Analysis Service, December 2008.

The number of people expected to be at risk in the future is estimated as follows:

Number of people currently at risk and whose situation is not expected to improve in the short term

Number of additional people who will become at risk

Number of people currently at risk but whose situation is expected to improve to the extent that they are no longer at risk

When subtracting populations no longer at risk, the caution in **Section 4.3.3** must be taken into account. It may be possible to divide the coming year into periods of, say, three months, and identify the populations at risk in each period. The feasibility of such an approach depends on the situation, however. For example:

- in a slow-onset emergency, numbers are unlikely to vary greatly within a short period;
- in a rapid-onset emergency, such as localized flooding, the population at risk may decline substantially over the course of a few weeks or months.

Note that this approach increases the analytical, administrative and logistics workload substantially, as beneficiary numbers may vary from month to month. The added value of this level of fine-tuning and the implications for targeting should be carefully considered (see **Section 4.3.3**), as should the practical constraints.

4.5.1 Final estimate of numbers at risk

The information that relates shocks and opportunities to population groups is added to the information collected in Table 4.8 (in **Section 4.2**), as shown in Table 4.11.

Table 4.11: Overview of groups affected under different scenarios, examples						
Scenario and probability (5 = certainty)	Period	Opportunity or shock	Affected groups	Number of affected people (current + forecast)	Location of affected groups	
			Farmers	20 000	District A	
1. Situation will improve:	0-6 months Good harvest: opportunity Flood: shock		alood marrooti	Agricultural labourers	5 000	Districts A and B
probability 4			Grain traders	1 000	Town C	
		Flood: shock	Farmers	2 000	Villages along river	
			Farmers	20 000	District A	
2. Situation will deteriorate:	deteriorate:	Major crop failure due to late failure	Agricultural labourers	5 000	Districts A and B	
probability 2		of rains: shock	Grain traders	2 000	Town C	

Conducting a response analysis

In response analysis, the conclusions from the situation and forecast analyses are combined to identify possible interventions that can help to save lives and secure livelihoods. The outputs of response analysis include:

- identification of the factors related to risk;
- identification of the broad sectors and types of intervention required the entry points;
- review of the intervention plans and capacities of government and other actors, and identification of gaps in these;
- identification of a range of response options to fill the gaps, and the strengths, weaknesses, opportunities and threats (SWOT) associated with each;
- selection of the most appropriate response option(s);
- recommendations for interventions, including targeting criteria, timing, scale and duration.

As already noted, response analysis is usually based on the most likely scenario (see **Section 4.2**). However, if there is a worst-case scenario that is less probable than the most likely scenario but *nonetheless has a reasonable probability of occurring*, a contingency plan should also be made for this scenario (see Example 4.7 in **Section 4.2**).

Interventions may be aimed at saving lives, protecting livelihoods or a combination of both. They may be focused on addressing the current situation, preventing future deterioration of the situation, or both. For timing and resource allocation purposes, response recommendations are prioritized according to the following urgency of needs:

- First priority: current risks to lives.
- Second priority: current risks to livelihoods, and risks to lives in the near future.
- Third priority: risks to lives and livelihoods in the more distant future.

If the current situation and/or forecasts indicate that an intervention is needed, the next step is to identify response options (see **Section 5.4**). The process is as follows:

 Ongoing interventions, future plans and the capacities of other stakeholders – government, United Nations agencies, NGOs and civil society – are examined and compared with the assessed needs. Gaps are identified.

- Different types of intervention are identified as **options** for responding to the needs and filling the gaps.
- For each possible intervention, a SWOT analysis is undertaken. This examines the **strengths**, **weaknesses**, **opportunities and threats** (SWOT) associated with each intervention option.
- Using the results of the SWOT analysis, the most effective and appropriate intervention strategy is chosen. This strategy should be proposed in the recommendations section of the assessment report.

Examples of interventions that might be used in a food security or nutrition crisis are given in Example 4.14.

Example 4.14: Intervention options for food security and nutrition crises

- Food distributions (general or targeted).
- Cash and voucher transfers.
- Food for work, cash for work.
- Supplementary or therapeutic feeding to malnourished individuals: pregnant and lactating women, emaciated children, people suffering from HIV / AIDS, tuberculosis, etc.
- Institutional feeding.
- Food for education.
- Health/nutrition education programmes, such as nutrition and food preparation training.
- Health programmes, such as immunization, vitamin A and iron supplements.
- Agricultural programmes, such as seed and tool distributions or fairs, fodder distributions, restocking.
- Other non-food interventions, such as water supply, provision of household items, market development.

Note: All relevant options should be considered, even those that do not fit the competence or mandate of the organization carrying out the assessment. In such cases, analysis should be disseminated to organizations with the requisite capacities.

For a more detailed discussion of response options see **Section 5.4**.

5.1 Factors related to risks to lives and livelihoods

To design an effective response, it is necessary to identify the factors that cause risk, both current and future.

For groups that are **currently at risk and whose situation will not change** in the short term, the main factors of risk are identified during the situation analysis (see **Chapter 3**).

For groups that are **expected to become at risk in the future**, risk-related factors are identified in the forecast analysis (see **Section 4.3.2**).

The factors that are directly related to risk should be defined as specifically as possible **and should relate directly to the situation and forecast analyses**. Factors that **contribute to risk** are categorized as immediate, underlying or basic, using the Food and Nutrition Security Conceptual Framework as a guide (see **Part I**). This categorization is useful when determining the types of intervention that are appropriate to the level of urgency:

- Immediate factors have a direct impact on lives and livelihoods. If dietary intake or health status is at a critical level, *immediate* action may be needed to save lives.
- **Underlying factors** may have an indirect impact on lives, but a direct impact on livelihoods. If they are not addressed, there is a danger that the situation will deteriorate, possibly leading to risks to lives in the future.
- **Basic factors** are long-term, structural issues. These are not normally addressed by EFSA response options, but if serious structural problems are identified, they should be recorded in the EFSA report, and relevant stakeholders such as government should be notified.

Factors that **decrease risk** should also be noted. These include the **capacities** of the affected groups and the **opportunities** identified in **Section 4.1**.

For each at-risk group, summarize the factors that increase and decrease risk, together with the associated livelihood characteristics, institutions and processes. An example is given in Table 4.12, which can also be used as a template.

Fable 4.40. Fasteria and stated with visit farmeria

Table 4.12: Fact	ctors associated with risk for various groups, with examples				
Group with lives at risk now or in the future	Factors exacerbating or alleviating risk	Associated livelihood assets and strategies	Associated institutions and processes		
		Factors increasing risk			
	Food utilization: - lack of breastfeeding - high prevalence of diarrhoea - poor quality of complementary feeding - poor quality of water	 Mothers' limited knowledge about feeding and hygiene practices 	 Low staffing in health centres 		
Severely wasted children under 5	Food access: - lack of income to purchase diverse foodstuffs	 Lack of agricultural land Women's multiple responsibilities: income generation, caring for children, cooking, etc. Low access to education 	 Environmental degradation of farmland Few employment options available Poor terms of trade between wage rates and food prices 		
	Factors decreasing risk				
	Food availability: - imminent abundance of nutritious wild plants with onset of rains	 Consumption of wild plants is part of usual livelihood strategy 			
	Food accessibility: - economy gradually diversifying	 Households gradually diversifying livelihood strategies to adapt to changing economic environment 	- Government loans to small businesses		
			(cont)		

(cont)
-------	---

Group with lives at risk now or in the future	Factors exacerbating or alleviating risk	Associated livelihood assets and strategies	Associated institutions and processes	
		Factors increasing risk		
	Food availability: - low production in area due to failed rains - few food imports	 Limited livelihood strategies: only farming 	 Poor roads increase transportation costs and discourage traders from moving food into the area High import tariffs reduce amount of food brought into the country 	
Households with low levels of food security	Food accessibility: - lack of own production - lack of income to buy food at raised prices	 Lack of able-bodied workers due to high disease prevalence Lack of alternative skills to diversify income 	 Poor preventive and curative health services Poor education reduces knowledge of health issues and limits income- related skills 	
	Factors decreasing risk			
	Food availability: - arrival of rains - improvement of physical and market infrastructure	 Diversification of the economy 	 Reduction of tariffs and other taxes that constrain the movement of food Improvement of physical infrastructure 	
	Food accessibility: - opportunities for income diversification	 Training in income- generating activities Access to credit 	 Introduction of vocational training programmes Introduction of credit programmes for small businesses 	

5.2 Entry points for interventions

Entry points are the sectors and broad types of intervention that can be used first to address the needs identified during the analysis phase of the EFSA. They also provide a basis for analysing interventions managed by other organizations (see **Section 5.3**).

The following are possible entry points for the examples given in Table 4.12:

Severely wasted children under 5:

- Water sector: Improvement of water quality and quantity through emergency delivery, treatment and storage systems.
- **Health services**: Deployment of additional health staff and equipment; establishment of complementary feeding programmes.
- **Care practices**: Dissemination of information regarding the benefits of hygiene and breastfeeding.

Food-insecure households:

- Health services: Establishment of preventive and curative health services.
- Vocational training: Training in activities that enable households to diversify their income sources.
- **Credit**: Provision of loans to help people buy productive assets, both agricultural and non-agricultural.

These programmes are more effective if they are supported by such government interventions as:

- **Policy changes**, reduction of taxes or changes in regulations that constrain the free movement of food.
- Investment in services, particularly health and education.

5.3 Other stakeholders' interventions, and remaining gaps

Before a response can be planned, the existing and planned activities of government and other agencies must be taken into account, to prevent duplication, identify gaps and ensure they are covered, and avoid incompatible programme responses, such as one agency undertaking food for work while another carries out free food distribution in the same area.

Relevant agencies are identified through stakeholder analysis, and consulted. The subsequent selection of agencies to collaborate on interventions is then based on the entry points, as described in **Section 5.2**. For example, for the emergency described in Table 4.12, entry points were identified in water, health services and care practices. Organizations working in these sectors should be approached for potential partnership activities. Box 4.10 lists some agencies that may be stakeholders in interventions resulting from an EFSA response analysis.

Box 4.10: Typical stakeholders in food security response programmes

- Government national, regional and local
- Non-State authorities, such as in situations of civil conflict
- WFP
- OCHA
- FAO
- UNICEF
- UNHCR
- International Committee of the Red Cross (ICRC)
- International Federation of Red Cross and Red Crescent Societies (IFRC)
- National Red Cross or Red Crescent societies
- International NGOs
- National NGOs
- Donors

The *capacity* of governments and agencies to *fulfil* their plans is assessed, as is their *flexibility* for changing plans if necessary. Whenever possible, details of the planned activities are discussed with the agency concerned, including the following issues:

• **Financial resources**: Does the agency have the necessary money, or is it waiting for funds?

- **Material resources**: Does the agency have the necessary goods and equipment, such as food for distribution, vehicles, etc.?
- Human resources: Have all the necessary personnel been deployed?
- Logistics: How will the operation function?

Detailed discussion of these and other topics with the government department or agency concerned should clarify the feasibility of its plans and can also form the basis for strong operational partnerships.

It can be difficult to estimate government capacity because of decentralization and the involvement of several departments. Indicators reflecting the macroeconomic situation and the government institutions and budgets allocated to disaster preparedness and response can be used for this purpose. For detailed advice, see Technical Guidance Sheet No. 13.⁶²

Information on different stakeholders' responses may be presented in a table similar to Table 4.13.

Table 4.13: (Ongoing and pla	nned response inter	ventions, wit	h examples
Actor	Type of intervention	Type and number of beneficiaries	Place of intervention	Duration of intervention (start to finish)
Ministry of Social Affairs	Subsidized food	To be determined	- Area B - Area D	One-off food delivery in
WFP	General food aid distributions of full rations	Vulnerable households: - with less than 1 ha - female-headed - the poorest according to leaders Total: ~ 50 000 people	- Area A - Area B - Area C - Area D	From to
	Supplementary feeding: • rations for children • take-home full rations for households	Moderately malnourished children Total: ~ 3 500 children; ~ 500 households	- Area B - Area D	From to
Religious institution	Targeted food aid distributions: ~ 3/4 ration	Vulnerable households identified by community	- Area B - Area C - Area D	

^{62.} Technical Guidance Sheet No. 13 Analysing National Capacity to Respond to Food Security Crises, WFP Food Security Analysis Service, September 2008.

The planned and ongoing activities of other agencies are compared with what is required to address the factors related to risks to lives and livelihoods identified in **Sections 5.1** and **5.2**. Information can be summarized in a table similar to Table 4.14.

Table 4.14: Summary of interventions and gaps for groups at risk				
Group at risk now and in the future	Ongoing interventions	Planned interventions	All needs covered? If not, what are the gaps?	
	What is being provided?	What will be provided?		
	To whom/how many?	To whom/how many?		
At-risk group A	By whom?	By whom?		
	Since when/until when?	When/for how long?		
	Where?	Where?		
	What is being provided?	What will be provided?		
	To whom/how many?	To whom/how many?		
At-risk group B	By whom?	By whom?		
	Since when/until when?	When/for how long?		
	Where?	Where?		

If all the risk factors identified in Section 5.1 are being addressed adequately by other agencies, there is no need to intervene at this stage. The situation should be monitored to ensure that any unmet needs arising in the future are identified promptly. This should be a key recommendation of the EFSA report.

If Table 4.14 indicates that some needs are not being addressed by other agencies, an additional or complementary response is necessary, as described in the following section.

5.4 Response options⁶³

In the EFSA report, response options are examined for the groups requiring assistance (see **Section 5.1**) that is not being provided by government or other agencies (see **Section 5.3**). Response options should be directly linked to the risk factors and groups identified in the situation and forecast analyses, taking into account the affected groups' capacities and other agencies' responses. As much as possible, the affected people should participate in planning the response, including women, the elderly and disabled people.

PART IV

^{63.} The following guidance is not a project planning guide. This section presents generic response options and explains the circumstances under which each is appropriate. It also provides an example of the level of detail expected in an EFSA report. Operational planning requires a comprehensive approach that is beyond the scope of this handbook.

The procedure for identifying response options is as follows:

- 1. Identify the number of people requiring food assistance.
- 2. Facilitate a workshop for identifying programme options with programme and other key WFP and non-WFP actors.⁶⁴
- 3. Use a matrix or SWOT analysis to identify all possible modalities and activities.

5.4.1 Identification of response options

There are many ways of addressing a food security or malnutrition problem. The most appropriate response is *highly context-specific* and depends on:

- the type of emergency, rapid or slow-onset, and the stage of the emergency at the time of response, early or mid-cycle;
- the pre-emergency situation the status of infrastructure and services, level of education, etc.;
- the habits, priorities and culture of the affected population;
- the degree of access to the affected area;
- the quality of infrastructure;
- food availability and market conditions in the affected area;
- the resources available financial, human, logistics, etc.;
- the range of feasible partnerships, such as with government, United Nations agencies and NGOs;
- the political and economic environment; and
- the security situation.

Each response must be planned according to the particular circumstances and must be explicitly linked to the needs and gaps identified in the analysis.

The first level of screening response options is to categorize the interventions required according to the type of risk factor that they address: food availability, access and/or utilization. A second level of screening defines the level at which each intervention can take place, based on whether it addresses an immediate, underlying or basic factor of risk (see the Food and Nutrition Security Conceptual Framework in **Part I**).

The following are sectors and broad types of intervention that address food availability, access and utilization factors:

- Food availability: Interventions to support agricultural production, both crops and livestock, the movement of food between deficit and surplus areas, food distributions, etc.
- Food access: Interventions to support income generation, such as public works and food/cash for work, income transfers, such as cash/voucher distributions,

^{64.} Collaboration within programmes is essential for defining response options: programme and assessment staff must work together from the onset of the assessment, when information requirements are defined, to the analysis. During the workshop, entry points are identified, and their feasibility, etc. evaluated by the WFP units that will be required to implement interventions.

- **Food utilization**: Interventions to improve health care, water, sanitation, shelter, nutrition knowledge and care practices, child-care services, etc.
- **Malnutrition**: Interventions to improve food consumption therapeutic and supplementary feeding programmes, school feeding, food distribution.

Additional examples of response options for addressing malnutrition and food insecurity problems are given in Table 4.15. Some interventions can cover more than one food security issue. For example, food distributions might ease problems of both food availability and food access. Responses in Table 4.15 are categorized according to their most common application. More detailed guidance on the circumstances under which each option might be applied, and the advantages and disadvantages of each, is given in **Annexes 4** and **5**.

Description		
onses to food availability problems		
Free rations to households in need – general distribution to all households in area or targeted distribution to households in specific groups		
Selected food commodities made available to traders and retailers to sell at controlled prices		
Reduction of logistics bottlenecks, such as through road repair, or provision of credit to traders		
Food ration as payment for work – can be used as method for self-targeting, with only those who really need the food being willing to work for it. Is also a response to food access problems		
Food as an incentive to individuals from food-insecure households to undertake training in skills that will help them improve their own food security. Is also a response to food access problems		
Responses to food access problems		
Food given to orphans and vulnerable children, such as in situations of high HIV/AIDS prevalence		
Provision of nutritionally balanced and fortified meals to children at school, and of take-home rations to compensate parents for sending their children to school		
Food provided to social institutions, such as orphanages, homes for the elderly or disabled, hospitals and health centres		
Cash distributed to households in need – general distribution to all households in area or targeted distribution to households in specific groups		

Table 4.15: Food and non-food responses in terms of factors of risk

(...cont)

Cash for work	Cash as payment for work – can be used as method for self-targeting, with only those who really need the money being willing to work for it	
Food vouchers	Distribution of vouchers that can be exchanged for food and other essential items	
Non-food transfers	Provision of non-food items, such as soap and blankets, or services, such as water, schooling and health care	
Non-food support to livelihood activities	Provision of productive inputs and services to maintain, rebuild or restore capital assets for food-insecure but economically active people	
Exchange with produce	Food given to households in return for produce that they are not able to sell at reasonable prices, such as livestock	
Responses to food utilization problems		
Food preparation materials	Provision of cooking equipment, fuel, water, etc.	
Nutrition, education, health, water and sanitation interventions	Improvement of feeding and care practices through, for example, prevention of nutrient loss during food preparation and prevention and treatment of diarrhoea and other diseases that affect nutrient absorption and utilization	
	Responses to malnutrition	
Therapeutic feeding	Medical and nutritional treatment to save the lives of severely malnourished individuals	
Supplementary feeding	Distribution of food to supplement the energy and nutrients available from the basic diet of individuals who have special nutrition needs or are malnourished	
Public health measures	Measures to improve sanitation, water supply, health care services, etc.	
Food fortification	Provision of food fortified with nutrients, particularly vitamins and minerals, when the diet is deficient in these respects	
Nutrient supplementation	Distribution of nutrient supplements, such as vitamin A capsules, when the diet is deficient in these respects	

Wherever possible, interventions that build on existing programmes should be selected, to speed up the implementation process and make use of established capacities and experience. Some interventions will be outside the mandate of WFP. If the EFSA indicates that such interventions constitute the most effective response, this should be stated in the EFSA report, to be shared with agencies that have the relevant competence and capacity. Response options should also be discussed with partner organizations.

Figure 4.1 shows a decision tree for determining the most appropriate type of response options. This decision tree can be used as a guide for drawing on contextual and empirical knowledge to solve issues such as constraints to market supplies, physical access and market linkages. The decision tree does not explicitly



include gender, but gender should be considered a critical factor throughout the entire assessment, analysis and response planning process.

5.4.2 SWOT analysis

Having identified a series of response options, each must be analysed in terms of its strengths and weaknesses, and the opportunities and threats that it presents. This is known as SWOT analysis. A SWOT analysis should be undertaken for each of the recommended response options.

PART IV

In a SWOT analysis **strengths** and **weaknesses** reflect the appropriateness and feasibility of the response option. The following criteria should be taken into account when assessing the **appropriateness** of a response option. The response should:

- address the factors that have been identified as contributing to risk;
- reflect the needs and priorities of the affected population, disaggregated according to sex, age, etc.;
- be compatible with local society and customs; and
- be compatible with the interventions of the government or other agencies.

The response should *not*:

- lead to dependency on aid for any sector of the population;
- have a negative impact on the local social, environmental or economic situation

 for example, a large food distribution might discourage agricultural production;
- divert people from other important tasks, such as productive activities, caring, collection of water and fuel;
- expose the population or agency staff to security risks; or
- stigmatize people for example, by targeting people with HIV/AIDS or from certain ethnic groups.

The following criteria should be taken into account when assessing the **feasibility** of a response option:

- Targeting criteria should be realistic, given the social and cultural factors and the time available.
- It should be possible to undertake the response with the resources available. Human resources, including expertise, financial and material resources should all be considered.
- The response must be implemented in a timely manner, given the urgency of the situation.

In a SWOT analysis **opportunities and threats** reflect the external factors that may affect the response. These are context-specific. The following are some examples:

Opportunities

- The introduction of new government policy that facilitates market functioning.
- The end of the wet season and the improvement of transportation.
- The signing of peace agreements.
- The harvest.

Threats

- Government policies that limit the scope of trade or aid programmes.
- Reduction of donor interest in the country.
- Deterioration of security.
- Lack of key programme resources such as fuel.

The strengths, weaknesses, opportunities and threats are combined in a matrix to assist the comparison of response options and evaluate the relative merits of each. Example 4.15 illustrates how a SWOT analysis may be conducted.

Example 4.15: SWOT analysis of response options

An EFSA has been undertaken in a rural area affected by recent floods. The area has always been poor, with some households suffering from chronic food insecurity. Nutrition surveillance over the last ten years indicates a gradual upward trend for malnutrition among children under 5. The local economy is based on agriculture, food processing and light industry.

The EFSA reveals the following:

- Approximately 20 percent of the population is food-insecure. Food consumption patterns show that both energy and micronutrient intakes are alarmingly low in this group.
- Among the remaining 80 percent of the population, food consumption is acceptable.
- Prices of food in the local market have increased.
- Opportunities for income generation are lower than usual. Farmland has been swamped, and some factories and processing plants have been put out of action.
- Farms that are not close to the river have recorded a good harvest because their fields were not flooded.
- Physical access to the area is difficult, as the flood destroyed a bridge on the main road.

The prospects for the next three months are poor, for the following reasons:

- The flood waters will take several weeks to recede.
- No harvest is expected before next year.
- There is no tradition of seasonal migration to look for work.
- The government's capacity to respond is weak, and the only NGO working in the area concentrates on supplementary feeding programmes and nutrition education for mothers and pre-school-age children.

In view of the situation, a targeted food distribution to address food shortage at the household level is proposed. A SWOT analysis is undertaken.

Note: The strengths and weaknesses reflect aspects that are under the control of the implementing agency – primarily programme design – while the opportunities and threats concern external issues that are outside the control of the implementing agency.

Strengths	Weaknesses	Opportunities	Threats
 Addresses the food consumption problem Helps bring down the price of food, thereby improving food access Partially substitutes lost earnings Could act as a catalyst in rehabilitating the local economy: increased household purchasing power and employment generated through logistics operation 	 May be difficult to target the intended 20% of the population Logistics complications may lead to food arriving late and disrupting the market If too much food is distributed, traders may be discouraged from bringing in commercial supplies, and farmers whose land was not flooded may not be able to sell their produce at good prices 	 Food for the distribution could be bought locally from farmers who are not close to the river The food distribution could be undertaken in collaboration with the NGO, making it possible to address both bulk food deficit and malnutrition 	 Weak government capacity means that coordination and support are likely to be poor Transportation of food will be difficult because of the broken bridge and flooded land Slowly receding flood waters might lead to water-borne disease, reducing the benefits of improved food consumption

In Example 4.15, it is evident that a food distribution could bring substantial benefits, but that these depend on a number of conditions, especially the following:

- An effective targeting system must be established.
- The amount of food needed and the duration of the distribution must be analysed realistically.
- The feasibility of the operation must be assured and logistics constraints taken into account. Late distribution might be worse than no distribution at all: it would not provide assistance when needed, and by the time the food arrives, the situation may have improved to the extent that food will disrupt the local market.
- Complementary health care activities should be implemented to reduce the incidence of water-borne disease and to maximize the benefits of the food distribution on food consumption.

This type of SWOT analysis is undertaken for each of the response options identified in **Section 5.4.1**. Options that do not comply with the appropriateness and feasibility criteria outlined in this **Section 5.4.2** are discarded. The remaining response options are ranked, as described in **Section 5.4.3**.

The "do-no-harm" principle is essential in the SWOT analysis of response options (see Box 4.11), particularly in conflict situations, where a specific conflict analysis is required. See *Methods and Tools for Conflict Analysis*⁶⁵ for more details.

Box 4.11: Conflict analysis and the do-no-harm principle

A badly planned response may be worse than no response at all if it harms the local population. Examples of harmful responses include:

- distribution of items that attract looters and put recipients in danger;
- food distributions that disrupt local markets to the extent that the livelihoods of farmers and traders are put at risk;
- distributions that necessitate long and dangerous journeys for recipients, for example, to collect distributed items.

Any proposed intervention should be analysed for its **potential negative effects**, as well as its benefits.

5.4.3 Ranking and prioritization of response options

The ranking of response options requires good judgement and a sound knowledge of the context. In general, the interventions that most fully comply with the criteria outlined in **Section 5.4.2** are the best options.

Different response options can be combined in one programme, either simultaneously or sequentially. For example:

^{65.} Methods and Tools for Conflict Analysis, WFP Transition Unit, August 2007.

PART IV

- a general food distribution can be combined with the provision of supplementary rations for pregnant and lactating women and malnourished children;
- a cash-for-work scheme can be combined with market support interventions and food for work implemented at different periods of the year.

The same intervention might also assist more than one target group. The target groups (see **Section 5.4.4**) are linked to the proposed interventions in a table similar to Table 4.16.

Table 4.16: Targeted interventions, examples	
Affected (target) group	Examples of interventions
Under-5 children whose lives are at risk: severely wasted	 Therapeutic feeding to children Supplementary feeding to mothers Emergency provision of clean water Targeted distribution of cooking materials
Households whose livelihoods are at risk: experiencing severe depletion of productive assets by distress sales, and of human assets by rapidly deteriorating food access	 Targeted general food distribution to affected households Provision of seeds and tools Provision of fortified on-site school feeding

The EFSA report does not need to include all the details of the forecast and response analyses, but the logic of each recommended intervention should be clearly explained. It is also important to explain why other proposed responses are not recommended. The recommended interventions should be linked explicitly to:

- the groups whose lives and livelihoods are at risk, and the factors of risk identified during the situation and forecast analyses;
- the context markets, agro-ecology, social circumstances, etc.;
- the security and access situation;
- operational constraints time, human resources, funding, etc.

5.4.4 Targeting

The EFSA report should provide recommendations about whether or not targeting is appropriate and, if so, the form that it should take. Targeting of assistance is based on the groups defined as being at risk in the situation and forecast analyses.

Targeting may be applied at different levels:

- Geographical targeting: All people living in a specific area receive assistance.
- Household targeting: All households fulfilling certain criteria receive assistance, based on the profiles of groups whose lives and livelihoods are at risk, such as IDP households or female-headed households.
- **Individual targeting**: *Within* households, individuals whose lives are at risk receive assistance, such as malnourished children or pregnant and lactating women.

• **Institutional targeting**: Schools, hospitals and other institutions receive support to improve food access and promote household and individual asset development and retention.

Whatever the approach to targeting, **practical criteria** must be applied to identify the people who qualify for assistance. Targeting criteria must be:

- easily understood and accepted by programme staff and the affected communities, otherwise the targeting is unlikely to be successful;
- observable and measurable, so that they can be monitored objectively, otherwise there will probably be lengthy debates about who is eligible for assistance and who is not;
- *specific to the target groups*: attributes that are also possessed by people outside the target group are not useful as criteria.

In some cases, **targeting may not be appropriate** because either everybody in the area needs assistance, or the costs and complications of targeting outweigh the benefits. Examples of the second possibility include:

- situations in which it is very difficult to define targeting criteria that are sufficiently
 observable and measurable, when the cost of targeting in terms of staff time may
 be greater than the cost saving achieved by limiting assistance to certain groups;
- communities in which the principle of targeting is not accepted, because the culture places a strong value on equality and people do not accept that some members of the community should receive assistance while others do not.

The following are approaches that can help address these issues:

- **Community targeting**: Community representatives decide who will receive assistance. This works well if the representatives are genuinely acting on the entire community's behalf. If they are not, there is a danger of substantial inclusion and exclusion errors.⁶⁶
- Self-targeting: Some types of intervention, such as food for work, are based on the principle that the target groups select *themselves*. The theory is that only those who really need the assistance will undertake the work required to receive it. However, such projects are often implemented in areas with high unemployment, where the work implemented cannot absorb all the people willing to work. In such cases, other targeting approaches need to be used to select workers.
- Blanket assistance: If the situation is very severe, such as after a tsunami, and the period of assistance provision is short, it may be cost-effective and socially beneficial simply to provide assistance to everyone.

^{66.} A guidance note on community-based food aid targeting in complex emergencies is being prepared by WFP's Policy, Planning and Strategy Division (Humanitarian Policy and Transitions). It will be available in mid-2009.

chapter 6

Formulating recommendations for interventions and follow-up

In the EFSA report (see **Part V**), the conclusions of the situation, forecast and response analyses are summarized in the recommendations section. The following detailed operational information is required for the selected response option(s):

- the type(s) of intervention recommended;
- the level of assistance: quantities of resources to be provided and frequency of provision;
- the target group(s) and institution(s), including descriptions and numbers of beneficiaries;
- the priority geographical areas;
- the duration of the interventions, including start and finish dates and exit strategy;
- the mechanisms for coordination with other agencies and the government.

The EFSA report may also include recommendations for interventions that WFP will not implement, such as those focusing on protection, agriculture, health, water, sanitation, shelter, education, and capacity-building and training. For this type of recommendation, the level of detail in the EFSA report depends on the composition of the assessment team. If the team includes specialist partner agencies, detailed recommendations may be feasible. If not, the EFSA report should include broad recommendations to be shared with agencies that have the relevant expertise. Although EFSAs are likely to result in recommendations for intervention carried out by WFP and/or partner agencies, they must remain open to the possibility of *not* intervening in circumstances where activities may be unsafe, detrimental to the affected community, or otherwise ineffective in addressing identified issues.

The EFSA report should also include recommendations for **follow-up assessment and monitoring**:

- If the situation is changing quickly, such as during the first days after a forced displacement, regular follow-up assessments will be needed. During a slow-onset emergency, a rapid assessment may be undertaken to ascertain whether or not the situation warrants an emergency intervention. If so, it is likely that an in-depth EFSA will be recommended. The EFSA report should provide specific recommendations concerning the timing and focus of follow-up assessments.
- Situation monitoring should be undertaken periodically after an EFSA. This will

show whether or not the situation is evolving as predicted in the scenario, and how effective any response interventions have been. Monitoring is usually less time- and resource-intensive than assessment, because it is based on selected indicators rather than a full process of primary data collection. The EFSA report should specify the indicators to be monitored and the frequency with which they should be collected.

Monitoring schedules should also be specified. Table 4.17 provides an example of a monitoring schedule.

Table 4.17: Monitoring data and schedule, example			
Information / aspect to monitor	Data required	Sources	Timing
Nutrition status	 Prevalence of wasting among children 6–59 months 	Community health centresNGO surveys	Monthly
Coping mechanisms	 Excessive out-migration Excessive animal sales Withdrawing children from school, etc. 	 Community key informants Markets Schoolteachers and parents 	Every two months
Harvest	Yields of staple crops	 Community key informants Local and central Ministry of Agriculture staff NGOs 	At harvest time
Market prices	 Prices of staple foods Prices of vegetables Prices of fuelwood 	 Community key informants Traders Local and central Ministry of Trade staff 	Every two weeks or monthly
Cross-border trade	 Prices on both sides of the border Volumes crossing the border 	 Traders Local Ministry of Trade and Customs staff 	Monthly
Livestock	 Prices of livestock Health condition of livestock Condition of pastures 	 Community key informants Traders Local and central Ministry of Agriculture staff 	Monthly in dry season

Part V Reporting EFSA results

Contents of Part V

1 Intro	duction	213
2 Rep	ort structure	214
3 Rep	ort quality monitoring	218
4 Com	municating the results of the EFSA	219
4.1	Users of the EFSA report and communication channels	219
4.2	EFSA executive brief	220
4.3	Presentation workshop or meeting	221
4.4	Dissemination by e-mail and hard copy	222
4.5	The news media	222
4.6	Websites and newsletters	223
4.7	Advocacy for non-food recommendations	223

Introduction

The report is the end-product of the EFSA. It provides a synthesis of the results for sharing with those who did not take part in the assessment. Users of the report may include decision-makers within WFP or partner organizations, and other key stakeholders such as government counterparts and potential donors. It is essential that the assessment analysis and recommendations are conveyed clearly and concisely so that stakeholders can understand the situation and take appropriate action.

This final **Part V** of the EFSA Handbook:

- outlines the report structure;
- explains the EFSA quality monitoring system, which is based on a review of the report;
- provides guidelines for communicating EFSA conclusions and recommendations.

Report structure

The EFSA report structure is given in Table 5.1. When preparing reports, the following points should be kept in mind:

- Not all the items in Table 5.1 will be needed in every assessment report: only the relevant issues should be included.
- It is very important that the methodology and constraints are clearly explained in Section II of the report. This ensures transparency and provides readers with a means of gauging the reliability of the information contained in the report.
- The report should be thorough but concise. Ideally, it should not exceed 30 to 40 pages, with additional details presented as annexes.
- The executive summary should be no longer than four pages.
- The sections of the report should be interlinked, and the analytical process must be evident. It must be clear to the reader how the assessment team reached its conclusions.
- Findings and conclusions should be illustrated by clear graphs and maps, when relevant.

Note: The outline of contents given in Table 5.1 is indicative. Actual contents depend on the objectives of the EFSA and the terms of reference.

Table 5.1: EFSA report format		
	Outline of contents	
Executive summary	 Reasons for doing the EFSA – main objectives of the EFSA Brief description of areas covered Extent and severity of malnutrition Extent and severity of household food insecurity, including numbers Extent of current risks to lives and livelihoods, including any distress coping mechanisms being employed Locations of food-insecure households, possibly including a map Characteristics – livelihoods, gender, etc. – of food-insecure individuals and households Factors contributing to food insecurity, risks to lives and livelihoods Likely evolution of the situation Total needs for assistance and what is already planned by other organizations Recommended responses, for whom and for how long Requirements for monitoring, periodic reassessment and contingency planning 	

	Outline of contents
I. Background	 Type of crisis, date/period of the event, evolution of the situation; assistance provided since the last assessment, when there is an ongoing operation Population groups and areas affected General context: economic, agricultural, political, security, etc. Rationale for the EFSA: Why was it carried out – as a first assessment following a sudden-onset crisis or warning of a slow-onset crisis, in preparation for a new EMOP or PRRO? Type of EFSA: When did it take place? Was it a first-time assessment (initial, rapid or in-depth) or a reassessment (rapid or in-depth)?
	2.1 ObjectivesMain objective and specific objectives of this EFSA
II. Objectives and methods of the EFSA	 2.2 Methods Partnerships and consultation process before, during and after fieldwork: government counterparts, agencies involved Local adaptation of the Food and Nutrition Security Conceptual Framework: Who was involved among the partners? What hypotheses were used to design the EFSA and select the data to collect? Sources of data – secondary reviews, households, key informants, focus groups – and type of data collected from each source; describe the data collection instruments used in an annex. Sampling approach: population groups and areas covered, random/purposive sampling, modalities of sampling; describe the sampling procedures used in an annex, if required Type of analysis: statistical and/or qualitative data; describe the analysis in an annex, if required Team composition and training Limitations of the methods employed, such as sampling, coverage and biases
III. Mortality and nutrition situation Note: This section can come after, or be integrated with, Section V if no nutrition status data were collected directly during the EFSA and only secondary data are used	Note: mortality data are presented only when relevant and available. 3.1 Current mortality rates and severity at the population level: • Current CMR and U5MR: dates of collection; primary or secondary data • Severity indicated by CMR and U5MR compared with standard thresholds • Trends in CMR and U5MR in recent months, when appropriate
	 3.2 Nutrition status Prevalence of malnutrition – acute, chronic – from direct anthropometric measurements or secondary data Severity indicated by population-level malnutrition rates compared with standard thresholds and local seasonal norms Trends in malnutrition rates, when appropriate
	 3.3 Health services and health status Effect of the crisis on access to health services Access to health services and constraints: physical, economic, staffing, supplies, equipment Prevalence and causes of main diseases Coverage of immunization against main infectious diseases Association/correlation between disease and nutrition status
	 3.4 Water, sanitation and housing conditions Effect of the crisis on access to and quality of water Effect of the crisis on sanitation and hygiene Effect of the crisis on housing Sources of water – quality of water Constraints to access to water Association/correlation between water access/quality and nutrition status Sanitation systems Association/correlation between sanitation systems and nutrition status Housing conditions Association/correlation between housing and nutrition status

(cont...)
(...cont)

	Outline of contents
	 3.5 Care practices Current infant and young child feeding practices: prevalence of breastfeeding, age of introduction of complementary foods, etc. Main infant and young child caregiver(s): age, education, etc. Association/correlation between care factors and nutrition status
	 3.6 Household food security Association/correlation between household food security and nutrition status (details on household food security are given in Section V)
IV. Markets	 4.1 Market structures and supplies Effect of the crisis on market structure, access to markets, level of market activity, and supplies to markets Location of markets in the area(s) Trade activity level: number of traders, types of commodity, etc. Level of supplies of the main staples: volumes available in markets and traders' stocks; comparison with the past: previous months and the same season in previous years
	 4.2 Levels and trends of market food prices Current market prices for food, changes from seasonal norm, trends
	 4.3 Main constraints to trade Movement of commodities from surplus to deficit areas: occurs/does not occur Constraints to trade from traders' viewpoint
	 5.1 Analysis of food security Food consumption Food access Food security groups
V. Household food security situation	 5.2 Livelihood characteristics of food-insecure households Human assets: gender, size, education (attendance, level), health Natural assets: crop and livestock/fisheries production – levels, damage, uses for consumption and income Physical assets: domestic and productive assets – ownership, sales Economic assets and strategies: income sources, migration/remittances, savings, expenditures (food and non-food), sources of food, debts Social assets: support structures, solidarity mechanisms, ongoing assistance programmes
	 5.3 Coping mechanisms Coping strategies used to obtain access to food and income Risks to lives and livelihoods caused by these strategies, if any Associations between food security status and coping strategies
VI. Main problems and priorities of	6.1 Main problems
households and key informants	6.2 Priorities for assistance identified by the population
VII. Conclusions on the nutrition and food security situation	 7.1 Summary of the food security and nutrition situation analysis Severity of food insecurity based on food consumption, access and coping strategies Severity of malnutrition at the population level 7.2 Groups most affected by food insecurity and risks to lives and livelihoods Summary characteristics of the food-insecure and/or malnourished Distinction between chronic and transitory food insecurity Main factors associated with food insecurity and risks to lives and livelihoods Tentative targeting criteria Estimated numbers of food-insecure people
	· Estimated numbers of food-insecure people

	Outline of contents
VIII. Scenarios and response options	 8.1 Forecasts and scenarios Shocks expected in the next 3, 6, 12 or more months Opportunities expected in the next 3, 6, 12 or more months Groups likely to be affected by predicted shocks and opportunities Most likely scenario, including shocks, opportunities, affected groups Worst-case scenario for preparedness and contingency planning, including shocks, opportunities, affected groups 8.2 Response options in the short and medium terms Options for short-term, early interventions Options for medium-term interventions
	 8.3 Analysis of response options: strengths, weaknesses, opportunities and threats SWOT analysis of the different options
IX. Recom- mendations	 9.1 Recommendations for interventions Interventions recommended to save lives and livelihoods, based on the results of the SWOT analysis Target beneficiaries, including targeting criteria and mechanisms Type(s) of assistance, quantities, start/finish dates
	 9.2 Recommendations for monitoring, reassessment and contingency planning Indicators to be monitored over the coming months to evaluate the evolution of the food security and nutrition situation Timing and type of reassessment, if appropriate Need for contingency planning
	Terms of reference
	List of team members
	Data collection tools: questionnaires, checklists
	Details of sampling procedures, if needed
	 Maps showing distribution of the affected population and severity of food insecurity
Annexes	List of reference documents (secondary data)
	Summary of stakeholder consultations
	 Optional annexes Description of households' food availability, access and consumption situation: overall, by livelihood group, by geographical area Description of linkages among food security, health and care factors Description of livelihood assets, strategies and context of households: overall, by livelihood group, by geographical area

Report quality monitoring

The report must be submitted to a quality check to ensure that:

- it contains all the required information;
- the analytical process is clear, and the reader can understand how the conclusions were reached.

At WFP, the quality check is carried out by experienced assessment officers posted in the region or at Headquarters, or by a qualified senior staff member of the country office who was not directly involved with the assessment. The quality check involves completing a quality monitoring checklist (QMC) in electronic format.

The results of the QMC are fed back to the team who carried out the EFSA, so that they can contribute to the learning process and the improvement of future EFSAs. WFP uses the QMC to decide whether a report is already suitable for posting on the public website (see **Section 4.6**) or will require further adjustment.

Communicating the results of the EFSA

4.1 Users of the EFSA report and communication channels

The EFSA report should enable decision-makers for WFP and its partners to understand the nature of the crisis and the types of intervention that may be helpful. It may also help donors to decide on the allocation of resources. It is therefore crucial that the assessment conclusions and recommendations are communicated clearly.

EFSA reports may be used by a wide range of stakeholders, as shown in Box 5.1.

Box 5.1: Potential users of an EFSA report

WFP

- Country directors require reliable and timely information to make decisions about crisis response.
- Managers at the country, regional and Headquarters levels need information that can be shared with governments and humanitarian partners. This is crucial for operational coordination and transparency.
- Programme officers need information that will help them to design or reorient interventions.
- Fundraisers need carefully researched and well-argued information to present to donors.

Other agencies and partners

- Host governments use the information for their own programmes and policy-making.
- United Nations and NGO partners use EFSA information when designing programmes.
- WFP implementing partners need to know the rationale behind the programmes that they are undertaking.
- Donors need accurate information when allocating resources.
- The media need reliable information to ensure accurate and balanced reporting.

Information can be communicated through a variety of channels:

- an EFSA executive brief;
- in-country workshops and presentations to stakeholders;

- dissemination of the EFSA report and executive brief by mail and e-mail;
- news media television, radio and newspapers;
- the websites of WFP and partners who participated in the EFSA;
- advocacy, when recommended interventions are outside the scope of WFP and its partners.

Box 5.2: WFP procedures for communicating EFSA results

For an EFSA led by WFP, the country director is responsible for disseminating the results. This enhances transparency and strengthens the links between assessment recommendations and decisions about programming and funding.

For an EFSA in which WFP participated but did not lead, the country director should work with the EFSA leader to ensure that the report is finalized and of acceptable quality, and that it is disseminated rapidly.

4.2 EFSA executive brief

The executive brief is directed to an audience of decision-makers in humanitarian and other food security interventions. These people are non-specialists and do not need to know the details of the analysis, but they must have a clear and accurate overview.

The executive brief is a stand-alone document. It differs from the executive summary presented at the beginning of the report. The executive brief must provide its audience with the information needed to make decisions on programming, funding allocation, targeting, etc. The information must be tailored to fit these needs. It must be both concise and complete.

The executive brief is prepared by the EFSA team leader and cleared by the country director. It is prepared as soon as the assessment analysis and recommendations are complete. *It can be prepared before the final report is complete*, and used before publication of the full assessment report, perhaps as a pre-alert for donors and partners.

The executive brief should not be more than three pages long. It should be written in clear, non-technical language, and should include visual instruments such as maps, tables, charts and graphics. It should also include the names and contact details of the people responsible for the assessment. An outline of its main contents is shown in Table 5.2.

Table 5.2: Conten	ts of the executive brief
	Summary of contents
Nature of crisis	Rapid-onset or chronicType: earthquake, conflict, etc.
Impact on food and nutrition security	 Summary of impact on nutrition, food availability, food access and food consumption
Affected population	 Estimated numbers of food-insecure people and households Description of food-insecure groups, the malnourished and their location
Reasons why people are at risk	 Main factors associated with food insecurity and risks to lives and livelihoods
Most likely future scenario(s)	 Explanation of the most likely evolution of the situation and the groups that will be affected Projected future numbers – in 3, 6 and 12 months – of food-insecure people
Recommended responses	 Type(s) of response, including summary of inputs required Objectives Target group(s) and numbers Start and finish dates

The template for an executive brief is available in Annex 6.

4.3 Presentation workshop or meeting

For each EFSA, a final workshop or meeting should be convened at which to present the results to partners, who include representatives of:

- government ministries;
- United Nations partners;
- national and international NGOs working in relevant sectors;
- WFP implementing partners; and
- donor agencies.

The purpose of the meeting is to:

- present the key findings and recommendations from the EFSA, and answer partners' queries on these;
- receive feedback and suggestions from the meeting participants;
- encourage partners' endorsement of the recommendations, to the extent possible, adjusted as a result of the meeting if required.

The meeting is an opportunity to develop a common understanding of the problem. It takes place after the conclusions and recommendations have been defined, but before final decisions about responses have been taken. This means that feedback from the meeting can be taken into account when finalizing the recommendations. EFSAs generally include recommendations for both food and non-food responses. At this meeting, WFP can explain the need for non-food responses to other stakeholders that may be able to contribute.

The meeting should be scheduled as far in advance as possible, to ensure good participation. For EFSAs in slow-onset chronic emergencies, it may be possible to organize a series of decentralized workshops and meetings at the district level in order to broaden the consultative base.

A draft of the executive brief, or of the full report if it is ready to be shared, should be distributed prior to the meeting to ensure that participants can familiarize themselves with the information before they meet. The executive brief and report will be amended after the meeting, on the basis of the feedback received.

4.4 Dissemination by e-mail and hard copy

In WFP, the country director is responsible for disseminating the EFSA report (see Box 5.2). Once finalized, the report and executive brief are e-mailed to all stakeholders and partners. This is done systematically, following compilation of a list of agencies at the national, regional and Headquarters levels. Hard copies may be delivered or mailed to the most significant stakeholders.

4.5 The news media

The media are an important channel for disseminating information. When they exist, agency public information officers at the national and regional levels should be aware that the EFSA is being undertaken, and should be updated about progress, conclusions and recommendations. They will ensure that the information is provided to the media through the right channels, when appropriate.

Information may be passed to the media through a variety of means, including:

- press releases;
- media advice notes;
- news briefings;
- press conferences;
- websites.

A press release is the most common way of informing the media about EFSA results. It constitutes a straightforward message, uses non-technical language and is tailored to journalists' needs. It is prepared by the public information officer according to specific guidelines.

The country director decides whether or not to issue a press release. This decision is based on the significance of the results, their newsworthiness, the political context, and the need for funds, among other factors. A press release may be issued in the affected country only, or also at the regional and international levels.

4.6 Websites and newsletters

Once it has been cleared by the country director, the EFSA report is usually sent to the relevant Headquarters unit. In WFP, the Food Security Analysis Service must receive all EFSA reports. After a report has gone through the QMC (see **Chapter 3**), the Service is responsible for posting it on WFP internal and public websites.

Reports may also be posted on other websites, such as:

- ReliefWeb, which is run by OCHA: reports can be submitted by e-mail to submit@reliefweb.int;
- Humanitarian Information Centres (HICs), managed by OCHA in disaster-affected countries;
- United Nations country team (UNCT) websites in the country concerned.

4.7 Advocacy for non-food recommendations

An EFSA may recommend both food and non-food responses to a crisis. This is because food and nutrition insecurity is multifaceted, and interventions may be required in such areas as education, hygiene promotion, water and sanitation, infrastructure development and agriculture.

WFP may not be able to implement some of the response recommendations, and must therefore encourage the government or other organizations to do so. The presentation meeting (see **Section 4.3**) is the starting point for discussions with possible collaborating partners. This should be followed by more specific meetings with interested agencies. The purpose of such meetings is to:

- provide a detailed explanation of the reasons why particular interventions have been recommended;
- emphasize the complementary nature of different food and non-food interventions;
- seek commitments from other organizations regarding interventions in specific sectors.

It is usually necessary to organize a series of such meetings.





i					
Ē	City/village name: City/village co	City/village code: $ _{-} $ Region code: $ _{-} $	code: _		
	Neighbourhood/street name :	Neighbourhood/street code: _ _ _	- - - :		
Clu	Cluster code: _ HH questionnaire number: _ _				
<i>da</i>) Enu	Date : _ _ / _ / _ / _ _ _ _ Team number: _ _ <i>day month year</i> Enumerator names :/	_			
Cor	Consent:				
We fam	We are conducting a survey of the food security situation of families in We would like to ask you some questions about your family and measure the weight and the height of all children under 5 years of age living in the family [OPTIONAL]. The survey usually	. We v years of age living	We would like to ask you some questions about your living in the family [OPTIONAL]. The survey usually	<pre>< you some que OPTIONAL]. TI</pre>	stions abo he survey
take Any you imp	takes 15 to 20 minutes to compete IADUST AS APPROPRIATE.]. Not information that you provide will be kept strictly confidential and will not be shown to other people. Your participation is voluntary and you can choose not to answer any or all of the guestions if you want. However, we hope that you will participate since your views are important. Do you have any guestions? May we begin now?	not be shown to ot wever, we hope th	her people. You at you will partic	ır participation i ipate since you	s voluntar Ir views ar
Т 	I – HOUSEHOLD COMPOSITION AND SCHOOL ATTENDANCE				
1.1	Who is the head of household? Is it a man or a woman?	1 = Male / 2 = Female	iale		_
1.2	How old is the head of household?		-	years	ş
1.3	What is the marital status of the head of household? [Omit this question if baseline exists and status/profile is not expected to have changed] 1 = Married 2 = Widowed 3 = Separated/divorced 4 = Cohabiting 5 = Nev marined 2 = Widowed 3 = Separated/divorced 4 = Cohabiting 5 = Nev	te <i>d to have</i> ng 5 = Never			
4.1				people	ole
Р	How many children and adults are currently living in the household?	Male		Fe	Female
0-5	0-5 years	1.5		1.6	
6-1	6-11 years [ADJUST AGE LIMITS AS APPROPRIATE]	1.7		1.8	
12-	12-17 years [ADJUST AGE LIMITS AS APPROPRIATE]	1.9		1.10	
18	18-63 vears [ADJUST AGE LIMITS AS APPROPRIATE]	1.11		1 1 2	

Annex 1: Sample household questionnaire

→ If Yes or 99, go to Section II → If Yes or 99, go to Check the skip Boys Boys 20 1.23 1.14 Girls Girls at Are there adult (18–59 years) members who have been sick or not fully functional for 0 = No / 1 = Yes 99 = not applicable 1.22 How many pregnant women are there in the household? Write "0" if there are none How many lactating women are there in the household? Write "0" if there are none 1.19 99 = not applicable 0 = No / 1 = Yes 5 = Absent teacher/poor-quality teaching 6 = Poor school facilities (building, toilets, etc.) 5 = Absent teacher/poor-quality teaching 6 = Poor school facilities (building, toilets, etc.) 1.13 What was the main reason for these children not attending regularly? What was the **main reason** for these children not attending regularly? 7 = No schools for girls/female teachers available 8 = No boarding school available 7 = No schools for girls/female teachers available 8 = No boarding school available Did ALL primary school-aged children regularly attend school in least 3 months over the past 12 months? Write "0" if there are none ADJUST RECALL PERIOD AS APPROPRIATE, SUCH AS THE Did ALL secondary school-aged children regularly attend attendance is an issue or if food aid/food rations are given to START OF SCHOOL YEAR, OR THE BEGINNING OF THE THE START OF SCHOOL YEAR, OR THE BEGINNING OF ADJUST RECALL PERIOD AS APPROPRIATE, SUCH AS 1 = Sickness/handicap 2 = Cannot pay school fees, uniforms, textbooks 1 = Sickness/handicap 2 = Cannot pay school fees, uniforms, textbooks 11 = Child works for cash or food (e.g. casual work, petty trade, begging) 11 = Child works for cash or food (e.g. casual work, petty trade, begging) Elderly (+ 63 years) [ADJUST AGE LIMITS AS APPROPRIATE] 10 = Child works for household unpaid (e.g. farming, family business) 10 = Child works for household unpaid (e.g. farming, family business) 4 = Early marriage (girl) 3 = Cannot pay transportation/far away 4 = Early marriage (girl) 9 = Domestic household chores unpaid (e.g. child care, washing) 9 = Domestic household chores unpaid (e.g. child care, washing) [Consider asking this question if secondary school encourage girls' attendance at secondary schools] ADJUST POSSIBLE ANSWERS AS APPROPRIATE ADJUST POSSIBLE ANSWERS AS APPROPRIATE] 13 = Other reasons (specify) 12 = Not interested 13 = Other reasons (specify) school in the past 6 months? 3 = Cannot pay transportation/far away the past 6 months? THE CRISIS] 12 = Not interested = Not applicable = Not applicable **CRISIS** 1.15 1.16 1.17 1.18 5 6 6

If answer is 5 or 6, go to 2.9 If answer is 1, go to 2.3 → If Yes, go to 2.9 If No, go to 2.9 → If No, go to 2.9 \uparrow 3 = Death of animals for climatic reasons (drought, floods) 4 = Lack of pasture or water for animals **1** = Resident **2** = Displaced less than 3 months ago **3** = Displaced 3–6 months ago **4** = Displaced ADJUST RECALL PERIODS AS APPROPRIATE, CONSIDERING THE TIMES OF THE FIRST 2 = private dwelling mostly in non-durable material (planks, plastic, mud, Ask this question if it is relevant for the particular assessment and context. Adjust 1 = Insecurity in place of origin (conflict, threats – may include loss of house, land, animals) 0 = No / 1= ILOCAL CURRENCY] 0 = No / 1 = Yes 0 = No / 1 = Yes Yes 1 = private house mostly in durable material (brick, cement) 0 = No / 1 = Has your debt for rent increased over the past 6 months? (ADJUST RECALL PERIOD AS APPROPRIATE, FOR EXAMPLE REFER TO THE BEGINNING OF THE CRISISI 2 = Loss or lack of harvest for climatic reasons (drought, floods, landslide) 5 = Loss or lack of economic opportunities/lack of labour in place of origin 5 = room(s) in a collective centre/public building Yes 8 = To receive health treatment 9 = For education/study purposes 4 = room(s) in a shared house or shared flat 6 = To graze/water animals 7 = To join other family members 6 = tent /plastic sheeting/shelter in camp ADJUST POSSIBLE ANSWERS AS APPROPRIATEI When did you move here (displaced or resident)? What was the main reason for you to come here? - HOUSING, WATER AND ENERGY ACCESS 3 = flat in multi-storey building How much rent do you pay per month? Are you currently in debt for rent payment? 6-12 months ago 5= Displaced >12 months ago Do you have to **pay rent** for your dwelling? DISPLACEMENT AND THE ASSESSMENT Are you the owner of your dwelling? = other (specify) wording if there is displacement] 10 = For another reason (specify) grass) AS APPROPRIATE] note the type **Observe** and CATEGORIES of dwelling **ADJUST** 2.6 2.2 2.3 24 2.5 2.7 20 10 5 =

[ADJUST RECALL PERIOD AS APPROPRIATE, FOR EXAMPLE REFER TO THE BEGINNING OF THE CRISIS; CONSIDER SEASONAL ISSUES IF APPROPRIATE (eg. SOURCE OF WATER)]		At the moment		6 months ago
[OPTIONAL: for when the crisis is expected to have affected the following] Where do household members go to use a toilet ? 1 Flush latrinefoilet with water 2 = Traditional pit latrine (no water) - Open pit (no roof or no walls) 3 = Communal latrine 4 = Noneibush	50		2.10	
Where do you obtain water for drinking? 1 = Safe source (piped water, public tap, tube well/borehole, protected well, protected spring water, rainwater, bottled water) 2 = Unsafe source (river, unprotected well, unprotected spring water, canal)	2.11	 If answer is 1, go to 2.15	2.12	 If answer is 1, go to 2.15
Do you treat the water before drinking it? 0 = No 1 = Yes, with chlorine tablets 2 = Yes, by boiling 99 = Do not know	2.13		2.14	
[OPTIONAL: for when the crisis is expected to have affected the following] following] How long does it take to reach the main source of drinking- water and come back (in minutes walking)? <i>Write "0" if within the</i> house or dwelling.	2.15	mins	2.16	suim
What is your <u>main</u> source of fuel for <u>cooking</u> ? 1 = Wood 2 = Animal dung 3 = Electricity 4 = Gas 5 = Coal/charcoal 6 = Other (<i>specify</i>)	2.17	<u> </u>	2.18	
What is the main constraint to access to cooking fuel? 1 = Distance 2 = Insecurity 3 = Cost 4 = Other (specify)	2.19		2.20	
What is your <u>main</u> source of fuel for <u>heating</u> ? 1 = Wood 2 = Animal dung 3 = Electricity 4 = Gas 5 = Coal/charcoal 6 = Other (<i>specify</i>)	2.21		2.22	

Do you us ually c (own + rented)?	Do you usually cultivate (own + rented)?	ate	a home garden other land/fields	•	۶ ۳	0 = No / 1 = Yes	es	3.1 3.2											
ouse	hold doe	IS NOT (If the household does NOT usually cultivate a garden or other land, go to column <u>D</u> below, ask about food crops and continue with columns E and	a garı	den o	r othe	r land,	go to c	olumn	D bel	ow, ask at	out foo	d crop:	s and c	ontinue	e with column	is E an	ЧГ	
3.3	Did you	plant L	Did you plant LAST season?	= 0	. / oN ;	0 = No / 1 = Yes	ø	_	if No, į	go to c	olumn <u>C</u> I	below, a	sk abo	ut crop	s and c	If No, go to column $\overline{ extsf{C}}$ below, ask about crops and continue with columns D, E and F	colum	ns D, Ean	ц
			ASK ONLY ABOUT THE CROPS THAT ARE USUALLY CULTIVATED BY THE HOUSEHOLD	OUT TI	HE CR	OPS T	S THAT ARE U HOUSEHOLD	RE USUA	TLYC	NLTIVA	TED BY TH	Е	미			ш			
	Do you	-	LEAVE BLANK IF NOT CUL TIVATED (DO NOT WRITE ZERO)	IF NO	T CUL	TVAT	00) 03	NOT WF	NTE ZE	RO)		T	How much of	ch of	Цомо	How do how more		щ	
ADJUST	USUALLY	۲,										th€	these products	ducts	han	w much uo you have in stock			-
THE 6 (or	cultivate	cultivate		٩		-		: 00			o				Ū	currently		now long will the stocks last for	rue L
fewer) MAIN		1 shoin	What area of these crops	f these	e crop		what p	What proportion	_	in vor	How much of these		for family		g	(production,	5	family	5
APPRO-	1 = Yes			olante	d this		haive:	or une crop (harvest or area)	(e	produ	products do you		consumption	tion	bnu	purchase, gifts,	8	cons umption?	č
PRIATE]	If No, do not asi other questions for these crops	lf No, do not ask other questions for these crops	[DEFII MEASU	season? VE THE UNI RE PRIOR T SURVEY]	T OF THE		dam k	has been damaged or lost?	Ĵ.		DEFINE THE UNIT OF MEASURE PRIOR TO THE SUBVEVI	-	during the year? [DEFINE THE UNIT OF MEASURE PRIOR	year? THE PRIOR	DEF OF PR	DEFINE THE UNIT OF MEASURE PRIOR TO THE SUIDVEV	Writ	Write 0 if less than 1 month	n 1
					ļ	+	F		╉	ŏ	ושאער	5	TO THE SURVEY	RVEY]					
Tomatoes	3.4		3.5				3.6		%	3.7	Kg	3.8	_		3.9	Kg	3.10		mths
Onions	3.11		3.12				3.13		% 3	3.14	kg	3.15	_	kg	3.16	kg	3.17		mths
Potatoes	3.18		3.19				3.20		% 3	3.21	Kg	3.22	_	K	3.23	kg	3.24	_	mths
Wheat	3.25		3.26				3.27		% 3	3.28	Kg	3.29	_	Kg	3.30	kg	3.31	_	mths
Sorghum	3.32		3.33				3.34		% 3	3.35	Kg	3.36	_	K	3.37	kg	3.38]	mths
			-																
	Э.	3.46 V	Will you plant NEXT season?	XT sea	son?						0 = No / 1 = Yes			Yes, g	If Yes, go to 4.1				
	If yc	ou will I	If you will NOT plant nextseason, why not? [ADJUST CATEGORIES AS APPROPRIATE, FOR EXAMPLE CONSIDED INSECTION: ISSUES ALSON ACEMENT FTC1	eas or	vhy,	not?	(ADJU	ST CATE	GORIE	S AS AI	PROPRIA	щ	= 0	0 = No/ 1= Yes	Yes				
	3.47	7 Gard	Garden or land too far away	away	2	5			2	-									
	3.48		Insecurity when cultivating or harvesting	ating o	r harv	esting													
	3.49		Land under water/washed away	shed av	way														
	3.50		Lack of seeds/seedlings	gs															
	3.51		Lack of irrigation or drought conditions prevent cropping	ought	condit	tions p	revent	cropping	Ē					ł					
	3.52		Lack of fertilizer/poor soil fertility	soil fer	tility														
	3.53		Lack of pesticide/crop disease	diseat	se														
	3.54		Lack of tools/farm machinery	chinen	٨														
	3.55		Lack of storage facilities/losses	es/los	ses														
	3.56		Lack of workforce																
	3.57	_	Lack of market for produce sales	duce s	sales														
	3.58	_	Will be absent during planting season	pla ntin	id sea:	u os								_					
					20		ļ		ļ	ļ			ļ		ļ				

IV – <u>ANIMALS</u>

Ask questions column by colum r [[F IMPORTANT IN THE CONTEX DRAUGHT/TRANSPORT ANIMAL buffaloes]]	Ask questions column by column (JF IMPORTANT IN THE CONTEXT, ADD PIGS, DRAUGHTTRANSPORT ANIMALS (e.g. horses/donkeys, buffeles)]		Cattle		Sheep and/or goats		Poultry
ow many of th	How many of these animals do you CURRENTLY own?	4.1		4.2		4.3	
Did you own the 0 = No / 1 = Yes ADJUST RECALL REFER TO THE B	Did you own these animals 6 months ago? 0 = No / 1 = Yes [ADJURT EGALL PERIOD AS APPROPRIATE, FOR EXAMPLE [REFER TO THE BEGININING OF THE CRISS]	4.4	If No, go to 4.2	4.5	If No, go to 4.3	4.6	 If No, go to 4.25
Have you sold a 0 = No / 1 = Yes ADJUST RECALL REFER TO THE BI	Have you sold any animals during the past 6 months ? 0 = N.1 = Yes ADJUGST RECALL PERIOD AS APPROPRIATE, FOR EXAMPLE REFER TO THE BEGINNING OF THE CRISIS]	4.7	If No, go to 4.20	4.8	<i>I</i> f No, go to 4.21	4.9	If No, go to 4.25
lf yes, have you 0 = No / 1 = Yes	lf yes, have you sold reproductive females ? 0 = No / 1 = Yes	4.10		4.11		4.12	
What was the main reason 1 (reproductive or not)? 1 = Need of money 2 = cld agi 3 = Infertility 4 = Lack of water 6 = Other reason (<i>specify</i>)	What was the main reason for selling animals (reproductive or not)? Ta Need of money 2 = 0/3 age/sickness 3 = infertility 4 = Lack of water 6 = Unter reason (<i>specify</i>)	4.13		4.14		4.15	
Did you sell mor 0 = No / 1 = Yes	Did you sell more animals than usual for this time of year? 0 = No / 1 = Yes	4.16	_	4.17		4.18	_
Have some anim 0 = No / 1 = Yes ADJUST RECALL REFER TO THE B	Have some animals died during the past 6 months ? 0 = No.1 = Yes [ADJURT REGALL PERIOD AS APPROPRIATE, FOR EXAMPLE [REFER TO THE BEGINNING OF THE CRISS]	4.19	<i>If No, go to 4.2</i>	4.20	If No, go to 4.3	4.21	If No, go to 4.25
What was the main ca [ADJUST CATEGORIES , [ADJUST CATEGORIES , [1 = Low temperatures 2 = Old age/sickness/misc 4 = Lack of water 6 = 6 = Other reason (specify)	What was the main cause of death ? [AD.UDT CATEGORIES AS AFPROPRIATE] = Low temperatures 2 = Old ageisickness/miscarriage 3 = Injury 4 = Leck of varet = 6 = Leck of rodder/animal feed 6 = Chiter reason (specify)	4.22		4.23		4.24	
hat are your n	What are your main difficulties with animal raising ? [AJUST CATEGORIES AS APPROPRIATE]	CATEG	ORIES AS APPROPRIATE]				0 = No 1 = Yes
4.25	Lack of money to buy animals						
4.26	Lack of water						
4.27	Lack of pasture/animal feed expensive						

4.28	Lack of shelter for animals					
4.29	Animal diseases/lack of veterinary services					
4.30	Lack of workforce to look after animals					
4.31	Lack of market to sell animals/products					
4.32	Thefts/looting					
4.33	Other constraint (specify)					
- INCOME S	V – <u>INCOME SOURCES, KINSHIP SUPPORT AND ASSETS</u>					
ADJUST RECAL	ADJUST RECALL PERIOD AS APPROPRIATE, FOR EXAMPLE REFER TO THE BEGINNING OF THE CRISIS]	บี	Currently		6 months ago	
How many hor	How many household members earn income? 5.1			5.2		
What is your m What comes s	What is your most important source of income? What comes second? [ADJUST INCOME CATEGORIES AS APPROPRIATE]		Most important source	-	Second source	
T = Sale of vegetables or fulls = Sale of vegetables or fulls = Sale of vegetables or fulls = Sale of cotton, tobacco, or = Sale of cotton and sale = Sale of and sale = Sale of handcrist = Sale of handcrists = Sale of handcrists = Sale of handcrists = Petry table day lobor, cash = Household allowances, ur = Sale soft and cash = Household allowances, ur = Sale soft and cash = Household allowances, ur = Sale soft allowances, ur = Sale soft allowances, ur = Sale soft and cash = Household allowances, ur = Sale soft allowances, ur = Sale s	 Sale of cereal production (wheat, maize, barley) Sale of cereal production (wheat, maize, barley) Sale of vogeniles or fruits Sale of vogeniles or fruits Sale of contron, tobacco, or other cash crop Sale of contron standarding to the control of an other cash Sale of the cash Sale of	ື ທີ່]	τ <u>ς</u>	Ī	
What share of your total inc Use proportional piling if needed Total may not =100% if there are	What share of your total income do these two sources of income provide? Use proportional piling if needed Total may not =100% if there are more than 2 sources of income, but should not be	de? 6 5.5	%	5.6	%	
Are these two	1 = Temporary/cseual 2 = Seasonal 3 = Stable/permanent	5.7	%	5.8	%	

5.19	Are there members of your household who are not working but are currently looking for work?	hold who are not w	orking bu	t are c u	urrently looking		0 = No / 1 = Yes		'	_
5.10	Has your income changed in the past 6 months? [ADJUST RECALL PERIOD AS APPROPRIATE, FOR EXAMPLE REFER TO THE BEGINNING OF THE CRISIS] 1 = No change / 2 = Decreased / 3 = Increased	<pre>past 6 months? OPRIATE, FOR EX/ change / 2 = Decrea;</pre>	AMPLE RE	FER TO Icreased	THE		If N	o change	 If No change, go to 5.12	5.12
5.11	By how much has it changed (decreased or increased)? Estimate the	creased or increas <i>Estimat</i> e	ed)? • the propol	rtion (in	increased)? Estimate the proportion (in %). Use proportional piling if needed	ial piling it	^r ne eded			%
5.12	When you are in need, can you receive food from relatives, neighbours or friends?	sceive food from	relatives,	neight	ours or friends	\$				
5.13	Have you received such support in the past 6 months? [ADJUST RECALL PERIOD AS APPROPRIATE, FOR EXAMPLE REFER TO THE BEGINNING OF THE CRISIS]	: in the past 6 mol OPRIATE, FOR EX/	nths? MPLE REI	FER TO	THE BEGINNING	OF	0= No 1= Yes			
5.14	Are you supporting relatives with food or with cash at the moment?	th food or with ca	sh at the	mome	nt?					
5.15	Do you have family members who have migrated out ?	have migrated o u	ıt?		0 = No 1 = Yes	= Yes		If No	 If No, go to 5.23	23
				2	Migrant No.1		Migrant No.2		2	Migrant No.3
hen = Les = 0.1	When did they leave? 1 = Less than 1 month ago 2 = Less than 6 months ago 3 = 6-12 months ago 4 = More than 1 year ago 9 = No 2 ^m incrant, or no 3 ^m incrant	ionths ago go		5.16		5.17			5.18	
hyd	Why did they leave?				-		-	_		-
= Wo	1 = Work 2 = Studies 3 = Health treatment 4 = Family reunion 5 = Insecurity/threats 99 = No 2 nd migrant, or no 3 rd migrant	4 = Family reunion no 3 rd migrant		5.19	-	5.20			5.21	-
5.22	Do any of the migrants send money back?	ey back?			0 = No 1 = Yes	= Yes				
o yo	Do you currently own?	0 = No 1 = Yes	Do you	curren	Do you currently own?					0 = No 1 = Yes
5.23	Fridge		5.24	Farm	Farm machinery (tractor, other)	or, other	(_
5.25	Oven (electric, gas)		5.26	Agricu	Agricultural tools (machete, hoe, etc.)	hete, ho	e, etc.)			
5.27	Television		5.28	Bicycle	e					
5.29	Satellite dish		5.30	Motorbike	bike					
5.31	Radio		5.32	Car, t	Car, taxi, truck					
5.33	Cell phone		5.34	Cash,	Cash, other savings (e.g. jewellery)	e.g. jewel	llery)			
5.35	Sewing machine		5.36	Bank	Bank account					

VI – EXPENDITURES AND DEBTS

10 100	In local currency	WEEK fo	WEEK for FOOD?		How much d	How much did you spend LAST MONTH on:	I LAST MON	TH on:
6.1	Bread, wheat [ADJUST STAPLE FOODS AS APPROPRIATE]			6.2	Fuelwood/charcoal	harcoal		
6.3	Rice	_	_	6.4	Gas/electricity	sity		
6.5	Cassava	_		6.6	Kerosene/petrol	etrol		
6.7	Potatoes			6.8	Housing (rent, repairs)	nt, repairs)		
6.9	Oil, butter, ghee	_		6.10	Water			
6.11	Sugar, honey, jam		_	6.12	Health			
6.13	Milk, cheese, yogurt			6.14	Education			
6.15	Meat, fish, eggs	_		6.16	Transportat purposes)	Transportation (for private purposes)	fe	
6.17	Vegetables, fruits	_	_	6.18	Clothing, shoes	loes		
6.19	Groundnuts, nuts			6.20	Soap, hygiene items	ene items		
6.21	Other food expenditures (condiments, beverages, etc.)			6.22	Ceremonie	Ceremonies, entertainment	ent	
6.23	Meals outside, snacks, beverages consumed outside home	les		6.24	Reimbursement debts	ment debts		
				6.25	Any other n	Any other non-food expenditures	enditures	
6.26	Have your expenditures changed compared with 6 months ago? [ADJUST RECALL PERIOD AS APPROPRIATE, FOR EXAMPLE REFER TO THE BEGINNING OF THE CRISIS] 1= No change 2 = Decreased 3 = Increased	ed compa PROPRIA sed 3 = In	ired with 6 months TE, FOR EXAMPLE F creased	ago ? REFER T	O THE BEGIN	INING OF THE		
Whic	Which expenditures have changed?	9.27	Food			6.28 Health	alth	
1 = No = De	1 = No change 9 = Decreased 3 = Increased	6.29	Energy (cooking, heating, lighting)			6.30 Edu	Education	
í		6.31	Housing			6.32 Tran	Transportation	

months → If No, go to 6.36 → If No, go to 7.1 ł **0** = No **1** = Υes In how many months do you think you will be able to repay your debts or credit? 1 = Relatives (excluding remittances from migrants abroad)
 2 = Traders/shopkeepers
 3 = Bank/credit institution/micro-credit project
 4 = Money lender Who is the main source of credit for all debts and loans (recent and old)? Have you have contracted **new debts or credit in the past 6 months**? ADJUST RECALL PERIOD AS APPROPRIATE, FOR EXAMPLE REFER TO 8 = To buy or rent a flat/house 3 = To pay school, education costs 4 = To buy agricultural inputs (seed, tools...) **10** = to pay ticket/cover travel for migration **11** = Other reason (*specify*) 6 = To buy animals **5** = Landlord (more than 1 month behind in rent) **6** = Other (*specify*) Do you have any **debt or credit to repay** at the moment? 2 = To cover health expenses What was the main reason for new debts or credit? **9** = To buy clothes, shoes **9** = To pay for ceremonies 5 = To buy animal feed, fodder, veterinary THE BEGINNING OF THE CRISISI 7 = To buy or rent land 1 = To buy food ။ က 6.33 6.34 6.35 6.37 6.36

г															1									Γ			7
	Children 1–5 years			Where do you USUALLY obtain	most of this food?	pass for each rood isted, even in not consumed in the past 7 days]	2 = central market purchase	or food urs/relatives	ervices	eaten															-	_	
-	ວ່	7.2	7.4	do you	most of	in the p	= centra	= Work 1 neiahbo	goods/s	ally not ∈																	
	Adults		 	Where		Consumed	1 = Own crop/garden production 2	3 = Local market/shop purchase 4 = Work for food 5 = Borrowing/debts 6 = Gifts from neighbours/relatives	 7 = Food aid 8 = Barter or trade of goods/services 9 = Scavenging for food/wild food collection 	99 = Not eaten in past 7 days or usually not eaten	7.7	7.10	7.13	7.16	7.19	7.22	7.25	7.28	7.31	7.34	7.37	7.40	7.43	7 46	2	7.49	
	۹	7.1	7.3	the	e of	54.	arden pr	et/shop p debts 6	8 = Barte J for food	in past 7															ī	_	
			3 II	What was the	main source of	7 DAYS?	/n crop/g	sal marke rrowing/o	od aid avenging	ot eaten	_	_	_	_	<u> </u>	_	_'		_				_		<u>-</u>	<u>'</u>	
			Same /	Å	mai		1 = 0	3 = Lo	7 = Fo 9 = Sc	N = 66	7.6	7.9	7.12	7.15	7.18	7.21	7.24	7.27	7.30	7.33	7.36	7.39	7.42	7 45	2	7.48	
			1 = Less / 2 = Same / 3 More	of the	as the		1 dav	ays	ays																		
		ten by:	1 = Le More	many c	AYS w		en 1=	3 = 3 days	7 = 7 days							-								_	Ī		
NO		als were eat	:h usual?	On how many of the	PAST 7 DAYS was the		0 = Not eaten 1 = 1 dav	2 = 2 days	6 = 6 days		7.5	7.8	7.11	7.14	7.17	7.20	7.23	7.26	7.29	7.32	7.35	7.38	7.41	7 44		7.47	
VII - FOOD CONSUMPTION		Yesterday, how many meals were eaten by	How does this compare with usual?		Do NOT count small	quantities (less than 1 teasnoon/nerson)	ADJUST THE LIST AS	APPROPRIATE, BUT KEEP TO A MAXIMI M OF 13-14	ITEMS]		Bread, wheat	Rice	Potatoes	Pasta, biscuits	Beans, lentils, peas, nuts	Vegetables	Fruits	Meat, offal	Eggs	Fish	Milk, cheese, yogurt	Sugar, honey, jam	Oil, fats	Street food 1 (specify which was the most	frequent)	Street food 2 (specify which was the 2 nd most frequent)	

ANNEXES

6 months? [ADJUGT RECALL PER EXAMPLE REFER TO THE BEGIN DE <u>NOT</u> ifst, leave the househinporta housshold to rank the 3 most importa [ADJUST THE LIST AS APPROPRIATE] = Loss employment/reduced salary 2 = Loss employment/reduced salary 2 = Death of household member/funerals 5 = High fuel/transportation prices 6 = F 7 = Death of noteshold member/funerals 9 = Electricity[39 se cuts 10] = Insecutiv/Ins 91 = Poor harvest/inturlet 10 = Fruvior	6 months? (ADUUGT RECALL PERIOD AS APPROPRIATE, FOR EXAMPLE REFER TO THE BEGINNING OF THE CRISIS] household is aver the household to answer spontaneously, then ask the household to ank the 3 most important				o dillocatio
Do NOT [ist, leave th household to rank t [ADJUST THE LIST / [ADJUST THE LIST / [ADJUST THE LIST / [ADJUST HE LIST /] = Loss employment = High fueltranspoi = Electro impusso = Electricity/gas cut	he household to answer spontaneously, then ask the the 3 most important				
 1 = Loss employment 3 = Death of househc 5 = High fue/transpoi 6 = Debtio reimburse 9 = Electricity/gas cut 11 = Proor hardwast/dimension 	AS APPROPRIATE]				
 5 = High fuel/transpoi 7 = Debt to reimburse 9 = Electricity/gas cut 11 = Pour harvest/drr 	salth expenditures 8.1	8.2		8.3	
9 = Electricity/gas cut 11 = Poor harvest/dro	= High fue/transportation prices 6 = Rent payment = Debt to reimburse 8 = Irregular/unsafe drinking-water				
	9 = Electricity/gas cuts 10 = Insecurity/thefts 11 = Poor harvest/drought 12 = Environmental problems (pollution, industries)				
13 = Floods, heavy rains, landslides 1 99 = No 2 nd or no 3 rd difficulty mentioned	13 = Floods, heavy rains, landslides 14 = Other shock (<i>specify</i>) 99= No 2 nd or no 3 rd difficulty mentioned				
Over the pa	Over the past 6 months, have there been times when you did not have enough money to	ough mone			
8.4 buy food or [ADJUST REC CRISIS]	buy food or cover other essential expenditures (health, cooking fuel, school, etc.)? PADUST RECALL PERIOD AS APPROPRIATE, FOR EXAMPLE REFER TO THE BEGINNING OF THE CARISIS	etc.)? GINNING OF		0 = No 1 = Yes	
Over the past 6 m [ADJUST RECALL P CRISIS]	Over the past 6 months, did you have to do any of these things to meet food and other needs? (ADJUST RECALL PERIOD AS APPROPRIATE, FOR EXAMPLE REFER TO THE BEGINNING OF THE CRISIS]	id other nee IG OF THE		0= No / 1= Yes	= Yes
8.5 Consume se	Consume seed stocks held for next season				
8.6 Decrease ex	Decrease expenditures for fertilizer, pesticide, fodder, animal feed, veterinary care, etc.	care, etc.		_	
8.7 Sell domest	Sell domestic assets (radio, furniture, fridge, TV, carpet, etc.)				
8.8 Sell product	Sell productive assets (farm implements, sewing machine, motorbike, land, etc.)	tc.)			
8.9 Sell more ar	Sell more animals than usual				
8.10 Decrease ex	Decrease expenditure for health care			_	
8.11 Take childre	Take children out of school			_	
8.12 Seek alterna	Seek alternative or additional jobs, work additional hours			_	
8.13 Increase the	Increase the number of members out-migrating for work and/or food				
During the past 7 c	During the past 7 days, has anyone in your household done any of these	0 = Never 1 = Seldom	0 = Never 1 = Seldom (fewer than 15 out of 30	15 out	of 30
things? [NOTE THA ON THE COPING ST	things? NOTE THAT THE 7-DAY RECALL IS ACCORDING TO THE LATEST GUIDANCE ON THE COPING STRATEGY INDEX FOR FOOD-RELATED STRATEGIES	days) 2 = Often (r 3 - Americ	days) 2 = Often (more than 15 out of 30 days) 2 = Alviori	5 out of 3	0 days)
8.14 Rely on less	Rely on less preferred and less expensive food	0 - AW499	-		
0 15 Device food	-				

8.16	Receive food gifts/rely on food help from friends or relatives	or relatives			
8.17	Limit portion size at meals				
8.18	Restrict consumption by adults so that small children can eat	ren can eat			
8.19	Reduce number of meals eaten in a day				
8.20	Pass entire days without eating				
8.21	Purchase food on credit, incur debts				
Are y Speci THE L	Are you currently receiving this assistance? Specifically ask about each type listed [ADJUST THE LIST AS APPROPRIATE]	0 = No 1 = Yes	[ADJU APPF	[ADJUST THE LIST AS APPROPRIATE]	0 = No 1 = Yes
8.22	Food for schoolchildren (eaten at school or take-home)		8.23	Free seeds, fertilizer (including programmes with pay back)	
8.24	Food for young/malnourished children or for pregnant/lactating women	_	8.25	Free agricultural tools	
8.26	Free food ration for the household		8.27	Free fodder, animal feed	
8.28	Food for work		8.29	Free veterinary services	<u> </u>
8.30	Cash for work		8.31	Plastic sheeting, tent, other housing material	
8.32	Cash transfers (grants, child subsidies, family subsidies, etc.) from NGOs, other agencies or government social assistance programmes		8.33	Cooking utensils, stove, kitchen set	
8.34	Free health care/drugs	-	8.35	Clothes, blankets	
8.36	Micro-credit		8.37	Mosquito nets	
8.38	Free seeds, fertilizer (including programmes with pay back)		8.39	Cooking utensils	
8.40	Free agricultural tools		8.41	Jerry cans	_
8.42	Free fodder, animal feed		8.43	Other assistance (<i>specify</i>)	

Do <u>NOT</u> list, leave the household to answer spontaneously, then	1 st pri	ority	2 nd F	1 st priority 2 nd priority	3 rd	3 rd priority
ask the household to rank the 3 most important ones						
1 = School feeding 2 = Food for voune/malnourrished children or other vulnerable groups						
3 = Food for the household in general						
4 = Cash transfers 5 = Credit						
6 = Employment 7 = Increased wage, increased pension						
8 = Seeds 9 = Fertilizer						
10 = Pesticides 11 = Agricultural tools		_	10	_	0 10	_
12 = Fodder, animal feed 14 = Veterinary services	0.4 1 1		0.4.0		0.40	
15 = Irrigation 16 = Water for domestic use						
17 = Health services (infrastructure, staff), treatment (subsidies, free drugs)						
18 = Clothes, shoes 19 = Roads 20 = Public transportation						
21= School infrastructure/teachers 22 = Housing/shelter (repairs, new)						
23 = Fuel (subsidies, donated) 24 = Security 25 = Other assistance (specify)						

Ask only about children und	Child I	<u>rs - Complete 1st child</u> No. 1	(column Child I		t child Child N	lo. 3
Sex of child - 1 = Male / 2 = Female	9.1.1	_	9.2.1	_	9.3.1	_
Date of birth [If supported by a document or checked – If no document or not known, estimate age in months]	9.1.2	/// day / month / year	9.2.2	/// day / month / year	9.3.2	/// day / month / year
Age in months [Estimate if date of birth is uncertain or unknown]	9.1.3	months	9.2.3	months	9.3.3	 months
Has the child been sick in the 2 weeks before the survey? 0 = No / 1 = Yes	9.1.4	 If No, go to 9.1.9	9.2.4	 If No, go to 9.2.9	9.3.4	ا If No, go to 9.3.9
What was the child's main sickness? 1 = Fever 2 = Repeated coughs/breathing difficulties 3 = Diarthoea 4 = Measles 5 = Other	9.1.5	1_1	9.2.5	1_1	9.3.5	_
Was the child taken to a health service? 0 = No / 1 = Yes	9.1.6	 If Yes, go to 9.1.8	9.2.6	اا If Yes, go to 9.2.8	9.3.6	 If Yes, go to 9.3.8
If the child was <u>not</u> taken, why? 1 = Not serious 2 = Far away/lack transport 3 = Lack money 4 = Does not like/distrusts 5 = Other source of treatment was used	9.1.7	1_1	9.2.7	1_1	9.3.7	_
6 = Other (specify) [Optional (see with UNICEF)] If the child had diarrhoea, did you give him/her oral rehydration solution (ORS)? 0 = No / 1 = Yes 99 = Did not have diarrhoea	9.1.8	1_1	9.2.8	_	9.3.8	
Did the child receive a vitamin A capsule during the past 6 months? [Show capsule!] 0 = No / 1 = Yes	9.1.9	1_1	9.2.9	_	9.3.9	_
Did the child receive breast milk yesterday? 0 = No / 1 = Yes	9.1.10	_	9.2.10	_	9.3.10	_
[Questions from 9.1.11 to 9.1.27 are optional and useful mostly if there is capacity to analyse the standard infant feeding indicators] Did the child receive animal milk yesterday? 0 = No/1 = Yes	9.1.11	1_1	9.2.11	_	9.3.11	_
How many times did the child receive breast milk or animal milk yesterday?	9.1.12	_	9.2.12	_	9.3.12	_
Did the child receive solid, semi- solid or soft foods yesterday? 0 = No / 1 = Yes	9.1.13	 If No, go to 9.1.28	9.2.13	 If No, go to 9.2.28	9.3.13	 If No, go to 9.3.20
How many times did the child receive food yesterday?	9.1.14	_	9.2.14		9.3.14	I

IX - UNDER-5 CHILDREN HEALTH AND NUTRITION STATUS

What food items did the child eat	yesterda	y? 0 = No / 1 = Yes				
Wheat, bread, rice, pasta, biscuits	9.1.15	_	9.2.15	_	9.3.15	_
Potatoes	9.1.16	_	9.2.16	_	9.3.16	_
Beans, peas, lentils, nuts	9.1.17	_	9.2.17	_	9.3.17	_
Milk, cheese, yogurt	9.1.18	_	9.2.18	_	9.3.18	_
Meat, liver, kidney, chicken, fish	9.1.19	_	9.2.19	_	9.3.19	_
Eggs	9.1.20	_	9.2.20	_	9.3.20	_
Yellow/orange vegetables (pumpkin, carrots) and/or fruits (apricots, plums)	9.1.21	_	9.2.21	_	9.3.21	_
Other vegetables and fruits, including fruit juices	9.1.22	_	9.2.22	_	9.3.22	_
Tea	9.1.23	_	9.2.23	_	9.3.23	_
Plain or sugary water	9.1.24	_	9.2.24	_	9.3.24	_
Infant formula	9.1.25	_	9.2.25	_	9.3.25	_
Porridge	9.1.26	_	9.2.26	_	9.3.26	_
Vitamins, mineral supplements and/or medicine	9.1.27	_	9.2.27	_	9.3.27	_
[Anthropometric measurements are optional and must be done only if quality (training, supervision, equipment) and analytical capacity are ensured. MUAC is the simplest.] Has the child got bilateral oedema? 0 = № / 1 = Yes	9.1.28		9.2.28	_	9.3.28	_
What is the child's weight	9.1.29	· kg	9.2.29	. kg	9.3.29	. kg
What is the child's height	9.1.30	_ _ _ . _ cm	9.2.30	_ _ _ . _ cm	9.3.30	_ _ _ . _ cm
What is the child's MUAC	9.1.31	_ _ · _ cm	9.2.31	_ _ . _ cm	9.3.31	lll . ll cm

Annex 2: Sample key informant questionnaire

Ann	ex 2 : Sample key inf	ormant questio	nnaire Questionnaire ID: _ _	_ _	_ _ _ _	_
-			City/village code: _ Region c		1-1	
Neig [INCI Clus	hbourhood/street nam UDED IN URBAN ASSESS ter code : _ _ Key	e: MENTS ONLY <mark>]</mark> Informant question	Neighbourhood/street code	: _	_ _	
day	: _ _ / _ _ / _ month year					
	merator names:	ite (latitude)	X-coordinate (longitude)			
I- <u>D</u> E						
1.1	How many people are APPROPRIATE]	living in this villag	e/neighbourhood? [SELECT LOCATION AS	\$		_
1.2	How many RESIDENT [SELECT LOCATION AS		living in this village/neighbourhood?			II
1.3	How many DISPLACE [SELECT LOCATION AS		e living in this village/neighbourhood?			
1.4		RECALL PERIOD A	left the village/neighbourhood S APPROPRIATE, FOR EXAMPLE REFER TER]	-	No <i>l</i> Yes	_
1.5	Who has left most?		1 = Men 2 = Women and children 3 = Whole families 4 = Do not know			_
1.6	Where have they gone, mainly?		ring villages 2 = To neighbouring cities		,	_
1.7	During the past 6 mor	ths, have people [ADJUST RECALL	arrived to settle in the PERIOD AS APPROPRIATE, FOR	0 =	No / Yes	_
1.8	Who has arrived most	?	1 = Men 2 = Women and children 3 = Whole families 4 = Do not know			_
1.9	From where do they come, mainly?		ouring villages 2 = From neighbouring o bital 4 = Returnees 5 = Do not know	cities	i	_

II - HOUSING AND INFRASTRUCTURE These questions to be asked when disasters/crises have affected housing and infrastructure

2.1	How many houses have been completely destroyed? (or ask for the proportion of houses, using proportional piling if appropri	ate)	(or	%)
2.2	How many houses have been partially damaged? (or ask for the proportion of houses, using proportional piling if appropri	ate)	(or	%)
2.3	How many households are living in collective centres (schools, othe	r buildings)?	-	
2.4	How many households are living in camps (tents, plastic sheeting)?			
2.5	How many households are hosted by resident households?			
2.6	Has the electricity network been affected?			_
2.7	Have the water supply facilities been affected?			_
2.8	Have the sanitation facilities been affected?	1 = Not affected 2 = Moderately a	affected	_
2.9	Has the telephone network been affected?	3 = Severely affe 4 = Destroyed	ected	_
2.10	Has school infrastructure been affected?	5 = Not application not exist before)		_
2.11	Has health services infrastructure been affected?			_
2.12	Has market infrastructure been affected?			_

III - SOURCES OF FOOD AND INCOME AND MARKETS

	urces of income for househol			то	0	Current		6 r	nonths ago		
BEFORE THE DISASTER]	AS APPROPRIATE, FOR EXAMPL				3.1	1 st inc	ome:	3.2	1 st income:		
WITH THOSE USED IN TH	S APPROPRIATE – MAKE SURE TI E HOUSEHOLD QUESTIONNAIRE]		LIGNED	3.1		_	3.2	_		
	production 2 = Sale of vegetable 4 = Self-employed (taxi, carpente		ts			2 nd inc			2 nd income:		
5 = Non-agricultural wage la 6 = Government employees	bour (store work, restaurant waiting salary 7 = NGO, private company	, bank, s	ser	vices, etc.)	3.3		_	3.4	_		
8 = Sale of handicrafts 9 = 10 = Petty trade 11 = Pen	Sale of animals/animal products					3 rd inc	omo:		3 rd income:		
12 = Daily or casual worker	13 = Sale of homemade food pro	ducts			3.5			3.6			
14 = Remittances 15 = Ot 99 = No 2 nd or no 3 rd source	her (specify):					_	-1				
	are engaged in the various			Currently			6 r	nonths	ago		
	ee answer to previous question)	3.7		% 1 st in	come	3.8	.	%	o 1 st income		
2nd source of ⁱ ncome (s	see answer to previous question)	3.9	1	% 2 nd in	come	3.10	.	%	2 nd income		
3 rd source of income (se	ee answer to previous question)	3.11		% 3 rd in	come	3.12		%	3 rd income		
[ADJUST RECALL PERIOD EXAMPLE REFER TO BEF				Currently			6 r	nonths	nths ago		
How do households ob	tain most of their food?			1 st sourc	e:			1"	rce:		
[ADJUST CATEGORIES AS THEY ARE ALIGNED WITH	S APPROPRIATE – MAKE SURE	3.13	3	1 1		3.14		I	1		
HOUSEHOLD QUESTIONN	IAIRE]							Ŀ	<u> </u>		
1 = Own crops/garden production 2 = Purchase 3 = Work for food 4 = Borrowing food/debts 5 = Gifts from neighbours/relatives 6 = Food aid 7 = Barter or trade of goods			5	2 ⁿ rce:		3.16		2"	^{id} so		
			5	_		5.10		.	_		
What proportion of hou		Currently			6			6 months ago			
sources of food? 1 st source of food (see a	answer to previous question)	3.17 % 1 st so 3.19 % 2 nd so Currently Purchase at			ource	3.18	I	% 1 st source			
2 nd source of food (see	answer to previous question)				ource 3.20		% 2 nd source				
Where do most househ	olds go to purchase food?				6 r			6 months ago Purchase at:			
1 = Local market within the 2 = Local market in neighbor	village (or city neighbourhood) uring village (or neighbourhood)										
	lage (or city neighbourhood)	3.21		1.1		3.22		I	1.1		
	le at the local (nearest) market	? [SELE	EC.	T UP TO 8	Cu	Currently compared with 6 months ago					
Potatoes					3.23	3	<u>د</u>	agu 	1		
Wheat					3.24	4		- <u>i</u> -			
Sorghum	4 - N				3.25						
Lentils	1 = None or almost none2 = Less than half the usual				3.26			<u> </u>			
Vegetables	 3 = About half the usual 4 = Almost normal or as usual 				3.27			<u> </u>	. <u>.</u> 		
Milk, dairy products					3.28						
Meat					3.29			<u> </u>			
What are the trends for UP TO 8 MAIN FOODS AS	food prices at the local (neare	est) ma	rke	t? [SELECT				ared wi	ith 6 months		
Potatoes					3.30	D					
Wheat					3.31	1					
Sorghum	1 = Doubled or more compared to		hs a	ago	3.32	2					
Lentils	2 = Increased but less than double 3 = No change/almost no change	ed			3.33	3		<u> </u>			
Vegetables	4 = Decreased compared with 6 m	nonths a	igo		3.34	4			.		
Milk, dairy products					3.35						
					3.36						

What are the trends for lives (nearest) market ? [SELECT U					Curre	ntly co	mpared with 6 months ago	
Cattle (specify sex, age, race if appropriate)	1 = Doubled or i	more compared wi	ith 6 mo	nths ago	3.37			
Sheep (specify sex, age, race if appropriate)	3= No change/a	2 = Increased but less than doubled 3= No change/almost no change				_		
Goats (specify sex, age, race if appropriate)	4= Decreased c	ompared with 6 m	onths a	go	3.39	_		
What are the wage levels fo [SELECT UP TO 4 TYPES OF L				Currently	,		6 months ago	
Agricultural casual labour (e.s	g. weeding)					3.41		
Non-agricultural casual labou construction)	ır (e.g.	CURRENCY] per day	3.42			3.43		

IV - CROPS AND LIVESTOCK

	Ton	natoes	0	nions	Pot	atoes	W	heat	So	rghum	Le	entils
What proportion of households usually cultivate? Use categories if easier (1 = None, 2 = Less than half the HHs, 3 = About half, 4 = more than half, 5 = practically all HHs)	4.1	_ (or _)	4.2	(or)	4.3	 (or _)	4.4	_ % (or _)	4.5	(or	4.6	(or
What is the estimated damage to these crops? Instead of % of damage, use categories if easier (1 = No damage, 2 = Less than half crop damaged, 3 = About half, 4 = More than half, 5 = Practically all crop damaged)	4.7	_ % (or _)	4.8	_ % (or _)	4.9	_ % _)	4.10	 %)	4.11	 %)	4.12	(or
What will be the mai [ADJUST CATEGORIE THOSE USED IN THE Loss of land (floods,	S AS A HOUSE	PPROPRI. EHOLD QU	ATE – M JESTIO	MAKE SUR NNAIRE]					4.19	0 = No	/ 1 = Yes	•
Loss of fartilizer	lanusii	ues, occu	ipied b	y outers)					4.19		<u> _ </u>	
Lack of pesticides									4.21		<u> </u>	
Lack of agricultural to	ools								4.22			
Lack of irrigation	00.0								4.23		<u> </u>	
Lack of workforce									4.24			
Insecurity when read	hina fi	elds							4.25			
[SELECT UP TO 4 MA MAKE SURE THEY AF HOUSEHOLD QUEST	IN ANIN RE ALIG	MAL SPEC					Cattle		Sheep a goa		Po	ultry
What proportion of h animals? Use catego of HHs, 3 = About half,	ouseho ries if e	olds usua asier (1 = l	Vone, 2	= Less tha		4.26	 (or	_ % 4.	27	_ % or _)	4.28	% (or)
What are the estima Instead of % of damage = Less than half animal = Practically all animals	e, use c Is lost, 3	ategories i	, f easier	(1 = No lo:	sses, 2	4.29	(or	_ % _ % _) 4.	30 I	% or)	4.31	% (or)
What will be the mai households next se	ason	[ADJUS	T CATE	GORIES A	S APPF	ROPRIATE	E – MAK	E SURE		0 = No	/ 1= Yes	
THEY ARE ALIGNED \		HOSE USE too expe		HE HOUSE	EHOLD	QUESTIO	NNAIR	=]	4.32	1	1 1	

Lack of water	4.33	_
Lack of veterinary services/animal diseases	4.34	_
Lack of animal shelter	4.35	_
Lack of workforce	4.36	_
Lack of market for selling animals or animal products	4.37	_
Insecurity when moving to pastures or migrating with animals	4.38	_

V - HEALTH AND EDUCATION

[ADJUST RECALL PERIOD AS APPROPRIATE, FOR EXAMPLE REFER TO BEFORE THE DISASTER]	С	urrently	6 m	onths ago	
Where do most households go for health treatment? 1 = Health centre within the village (or vity neighbourhood) 2 = Health centre in neighbouring village (or neighbourhood) 3 = Health centre/hospital in nearest district (or city) 4 = Mobile/outrerach clinic 6 = Traditional healer 7 = Home made treatment	5.1	_	5.2	_	
How long does it take to walk to the nearest health centre?	5.3	min	5.4	min	
What are the main diseases affecting the population? 1 = Acute respiratory infections 2 = Malaria 3 = Diarthoea 4 = Skin problems 5 = Other (specify)	5.5	_	5.6	_	
What are currently the main constraints for households to receive health treatment? [ADJUST CATEGORIES AS APPROPRIATE] Do NOT read the categories, leave the respondent to answer spontaneously		0 = No /	1 = Ye	es	
Distance	5.7		_		
Lack of money to pay for drugs and/or treatment	5.8		_		
Lack of drugs within the health centre	5.9				
Lack of doctors and other health agents	5.10		_		
Insecurity when reaching the centre	5.11	_			
[ADJUST RECALL PERIOD AS APPROPRIATE, FOR EXAMPLE REFER TO BEFORE THE DISASTER]	Cı	irrently	6 m	onths ago	
Where do most children go to primary school? 1 = Primary school in the village (or city neighbourhood) 2 = Primary school in neighbouring village (or neighbourhood)	5.12	_	5.13	_	
How long does it take to walk to the nearest primary school?	5.14	_ min	5.15	_ min	
Currently, what are the main constraints for households sending their girls and their boys to primary school? IADJUST CATEGORIES AS		0= No /	1= Ve		
APPROPRIATE] Do <u>NOT</u> read the categories, leave the respondent to answer spontaneously		Boys		Girls	
	5.40	1 1	5.17		
Distance	5.16				
Distance Lack of money for clothing, uniform, textbooks etc.	5.16		5.19		
			5.19 5.21		
Lack of money for clothing, uniform, textbooks etc.	5.18				
Lack of money for clothing, uniform, textbooks etc. Lack of teachers	5.18 5.20		5.21		
Lack of money for clothing, uniform, textbooks etc. Lack of teachers Poor school facilities	5.18 5.20 5.22		5.21 5.23		

7 = Human epidemics 8 = Lack of health care 1st problem 9 = Electricity cuts 10 = Animal disease 11 = Locust infestation 1st problem 12 = Violence, insecurity 13 = Arrival of displaced people, or refugees 1st problem 1st problem 14 = Increased difficulties with migrating and receiving remittances 1st problem 1st problem	6.2	 2 nd proble	 6.3	3 3 rd problem	
Do you think that the problems of people in the village/neighbourhood are different from last year at the same period?	6.4			or 2, go to .6	
1 = Fewer problems 2 = Same as last year 3 = More problems than last year What would you say is the main reason for the worsening of the situation? 1 = Price rises 2 = DroughVirregular rains 3 = Floods 4 = Other climatic event (snow, hail, winds, frosts) 5 = Landslides/erosion 6 = Crop pests and diseases 7 = Livestock diseases 8 = Human disease/epidemics 9 = Insecurity/violence 10 = Restricted access to markets 11 = Unemployment 12 = Other reason (specify)	6.5		L]	
When was the last time that a crisis comparable to this one occurred? Write 9999 if cannot answer	6.6	L	_ _	_ year	
Which population groups face the most problems with access to food and income? Do <u>NOT</u> read the categories, leave the respondents to answer spontaneously [ADJUST CATEGORIES AS APPROPRIATE]		0=	No / 1 =	Yes	
Large families	6.7		Ŀ	_	
The elderly, pensioner	6.8		 		
Households with disabled members	6.9				
Households headed by a woman	6.10)			
Households with orphans	6.11				
Households with no land	6.12				
Households with no animals	6.13			_! _!	
Households with no migrants sending remittances, or no migrants at all	6.14				
At present, what are the priority requirements for the majority of households? Do <u>NOT</u> read the categories, leave the key informants to answer spontaneously Once done, ask the key informants to rank the 3 most important ones [ADJUST CATEGORIES AS APPROPRIATE – MAKE SURE THEY ARE ALIGNED WITH THOSE SURVEY (triangulation)] 1 = School feeding	EUSEI	D IN THI	E HOUS	EHOLD	
2 = Food for young/malnourished children or other vulnerable groups 3 = Food for households in general 4 = Cash transfers 5 = Credit 6 = Employment 7 = Increased wages, increased pensions 8 = Seeds 9 = Fertilizers 10 = Pesticides 11 = Agricultural tools 12 = Fodder, animal feed 14 = Veterinary services 15 = Irrigation 16 = Water for domestic use 17 = Health services (infrastructure, staff), treatment (subsidies, free drugs) 18 = Clothes, shoes 19 = Roads 20 = Public transportation 21 = School infrastructure/teachers 22 = Housing/shelter (repairs, new) 23 = Fuel (subsidies, domations) 24 = Security 25 = Other assistance (specify)	pr	 2 nd iority	6.17	 3 rd priority	
Is this assistance currently provided in the [ADJUS' villagelneighbourhood? OF No SURE IT [ADJUST THE LIST AS APPROPRIATE – MAKE SURE IT IS SIMILAR TO THE ONE USED IN THE HOUSEHOLD HOUSEN SURVEY (triangulation)] (triangulation)	PRIATI IS SIN E USE HOLD	e – Mak Vilar T D in Th	Е О IE	0 = No 1 = Yes	
6.18 Food for schoolchildren (eaten at school or 6.19 (including p			with		

6.20	Food for young/malnourished children or for pregnant/lactating women	_	6.21	Free agricultural tools	_
6.22	Free food ration for the household		6.23	Free fodder, animal feed	
6.24	Food for work		6.25	Free veterinary services	
6.26	Cash for work	_	6.27	Plastic sheeting, tent, other housing material	_
6.28	Cash transfers (grants, child subsidies, family subsidies etc.) from NGOs, other agencies or government social assistance programmes	_	6.29	Cooking utensils, stove, kitchen set	_
6.30	Free health care/drugs		6.31	Clothes, blankets	
6.32	Micro-credit		6.33	Mosquito nets	1_1
6.34	Free seeds, fertilizer		6.35	Cooking utensils	1_1
6.36	Free agricultural tools		6.37	Jerry cans	1_1
6.38	Free fodder, animal feed	_	6.39	Other assistance (specify)	

Annex 3: Sample checklist for focus group discussions

Questionnaire number: _ _ _	
City/village name:	_City/village code: _ Region code: _
Neighbourhood/street name : [INCLUDED IN URBAN ASSESSMENTS ONLY]	Neighbourhood/street code: _ _ _
Cluster code: _ _ HH questionnaire numbe	r: _ _ _
Date: _ _ / _ _ / _ _ _ _ Tear <i>day month year</i>	n number: _ _
Enumerator names :	1
Instructions to facilitators:	
 Ask the village/neighbourhood leader to help you relatively private area 	gather from 5 to no more than 12 people in
relatively private area. In each village, hold one discussion with only mer men and women if acceptable. [DECIDE THE AF BEFORE THE SURVEY STARTS, TAKING INTO A Each discussion should not last for more than 2 hou	PROPRIATE COMPOSITION OF THE GROUPS
Selection of the participants should be guided by people in the group should represent the average I neither the poorest of the poor, nor the most educa PROFILE OF PARTICIPANTS BEFORE THE SU SURVEY OBJECTIVES].	the village/neighbourhood. As far as possible, the nouseholds living in the village/neighbourhood (i.e ated, well-off and influential people). [DECIDE TH
 Local authorities and leaders should be asked no interviews will be conducted with them (use the key 	
 Inform participants that the aim of the discussion is living in the area and how they handle the various important for understanding what the main constra- these, and what remains problematic. 	difficulties they may be facing. The outputs will b
 Be clear that participation in the meeting does not r to this particular village/neighbourhood, nor that pa results will be communicated to higher-level people 	rticipants will be targeted for support. However, th
 As soon as discussion of the topic has finished, fa standard templates (included). 	cilitators should take time to summarize it into th
 If appropriate (check with village/neighbourhood However, emphasize that during the discussion th should reflect the situation of the majority of housel 	ey should not speak only on their own behalf, bu
 Start with two or three general questions to put peo 	ple at ease, for example:
l will introduce myself and then I would like each of yo /neighbourhood and whether you have participated in	

	What do most people in the village/nei most of their incomes? [As much as possible, align the answers household and/or key informant quest/onnaires]	What do most people in the village/neighbourhood do to obtain most of their incomes? <i>As much as</i> possible, <i>align</i> the answers with categories used in the household and/or key informant questionnaires]		
	Answers may include: all of a start of more of production (staples) - Sale of own food production (staples) - Sale of animals - Sale of animals products (milk, cheese, meat, ' - Casuel labour (frregular) - Regular employment (salared e worke) in agin - Regular employment (salared e employee) in a - Government work (teacher, health agent, e) in - Government work (teacher, health agent, e) in - Remitances from family members elsewhere - Pensions, etc.	Answers may include: • Sale of cash crops (cotton, tobacco, coffee, horiculture, etc.) • Sale of cash crops (cotton, tobacco, coffee, horiculture, etc.) • Sale of animals pound (milk, cheese, meat, eggs, etc.) • Casual labour (irregular) • Regular employment (salaried worker) in a private company • Regular employment (salaried worker) in a private company • Comment work (leacher, health agent, acc.) • Reministraces from family members elsewhere in the country or abroad • Prensions. etc.		
	Are there populatio sources or are earn [ADJUST THE RECALL REFER TO THE BEGINN	Are there population groups which have lost their income sources or are earning less than they did 6 months ago? (ADUUST THE RECALL PERIOD AS APPROPRIATE, FOR EXAMPLE REFER TO THE BEGINNING OF THE CRISIS]	Groups whose income decreased	Main reasons
	If yes, which groups, and why?	and why?		
	Population groups may be:	Reasons may be:		
1.2	Farmers	Loss of land, increased cost of inputs, insecurity, etc.		
	Cattle owners	Loss of animals, animal diseases, lack of water, etc.		
:	 Pensioners 	Pensions paid irregularly, too low, etc.		
4	 Agricultural labourers 	No jobs, decreased wages, etc.		
	 Teachers 	Low salaries, school closed etc.		
i	 Health staff. etc. 	Low salaries. health centre destroved. etc.		

ō	
E	
Ъ	
R	
s	
N	
5	
Ō	
ō	
0	
7	
=	

	How do most peopl obtain their food? [As much as possible, household and/or key in	How do most people in the village/neighbourhood currently babit heir food? Tood? As much as possible, aign the answers with categories used in the household and/or key informant questionnaires]		
	Answers may include: Consumption of their own stocks (staple) Purchase at market (payment) Purchase at market or shops on credit Work for food	Answers may include: • Consumption of their own stocks (staples, fruits) and animal products • Curchase at market (payment) • Purchase at market or shops on credit • Work for food		
	 From relatives, friends or neighbours From humanitarian assistance From government social programme Scavenging, collecting wild food, hunn Begging, etc. 	From relatives, itends or neighbours From Mumminatian assistance From government social programme Begging, eits.		
	Are there population groups the procuring their food? If ves. which groups, and why?	Are there population groups that currently face difficulties procuring their food? If ves. which groups, and whv?	Groups with difficulties procuring food	Main reasons
	Population groups may be:	Reasons may include:		
-	 Farmers 	Harvest failed, no food stocks, no sales so no income to purchase food, etc.		
c	Cattle owners	Animal death, no sales so no income to purchase food, etc.		
	 Agricultural labourers 	No jobs, so no income to purchase food; market not accessible owing to insecurity, etc.		
	 Casual workers 	Insufficient income to purchase food		
	Pensioners	Pension not paid, so no money to purchase food, etc.		
	Teachers	Salary has not kept pace with price increases, job loss so no income to purchase food, etc.		
	 Displaced people 	Arrived too late to benefit from last food distribution, etc.		

 How do affected population groups cope with (respond to) their income and food difficulties? A much as possible, align answers with categories used in the household and/or key informant questionmains 		More severe		
2) Which of these behaviours are considered different from previous periods of difficulty?	Behaviours for coping with income and food difficulties	 Same as before Less severe 		
Answers may include:				
 Increase migration, more than usual 				
Enter into debts				
Sell animals				
 Sell personal belongings (furniture, household items, jewellery) 				
Use savings				
Decrease food consumption				
 Decrease expenditures for agricultural inputs 				
 Decrease expenditures for clothing 				
 Take children out of school 				
 Ask help from relatives 				
• Beg				
 Ask support from humanitarian institutions 				
 Move away (if this is mentioned, identify where to), etc. 				
When do you think that the population groups who have income or food difficulties will recuperate their pre-shock	Groups with income and food consumption difficulties (see 1.2 and 2.2)	Time to recover		
income or food consumption level?				
Answers should be specific to each population group identified in 1.2				
and 2.2, and may include: • In the next 3 months				
In the next 6 months Mhan the hervest comes in XXX months				
 vuren ure narvest corries, in XXX montris At least a year 				
 May not recover, etc. 				
	receive? As much as possible, align answers with categories used in the household and/or key informating questionnaires and/or key informating questionnaires Support from relatives, friends, neighbours Support from relatives, friends, neighbours Covernment cashYsocial assistance (allowances, monthly benefits, etc.) Government cashYsocial assistance (allowances, monthly benefits, etc.) Government cashYsocial assistance Government food assistance Covernment food assistance assistance from humanitarian agencies/NGOs Coat assistance from humanitarian agencies/NGOs Food vouchers from humanitarian agencies/NGOs Food vouchers from humanitarian agencies/NGOs Food for realhourished children Food for real food food food food food food food foo			
-----	--	--	------------------------	------------
	 Elc. What else would these population groups need to recover their pre-shock income and food consumption levels? 	Groups with income and food consumption difficulties (see 1.2 and 2.2)	Additional required	assistance
	As much as possible, align answers with categories used in the household , and/or key informant questionnaires			
	Answers should be specific to each of the population groups identified in 1.2 and 22, and may include: Free food - Cash transfers - Increased persions, allowances - Sobs, employment - Better salaries - Better salaries - End			
	 Agroutural moute Equipment (fishing gear, farm machinery, sewing machine, etc.) House reconstruction Assistance with free health care, drugs etc. Credit Etc. 			
- 1	Are there population groups that do not receive any Grou assistance, and whv?	Groups with difficulties and NOT receiving assistance	Main reasons	

	Displaced people	Arrived too late to be registered			
	Women-headed households	Do not know where to go, have no time to queue at distribution points, etc.			
	Landless labourers	Have no land, so do not benefit from agricultural inputs, and no other programme			
	Disabled, elderly living alone	exists No one to assist them with obtaining assistance			
	MAIN DIFFICL	IV – <u>MAIN DIFFICULTIES AND PRIORITIES</u>			
+	What have beevilated willage/neighbo	What have been the main shocks and difficulties faced by the ovillage/neighbourhouch inhabitants in the past year? [ADUUST			
	As much as poss	As much as possible, align answers with categories used in the household			
	and/or key inform Answers may in	and/or key informant questionnaires Answers may include, for example:			
	Drought	•			
	 Floods, landslides 	des			
	Cold temperatures	res			
	High fuel prices	0 0			
	 Epidemics Lack of health are 	are			
	 Electricity cuts 				
	 Animal disease Locust infestation 	100			
	 Violence, insecurity 	urity			
	 Arrival of displa Increased diffic 	Arrival of displaced people, refugees Increased difficulties with migrating and receiving remittances			
	 Loss of employment Etc. 	ment			
	Are the curren	Are the current shock and difficulties different from in		Curront difficultion	More severe Same as hefere
	Previous crises r Specify for each of	previous crises r Specify for each current difficulty if there are several		diffectives	Less severe
	simultaneously	simultaneously drought + displacement; high food prices +			
	hurricane; etc.)				
-	If more severe crisis was as s	If more severe than previous crises, when was the last time a crisis was as severe as this one?	Write the years mentioned		
-	What are th	are the priorities for the village/neighbourhood	1 st priority	2 nd priority	3 rd priority

inhabitants to prevent future crises? Write answers in the blank spaces in the table and ask participants to prioritize 3 of them As much as possible, align answers with categories used in the household	Anisares may include: Repair of roads, bridges • Village food stores	Credit School to improve education Construct in the construction	Provision of yours Provision of yours Repeat of Water sources	
---	--	--	---	--

Annex 4: Main advantages and disadvantages of selected response options

Type of response	Advantages	Disadvantages
Addressing probl	ems of inadequate food access for s	ome or all households
Food transfers: free food distribution, food for work (FFW)	 Food addresses consumption and nutrition deficiencies directly. It is likely to be consumed because it is less easily converted than cash. At the household level, women usually manage food. Food transfers are effective and necessary where markets are unable to meet all the needs of the population. 	 Food distribution is often costly because of high overhead costs for transport, storage and administration. Food is bulky and difficult for beneficiaries to transport; when distribution points are not close to their homes, beneficiaries may incur significant transport costs. A lengthy supply chain before the food arrives at a distribution point may lead to delays in distribution point may lead to delays in distributing the food. If food distribution is large scale or continues for a long time, and local farmers are trying to sell their own production, it can distort the local market and economy, reduce local producers' income and be a disincentive to future local food production. Available food aid commodities may not meet all the nutrient requirements of populations or satisfy local preferences and tastes. Food may be used as a weapon in conflict situations, and warring factions may impose administrative obstacles.
Cash transfers: cash distribution, cash for work (CFW)	 Cash allows households to define their own spending priorities. Where some foods are available at the market, cash can complement food distribution by enabling beneficiaries to make up for a lack of variety in food rations. Cash transfers can help people recover livelihoods and support survival, as part of a mixed recovery package. Cash injections can stimulate the economy and encourage productivity. Cash is more cost-efficient than other response options, as distribution costs are minimal. 	 Cash cannot be used in a barter economy, only in a cash economy. Targeting may be difficult, as cash is attractive to everyone. Cash injections can lead to inflation, unless market traders are ready and able to increase supplies immediately. Women are normally responsible for feeding the family, but may not have control of the money and how it is spent. Donors may be less willing to provide cash than food for distribution, especially when large quantities are required.
Public works projects: FFW, CFW	 Works projects provide employment to the unemployed, so have important psychological as well as economic benefits. They can help improve food 	 Large-scale works projects are difficult to organize quickly under acute crisis conditions – administrative demands are often unmanageable in an emergency context. They are best suited to slow-onset crises, the

Annex 4 - Main advantages and disadvantages of selected response options

[|] The costs involved in transporting, storing and handling food commodities from the point of origin to the distribution site are often several times the purchase cost of the commodities, especially when long distances and difficult logistics are involved.

Annex 4 – Main advantages and disadvantages of selected response options

	 access and availability for the whole community by creating a communal asset via FFW or CFW, or the conditions for recovery via food for recovery (FFR). Some works projects are self-targeting – only those in real need participate. 	 recovery phase after sudden natural disasters, and protracted complex emergencies, if security conditions permit. Works projects do not help households where there are no able-bodied people, which are often the poorest, or where all able-bodied people are fully occupied, for example, preparing land and planting for the next harvest. They may increase the burden on women, if they have to undertake the work.
Supporting livelih	oods	
Livelihood support projects	 Livelihood support projects protect long-term food security and promote recovery and self-reliance. They build on existing resources at the household and community levels. They address some of the root causes of food insecurity and poverty, thereby acting as preventive measures against future food insecurity. 	 Livelihood support projects may not address the needs of destitute households that have no capital assets and/or are not economically active. They may not be appropriate in situations where people are cut off from their livelihood base, although training to develop skills for future livelihoods may be appropriate in such cases. They can be difficult to implement in situations of conflict or political crises, but this is not impossible. They require careful planning, good management and often a small-scale pilot phase to start with. They may be difficult to implement on a large scale.
Addressing nutrit	ion problems or problems of inadequ	uate food access for certain individuals
School feeding	 If children/youths are not getting enough food at home, the food they receive at school increases the quality and quantity of their food intake, provided it is not matched by an equal decline in home food intake. If the food is fortified, it provides children with micronutrient boosts. School meals can reduce hunger and help children to concentrate at school. School feeding can maintain and increase school attendance, and reduce and prevent drop-out, especially during crisis periods. This improves schools' educational impact and effectiveness as a channel for health-related measures such as de-worming. 	 Exclusion errors may occur: school feeding benefits only those children who attend school. Some – sometimes many – of the children from the poorest and most food-insecure households do not attend school, so cannot benefit from a school feeding programme. Inclusion errors may occur: school feeding targets all the children attending a school. Some children from food-secure households may therefore receive support.

** A Critical Review of Approaches to Assessing and Monitoring Livelihoods in Situations of Chronic Conflict and Political Instability (SCCPI), by S. Jaspars and J. Shoham, Oversees Development Institute (ODI) Working Paper No. 191, 2002. London, ODI.

Annex 4 – Main advantages and disadvantages of selected response options

	 School feeding and its impact on school attendance and learning can foster a sense of normality within the community. In displaced communities, school feeding can target school-age children whose basic nutrition needs are not met by the general ration. 	
On-site supplementary feeding	 Staff observe food being consumed and have more regular contact with mothers/carers so can support recovery more effectively. 	 On-site feeding incurs an opportunity cost, in terms of time for carers. It is more costly for the implementing agency. There is the risk of spreading infections in crowded feeding centres. Beneficiaries may receive less food at home as a result of receiving the supplement.
Take-home supplementary feeding	 There is less opportunity cost in terms of time for mothers/carers than in on-site feeding. It is less costly for the implementing agency, and more rapid to implement. 	 The food is likely to be shared within the household; extra food has to be provided to allow for this. There is a possible security risk.

Annex 5: Main response options and information required to judge their appropriateness

Response option	When it may be appropriate	Information required to establish appropriateness
	In a situation of acute/transitory food insecurity where:	 Previous and current nutrition status of population; previous and current morbidity patterns; previous and current mortality rates, both crude and for children under 5. Presence or absence of structural food aid.
	 all, or a significant proportion, of households lack access to food; and 	 Current levels of household access to food: quantity/type of food obtained through normal mechanisms; current consumption levels; sustainability of households' coping stratedies.
Free general or targeted food distribution:	 there is a lack of food availability; and 	 Current and potential availability of food: whether there is a food deficit at the local, sub-national and/or national level(s): reasons for deficit(s), trends and elasticity of food supply over the year, and factors affecting food supply tends; presence of alternative sources of food supply: extent to which a deficit could be filled by local actors such as government and/or commercial traders, and the possible role of humanitarian actors in supporting the government or the private sector.
provides households with free rations of food (see	 alternative ways of assisting people's access to food would be unpractical or unreliable and/or take too long, because the situation is urgent; and/or 	 Practicality of implementing alternatives to a general distribution: whether infrastructure and conditions are in place to implement a non-food aid project rapidly; whether partners, equipment and technical inputs are available, and the security strated on project and the security
following sections on ration types)	 the characteristics of the affected population mean that many target households would not be able to participate in an employment- based project, i.e., they are mainly women, children, the eldenty and/or the sick; <i>and</i> 	 Demographics of affected population. Health status of affected population.
	 the population has no time to engage in other activities; people are not underemployed, in that they are engaged in their usual livelihood activities such as preparing fields and planting for the next harvest; and their is no surplus labour. 	 Daily, seasonal calendars of activities engaged in by men, women and children in the affected population.
	In addition, food distributions may be appropriate for short periods, of one to two possible hunger without knowing whether the above conditions have been met.	In addition, food distributions may be appropriate for short periods, of one to two weeks for example, when there is reason to fear possible hunger without knowing whether the above conditions have been met.

Activity structure. Security structure. Security structure. The target population's food habits, commodity preferences and acceptable substitutes, and familiarity with/acceptance of the commodities available for distribution. Households' access to fuel for preparing the proposed foods, and their ability to store food. Micronutrient requirements, to decide on the need for food forfication. Potential distribution points and an appropriate distribution system. Media and community networks available to publicize ration entitlements and distribution arrangements. Needs of host population, when distributions are planned for displaced people. Needs of the trans to decide a community networks are planned for displaced people. Media Micronutient or the propriate Media appropriate of the displaced people. Media Men it may be appropriate	Current levels of household access to food: and throuseholds lack access to food; and current consumption levels;	food availability in the area is	imited in quantity an <i>a/or</i> variety, and there is no indication that this will change; <i>and</i>	•	aver-course perçus avio are . Composition of food-insecure households: average number of able-bodied people per household; percentage of lac, there is surplus abour in Seasonal labour patterns, periods when no work is available.	•	public works projects are
ppulation access to requiren fibution p mmunity acity – st – roads, t populati	In situati househc	• fooc limit	will	 foot abl€ 	une i.e., taro	targ	• pub

	 the assets created will be properly managed and mantained after completion of the project, or 	 Presence of community and/or government arrangements to ensure ongoing management and maintenance.
	 following a sudden disaster when. there is need for debris removal and general clean-up operations, labour-intensive repair of rural roads, small embankments or other public infrastructure; and the population has the capacity to undertake the required work without outside technical supervision. 	 Nature and extent of physical damage, and need for initial, labour-intensive clean-up and repair action. Availability of able-bodied people in food-insecure households to participate in short-term community rehabilitation activities.
Additional infol • Number • Availabil	 Additional information required for designing a FFW project: Number of target beneficiaries/workers. Availability of public works benefiting the community and long-term food security. 	/ project: unity and long-term food security.
When w Location	When works can/should be undertaken, seasonal considerations. Locations of possible public works sites a deministrative and mon	When works can/should be undertaken, seasonal considerations. Locations of possible public works stes.
Security	oceperates or participations, equipments, administrative and monitoring systems - to Gender roles and their implications for narticipation in public works at particular times.	istrative and internet gasteries - to plan and important the activities.
 Local data Attractiv 	Local daily wage rates and the transfer value of c Attractiveness of the available food commodities.	Local daily wage rates and the transfer value of commodities, to determine the FFW wage rate in food. Attractiveness of the available food commodities.
 Media a Infrastru Ownersi Possible 	Media and community networks available to publicize the project. Infrastructure - roads, storage capacity, milling facilities - to hanc Ownership of the asset(s) created. Possible environmental impact of the public works projects.	Media and community networks available to publicize the project. Infrastructure – roads storage capacity, milling facilities – to handle the delivery of food commodities and materials for public works activities. Powneship of the asset(s) created. Possible environmental impact of the public works projects.
Where there are vithin the host p	Where there are large numbers of displaced people, employment opportunities should within the host population; activities should not be at the expense of the local population.	Where there are large numbers of displaced people, employment opportunities should be made available to both displaced people and food-insecure groups within the host population; activities should not be at the expense of the local population.
Additional infor Number Locatior Capaciti Local da	 Additional information required to design a FFR project: Number of target beneficiaries/workers. Locations of possible public works sites. Capacities of partners to plan and manage distributions. Local daly wage rates and transfer value of commodifie. 	al information required to design a FFR project: Number of target beneficiaries/workers. Locations of possible upolic works sites. Copacities of partners to plan and manage distributions.

Nature and seasonality of the income-generating activities of able-bodied household members actors such as government and/or commercial traders, and the possible role of humanitarian State of the economy, risk of inflation if cash is injected; market prices and integration of local Current conditions and requirements for rehabilitation/reconstruction of infrastructure, physical Capacities of partners to plan and manage projects – availability of necessary material inputs, potential alternative sources of food supply: extent to which a deficit could be filled by local - whether there is a food deficit at the local, sub-national and/or national level(s); reasons for deficit, trends and elasticity of food supply over the year, and factors affecting food supply Availability of labour-intensive projects that would contribute to long-term food security and Composition of food-insecure households: number and gender of able-bodied people per Local daily wage rates and cost of living, including medical and funeral costs where the impact of HIV/AIDS is high, to set CFW wage rate. household; percentage of households with no able-bodied member available to work. Information required to establish appropriateness quantity/type of food obtained through normal mechanisms; actors in supporting the government or the private sector. sustainability of households' coping strategies Current levels of household access to food: services for food-insecure households. current consumption levels; markets within a wider system. equipment and technical staff. Current availability of food: assets or the environment. trends; Additional information required for designing a CFW project: • . • • . public works projects are required; and a depressed economy needs a cash the risk of inflationary pressure is low the necessary non-food inputs, such households lack access to food; and unemployed or underemployed, i.e. food-insecure households include food is available in the area; and When it may be appropriate there is surplus labour in target -ocation of public works to be undertaken. able-bodied people who are supervision can be assured. as equipment and technical Number of target beneficiaries/workers. Timing and duration of activities. households; and In situations where: injection; and Timing of payment. Security situation. • • • • • Gender roles. . Response option Cash for work (CFW)

Media and community networks available to publicize the project.

Likely duration of project.

Response option	When it may be appropriate	Information required to establish appropriateness
	In situations where: • food is available in local markets but households lack mans of purchasing it without depleting essential assets; or	 Market prices of the usual staple and other, less preferred staples. Price trends. Income and purchasing power of food-insecure households. Presence of vulnerable groups who do not participate in economic activities.
Cach transfor	 the costs of procuring and transporting food to the affected area are high, but traders would respond to market demand: or 	 Costs of delivering and distributing food aid. Competitiveness and integration of markets. Ability of traders to respond to increased demand.
programmes: cash distributed to	 mobilizing food aid would take a long time; or 	Lead times for the delivery of food aid.
target beneficiaries	•	Prospects for economic recovery without intervention.
	 there is little risk of inflation due to an injection of cash; and 	Risk of inflation.
	 capacity is available for managing the programme; and 	 Administrative capacity to implement programme, such as the banking system. Capacity for monitoring and accounting.
	 donors are willing to support a cash distribution programme. 	The policies of potential donors.
	In situations similar to those for cash transfer programmes but where:	
Food vouchers: vouchers		The policies of potential donors.
alstributed to target beneficiaries	•	The availability of retailers willing to cooperate.

Location and nature of logistics bottlenecks that inhibit the movement of food from other parts The credit-worthiness of the traders and the existence of a mechanism – or the possibility for The feasibility of repairing/improving infrastructure to enable greater quantities of food to be The constraints, other than logistics, that prevent traders who normally bring food into the Experience of cash transfer programmes is relatively limited, so there are a number of unknowns regarding this type of programme, including: establishing one – that makes credit available to traders and assures repayment. Information required to establish appropriateness Problems of market integration affecting specific areas and produce. the necessary level of purchasing power, and the distance from the supply that ensures an inflow of food or other items; Options for assistance directly targeting food-insecure households area from doing so, or from increasing the quantities that they bring in. Market prices of the usual staple and other, less preferred staples. Terms of trade of livestock against grain and other essential items Price differential between main staple and low-cost staple. Numbers of retailers willing to participate in the scheme. Existence of milling and storage capacity Capacity for fortifying low-cost staple of the country into the affected area. Prices of livestock, and price trends. Poor farmers' on-farm seed stocks Price trends – rate of inflation Poor farmers' access to food. Livestock death rates moved into the area. . • • 0 0 . • . • . • • • • . retailers are interested in participating in the price of the usual staple is no longer traders who would otherwise bring food rural producers of cash crops are unable to feasible, especially in urban areas; and into the area are unable to do so owing bringing sufficient food into the area; or drought has led to deaths among cattle. marketing systems have collapsed, so affordable to many, owing to shortage how prices behave following an injection of cash; farmers face an acute food shortage and would eat their seed stocks;^{*} or there is need to revitalize the milling a dramatic fall in the prices paid for to lack of finance (credit) or lack of infrastructure prevent traders from When it may be appropriate nardship among pastoralists;" or damaged roads, bridges or other ivestock and consequent acute targeted general rations are not access to fuel and spare parts. sell their produce and buy food.[®] and resulting inflation; and issues related to cash transfer programmes the scheme; or In situations where: In situations where: In situations where: sector . • 0 0 lower-status food is Response option Exchange against exchange for their Market assistance made available to made available to retailers to sell at which there is no affordable prices support: logistics reduced or credit own produce for bottlenecks are Market svstem households in affected rural programme:" food given to ocal market produce raders •

- In this kind of situation, grain may be given in exchange for animals destocking and the meat used in other programme activities.
 - In this kind of situation, grain may be given in exchange for seed, which is stored and distributed in time for the next planting.
 - æ
- See MAPP-IMalawi case study on the CD-ROM for a brief description of such a programme, implemented by CARE, and supported by USAID. For example, in Mozambique in the 1980s, grain was given in exchange for cashew nuts. 2

the revert of cash introver at which initiation becomes internation, how beneficiaries of different gender, economic status, etc. si security risks in situations of conflict and political instability.	 Issues related to exchange against produce programmes Rates of exchange should normally be fixed, taking in The produce received can be used in local relief profucks that bring relief food into the area. 	<u>Non-food support to livelihood activities</u>	Response option When it may be appropriate	Non-food support . targeted households have access to the institution the natural resource base; and	•	•	the capital assets of cond-insecure but cood-insecure but economically active • this lack limits, or will limit, production. individuals and households	 Additional information required for designing a livelihood support project: Number and geographical spread of beneficiaries. Likely duration of project. Security situation. Gender roles. Sessonit calendar of livelihood activities. Timing of distribution of project inputs. 	Media and community networks available to publicize project. Partners' capacity - staff, equipment, administrative and monitoring systems. Environmental limplications of the project. Access to functioning markets and banks, particularly for micro-finance projects	Productive inputs may include seeds, tools, irrigation pasture, and financial services, such as emergency la nouseholds.
ue revo do can innova a windrimizatori ne occuries incertoscie. Towo beneficiaries of different gender, economic status, etc. spend cash in different circumstances – emergency/non-emergency; security risks in situations of conflict and political instability.	clated to exchange against produce programmes Rates of exchange should normally be fixed, taking into account the relative prices on local markets both before the emergency and at present. The produce received can be used in local relief programmes or be transported to urban markets where it can be sold. It may be back-loaded on the trucks that bring relief food into the area.		Information required to establish appropriateness	 Household access to the natural resource base, ownership patterns. 	Households' stocks of productive inputs, in normal times and now. Households' ability to buy productive inputs – expenditure patterns, assets, access to credit.	•	 Production levels, normal and at present. Factors affecting production levels: access to the natural resource base, labour availability, security situation and/or access to and availability of productive inputs. 	ood support project:	project. nd monitoring systems. for micro-finance projects.	Productive inputs may include seeds, tools, irrigation, fodder or other livestock inputs. Services may include veterinary care, extension, improved access to pasture, and financial services, such as emergency loans for rebuilding/restoring the capital assets of food-insecure but economically active individuals and households.

Response option		When it may be appropriate		Information required to establish appropriateness
	s •	In situations where: • school attendance is low and a school meal would encourage attendance, among girls as well as boys; or	••	School enrolment and attendance rates for girls and boys. Opportunity cost to households for sending children to school, and extent to which food can be an incentive to sending children to school regularly.
	•	educational performance is low because children attending school are hungy, do not get enough food at home and/or have a diet lacking in variety and low in essential micronutrients; or	• •	Children's food consumption – the access to food of schoolchildren's families, patterns of intra-household distribution and diet diversity. Educational performance and the reasons for poor performance, as reported by trachers.
School feeding: provision of a	•	regular attendance at school could help children to overcome psycho-social trauma or reduce their exposure to abuse and expolration, and school feding could encourage and enable such regular attendance, <i>and</i>	•••	The degree of stress children have suffered and the extent of psycho-social trauma, as reported by social workers and trained teachers. The risks of abuse and exploitation of children.
nutritionally balanced meal or snack to	••	schools have the basic facilities to store and prepare food, and school teachers and parents' committees are willing and have the capacity to organize the preparation and distribution of the food, and	•	Schools' capacity to store and prepare food: availability of staff, cooking fuel, cooking utensils and storage space, access to clean water, and good sanitation.
cnildren/youths at school; provision of take-home rations to children/youths		If there is a general ration distribution programme, school feeding is planned as an integral part of the overall food assistance strategy.	••	The overall food security assistance strategy. Nutrition requirements of school-age children.
at school	ë 4.	Preschool feeding may be appropriate in situations where: • preschool-age children have particular nutritional needs that are not met by a standard general ration. <i>and</i> • preschool-age children attend preschool.	••	Nutrition requirements of preschool-age children. Schools' capacity to store and prepare food: availability of staff, cooking fuel, cooking utensils and storage space, access to clean water, and good sanitation.
	• • • old	Exceptionally, school feeding may be appropriate in situations where: • older children/youths have particular nutrition needs that are not met by a standard gener faction; • older children/youths attend school.	• ••	Nutrition requirements of older children/youths, for their height and level of activity. Existence of schools. Profile of school pupils.

- The number of safe learning spaces.
- The numbers of functional and non-functional schools in the area.
 - The geographical and social coverage of schools. Number of children to be fed.
- Likely duration of the intervention.
- Food preferences and the availability of acceptable foods that can be prepared at school.
 - Gender roles and girls' specific needs.
- Environmental impact, such as of the type of stoves used. Communities' willingness to participate.
 - Security situation.
- Infrastructure roads, storage capacity, milling facilities, water and sanitation provision.

Response option		When it may be appropriate		Information required to establish appropriateness
Food to other social service institutions: such as orphanages; homes for the elderly or	E•••	 In situations where: social service institutions do not have the resources to social service institutions do not have the resources to assure adequate food for in-patients or residents; <i>and</i> residents families do not have access to enough food to feed them; <i>and</i> 	• • •	Quantity and quality of food available in each institution relative to number of residents, diet of residents. Nutritition and health status of residents. Current levels of food access for residents families: • quantity/type of food obtained through normal mechanisms; • sustainability of families coping strategies.
nanucappeu, nospitals and health centres	•	staff and the facilities to store and prepare food are assured at each institution.	•	Each institution's kitchen facilities, storage space, cooking utensils, and staff available to prepare and cook food.
	'n	In situations where:		
Neighbourhood care	•	there are large numbers of orphans and vulnerable		
programmes (NCPs): ^B		children owing to the HIV pandemic; and	•	Numbers of orphans and vulnerable children in the community.
for unaccompanied	•	communities are under strain to provide care for these	•	Evidence of higher rates of malnutrition among these children.
children, orphans and		children, and many preschoolers have little food provision;	•	Evidence of higher rates of malnutrition among households supporting
where there is high	•	NCPs will afford protection to this vulnerable age cohort.	•	Evidence of sexual risk among this age cohort.
prevalence of HIV/AIDS		and will help strengthen village-level food security through agricultural schemes and orphan and vulnerable children		

NCP and orphans and other vulnerable children programmes have been developed in the context of PRROS in the Southern Africa region.

C;

the household? Are certain foods reserved for children? For women? the household? Are certain foods reserved for children? For women? Partners' capacity to establish and manage supplementary feeding below mean energy requirements, crude mortality rate > 1 per 10 Patterns of intra-household distribution: How is food shared within below mean energy requirements, crude mortality rate > 1 per 10 Patterns of intra-household distribution: How is food shared within Targeted households' access to cooking fuel and cooking utensils. 000 per day, an epidemic of measles or whooping cough, or high specifically, percentage of children with < -2 SD or below 80% of specifically, percentage of children with < -2 SD or below 80% of Presence of other aggravating factors, including: a general ration Presence of other aggravating factors including: a general ration Capacity of partners to screen/select malnourished individuals. Capacity of partners to screen/select malnourished individuals. 000 per day, epidemic of measles or whooping cough, or high Prevalence of malnutrition among children aged 6 59 months, Prevalence of malnutrition among children aged 6 59 months Presence of oedema among children aged 6 59 months Presence of oedema among children aged 6 | 59 months. Extent of the area within which target ben eficiaries live. Information required to establish appropriateness prevalence of respiratory or diarrhoeal disease. prevalence of respiratory or diarrhoeal disease. median weight for height. median weight for height. sites in many locations. Security situation. . . • • • . of feeding sites close to the homes of the target beneficiaries, so that opportunity costs for carers are low. Ъ ъ households lack cooking fuel and/or cooking utensils; or 10 14% of children under 5 are < -2 SD or below 80% there is reason to believe that a take-home ration would insecurity levels are such that beneficiaries are not safe 10 14% of children under 5 are < -2 SD or below 80%. members, which is taken into account in the standard it is possible to establish decentralized supplementary there is reason to believe that targeted individuals will 5 9% of children under 5 are < -2 SD or below 80%, the ration size allows for some sharing among family 5|9% of children under 5 are < -2 SD or below 80% it is possible to target malnourished individuals; and median weight for height and there are aggravating median weight-for-height and there are aggravating recommended ration size for take-home distribution. In situations where a general distribution is ongoing and In situations where a general distribution is ongoing and food preparation at the household level is difficult consume a significant proportion of the food; and it is possible to identify and target malnourished going home with the ration; and of median weight-for-height; or median weight for height; or When it may be appropriate be widely shared; or individuals; and factors; and factors; and • • • • • ٠ • . • from therapeutic feeding from therapeutic feeding malnourished becoming malnourished becoming including children under and other malnourished home ration in addition severely malnourished, severely malnourished, ready-to-eat food to be under 5, older children. pregnant and lactating ration: aims to prevent pregnant and lactating home ration: mothers Malnourished children addition to the general eaten on the spot, in feeding with a *take*malnourished people provided with a taketo the general ration; people provided with children discharged children discharged aims to prevent the Response option children and other programmes, and including children programmes, and 5, older children, feeding on site: supplementary of malnourished supplementary the moderately moderately **Fargeted Targeted** women vomen

Licitude actividear	Ş	When it may be appropriate	Information required to establish appropriateness
	ln s	In situations where a general distribution is ongoing and:	 Dravalance of malnutrition among childran aged 6 [50 months
	•	15% of children under 5 are < -2 SD or below 80%, of	specifically. percentage of children with < -2 SD or below 80% of
Blanket		median weight for height; or	median weight for height
supplementary	•	10 14% of children under 5 are < -2 SD or below 80%, of	Presence of oedema among children aged 6 59 months
feeding with <i>take</i> -		median weight for height and there are aggravating	Dresence of other andravating factors including: crude mortality rate
home ration: all		factors; or	> 1 per 10 000 per day: an epidemic of measles or whooping could
children and other	•	during the early stages of an acute crisis, before a reliable	or high prevalence of respiratory or diarrhoeal disease; or a general
individuals provided	_	province can be established for an adequate general ration; and	ration not yet adequate to cover people's mean energy requirements.
with a ration to take	•	there is reason to believe that target individuals will	
home, in addition to the general ration		consume a significant proportion of the food; and	 Patterns of intra-household distribution: How is food shared within
	•	the ration size arrows for some snaming with farming members, which is taken into account in the standard	the household? Are certain foods reserved for children? For women?
		recommended ration size for take-home distribution.	
	2	la eitustione where a general distribution is on going and:	 Prevalence of mainturnion among children aged o [39 months, choosefoolly horizonthade of children with < 3 SD or holizon 2002 of
Blanket		and the second of the second o	specifically, percentage of cilitaten with < -2 ou of below ou 70 of madian weight for height
supplementary	•	10% of children under 0 are > -2 ou of below ou % of median weicht for heicht: or	 Presence of oedema among children aged 6/50 months
feeding on site: all		10 14 40% of children under E are / 2 CD or holow 900% or	
children and other	•		 Presence of other aggravating factors including: a general ration
nutritionally vulnerable		or median weight for height <i>and</i> there are aggravating factore: מאל	below mean energy requirements, crude mortality rate > 1 per 10 000 per day on exidemic of measiles ervideoning course or the
with ready-to-eat food			provide the second of the seco
eaten on the spot, in	•	there is reason to believe that a take-home ration would	 Patterns of intra-household distribution: How is food shared within
addition to the general		be widely shared; or	the household? Are certain foods reserved for children? For women?
ation	•	food preparation at the household level is difficult,	 Household access to cooking the and cooking utensils.
	_	households lack cooking fuel and/or cooking utensils; or	
	•	insecurity levels are such that beneficiaries would not be	Security situation
		safe going home with the ration.	
dditional information	requ	Additional information required for designing a supplementary feeding programme:	nme:
Number of bene	ficial	Number of beneficiaries and their location.	
 Likely duration of the intervention. 	f the	e intervention.	
 Acceptability of t 	hec	Acceptability of the commodities that could be available for the programme.	
 Potential sites fc 	Ins J	Potential sites for supplementary feeding distribution or feeding centres.	
 Security situation. 	Ŀ.		
 Media and comr 	nuni	Media and community networks available to publicize information concerning supplementary feeding.	l supplementary feeding.
 Possible causes 	of n	malnutrition other than lack of food, such as diarrhoeal di	Possible causes of malnutrition other than lack of food, such as diarrhoeal disease due to lack of access to sanitation and potable water.
 Partners' capaci 	tv to	Partners' capacity to address underlying causes of malnutrition, other than lack of food.	ck of food.
 Partners' capaci 	। ≤.	- staff. skills. equipment. administrative and monitorinc	Partners' capacity - staff, skills, equipment, administrative and monitoring systems - to establish and manage supplementary feeding programmes.
including availat	lit	of cooking fuel. cooking utensils and storage space, acc	including availability of cooking fuel. cooking utensils and storage space, access to clean water and sanitation for on-site programmes.
	2000	Infrastructura – roade etorada canacity, milling facilities	
	τ		

S
ш
×
Ш
z
~
7
-

In struations where: • there are significant numbers of everety mainourished individuals • children and/or adults or an wident increase in numbers of severely mainourished individuals; and • case loads are sufficiently • concentrated to warrant establishing • concentrated to warrant establishing • TCs within all communities • concentrated to warrant establishing • trained health staff are available, or • there are significant unubers of • populations are widely dispersed and/or separated by hills, rivers, etc., • making it difficultion establish TFCs • making it difficultion establish trained • making and or setablish trained • making and or setabl	Prevalence of severe mainutrition among children aged 6 59 months, specifically, percentage with <-3 SD or below 70% of median weight for height and/or oedema. Prevalence of severe mainutrition among children aged 5 9.9 years, specifically, percentage with <-3 SD or below 70% of median weight for height and/or oedema. Percentage of adults aged 20 59.9 years with BMI < 16, excluding disabled people and pregnant women and/or oedema.
there are significant numbers of severely mainounished individuals everely mainounished individuals everely mainounished individuals, and everely mainounished individuals, and exact loads are sufficiently case loads are sufficiently concented to warrant establishing FFCs within all communities concented, and trained health staff are available, or an be made available to supervise all tFFCs. In situations where: the are significant numbers of severely mainourished individuals, and everely mainourished individuals, and everely mainourished individuals, and evereshipt indirect are widely fispersed and/or separated by hills, rivers, etc. and or separated by hills, rivers, etc.	sentage with <-3 SD or below 70% of median weight for height and/or oedema. valence of severe mainutrition among children aged 5[9.9 years, specifically, sentage with <-3 SD or below 70% of median weight for height and/or oedema. sentage of adults aged 20[59.9 years with BM < 16, excluding disabled people pregnant women and/or oedema.
severely manourished individuals charactering manourished individuals charactering manourished individuals charactering manourished individuals, and ease loads are sufficiently concentrated to warrant establishing frecomerned; and concentrated to warrant establishing concentrated to warrant establishing frecomerned; and the load start are evaluable, or can be made available to supervise all thread to supervise all thread to supervise all characters are widely dispersed and/or separated by hils, rivers, etc., and/	ralence of severe mainutrition among children aged 5 [9.9 years, specifically, centage with < -3 SD or below 70% of median weight for height and/or oedema. centage of adults aged 20 [59.9 years with BMI < 16, excluding disabled people pregnant women and/or oedema.
 children and/or adults or an evident increase in numbers of severely mainourished individuals. and case loads are sufficiently concentrated to warrant establishing TFCs within all communities concentrad: and concerned: and concerned: and concerned: and trained health staff are available, or can be made available or trained health staff are available, or can be made available to supervise all TFCs. In situations where: the made available to supervise all the trained health staff are available, or can be made available to supervise all the trained manourished individuals / contant ender of severely manourished individuals and populations are widely dispersed and/or separated by hils, rivers, etc. 	entage with < -3 SD or below 70% of median weight for height and/or oedema. centage of adults aged 20∫59.9 years with BMI < 16, excluding disabled people pregnant women and/or oedema.
increase in numbers of severely mainourished individuals. and case loads are surficiently concentrated to warram testabishing TFCs within all communities concentrad. and concerned. and concerned of the concerned of	-sentage of adults aged 20[59.9 years with BMI < 16, excluding disabled people pregnant women an $d \prime or$ oedema.
mainourished individuals, and case loads are sufficiently concentrated to warrant establishing TFCs within all communities concerned, and trained health staff are available, or can be made available to supervise all TFCs. In situations where: the are significant numbers of the everety mainourished individuals. and children and/or adults or an evident in children and/or adults or an evident mainourished individuals. and or populations are widely dispersed and/or separated by hills, rivers, etc. madrost toffinct or establish TFCs	pregnant women and/or oedema.
 case loads are sufficiently concentrated for warrant establishing tFCs within all communities trained health staff are available, or concerned, and trained health staff are available, or can be made available to supervise all In situations where: In situations where the are significant numbers of severely malnourished individuals children and/or adults or an evident malnoase in numbers of severely malnoarish to dispersed and/or separated by hills, rivers, etc., and/or separated by hills, rivers, etc. 	
 concentrated to warrant establishing TFCs within all communities concerned; and trained heath staff are available, or trained heath staff are available, or can be made available to supervise all In situations where: these are significant numbers of severely malnourished individuals populations are widely dispersed and/or separated by hils, rivers, etc., mandors at and or the staff. 	
 resonance and communities trained neath staff are available, or concerned; and trained neath staff are available, or can be made available to supervise all In stituations where. these are significant numbers of severely manourished individuals children and/or adults or an evident increase in numbers of severely malroase are widely dispersed and/or separated by hills, rivers, etc., and or separated by hills, rivers, etc. 	Geographical distribution of cases of severe malnutrition.
 trained health staff are available, or can be made available to supervise all TFCs. In situations where: the are significant numbers of severely malnourished individuals children and/or adults or an evident malnourished individuals. and populations are widely dispersed and/or separated by hills, rivers, etc., madrost tidinctor establish TFCs 	
can be made available to supervise all TFCs. TFCs. In situations where: In situations where: there are significant numbers of severely malnourished individuals children and/or adults or an evident increase in numbers of severely malnourished individuals, and malnourished individuals, and populations are widely dispersed and/or separated by hills, rivers, etc., mandor severable to all concerned	
TFCs. In situations where: In situations where: everely mainourished individuals everely mainourished individuals children and/or adults or an evident increase in unders of everely mainourished individuals, and everely mainourished individuals, and and/or separated by hills, rivers, etc., making it difficut to estabilish TFCs and/or setsible h all on one end	Availability of trained health personnel.
 In situations where: there are significant numbers of	
 there are significant numbers of severely malnourished individuals children and/or adults or an evident increase in numbers of severely malnourished individuals, and populations are widely dispersed and/or separated by hills, rivers, etc. making it difficut estabilish TFCs 	Prevalence of severe malnutrition among children aged 6 [59 months, specifically,
severely malnourished individuals everely malnourished aduls or an evident increase in numbers of severely malnourished in dividuals, and epopulations are widely dispersed and/or separated by hills, rivers, etc., mador separated by hills, rivers, etc., machine it difficulto estabilish TFCs escashed to all concerned estable to all concerned estables are set on the estable of a set on	percentage with < -3 SD or below 70% of median weight for height and/or oedema.
children and/or adults or an evident increase in numbers of severely mainourshed individuals, <i>and</i> oppulations are widely dispersed and/or separated by hills, rivers, etc., making it diffront to estabilish TFCs	Prevalence of severe malnutrition among children aged 5 9.9 years, specifically,
 malnourished individuals, and malnourished individuals, and populations are widely dispersed and/or separated by hills, rivers, etc., making it dufficit to estabilish TFCs 	percentage with < -3 SD or below 70% of median weight for height and/or oedema.
mainourished individuals; and populations are widely dispersed and/or separated by hills, rivers, etc., mathing it difficut to establish TFCs accessible to all concerned	Percentage of adults aged 20 59.9 years with BMI <16 and/or oedema, excluding
 populations are widely dispersed and/or separated by hills, invers, etc., making it difficult to establish FFCs accessible to all concerned 	disabled people and pregnant women.
and/or separated by hills, rivers, etc., making it difficult to establish TFCs • accessible to all concerned •	
making it difficult to establish TFCs accessible to all concerned	
accessible to all concerned	Geographical distribution of cases of severe malnutrition.
	Nature of the terrain/topography.
ined households, so default rates are likely	
health workers to be high; <i>and</i>	
ity health workers are	
available to provide home-based	Availability of trained community health workers.
follow-up.	

Annex 6: Executive brief template



Add the logo of partner agencies which took part in the assessment

Executive Brief on the [country] Emergency Food Security Assessment

[See final section of this document for additional overall guidance on preparing this brief.]

Overview, scope and methods

- Introductory paragraph giving information on the context (general information on the country, crisis or chronic situation) and what triggered the assessment (shock).
- ✓ If an EMOP or PRRO is ongoing, a sentence stating its dates, number of recipients and interventions.
- Paragraph on the purpose of the assessment, particularly if it will inform the design of a WFP programme, listing the team members (agencies, NGOs and government ministries), what the assessment covers (entire country, specific regions or areas), and the dates of the fieldwork.

How was the assessment carried out? Provide some brief information on the methods used: secondary data analysis, household survey, key informant interviews, etc. State the numbers of villages visited and households interviewed, and where. State the numbers of children whose anthropometric measures were taken, and of mothers interviewed.

How many people are food-insecure or vulnerable?

- State the absolute number of people who are food-insecure and the percentage of the total population this represents. Provide a clear explanation of how the food insecurity is determined.
- State the absolute number of people who are vulnerable and the percentage of the total population this represents. Provide a clear explanation of how vulnerability is determined on the basis of household food consumption patterns and food access, and of what it means in this particular case.
- Provide a clear explanation of how the situation puts lives at risks.
- ✓ When possible, give an indication of how the situation has evolved since the last assessment or from the baseline, if any.
- State the percentages of malnourished individuals (stunted, wasted and underweight under-5 children, mothers, etc.), and the source(s) and date(s) of the data.

Where are the food-insecure and vulnerable people?



Who are the food-insecure and vulnerable people?

Provide a profile of food-insecure people describing their livelihood characteristics, gender(s), age(s), sources
of food, etc.

Why are they food-insecure and vulnerable?

Provide an overview of the context and causes of the crisis. In particular:

- Discuss the main causes of food insecurity: e.g. related to food availability, access and consumption/utilization. This part of the brief can be divided into these categories.
- Explain the consequences for lives and livelihoods.
- ✓ Highlight conclusions related to markets, regarding food availability and access.
- Highlight the most important EFSA conclusions about food consumption, access, coping strategies and income-earning opportunities, etc.
- Present data and conclusions on nutrition, when available. Specify whether malnutrition is related primarily to food security or to other factors, such as health, sanitation and care.

Is the situation likely to change in the coming months?

Add information on potential shocks and prospects, and the likely scenario for the coming months. Explain how the potential shock may affect people's livelihoods and food security situation. Specify whether this forecast will change the number and type of people whose lives or livelihoods are at risk.

Recommendations for interventions

- Summarize the main recommended food and non-food programming options, if possible stating how many people should be targeted under each, the suggested duration, and any targeting criteria, such as people without land, IDPs, etc. Explain briefly the linkages between the proposed responses and the causes of food insecurity.
- If cash-based or market responses are recommended, discuss which conditions need to be in place, such as market monitoring, whether there is prior experience/implementation capacity, and what fall-back responses should be considered, such as food for work if cash for work is not feasible.
- Include recommendations regarding complementary non-food activities, such as agriculture, water and sanitation, and market support.

Include recommendations for a follow-up assessment and/or monitoring to update the situation if required.

For more information on the EFSA, please contact:

[Name of country or regional director], WFP [country]: e-mail address: <u>name@wfp.org</u> [Name of WFP staff responsible for EFSA, i.e. EFSA team leader]: e-mail address: <u>name@wfp.org</u>

Additional guidance for preparing the executive brief

 Ideally, the executive brief should be two, and no longer than three, pages long. It should follow the established format to the extent possible. Remember to fill in all the areas in the form, including the headers and footers (date of release) and other sections highlighted in yellow.

2. The main audience is humanitarian decision-makers and managers, both within and outside WFP, who are familiar with food security issues: country directors, regional directors, donor representatives, representatives of

partner agencies, government representatives. Given the diversity of this group, avoid using WFP-only acronyms and overly technical language.

3. Make sure that the key conclusions and recommendations are presented clearly, convincingly and concisely.

4. The brief should be written by one of the main authors of the EFSA report, and should be prepared as soon as the main conclusions and recommendations have been finalized. It can be used to disseminate the main findings before the main report is finalized.

5. The brief should be written in the language of the primary audience, which is usually the same language as the assessment report. To ensure a wider audience of decision-makers, it should also be translated into English.

6. The brief should be based on the final assessment results as approved by the country/regional director. If the brief includes any tentative results, this should be clearly stated.

7. Ensure as much consistency a possible between the brief and any public document produced by the country office or Headquarters (including for WFP's Operational Reporting Group at Headquarters) on the same operation.

8. All executive briefs are posted to WFP's public website on the Food Security Analysis page; please be sure to send the final versions to the Food Security Analysis Service at Headquarters who will handle this task.

9. The brief should be cleared by the lead assessor and the country director and disseminated by the country office.

Glossary

Analysis plan

A plan, based on the key hypotheses to be tested, detailing how the data collected will be analysed. It may also guide which data need to be collected from primary and secondary sources.

Assets

Anything that is considered valuable or useful, such as a skill, a quality or a person.⁶⁷ In the Sustainable Livelihoods Framework, the following six categories of assets are defined:⁶⁸

- human: health and nutrition status, physical capacity, skills, level of education, etc.;
- social: household, gender, kinship and other networks, community groups, values and attitudes, etc.;
- financial: income, credit and loans, savings, liquid assets, etc.;
- physical: productive items, such as tools and equipment, stores, housing, livestock, infrastructure, etc.;
- natural: land, water, forests, etc.;
- political: power relationships, access to and influence over local and higher-level government processes.

Chronic food insecurity

A long-term or persistent inability to meet minimum food requirements.

Cluster sampling

A sampling technique in which the sample is defined in two or more stages. The sampling frame (or population of interest) is first divided into groups, or clusters, usually according to geographical area, location or village. Then a sample from these clusters is selected – usually households or individuals. Additional stages can take place. Sampling units in the second and subsequent stages are selected from within the clusters selected at the previous stage – for instance households from within a sampled village, or individuals from within a household, etc.

^{67.} Chambers Compact Dictionary, Chambers Harrap Publishers Ltd, Edinburgh, UK, 2005.

^{68.} Sustainable Livelihoods Approaches: progress and possibilities for change, D. Carney, DFID, London, 2002.

Community group discussion

A discussion with a mixed group of community members that includes men, women and young people from all subgroups within the community – village, camp, urban neighbourhood, etc.

Composite shock

The interaction of several different shocks.

Contextual information

Details of the processes that led to the current emergency and the reasons why it is occurring; factors that help to explain the emergency and give insights into the responses that may be appropriate.

Convenience sampling

Households and individuals are selected on the basis of ease of access.

Coping strategies

Activities to which people resort to obtain food, income and/or services when their normal means of livelihood have been disrupted.

Coping strategy index (CSI)

A quick and simple indicator of household food insecurity behaviour that reveals how households manage or cope with shortfalls in food consumption. Households' responses to such shortfalls are formulated into a simple numeric score reflecting the frequency and perceived severity of these coping behaviours. The higher the score, the greater the coping, and hence the higher the level of food insecurity.⁶⁹

Cross-tabulation

The process of correlating two or more indicators; it is used, for example, to gain insights into the prevalence and causes of malnutrition and food insecurity.

Data

The pieces of information collected from primary or secondary sources.

Direct sampling

Households and individuals are selected directly from the entire sampling frame.

Dominant opportunities and shocks

The opportunities and shocks that will have the greatest influence over how the food and nutrition security situation evolves in the coming months.

^{69.} The Coping Strategies Index, Field Methods Manual, Daniel Maxwell (Feinstein International Center, Tufts University), Richard Caldwell (TANGO International), CARE USA, Second Edition, January 2008.

Dynamic variables

Factors that can change quickly and that indicate transitory problems that may exacerbate existing chronic problems.

Emergency

A situation that causes widespread human, material, economic or environmental damage, threatening human lives and/or livelihoods and exceeding the coping capacities of the affected communities and/or government.

Entry points

The sectors and broad types of intervention where needs may first be addressed; they are identified in response analysis.

Focus group discussion

A structured discussion on a particular topic – the focus – with people who have knowledge and experience of that topic.

Food access

A household's ability to acquire food regularly through one or a combination of own home production and stocks, purchases, barter, gifts, borrowing and food aid.

Food access gap

The difference between household food expenditure and the cost of a minimum food basket, taking into consideration the proportion of food that is not purchased, such as home-produced food.

Food availability

The food that is physically present in the area of concern, through all forms of domestic production, commercial imports and food aid. This might be aggregated at the regional, national, district or community level.

Food availability gap

The shortfall between the food physically present in the area of concern and the food that is needed to feed the area's population.

Food consumption gap

The gap between the food consumption required to meet minimum nutrition needs and the actual food consumption, taking into consideration both minimal energy intake and diet quality in terms of essential nutrients.

Food consumption score (FCS)

A composite score based on the dietary diversity, food frequency and relative nutritional importance of the different food groups consumed. The higher the FCS, the more likely dietary diversity will be and the more likely that a household will achieve nutrient adequacy.

Food security

Exists when all people, at all times, have physical and economic access to sufficient, safe and nutritious food to meet their dietary needs and food preferences for an active and healthy life (World Food Summit, 1996).

Food utilization

Refers to: (i) households' use of the food to which they have access; and (ii) individuals' ability to absorb and metabolize nutrients – the efficiency of food conversion by the body.

Forecast analysis

Analysis to predict the nature of future opportunities and shocks, and the ways in which the situation is likely to evolve.

In-depth assessment

Undertaken when substantial time, access and resources are available. It uses rigorous methodologies that are adapted to the context, such as random, large-scale household food security and nutrition surveys, and household economy baseline surveys.

Indicator

A variable or combination of variables that gives insight into a particular aspect of the situation.

Information needs

The data that must be collected and processed from primary and secondary sources in order to fulfil the assessment objectives.

Initial assessment

Undertaken promptly following: (i) a sudden crisis; (ii) reports of deterioration in a long-term crisis; or (iii) improved access in an ongoing crisis. It is based mainly on secondary data and key informant interviews, but some quick field visits may be undertaken.

Key informant interviews

Interviews with individuals who have good knowledge about particular aspects of the community or the present emergency.

Livelihoods

The capabilities, assets – both material and social – and activities required for a means of living linked to survival and future well-being.⁷⁰

^{70.} Humanitarian Charter and Minimum Standards in Disaster Response, The Sphere Project, Geneva, 2004

Livelihood group

A group of people who share the same basic means of livelihood and life style – the same main subsistence and income activities, and social and cultural practices – and who face the same risks of food and nutrition insecurity.

Livelihood strategies

The ways in which households utilize and combine their assets to obtain food, income and other goods and services.

Nutrition security

A situation in which all individuals and households are food-secure, have good access to preventive and curative health care, and undertake healthy and sustainable care practices.

Observation method

The noting of visible and significant aspects of the affected area.

Primary data

Data collected during the assessment, from interviews with key informants, focus groups, households and individuals.

Proxy indicator

Used as an indirect means of measuring a variable that is difficult to measure or cannot be measured by itself. It provides information about a factor indirectly.

Purposive sampling (non-probability sampling)

Groups are selected for interview according to the researcher's choice. It does not involve random selection, so extrapolation of results to wider populations is not possible; its value lies in selecting information-rich cases for in-depth analysis related to the issues being studied.

Qualitative data

Observations that are categorical rather than numerical; they often involve attitudes, perceptions and intentions.

Quantitative data

Observations that are numerical.

Questionnaire

A series of questions that has been carefully formulated and ordered to provide information from individuals, households and communities. In a selected sample, the same individual, household or community questionnaire is addressed to each individual, household or community, respectively.

Random sampling (probability sampling)

Based on formal statistical theory, allowing reliable estimates to be calculated and minimizing bias; all members of the sampled population have a known, non-zero chance of being selected. Results can be extrapolated to the entire population with a degree of accuracy that depends on the sample size and the variability of the indicator.

Rapid assessment

Undertaken following an initial assessment in a sudden crisis, or as a component of a reassessment. It is based on a combination of secondary and primary data.

Resilience

The ability of a person or community to bounce back or recover after adversity or hard times, and to build positively on these adversities.

Response analysis

Analysis to determine the need, or otherwise, for an intervention and, when appropriate, to identify the most suitable types of intervention, the timing and the targeting criteria.

Sample

A selection of households or individuals from the total affected population.

Sampling frame

The area and population that the assessment covers.

Scenario

Describes a situation that could occur, based on a set of informed assumptions about the existing situation.⁷¹

Secondary data

Data collected from outside the current assessment. Examples include data collected by other agencies.

Secondary opportunities and shocks

Opportunities and shocks that are: (i) caused by the dominant event – cascade effects; (ii) unrelated to the dominant event but enhancing or mitigating its effects – synergistic effects; or (iii) subsumed by the dominant event.

Semi-structured interview

An interview based on a prepared series of questions and a checklist, the phrasing, order and form of which are not fixed.

^{71.} Contingency Planning and Humanitarian Action, a Review of Practice, R. Choularton, Overseas Development Institute (ODI) Network Paper No. 59, March 2007.

Shocks

Events that have negative impacts on food security and/or nutrition status. They can be natural or caused by human action.

Situation analysis

Analysis of the current food security and nutrition status of the population and the risks to lives and livelihoods.

Snowball sampling

Households and individuals are selected according to recommendations from other informants; each informant recommends the next set of informants.

Stage sampling

See cluster sampling.

Strengths, weaknesses, opportunities and threats (SWOT) analysis

A systematic appraisal of different response options, based on their strengths and weaknesses, and the opportunities and threats associated with them.

Structural variables

The underlying contextual factors in an area; they do not change quickly, and therefore influence livelihood outcomes for individuals and communities.

Terms of reference

Describe how the assessment will be carried out, and the roles and responsibilities of the assessment team; they are prepared at the start of the assessment.

Transitory food insecurity

A short-term or temporary inability to meet minimum food requirements, indicating a capacity to recover.

Triangulation

A process for comparing information from different sources to determine whether or not evidence converges.

Trigger

An event or series of events indicating that the nutrition or food security situation is deteriorating, or has already reached a level of crisis, and that an assessment is needed. Trigger indicators are based on data collected by early warning and food security monitoring systems, and consist of a few macro- and micro-level indicators that signal the need for a closer look at the situation.

Vulnerability

Exposure to risk and the lack of ability to cope with its consequences.

Index

Α

accuracy - **32**, 92, 96, 141, 188 analysis plan - 28, **38**, 55, **88** anthropometric survey - 41, 57, 90, 107 assessment - **31-34**, 116 asset(s) - **21-27**, 43, 159-167, 216 assumption(s) - 31, 101, 102, **175**

B

beneficiary/ies - 192 bias - 57, **96-104** blanket supplementary feeding - 266 body mass index (BMI) - **55-59**, 165, 267

С

camp - 54, 105-115 care practices - 24, 196, 216 cash and voucher transfers - 194 cash for work (CFW) - 200, 202 causality - 162 checklist - 119-135, 218, 246 child/children - 23-27, 57-61, 107-112, 160-174, 194-210 chronic food insecurity - 150, 162, 205 clinical status - 61 cluster - 107-112 cluster sampling - 107-112 communication - 115, 116, 219 community/ies - 25, 118, 119, 122, 208 community-based therapeutic care (CTC) - 267 community group discussion(s) - 122, 129 composite shock(s) - 177 conceptual framework - 15, 17, 20, 25-28, 50 confidence interval(s) - 98, 100 Consolidated Appeal(s) Process (CAP) - 34

consumer price index - 47, 72 context-specific threshold(s) - 49, 52 contextual information - 39, 45, 46, 153 contingency plan(s) - 176, 179 coping - 26, 75, 78, 156 coping strategies index (CSI) - 75, 76, 89 crop and food security assessment mission (CFSAM) - 13, 34 cross-tabulation - 53, 157 crude mortality rate - 57, 165-166

D

data analysis - 155, 159-160 debt - 75, 152 decision tree - **202-203** decision-making - 17, 47 dietary diversity - 62-63, 166-167 displacement - 146, 164-165, 167, 209 do no harm **206**

E

early warning systems - 33 EFSA types - **31** elderly - 23-24, 155 enumerator(s) - 113-129 exchange against produce - 261 extrapolation - 58, 107, **154**

F

factor(s) of risk - 200 famine foods - 67, 69 feasibility - 38, 204-205 feeding - 24, 134, 194-207 field assessment team fieldwork - 34 focus group(s) - 131, 138, 151 focus group discussion(s) - 42, 92, 122, 131 food access - 23-28, 66, 69, 73, 167-168 Food and Nutrition Security Conceptual Framework - 25 food availability - 23, 139, 174, 195 food consumption - 62, 66-67, 157, 168, 201 food consumption score (FCS) - 54, 62, 65 food distribution - 194, 201 food for education - 194 food for work (FFW) - 201, 208

food gap - **187-191** food groups - 62-63 food insecure - 150-160, 196 food-secure - 166 food security - **22**, 62, 24-26, 54, **165** food security indicator - 50, 62, 74-75, 106, 164 food self-sufficiency - **68**, **70** food transfers - 201-202 food utilization - **23-27**, 195, 201 food voucher(s) - 202 forecast - 146, **170-192** free food distribution - 197, 201

G

gap (see food gap) gender - 25, 50, 114, 130, 203 global acute malnutrition - 59, 165

Η

health programme(s) - 194 height for age - 59 historical timeline - 61, **140** HIV/AIDS - 56, 103, 146, 148, 174 household economy baseline survey(s) - 33 household food access - 25, 70 household (-level) food consumption - 54, 75, 168 household food security - **25**, 50, **62**, 157, 216 household survey(s) - 122, 125, **129**, 188 hub effect - 101 human resources - **113**, 198, 204 hypotheses - 26-27, 45

impact - 42-43, 148, **185**, 221 income - 53, 69-70, 185 in-depth assessment - **32-33**, 95, 167 information needs / requirements - 39, **41-45** initial assessment - **32**, 86, 94 institutional feeding - 194 Integrated Food Security Phase Classification (IPC) - 165-166 inter-agency assessment(s) - **33-34** internally displaced persons (IDPs) - 111, 154-156 intervention - **196-199** intervention type - 166, 194, 196, 219

J

joint assessment mission (JAM) - 34

Κ

key informant(s) - 94, 103, 122, 133 key informant interview(s) - **133**, 155

L

livelihood(s) - 20, 25, 185, 194 livelihood assets - 25, 44, 70, 167 livelihood group(s) - 148-149, 182-183 livelihood strategy/ies - 25, 67, 182-183 livelihood zone(s) - 105 livestock - 73, 210 local adaptation - 28, 53, 215 logistics - 88, 115, 198 long-term crisis - 32

Μ

malnourished individuals - 53, 150, 202 malnutrition - 28, 59-60, **162**, **164**, 167, 183, 201-205, market(s) - 71, 151, 158 market assistance programme - 201 market indicators - 71-72 market system support - 261 methodology - **92-93**, 157 micronutrient - 57, 61, 63 mid-upper arm circumference (MUAC) measurement - 51-59 mortality - 25, 50, 54, **56**, **164**

Ν

neighbourhood care programme (NCP) - 264 non-food intervention(s) - 194, 223 nutrient density - 63 nutrition crises - 152, 194 nutrition indicator - 50, 58 nutrition status - 25, **57**, 165

0

observation - 122, **136-137** ongoing crisis - 32 on-site supplementary feeding - 255 opportunities and shocks - **171**, 177 orphan(s) - 201

Ρ

pair-wise ranking - 141 partner(s) - 33-34, 215 physical access - 93, 159, 178, 202-203 pillars (food security) - 22 planning - 86-116 policies - 21, 47, 177, 204 post-conflict needs assessment (PCNA) - 34 pre-crisis data - 42, 52, 163 pregnant and lactating women - 194, 207 primary data - 55, 120-142 principal component analysis (PCA) - 53 probability proportional to size (PPS) - 106, 113 profile(s) - 50, 88, 157, 159, 181 proportional piling - 140, 155 protracted (crisis/crises) - 12, 87 proxy indicator(s) - 50, 60-61, 152 public health environment - 24, 157 public works project(s) - 253 purchasing power - 68-69, 178 purposive sampling - 97-106, 154

Q

qualitative data - 32, **155-163**, 181 quantitative data - 32, 79, **153**, **157-161**, 181, 188 questionnaire survey - 129, 131, 169 quick-onset emergencies - 94

R

random sampling - 96, **104**, 108 rapid assessment (EFSA) - **32**, 94 recommendation(s) - **209-210** refugee - 34, 54, 103, 150 remittances - **68-70**, 141, 189 resilience - **22** response analysis - **193-208** response options - **199-206**, 217 risk(s) - 20, **155**, **164**, 194

S

sample size - **98-99**, 107-110 sampling frame - **95-97**, 105-111 sampling methodology - 95-96 sanitation - 61, 159, 215

scenario(s) - 175-185, 192 school feeding - 201, 207 seasonal calendar - 139, 188 secondary data - 55, 89, 91, 124 sector-specific information - 135 security - 116 semi-structured approach - 132, 135 sequential EFSA - 17 severity - 77, 164, 215-216 shocks - 170-174, 185-192 situation analysis - 74, 152-169 slow-onset or protracted crisis/crises - 46, 95, 176 snapshot assessment - 94 snowball sampling - 103 source(s) of food - 67, 159 stage sampling - see cluster sampling stakeholder analysis - 197 strategic response framework - 166 stratification - 41, 105 structural variable(s) - 164 structured observation - 137 stunting - 59, 163, 165, 167 sudden-onset crisis - 24, 100, 215 supplementary feeding programme (SFP) - 201, 205 SWOT analysis - 203-206, 217 systematic sampling - 109

Т

take-home supplementary feeding - 255 targeted supplementary feeding - 265 targeting - **207-208** terms of reference - **81-89**, 214 terms of trade - 70, 72 therapeutic feeding programme (TFP) - 265 transect walk - **136**, 188 transitory food insecurity - 150, 162, 216 triangulation - 96, **150** trigger - **86-87**, 95, 134 type(s) of EFSA - **31** type of emergency - 31, 98, 175, 200 type of informant - 135 types of intervention -166, 194, 196, 219

U

urgency of the crisis - 43, 86, 93-94

V

value judgement(s) - 52 vulnerability - **22**, 26 vulnerable groups - 101-102, **148-150**

W

wasting - 58-59, 164 weight-for-height - 51, 58 weight-for-age - 58-59 woman/women 21-23, 47, 57-60, 132, 165, 195 worst-case scenario - 175, 179, 193

Reference List

WFP publications and joint publications

The following publications are listed in alphabetical order according to their titles or series names. Please note that in January 2008 WFP's Emergency Needs Assessment Service and its Vulnerability Analysis and Mapping Branch were merged into the Food Security Analysis Service.

A Manual: Measuring and Interpreting Malnutrition and Mortality. CDC and WFP, July 2005.

Estimating Population Size in Emergencies, by A. Henderson. WFP Emergency Needs Assessment Service, December 2006.

Food Consumption Analysis – Calculation and use of the Food Consumption Score in food consumption and food security analysis. WFP Vulnerability Analysis and Mapping Branch, January 2008.

Guidance Note, Strengthening Rapid Food and Nutrition Security Assessments. WFP Emergency Needs Assessment Service, July 2007.

Guidelines for Comprehensive Food Security and Vulnerability Analysis (CFSVA). WFP Food Security Analysis Service, January 2009.

Market Analysis in Emergency Food Security Assessments: Guidelines on Market Situation Analysis and Forecast and Response Protocol. WFP Emergency Needs Assessment Service, August 2007.

Market Analysis Tool: Terms of Trade. WFP Economic Analysis Unit, August 2007.

Methods and Tools for Conflict Analysis. WFP Transition Unit. August 2007.

Participatory Techniques and Tools - A WFP Guide. 2001.

Policies and Procedures for the Use of the Immediate Response Account (IRA). WFP Operations Department and Administrative Department Joint Directive, 18 October 2005. (OD2005/005, A 2005/009)

Strengthening Rapid Food and Nutrition Security Assessments. WFP Emergency Needs Assessment Service, July 2007.

Technical Guidance Sheet No. 1. Integrating Migration and Displacement into Emergency Food Security Assessments (EFSAs). WFP Emergency Needs Assessment Service, May 2007.

Technical Guidance Sheet No. 4. Trigger Criteria for an Emergency Food Security Assessment (EFSA) in Slow-Onset Crises. WFP Emergency Needs Assessment Service, November 2007.

Technical Guidance Sheet No. 5. Distinguishing between Chronic and Transitory Food Insecurity in Emergency Food Security Assessments (EFSAs). WFP Emergency Needs Assessment Service, December 2007.

Technical Guidance Sheet No. 6. Initial Emergency Food Security Assessments (EFSAs). WFP Emergency Needs Assessment Service, November 2007.

Technical Guidance Sheet No. 7. Area Method to Estimate Population Size and Demographics in Emergency Food Security Assessments (EFSAs), by A. Henderson. WFP Emergency Needs Assessment Service, September 2007.

Technical Guidance Sheet No. 9. Qualitative Data Collection and Analysis for Food Security Assessments, by L. Morinière. WFP Emergency Needs Assessment Service, September 2007.

Technical Guidance Sheet No. 10. Using the Delphi Method to Estimate Population Size and Demographics in Emergency Food Security Assessments (EFSAs), by A. Henderson. WFP Food Security Analysis Service, January 2008.

Technical Guidance Sheet No. 11. T-Square Method to Estimate Population Size and Demographics in Emergency Food Security Assessments (EFSAs), by A. Henderson. WFP Food Security Analysis Service, January 2009.

Technical Guidance Sheet No. 12. Complementary methods and tools for Emergency Food Security Assessments (EFSAs), by WFP Emergency Needs Assessment Service, December 2007.

Technical Guidance Sheet No. 13. Analysing National Capacity to Respond to Food Security Crises. WFP Food Security Analysis Service, September 2008.

Thematic Guidelines on Sampling. WFP Vulnerability Analysis and Mapping Branch, December 2004.

Publications by other organizations

The following publications are listed in alphabetical order according to publishing organization.

Active Learning Network for Accountability and Performance in Humanitarian Action. Participation by Crisis-Affected Populations in Humanitarian Action: A Handbook for Practitioners, 2003.

CARE USA. The Coping Strategies Index, Field Methods Manual, by D. Maxwell (Feinstein International Center) and R. Caldwell (TANGO International). Second edition, January 2008.

Chambers Harrap. Chambers Compact Dictionary. Chambers Harrap Publishers Ltd, Edinburgh, UK, 2005.

DFID. Sustainable Livelihoods Approaches: progress and possibilities for change, by D. Carney. London, 2002.

FAO. Integrated Food Security Phase Classification (IPC) User Guide. Draft-in-progress for feedback. July 2008.

IASC. Guidelines for Gender-based Violence Interventions in Humanitarian Settings, Focusing on Prevention of and Response to Sexual Violence in Emergencies. September 2005.

IFRC. Guidelines for Emergency Assessment, 2005.

IPC Global Partners. Integrated Food Security Phase Classification: Technical Manual Version 1.1. FAO, Rome, 2008.

ODI. A Critical Review of Approaches to Assessing and Monitoring Livelihoods in Situations of Chronic Conflict and Political Instability (SCCPI), by S. Jaspars and J. Shoham. ODI Working Paper No. 191. London, 2002.

ODI. Contingency Planning and Humanitarian Action, a Review of Practice, by R. Choularton. ODI Network Paper No. 59. London, March 2007.

ODI. Remittances during crises: implications for humanitarian response, edited by K. Savage and P. Harvey. Humanitarian Policy Group Report No. 25. London, May 2007.

Save the Children. Emergency Nutrition Assessment: Guidelines for Field Workers, 2004.

Save the Children. The practitioners' guide to the Household Economy Approach. Regional Hunger and Vulnerability Programme, Food Economy Group and Save the Children, 2007.

Sphere Project (The). Humanitarian Charter and Minimum Standards in Disaster Response. Geneva, 2004.

List of boxes

Box 1.1	Assets and strategies	20
Box 2.1	Feasibility	38
Box 2.2	Definition of terms used to develop an EFSA analysis plan	39
Box 2.3	Identification of priority information requirements using the	
	Food and Nutrition Security Conceptual Framework	43
Box 2.4	Proxy indicators of food security	50
Box 2.5	Cross-tabulation and comparison	53
Box 2.6	Calculation of the FCS	63
Box 2.7	Validation of the FCS	65
Box 2.8	Process for establishing the CSI	75
Box 2.9	Developing coping strategy indicators	78
Box 3.1	Rule of thumb for estimating sample size in purposive sampling	99
Box 3.2	Two-stage and simple random sampling	105
Box 3.3	Rule of thumb for estimating the sample size in a	
	household survey	107
Box 3.4	Rule of thumb for determining the number of clusters	
	in a random sample	108
Box 3.5	Common problems with primary data collection	120
Box 3.6	Main characteristics of questionnaires	123
Box 3.7	Main characteristics of semi-structured interviews	124
Box 3.8	Tips for semi-structured interviews	128
Box 3.9	Data collection in the context of sexual and	
	gender-based violence	130
Box 3.10	Tips for facilitating focus group discussions	132
Box 3.11	Rules of thumb for identifying key informants	134
Box 4.1	Chronic and transitory food insecurity	150
Box 4.2	Qualitative identification of population groups, households	
	and individuals facing food insecurity and malnutrition	155
Box 4.3	Defining the characteristics of households and individuals	
	facing food insecurity and malnutrition, using qualitative data	159
Box 4.4	Compilation of pre-crisis information using a focus group	
	discussion	163
Box 4.5	The IPC approach	166
Box 4.6	Estimation and use of the food consumption gap	168
Box 4.7	Estimation and use of the food access gap	169
Box 4.8	Opportunities and shocks	171
Box 4.9	Predicting opportunities and shocks	172
Box 4.10	Typical stakeholders in food security response programmes	197
Box 4.11	Conflict analysis and the do-no-harm principle	206
Box 5.1	Potential users of an EFSA report	219
Box 5.2	WFP procedures for communicating EFSA results	220

List of figures

Figure 1.1	The Food and Nutrition Security Conceptual Framework	25
Figure 3.1	Example of a seasonal calendar	139
Figure 3.2	Example of a historical time line	140
Figure 3.3	Example of proportional piling	141
Figure 4.1	Decision tree for response options	203

List of tables

Table 1.1	Inter-agency assessments	34
Table 2.1	Format for an analysis plan	39
Table 2.2	Template for collecting pre-shock livelihood information	42
Table 2.3	Sudden-onset crises and their impacts on nutrition status	
	and food security	42
Table 2.4	Slow-onset crises and their impacts on nutrition status	
	and food security	43
Table 2.5	Examples of contextual information	46
Table 2.6	Crude mortality rate	57
Table 2.7	Anthropometric and clinical indicators of the nutrition situation	59
Table 2.8	Examples of additional indicators that inform about	00
10010 210	nutrition status	61
Table 2.9	Template for calculating the FCS	64
Table 2.10	Food consumption groups and corresponding FCS	• •
	thresholds and energy intake levels.	66
Table 2.11	Examples of food access indicators	67
Table 2.12	Market indicators	71
Table 3.1	Secondary data sources	89
Table 3.2	Identification of information gaps, with examples	90
Table 3.3	Circumstances under which each type of sampling is used	97
Table 3.4	Table for identifying clusters	112
Table 3.5	Example of an EFSA schedule	117
Table 3.6	Common pitfalls and solutions in questionnaire interviews	126
Table 3.7	Form for recording observations, with examples	137
Table 3.8	Example of pair-wise ranking	142
Table 4.1	Stages of EFSA analysis	146
Table 4.2	Characteristics of households where malnutrition exists	157
Table 4.3	Examples of individuals and households who may face risks	
	to health and livelihoods, and possible reasons why	161
Table 4.4	Interpretation of the population-level severity of mortality	
	and nutrition status indicators	165
Table 4.5	Key reference indicators and thresholds used in the	
	IPC approach	167
Table 4.6	Documentation of opportunities and shocks, with examples	173
Table 4.7	Shocks and their potential impacts on food security	174
Table 4.8	Scenarios and their impacts, examples	180
Table 4.9	Shocks and their impacts on different groups, examples	182
Table 4.10	General process for determining the food gap	188
Table 4.11	Overview of groups affected under different scenarios,	
	examples	192

Table 4.12	Factors associated with risk for various groups,	
	with examples	195
Table 4.13	Ongoing and planned response interventions, with examples	198
Table 4.14	Summary of interventions and gaps for groups at risk	199
Table 4.15	Food and non-food responses in terms of factors of risk	201
Table 4.16	Targeted interventions, examples	207
Table 4.17	Monitoring data and schedule, example	210
Table 5.1	EFSA report format	214
Table 5.2	Contents of the executive brief	221

Resources on the Food Security Analysis Guidance DVD

The following documents, reference materials, handbooks and other resources are available on the Food Security Analysis Guidance DVD.

- EFSA Handbooks (2009 & 2005 editions) in English, French & Spanish
- CFSVA Guidelines
- VAM Thematic Guidelines (some also available in French)
 - Initial assessments
 - Urban assessments
 - Market analysis tools
 - Nutrition
 - Livelihoods
 - Food consumption
 - Estimating population numbers & sampling
 - Qualitative analysis
 - HIV/AIDS
 - Participation & gender
 - National response capacity
 - Other
- UN/Inter-Agency Assessment Guidelines
 - WFP/UNHCR Joint Assessment Guidelines (JAMs), March 2009
 - FAO/WFP Joint Guidelines for Crop & Food Security Assessment Missions (CFSAMs), January 2009
 - IASC Initial Rapid Assessment (IRA): Guidance notes & field assessment form
 - CAP Needs Analysis Framework, IASC CAP Sub-Working Group, April 2005
- Inter-Agency & Partners Food Security Analysis Guidelines
- Humanitarian Principles & Protection
- WFP Contingency Planning Guidelines
- WFP tools for EFSAs, CFSVAs & Inter-Agency Assessments
 - Example terms of reference (TORs)
 - Example checklists & questionnaires
 - Example executive briefs & template
 - Example reports
 - Quality Monitoring Checklist templates

Credits

Photographs

Cover, first line, from left to right: WFP / Kornelius Schiffer WFP / Sven Torfinn WFP / Elmer Martinez Cover, second line, from left to right: WFP / Marcus Prior WFP / Riccardo Gangale WFP / Rein Skullerud Cover, third line, from left to right: WFP / Caroline Chaumont WFP / Riccardo Gangale WFP / Luke Powell

Page 6:WFP / Edith ChampagnePage 15:WFP / Evelyn Hockstein-PolarisPage 35:WFP / Alejandro ChicheriPage 81:WFP / Amy HortonPage 143:WFP / Rein SkullerudPage 211:WFP / Leah RichardsonPage 225:WFP / Frederic Loward

Editor

Jane Shaw

Proof-reader Valerie Guidi

Graphic design

Indro Uttinacci

Printer Stilgrafica Srl

Production Caroline Chaumont Mark Bennett